

MAINSTREAMING UNIVERSAL DESIGN IN CAPE TOWN: FIFA 2010 WORLD CUP™-RELATED ACTIVITIES AS CATALYSTS FOR SOCIAL CHANGE

by

MUGENDI KANAMPIU M'RITHAA

Thesis submitted in fulfilment of the requirements for the degree

Doctor of Technology: DESIGN

in the Faculty of Informatics and Design

at the Cape Peninsula University of Technology

Supervisor:Professor David A.V. BoonzaierCo-supervisor:Professor Simeon E.H. Davies

Cape Town (December 2009)

DECLARATION

I, Mugendi Kanampiu M'Rithaa, declare that the contents of this thesis represent my own unaided work, and that the thesis has not previously been submitted for academic examination towards any qualification. Furthermore, it represents my own opinions and not necessarily those of the Cape Peninsula University of Technology.

7th December 2009

Date

Signed

ABSTRACT

Universal Design (UD), as it is known in the USA and elsewhere, is an inclusive approach to designing for the broader population and is rapidly gaining popularity amongst design practitioners and planners globally. Similar non-exclusive approaches have evolved in diverse parts of the world to counter the systemic disablement and exclusion of vulnerable/special populations of users perpetuated by traditional approaches to design. The transdisciplinary field of UD is informed by concilience in accommodating a wide range of related fields, such as education, landscape architecture. architecture, town and regional planning, industrial/product/three-dimensional design, furniture design, interior design, communication/information/graphic design, interaction design, human-computer interaction (HCI)/usability studies, and ergonomics/human factors engineering. UD proposes a collaborative systems approach that benefits from the synergies of cross-functionalism by approaching the diverse challenges facing society through socially responsible design. In so doing, UD can potentially impact such diverse issues as health, transportation, inclusive education, sports and recreation, entertainment, social welfare, inclusive employment, transgenerational/lifespan housing, inclusive tourism, accessibility, safety, and ecological concerns on sustainability.

The thesis adopts the framework of Activity Theory as a lens through which to investigate how UD can ensure social inclusion and participation by informing the design of products, systems, services, and built environments to accommodate the greatest possible number of users whilst still respecting their diversity. Specifically, the thesis embraces an interpretivist paradigm in its investigation of the relevance and applicability of UD strategies to the City of Cape Town. The research employs a qualitative mixed/multi-methods approach – including a survey of 147 design students at the Cape Peninsula University of Technology (CPUT), and interviews with key informants (design educators, professional designers, local government officials, disability sector practitioners, design activists, and politicians). The urban development projects associated with the FIFA 2010 Soccer World Cup™ offer an historic opportunity for an exploratory case study on how UD strategies can be effectively mainstreamed using this sporting mega-event as a catalyst to effect social change. Simultaneously, action research allows for experimentation in the classroom as well as participation in a 2010-related student competition to investigate the efficacy of UD in multiple contexts. A UD audit of a public park/botanical gardens is also included for discussion. Implications for tertiary-level education of design students, as well as the continuing professional development of graduate designers are discussed herein. The combined settings inform the prerequisites for adjustable UD strategies that are not only relevant and sustainable within Cape Town, but also nationally and much further afield.

KEYWORDS

- Accessibility
- Action Research
- Activity Theory
- Ageing
- Cape Town
- Concilience
- Continuing Professional Development
- Design Education
- Disability
- Diversity
- Equity
- Ergonomics / Human Factors
- Explanatory Legitimacy Theory
- FIFA 2010 Soccer World Cup™
- Inclusive Education
- Mega-Events
- Participation
- Social Inclusion
- Socially Responsible Design
- South Africa
- Sports Tourism
- Sustainability
- Ubuntu
- Universal Design (UD)
- Usability
- Vulnerable Populations
- Western Cape Province

IN-TEXT CITATION CONVENTIONS

Dates of publication and page numbers are provided for all authors whose ideas are quoted or paraphrased. Double quotation marks or indentations are used in the case of direct quotations. Exceptions to providing dates and page numbers are where the general focus of an author is referred to or material sourced from the Internet for which no dates and/or page numbers exist. In those instances where multiple references are cited with respect to related opinions/stances in the reviewed literature, the same are listed in chronological order.

ACKNOWLEDGEMENTS

I am because we are... I wish to express my sincere appreciation to a number of individuals leading up to the completion of this thesis:

- To my erudite supervisors, Professors David Boonzaier and Simeon Davies your experience, direction and constructive criticism kept me focussed and duly informed;
- To my enlightened parents for establishing a solid academic foundation upon which every successive learning endeavour (including this one) has been built;
- To my siblings and the extended family network (including KK) for your own unique and effectual expression of *ubuntu*, and for perpetually challenging me to achieve higher goals;
- To the executive members of my resident fan club; Doreen Kainyu, Martin Mwenda, and Michella Mwende: *asanteni* for believing in me (in spite of my acute occupational dementia). You are a fantastic team, and the best inspiration a man could wish for;
- To Adrienne Viljoen, the design *sangoma* whose passion for design (and vision for designers) on this continent has been a constant source of inspiration to me;
- To Alexi Prodromou for being such a conscientious and dedicated brother's keeper;
- To David Christer (my fellow biophile), for sharing my enthusiasm for UD, as well as for creating some of the visually pleasing graphics used in the latter part of this thesis;
- To Johan van Niekerk for your sense of optimism and wonderful friendship;
- To Jörn Messeter, for your timeous magnanimity at a critical juncture along this journey;
- To Laurine Platzky and Pazuna Stofile for availing of your resources and expertise;
- To Mel Hagen, the indefatigable design activist for being ever so instrumental in facilitating key contacts and in so doing you eased my research journey significantly;
- To Ziggy Strohbach and Alan Meru for steadfastly (and effectively) nudging me on;
- To Adriana Hornea, Alettia Chisin, Amelia Rorwana, Andre Meyer, Andrea Broom, Bart Verveckken, Belinda Verster, Brendon Knott, Bruce Snaddon, Catherine Trimble, Charles Gilbert, Chris Winberg, Christina Zonyane, Cora Motale, Courtley Pharaoh, Craig Finnan, C.V. Botha, David Makola, Diane Bell, D.J. Coetzee, George Mvalo, Jenny Penfold, Johann van der Merwe, Johannes Cronjé, Juana Jacobs, June Khuzwayo, Kamilla Swart, Karen Hector, Mari Lewis, Marianna Olivier, Marlon Parker, Maurice Dassah, Munira Allie, Nigel Tapela, Nina du Toit, Rael Futerman, Retha de la Harpe, Roelof Oelofsen, Shaun Pather, Stuart Warden, Vikki du Preez, Wayne Coughlan, and Zahied Mukaddam, for valued support;
- To Aguinaldo dos Santos, Ajay Kumar, Amos Wangombe, Anna Bernagozzi, Anthony Mbogho, Bill Moggridge, Bongani Ntombela, Byron Qually, Carla Cipolla, Carlo Vezzoli, Carlos Hinrichsen, Carol o'Brien, Craig Thompson, David Stairs, Debra West, Dick Henry, Elza Miles, Erica Elk, Esmé Kruger, Eugene Ngugi, Ezio Manzini, Fatima Darries, Felix Ochieng', Finzi Saidi, François Jégou, François Nsenga, George Teodorescu, Gianfranco Zaccai. Guy Davies, Håkan Edeholt, Hammond Njagi, Havi Murungi, James Odora, Jeremy Opperman, John Mayienga, John Thackara, Johnnie Petersen, Jorge Frascara, Kris Kumar, Kyalo Manthi, Lalit Narkhede, Lara Penin, Lilac Osanjo, Linda Mvusi, Lorraine Ambole, Lou Yongqi, Marcel Maré, Marie Hendricks, Micke Svedemar, Mikko Korpela, Monica Tompson, Nick Ndaba, Odoch Pido, Onesmus Njenga, Pelle Ehn, Peter Mweu, Phillip Thompson, Ralph Shepherd, Ricardo Gomes, Richard Perez; Richie Moalosi, Roberto Bartholo, Roelf Mulder, Roger Fisher, Saki Mafundikwa, Samuel Khumomotse, Sebenzile Matsebula, Sheila Mwihia, Shona McDonald, Shorn Molokwane, Sudhakar Nadkani, Sunday Ogundiyile, Thomas Ongolo, Thumbi Ndungu, T.V. Praseed, T.V.P. Chowdry, Uday Athavankar, Undere Deglon, Venny Nakazibwe, Vincent Daniels, Zayd Minty and myriads whose goodwill and generosity of spirit helped shape my worldview and fuel my passion for design (and life); and
- To all DRAW members; my trusty assistants Benjamin Bell, Colette Fransolet, Ewaldi Grové, Guillaume du Toit, George Hugo, Jared Odell, Chad Petersen and the many enthusiastic students I have been privileged to engage with thus far in my pedagogic sojourns...

The financial assistance of the CPUT University Research Fund (URF) towards this research is gratefully acknowledged. Opinions expressed in this thesis and the conclusions arrived at, are those of the author, and are not necessarily to be attributed to the Cape Peninsula University of Technology.

DEDICATION

"[We're all] products of history and community, of opportunity and legacy." – *Malcolm Gladwell (2008:285)*

The pervading influence and inspirational legacy of my predecessors in design – Selby Mvusi, Jessica Otieno-Gwalla, Nathan H. Shapira, and Sylvester J. Maina – is respectfully memorialised. Moreover, this project is dedicated to all those who have embraced and advanced the values of access, diversity, equity, humanity, inclusiveness and participation for all people, everywhere. *Godspeed...*

TABLE OF CONTENTS

Declaration	ii
Abstract	iii
Keywords	iv
In-text Citation Conventions	ν
Acknowledgements	vi
Dedication	vii
Table of Contents	viii
Glossary	XV

CHAPTER ONE: WHAT THE STUDY IS ABOUT

1.0	Introduction	1
1.1	Motivation for Choosing this Topic	2
1.2	The Research Problem in Context	2
1.2.1	Hypothesis of the Study	2
1.2.2	Basic Assumptions	2
1.3	Research Question	3
1.4	Research Objectives	3
1.5	Significance of the Research	4
1.6	Structure of the Thesis	5
1.7	Summary	6

CHAPTER TWO: AN AFRICAN PERSPECTIVE ON MEGA-SPORTING EVENTS

2.0	Introduction	7
2.1	Mega-Sporting Events	7
2.2	"The Beautiful Game"	8
2.3	A First for Africa	9
2.4	Implications for the Host Country	11
2.5	2010 and Cape Town	12
2.5.1	Building a Sustainable Legacy	20
2.5.2	Anticipated Outcomes	24
2.6	Summary	25

CHAPTER THREE: VULNERABLE POPULATIONS IN CONTEXT

3.0	Introduction	26
3.1	Vulnerable Populations	26
3.1.1	The Disabled	27
3.1.1.1	Definitions of Disability	30
3.1.1.2	Categories of Disability	31
3.1.2	The Elderly	32
3.1.3	Women	35
3.1.4	The Young	36
3.1.5	Foreigners and Visitors	38
3.2	Implications of Shifting Demographics	39
3.3	Summary	43

CHAPTER FOUR: RESPONSES TO THE NEEDS OF VULNERABLE POPULATIONS

4.0	Introduction	44
4.1	Models of Disability	44
4.1.1	Medical Model	44
4.1.2	Social Model	45
4.1.3	Architectural Model	46
4.1.4	Biopsychosocial Model	47
4.2	Responses to the Needs of the Vulnerable	48
4.2.1	The Legal Case	49
4.2.1.1	Disability-specific Legislation	49
4.2.1.2	Political Implications	53
4.2.2	The Human Rights Case	54
4.2.3	The Welfare Case	56
4.2.4	The Business Case	58
4.3	Global Trends	59
4.4	Summary	61

CHAPTER FIVE: UNIVERSAL DESIGN AND RELATED CONCEPTS

Introduction	62
Design Responses to Counter Exclusion	62
Accessible Design	62
Barrier-Free Design	63
Design-for-All	63
Design for Disability	64
"Extra-Ordinary" Ergonomics	64
Inclusive Design	66
Kyoyo-Hin	67
Transgenerational Design	68
Universal Access	69
Universal Design	70
Principles of Universal Design	73
Universal Design Applications in Education	74
Case Studies on Universal Design in Cape Town	76
Case Study One: UD for Product Design Students	76
Case Study Two: UD for Public Parks	80
The Making of a Legible Space	83
Universal Design and Wayfinding	84
Wayfinding in Context	88
Legible Kirstenbosch	89
Towards More Inclusive Interactivity	96
Implications for Public Parks	98
Case Study Three: UD in a Students' Design Competition	99
Summary on Case Studies	101
Convergence of Concepts	102
Summary	103
	Introduction Design Responses to Counter Exclusion Accessible Design Barrier-Free Design Design-for-All Design for Disability "Extra-Ordinary" Ergonomics Inclusive Design Kyoyo-Hin Transgenerational Design Universal Access Universal Design Principles of Universal Design Universal Design Applications in Education Case Studies on Universal Design in Cape Town Case Study One: UD for Product Design Students Case Study Two: UD for Public Parks The Making of a Legible Space Universal Design and Wayfinding Wayfinding in Context Legible Kirstenbosch Towards More Inclusive Interactivity Implications for Public Parks Case Study Three: UD in a Students' Design Competition Summary on Case Studies Convergence of Concepts Summary

CHAPTER SIX: THEORETICAL, CONCEPTUAL & ANALYTICAL FRAMEWORKS

6.0	Introduction	104
6.1	Activity Theory	104
6.2	Explanatory Legitimacy Theory	108
6.3	Other Concepts Informing this Study	109
6.3.1	Participatory Design	110

6.3.2	Design for Sustainability	110
6.3.3	Ubuntu	113
6.4	Summary	117

CHAPTER SEVEN: RESEARCH METHODOLOGY

7.0	Introduction	118
7.1	Contextualising Research Perspectives	118
7.2	Experimental Design and Sampling Method	121
7.3	Role of Researcher	123
7.4	Research Activities	123
7.4.1	Observational Methods	124
7.4.2	Database Methods	125
7.4.3	Subjective Methods	125
7.4.3.1	Personal Interviews	125
7.4.3.2	Questionnaires	125
7.5	Workshops and Focus Groups	126
7.6	Data Collection Methods	126
7.6.1	Coding Framework	127
7.7	Data Evaluation Methods	131
7.7.1	Testing of Pictograms	131
7.7.2	Case Studies and Exemplars	131
7.7.3	Students' Design Competition	132
7.8	Ethical Considerations	132
7.8.1	Ethical Issues Relating to Informants	132
7.8.2	Ethical Issues Relating to the Researcher	133
7.8.3	Ethical Issues Relating to the Sponsoring Organisation	133
7.9	Summary	134

CHAPTER EIGHT: DATA ANALYSIS AND DISCUSSION OF FINDINGS

8.0	Introduction	135
8.1	Identifying Universal Design Exemplars	135
8.2	Roles of the Principal Actors	135
8.2.1	Activity Analysis: UD in Academic Settings	136
8.2.2	Activity Analysis: UD in Public Parks	141
8.2.3	Activity Analysis: UD in the Public Domain	148
8.2.4	Thematic Analysis: Institutional Architect's Perspective	154
8.2.5	Thematic Analysis: Design Students' Perspective	156
8.2.6	Thematic Analysis: Indigenous Knowledge Expert's Perspective	160
8.2.7	Thematic Analysis: Politician's Perspective	163
8.2.8	Thematic Analysis: Consolidated Perspective	166
8.2.9	Findings from Student Survey	167
8.3	Implications for Principal Actors	185
8.3.1	Implications for the City of Cape Town	185
8.3.2	Implications for Professional Designers	187
8.3.3	Implications for Design Educators	193
8.4	Possible Champions for Universal Design	203
8.5	Mainstreaming of Universal Design in Cape Town	206
8.6	Summary	207

CHAPTER NINE: CONCLUSIONS AND RECOMMENDATIONS

9.0	Introduction	208
9.1	Revisiting Aims of the Research	208
9.2	Conclusions Drawn	208
9.3	General Conclusions and Proposed Strategies	210
9.4	Contributions to Knowledge	213
9.5	Limitations of Research	214
9.6	Implications for Further Research	215
9.7	Summary	216

REFERENCES

218

LIST OF FIGURES

Figure 2.1:	Logo of FIFA 2010 World Cup™	9
Figure 2.2:	Zakumi – the official 2010 mascot	11
Figure 2.3:	2010 logos for Cape Town and South Africa	14
Figure 2.4:	Central Cape Town's IRT hub	16
Figure 2.5:	Cape Town's complete IRT system – Phases 1-4	17
Figure 2.6:	Map of Cape Town showing major highways linking the airport and the stadium	18
Figure 2.7:	Map of central Cape Town showing the route of the proposed 2010 Fan Mile	19
Figure 2.8	Detailed map of central Cape Town showing the main 2010 FanFest venue	20
Figure 2.9: Figure 2.10a	Map showing the scale of development around Greenpoint Stadium Ongoing upgrading of transportation infrastructure – before	22
	construction	23
Figure 2.10b	Ongoing upgrading of transportation infrastructure – after completion	23
Figure 2.11:	Panoramic view of the completed Greenpoint Stadium	25
Figure 3.1:	Population pyramids for South Africa – 2000; 2025; & 2050	34
Figure 3.2:	Distribution of total South African population by age group and sex	36
Figure 3.3:	Population distribution by ethnic background in South Africa	40
Figure 3.4:	Population pyramids of the African and Coloured populations	41
Figure 3.5:	Population pyramids of the Indian/Asian and White populations	42
Figure 5.1:	The Inclusive Design Cube	67
Figure 5.2:	Kyoyo-Hin and Kyoyo Services	67
Figure 5.3:	The Third Age Suit	69
Figure 5.4:	The Universal Design Pyramid	71
Figure 5.5:	Paraffin container	79
Figure 5.6	Universal bathtub	80
Figure 5.7	Paths (intersecting)	85
Figure 5.8	Landmark (mountain) and node (sculpture)	86
Figure 5.9	Directional signage at nodal intersection	86
Figure 5.10	Destination signage (in green)	87
Figure 5.11	Situational warning signage (in red)	87
Figure 5.12	My Guide™ handset in use	90
Figure 5.13	Beacon for My Guide™ with cardinal direction	90
Figure 5.14	Braille Trail	90
Figure 5.15	Fragrance Garden	91
Figure 5.16	Principle 1: Equitable Use	92
Figure 5.17	Principle 2: Flexibility in Use	93
Figure 5.18	Principle 3: Simple and Intuitive Use	93

Figure 5.19	Principle 4: Perceptible Information	94
Figure 5.20	Principle 5: Tolerance for Error	94
Figure 5.21	Principle 6: Low Physical Effort	95
Figure 5.22	Principle 7: Size and Space for Approach and Use	95
Figure 5.23	Sample of submission for 2010 Students' Design Competition	101
Figure 6.1	1 st Generation Activity Theory	104
Figure 6.2	2 nd Generation Activity Theory	106
Figure 6.3	Elements of a work activity using an ActAD lens	107
Figure 7.1	The overall research design	119
Figure 8.1:	Actors within the educational environment	136
Figure 8.2:	Group activities within the educational environment	137
Figure 8.3:	Organisational level activities within the educational environment	138
Figure 8.4:	Societal level activities within the educational environment	138
Figure 8.5:	The educational environment: an overview	139
Figure 8.6:	Activity analysis of the educational environment	140
Figure 8.7:	Actors within the botanical gardens environment	142
Figure 8.8:	Group activities within the botanical gardens environment	143
Figure 8.9:	Organisational level activities within the botanical gardens	
	environment	144
Figure 8.10:	Societal level activities within the botanical gardens environment	145
Figure 8.11:	The botanical gardens environment: an overview	146
Figure 8.12:	Activity analysis of the botanical gardens environment	147
Figure 8.13:	Actors within the students' competition domain	148
Figure 8.14:	Group activities within the students' competition domain	149
Figure 8.15:	Organisational level activities within the students' competition	
	domain	150
Figure 8.16:	Societal level activities within the students' competition domain	151
Figure 8.17:	The students' competition: an overview	152
Figure 8.18:	Activity analysis: the students' competition	153
Figure 8.19:	Thematic analysis: an architect's perspective	155
Figure 8.20:	Thematic analysis: a design students' perspective	156
Figure 8.21:	Thematic analysis: an IKS expert's perspective	161
Figure 8.22:	Thematic analysis: a politician's perspective	164
Figure 8.23:	Thematic analysis: a comparative view	167
Figure 8.24:	Familiarity with "extra-ordinary" population groups	168
Figure 8.25:	Whether the informants had UD-related training during the	
	previous six months	169
Figure 8.26:	UD-related concepts that were most familiar	170
Figure 8.27:	UD-related concepts that were best understood	170
Figure 8.28:	Recognition of Principles of UD pictograms	172
Figure 8.29:	Whether UD is used extensively in the informant's design process	173
Figure 8.30:	Principles of UD that are most often used	174
Figure 8.31:	Reasons for non-use of Principles of UD	175
Figure 8.32:	Likelihood of using information on UD should the same be availed	1/6
Figure 8.33:	Preferred format for availing information on UD	176
Figure 8.34:	Actors perceived to be the most effective champions of UD	1//
Figure 8.35:	Degree of satisfaction with current levels of accessibility within	470
E laura 0.00	built environments, products, and/or systems used most often	178
Figure 8.36:	Perception of the likelihood of UD as adding to the total cost of a	470
Eiguro 9 27	design process/project	170
Figure 0.37.	Derection on the importance of designers engaging in socially	179
1 iyule 0.30	responsible design to counter various forms of discrimination	120
Figure 8 30-	Degree of confidence that informant will be practicing design in	100
i igule 0.39.	South Africa beyond the 2010 FIFA World Cup \mathbb{M}	180
Figure 8 40.	Degree of agreement that <i>ubuntu</i> is useful to design practice in	
9410 01401	South Africa	181

Figure 8.41:	Likelihood of greater adoption of UD should its value be showcased in 2010 projects	181
Figure 8.42:	Likelihood of compliance should implementation of UD in	400
Figuro 8 /3-	projects be mandated by law	182
Figure 0.45.	all design-related disciplines at university level	183
Figure 8.44:	Likelihood of implementing UD in projects if incentives were	
	available to reward designers	183
Figure 8.45:	Paralympian swimmer Natalie du Toit	204
Figure 8.46:	Paralympian sprinter Oscar Pistorius	205
Figure 9.1:	Summary of activities: the educational environment	210
Figure 9.2:	Summary of activities: the botanical gardens environment	211
Figure 9.3:	Summary of activities: the students' competition	212

LIST OF TABLES

Table 2.1:	FIFA World Cup™ facts and figures	8
Table 2.2:	Match schedule for 2010 World Cup™ Host Cities	13
Table 3.1:	Prevalence (%) of disability in South Africa by province	28
Table 3.2:	Prevalence (%) of disability in South Africa by ethnic background	29
Table 3.3:	Number of elderly people in the SADC region	35
Table 3.4:	Number of young people in the SADC region	37
Table 3.5:	Key terms used with respect to tourism	38
Table 4.1:	Conceptual dichotomy between medical and social perspectives	
	on disability	46
Table 4.2:	Contrasting medical, social and bio-social models of disability	48
Table 4.3:	Definitions and Interpretations of Terms used in the Promotion of	
	Equality and Prevention of Unfair Discrimination Act	50
Table 4.4	Cost of orphan care in South Africa (in Rands)	57
Table 5.1:	"Extra-ordinary" trends and changes in functioning	65
Table 5.2:	Comparison of definitions of UD and related concepts	72
Table 5.3:	Comparison of Principles of UD and Principles for Kyoyo-Hin	74
Table 6.1	Leont'ev's Three-Level Model	105
Table 6.2	Elements of an activity system	105
Table 6.3	Elements of Explanatory Legitimacy Theory	109
Table 6.4	Characteristics of social capital	116
Table 7.1	Delineation of paradigmatic assumptions and perspectives	119
Table 7.2	Experimental procedure	122
Table 7.3	Sample population and selection	126
Table 7.4	Thematic coding framework	128
Table 8.1	Summary of emerging themes based on coding framework	154

APPENDICES

254

Appendix A:	Disability Definitions, Models and Terminology	254
Appendix B:	Principles of Universal Design (Poster)	256
Appendix C:	International Universal Design Declaration	257
Appendix D:	Universal Design Audit Checklist	258
Appendix E:	Sample of Informed Consent Form	259
Appendix F:	Enrolments for Faculty of Informatics & Design	260
Appendix G:	Judging Criteria and Weighting for 2010 Green Goal Mouille Point	
	Promenade & Beachfront Student Landscape Design Competition	260
Appendix H:	Questionnaire for Institutional Architect	261
Appendix I:	Questionnaire for Focus Group with Interior Design Students	262

Questionnaire for Indigenous Knowledge Expert	263
Questionnaire for Politician	264
Questionnaire for Survey of Design Students	265
Questionnaire for Design Educators	270
Questionnaire for Professional Designers	272
Summary of Responses from UD Workshop at DEFSA 2007	
Conference	274
Sample SPSS Outputs	278
	Questionnaire for Indigenous Knowledge Expert Questionnaire for Politician Questionnaire for Survey of Design Students Questionnaire for Design Educators Questionnaire for Professional Designers Summary of Responses from UD Workshop at DEFSA 2007 Conference Sample SPSS Outputs

GLOSSARY

Terms/Acronyms/Abbreviations	Definition/Explanation
Accessibility	the extent to which a product, system or environment can be used by a variety of people, especially those with disabilities.
Accessible Design	design focussed on the principles of extending standard design to people with some form of performance limitation to maximise the number of potential customers who can readily use a product, building or service (ISO, 2001:2).
Activities of Daily Living (ADLs)	a set of six basic activities (bathing, dressing, toileting, transferring, continence, and feeding) that a person can perform in the process of self-care.
Activity Theory (AT)	a theory attributed initially to the Soviet psychologist, Lev Semyonovich Vygotsky and his colleagues. AT was further developed by other scholars, including Alexei Nikolaevich Leont'ev and later by Yrjö Engeström used in the analysis of various dynamics in human activities.
ADA	The Americans with Disabilities Act of 1990 (Public Law 101-336 passed on 26 th July 1990).
ADPD	(Secretariat of) the African Decade of People with Disabilities; based in Cape Town, South Africa.
Adaptable Design	design that can be changed to fit the user, rather than requiring the user to adapt.
Adaptive Technology	the use of technology to assist individuals who have difficulty accessing systems, products or environments using conventional methods.
Affordances	a concept from ecological psychology based on the paradigm of direct perception. They are specific combinations of the properties of substances and surfaces taken with reference to an observer (Raubal, 2007:1).
African Renaissance	a movement aimed at promoting the revival of an African philosophy and way of life called "Ubuntu" (humanness) as espoused by NEPAD and the AU.
Ageing	a continuous process of progressive change in all structures and functions of the body: the impact of such changes on a person's quality of life is largely dependent on the social and cultural milieu (WHO).
Aids	Acquired Immunodeficiency Syndrome; the most severe manifestation of infection with the Human Immunodeficiency Virus (HIV).

Anthropometry / Anthropometrics	the field that deals with the physical dimensions, proportions, and composition of the human body, as well as the study of related variables that affects them.
Assistive Technology / Assistive Device	piece of equipment, product system, hardware, software or service that is used to increase, maintain or improve functional capabilities of individuals with disabilities (ISO, 2001:2).
AU	African Union (formerly known as the Organisation of African Unity – OAU) composed of fifty-three states in Africa and modelled after the European Union (EU).
Barrier-Free Design	design that seeks to eliminate physical barriers to people with disabilities; often used synonymously with physical "accessibility" in built environments.
BRT	Bus Rapid Transit – a high quality bus-based system that delivers fast, comfortable, and cost-effective urban mobility through the provision of segregated right-of-way infrastructure, rapid and frequent operations, and excellence in marketing and customer services (Wright & Hook, 2007:11).
CoE	Centre(s) of Excellence. A CoE (supported by the DST and NRF) can be conceptualised as a co-ordinating body or organisation that provides strategic oversight and service in developing sustained human resource capital in South Africa (DST, n.d.; 2003; 2004).
Consilience	refers to the joining together of knowledge and information in multi-disciplinary across disciplines and cross-domain contexts to create a unified framework of understanding – it "offers a powerful mechanism for borrowing from other disciplines, thereby extending the scope of what can be known" (Dalcher, 2006:253).
DALY	one of the ways of assessing the impact of Aids other than through the number of deaths is to look at the burden of the disease as d isability a djusted l ife y ears (DALYs) (Barnett & Whiteside, 2002:282).
DDA	Disability Discrimination Act (of Australia and the United Kingdom).
DEAT	Department of Environmental Affairs and Tourism (South Africa).
Dependency Ratio	the ratio of the economically dependent part of the population to the productive part; the ratio is computed by dividing the number of people who are most likely to be dependent (the young: those under the age of 15, plus the elderly: those aged 65 and older) by the number of people in the working-aged population (ages 15 through 64) expressed as a percentage.

Design for Development	a concept promoted by the SABS Design Institute that recognises, accredits and awards cost-effective products that demonstrate developmental potential and real quality of life improvement in the lives of the people for whom the said products were designed.
Design for Disability	design philosophy and practice that specifically seeks to address the needs of people with disability.
Design for Sustainability (DfS)	a design practice, education and research that in one way or another, contributes to sustainable development (Vezzoli, 2007:39).
<i>Designing for the Broader</i> <i>Average</i>	Maria Benzton and Sven-Erich Juhlins of Ergonomi Design Gruppen have demonstrated that by applying the concept of the "pyramid of needs" it is possible to design for a broader average range of needs, including the special needs of specific groups within the population, resulting in products which offer a combination of performance, functionality and aesthetic appeal (Coleman, 1999).
Design Practitioners	a generic term that includes design educators, students of design, professional designers, and planners of design- specific policies. The professionals in this categorisation include Architects (and Architectural Technologists), Designers (Graphic/Information, Industrial/Product, and Interior), and Town and Regional Planners.
Disability	a physical or mental impairment that substantially limits one or more of the major life activities of such individual, a record of such an impairment, or being regarded as having such an impairment (WHO; Americans with Disabilities Act of 1990).
DPSA	Disabled People South Africa.
DSD	Department of Social Development (South Africa).
DST	Department of Science and Technology (South Africa).
DTI	Department of Trade and Industry (South Africa).
(The) Elderly	those members of the general population aged 65 and older; usually assumed to be dependents. Also referred to respectfully as Senior Citizens.
Emancipation	being freed; the condition or fact of being set free or freed from some restriction (Microsoft® Encarta®, 2009).
Empowerment	feeling of having the right to make one's own choices, and having the ability to act on them (ibid).
Ergonomics/Human Factors	the science of work: of the people who do it and the ways in which it is done; the tools and equipment they use, the places they work in, and the psychosocial aspects of the working situation (Pheasant, 1996:4).

"Extra-Ordinary" Ergonomics	design criteria responsive to physical, mental and sensory characteristics of human users; takes into account the needs of people with disabilities, as well as those experiencing the effects of natural processes such as ageing and pregnancy (Kroemer, 2006).
FIFA	a French acronym 'Fédération Internationale de Football Association' translated into English to read: "International Federation of Association Football".
Graying Markets/Products	the market segment or products relating to the elderly.
Handicap	as defined by the WHO: disadvantage(s) experienced by the individual as a result of impairment and disabilities.
HEI	Higher Education Institution – includes all tertiary level academic institutions (e.g. universities of technology).
ΗΙV	the virus that progressively weakens the immune system, ultimately leading to Aids.
Human Development Index	the HDI is a summary measure of human development. It measures the average achievements in a country in three basic dimensions of human development: a long and healthy life, as measured by life expectancy at birth; knowledge, as measured by the adult literacy rate (with two-thirds weight) and the combined primary, secondary and tertiary gross enrolment ratio (with one-third weight), and; a decent standard of living, as measured by GDP per capita (PPP US\$). These socio-economic indicators form the basis of the relative rankings of specific countries as published in Human Development Reports by the UNDP.
ICF	International Classification of Functioning, Disability and Health. In 2001, this replaced the previous WHO 'International Classification of Impairments, Disabilities and Handicaps (ICIDH)' of 1980 (Pasha & Pasha, 2006:4).
Impairment	as defined by the World Health Organization: any loss or abnormality of psychological, physiological, or anatomical structure or function; any loss of function resulting from injury or disease.
Inclusive Design	the design of mainstream products and/or services that are accessible to, and usable by, as many people as reasonably possible on a global basis, in a wide variety of situations and to the greatest extent possible without the need for special adaptation or specialised design; sometimes used as a synonym for Universal Design.
IRT	Integrated Rapid Transit system.
Life Expectancy	the average number of years a person can be expected to live under current mortality conditions.

(The Principle of) The Limiting User	the limiting user is that hypothetical member of the user population who, by virtue of his or her physical (or mental) characteristics, imposes the most severe constraint on the design of the artefact. In clearance problems the bulky person is the limiting user; in reach problems the small person is the limiting user (Pheasant, 1998:23).
LOC	the Local O rganising C ommittee – the national body responsible for organising the 2010 FIFA World Cup™ events locally.
Mainstream	main current of thought or behaviour; the ideas, actions, and values that are most widely accepted by a group or society, e.g. in politics, fashion, or music (Microsoft® Encarta®, 2009).
Mega Events	international events on a grand scale such as the Olympic Games and World Cup™ typically involving a large number of participants and spectators over a period of several weeks.
MDS	Minimum Data Set: a common set of data items, definitions and standards that should be used to collect and report data. These data should be comparable across geographic regions within the continent and over time (Kowal, Wolfson & Dowd, 2000:2).
<i>Millennium Development Goals</i>	the MDGs commit the international community to an expanded vision of development, one that vigorously promotes human development as the key to sustaining social and economic progress in all countries, and recognises the importance of creating a global partnership for development. The goals have been commonly accepted as a framework for measuring development progress and promotion of gender equality and empowerment of women, among other issues.
Mortality Rate	the number of deaths in a group of people, usually expressed as deaths per thousand.
NEPAD	The New Partnership for Africa's Development.
NRF	The National Research Foundation (South Africa).
The Owl Mark™	a quality mark developed by the Centre for Applied Gerontology (at the University of Birmingham, UK) that recognises and accredits good products for older people (Coleman, 1999).
Participation	the freedom of taking part in any socially acceptable and desirable activity thus promoting (within the participant) a sense of belonging and ownership.
Participatory Design	a design approach that actively incorporates the contribution and participation of (potential) end-users in the design process (Lewis, 2000).

Percentiles	percentile values (as used in normal or Gaussian distribution to depict anthropometric data) are typically more useful than the simpler arithmetic mean in showing the range of a specific dimension and may be used to indicate the number of measurements represented in a given population. For example, a 5 th percentile in stature would mean that 'exactly 5% of people are shorter than this' or 'there is only a one-in-twenty chance of encountering a person shorter than this' (Pheasant, 1996:17).
PVA	Public Viewing Area(s).
QALY	one of the ways health economists attempt to measure both quantity and 'quality of life' known as q uality a djusted life y ears (QALYs) (Barnett & Whiteside, 2002:282).
'Reasonable accommodation'	measures mandated by law to be effected in the planning, design and construction of built environments to accommodate people with disabilities with respect to physical accessibility.
Rehabilitation Design	an innovative approach developed by David Guynes and Patricia Moore to the design of health-care and rehabilitation facilities which focus on encouraging independence and an early return to the community as opposed to long-term institutional care (Coleman, 1999).
Retrofitting	the (post-production) practice of fitting or customising products/environments to suit specific applications.
SACN	South African Cities Network.
SADC	the Southern African Development Community; an economic grouping of 15 southern African states including Angola, Botswana, the Democratic Republic of Congo (DRC), Lesotho, Madagascar, Malawi, Mauritius, Mozambique, Namibia, Tanzania, Seychelles, South Africa, Swaziland, Zambia, and Zimbabwe.
Sports Tourism	refers to different sports that throughout history have been the source and/or the principal reason for travel – relative in distance according to the lifestyle epochs of history (Zauhar, 2004:6).
Sustainable Product Design	a design activity that takes into consideration and tries to balance the environmental, social and economic factors in creation of products.
тсс	Technical Coordinating Committee.
Third Age	synonymous with late-adulthood – the period in life when one is classified as being elderly (\geq 65 years). The First Age is childhood (\leq 14 years); whilst the Second Age is early through mid-adulthood (15-64 years).

Through Other Eyes	concept developed by Patricia Moore who in her twenties spent three years travelling in over 200 cities throughout US and Canada experiencing life as an 85 year old woman (Coleman, 1999).
Transgenerational Design	attributed to James Pirkl et al., this is an approach to design that specifically considers the mental, physical and sensory changes in ability that occur with age; also known as Lifespan Design (Coleman, 1999).
Ubuntu	a traditional African ideal that finds expression through mutually reaffirming communal interaction, mutual support, group solidarity and humanness.
UNESCO	United Nations Educational, Scientific and Cultural Organization.
Universal Design (UD)	addresses design in the context of needs for all people – of all ages and abilities; associated with Design for All, Design for the Broader Average, Design for Disability, Inclusive Design, Kyoyo-Hin, and Transgenerational Design (Coleman, 1999).
Usability	the extent to which a product or system is can be used by specified users to achieve specified goals with effectiveness, efficiency and satisfaction in a specified context of use (ISO 9241-11 in Baber, 2005:359).
User-Centred Design	the design of products that can be used by specific users to achieve desired goals with effectiveness, efficiency and satisfaction in their given contexts of use. It is participative; system orientated; and acknowledges human diversity (Pheasant, 1996:13).
Vulnerable Population(s)	includes the elderly, orphans, and vulnerable children; any category of the general population susceptible to disability, physical limitation, discrimination or economic disadvantage.
Wayfinding	a system that welcomes, orients and guides visitors in an around a community (Vrooman, 2007:9). Also defined as behaviour that is the purposeful, directed, and motivated movement from an origin to a specific distant destination that cannot be directly perceived by the traveller. It involves interaction between the wayfinder and the environment (Raubal, 2007:1).
WHO	World Health Organization.
(The) Young	those members of the general population aged 14 and younger; usually assumed to be dependants.
ZAR	South African Rands – the official currency of the country.

CHAPTER ONE WHAT THE STUDY IS ABOUT

1.0 Introduction

This thesis addresses the challenges of mainstreaming *Universal Design* (UD) in a developing context from the perspective of activities and projects associated with the hosting of the greatly anticipated *FIFA 2010 Soccer World Cup*TM in *Cape Town*. It will be the first time in FIFA's history for the sporting mega-event to be hosted on the African continent, and consequently South African host cities are mobilising significant capital (human and financial) to deliver a memorable experience for their visitors and residents alike. As expected, tensions arise in an emerging economy's need to deliver world-class facilities, against the existing stark social and developmental challenges that demand urgent redress. The vexing problem of engaging with global imperatives against pressing local realities is a recurring theme in this thesis.

In South Africa, the apartheid legacy of institutional exclusion and the concomitant ills of discrimination and stigmatisation adversely affected disabled people, non-whites, and women. The post-apartheid dispensation has marshalled a host of novel concepts to effect positive change in this regard. However, certain structural (or systemic) and functional (or attitudinal) obstacles remain for the ideals of access, social inclusion, and civic participation to find fuller and more sustainable expression (SAMA, 2006). It is against this backdrop that the potential role of Universal Design is interrogated based on the belief that UD offers a singularly appropriate platform for designers (students, educators and professionals) and policy makers to adequately engage the society in formulating more inclusive and equitable solutions for industrial design products, systems and built environments.

UD as a concept has its roots in the developed world as a response to developments therein. This thesis investigates the applicability of such a global concept in an African city. Further, in view of FIFA's stringent criteria for infrastructural and logistical compliance in host cities, and in the role of such a powerful international actor and its resultant impact on local affairs, particularly when viewed from a uniquely African perspective. The thesis adopts an Activity Theory analytical lens to propose adjustable strategies for the appropriate and sustainable mainstreaming of UD amongst designers, policy makers and other key actors in the city of Cape Town. Ultimately the emerging strategies could be adapted to fit developing contexts in South Africa and beyond.

1

1.1 Motivation for Choosing this Topic

The motivation for choosing this topic stems from the realisation of the historic import of hosting the FIFA 2010 World CupTM in South Africa. Further, the author has been offering a Universal Design module in the Department of Industrial Design, CPUT since 2005. The opportunity thus presented itself for an action research-inspired project at CPUT to evolve into a more comprehensive study that could potentially inform plans, policies, strategies, and services for the city of Cape Town. The benefits are two-fold: the possible expansion of the UD module to reach a larger number of design students; and the potential of enabling UD to influence/embrace a wider constituency of residents (and visitors) in the city of Cape Town. There is an intrinsic mutual relationship between CPUT and the City of Cape Town through employment opportunities for the students of the former, as well as joint research activities and consultancy projects between the two organisations. Further, this research is seen as a strategic and functional contribution towards the stated objectives of the *African Decade for People with Disability* (ADPD) which was initially set to end in 2009 (Phiri, 2006).

1.2 The Research Problem in Context

The main research problem revolves around the question of whether there are appropriate strategies for promoting the adoption of UD-related applications in Cape Town's plans for cohosting matches during the FIFA Soccer World Cup^{TM} in 2010. The fact that UD-related concepts emerged from industrialised countries necessitates interrogation of existing and anticipated strategies of UD promotion from the unique perspective of a city in Africa (such as Cape Town).

1.2.1 Hypothesis of the Study

In setting the objectives of this study, two tentative hypotheses have been formulated:

i) there is a relationship between the adoption of UD strategies and the inclusiveness (in terms of accessibility and usability) of the products, services, systems and built environments thus designed; and

ii) there are unique factors in Cape Town that present challenges to the effective adoption and implementation of UD strategies there – these challenges necessitate a dedicated response.

1.2.2 Basic Assumptions

At the philosophical level, this study assumes that the extent to which accessibility and usability of products and built environments is enjoyed by various cohorts of users (especially by members of vulnerable or special populations) depends on the degree to which appropriate UD strategies would have been incorporated in their design philosophy and praxis. On a more practical level, the study interrogates the relevance of the construction and completion of the Greenpoint Stadium as well as other projects and activities associated with the FIFA 2010 Soccer World Cup[™] vis-à-vis the potential for mainstreaming UD in Cape Town.

1.3 Research Question

The main research question and its sub-questions address the following: How can the concept of Universal Design be promoted in Cape Town in the context of the FIFA 2010 Soccer World Cup^{TM} projects and related activities?

- What kind of adjustable strategies are needed to implement the ideals of UD in an emerging economy context?
- What kinds of UD-related role players are there in South Africa? What UD-related policies are there in the country? What resources does the country have to support UD applications?
- Why has UD not been adopted by planners and professional designers on any significant scale previously?

1.4 Research Objectives

The overall objective of this study is to propose strategies for effectively mainstreaming UD considerations amongst design practitioners in an emerging economy context.

To achieve this purpose, the specific objectives have been addressed by doing the following:

- Evaluating the current level of UD awareness among selected designers and planners in Cape Town, especially in the light of FIFA 2010 World Cup[™]-related activities within the city;
- Assessing the applicability of UD strategies that have evolved outside the context of study;
- Proposing appropriate sustainable and adjustable strategies for mainstreaming UD applications in Cape Town with respect to the 2010 sporting mega-event; and
- Contributing to the strategic goals of the African Decade for Persons with Disabilities (1999-2009).

1.5 Significance of the Research

The concept of UD promotes inclusion of all people of all abilities, and consequently considers the breadth of human diversity across lifespan to create design solutions that work for all possible users. This in principle is a noble ideal. In reality however, one finds that a significant number of individuals in Africa in general, and in South Africa in particular, are patently disadvantaged (or even disabled) on account of poorly designed products and inaccessible built environments. This has exacerbated the stigma associated with various vulnerable populations, as well as perpetuating discrimination by excluding a significant number of potential users and consumers. UD has been formally adopted in a number of countries elsewhere in the world. Notwithstanding, there is a dearth of research on UD-related issues on our continent. This study seeks to contribute to such knowledge by investigating the applicability of UD in an emerging economy context. In this regard, it was necessary to examine design strategies from other parts of the world to assess their practicability and applicability to industrially developing (or majority world) contexts (Schumacher, 1973; Shapira, 1991; Parker, 2001; Sandhu, 2002).

South Africa is second only to India in the number of official languages. Though well-intended, the complexity of having 11 official spoken (and written) languages, as well as South African Sign Language (Howell, Chalken & Alberts, 2006:69) is itself a daunting reality from a practical point of view – this is further compounded by the needs of the multitude of foreign visitors and sports tourists expected in major South African cities once the FIFA 2010 World Cup™ mega event unfolds. The urgent need for effective and transparent wayfinding systems, to name one challenge, cannot be over-emphasised.

Further, the history of systemic oppression under *apartheid* provides a unique backdrop to addressing what is potentially an emancipatory vehicle for social inclusion. The status of disabled persons in this country enjoys an enviable measure of political support (Matsebula, Schneider & Watermeyer, 2006). However, the same cannot be said of the plight of other vulnerable/special populations such as the elderly, women, children and foreigners (who are unfamiliar with the local languages and socio-cultural norms). Proposed UD strategies would effectively provide mechanisms that create and sustain an enabling environment of equity for all denizens.

The research study focuses on evolving a strategy for mainstreaming UD applications with reference to the 19th FIFA 2010 World Cup[™] which is billed to be the biggest event yet on

African soil. It is expected that by benchmarking the Host City of Cape Town as a best practice exemplar of UD implementation, other cities on the continent would benefit from lessons learnt. Whereas the Olympics, Paralympics and Commonwealth events "have the most specific universal accessibility requirements", there is a dearth of similar information on specifications for accessibility vis-à-vis the FIFA soccer event (Swart, Wang, Turco,Vollmer, Knott, Parish & McCrae, 2008:8). Through "programmatic accommodation" Koncelik (1998:122) reports that Atlanta, Georgia effectively positioned itself as an accessible city after the 1996 Olympics and Paralympics. Similarly, Munich was linked with Cape Town for "promotional as well as learning purposes" (City of Cape Town, 2007:48). The anticipated mega event offers a historic opportunity as a case worthy of study and suffices as motivation for interrogating the impact of design intervention in the process of introducing new technical systems (Eason, 2005).

The proposed strategies could be readily adapted to similar realities on the continent, and further afield where, as Margolin (2007:115) challenges, *design for development* interventions should be targeted sensitively and "in the most effective ways" so as not to appear patronising or condescending towards local aspirations. Such strategies would promote collaboration and support peer review mechanisms among member states under the aegis of continental organs such as the *New Partnership for Africa's Development* (NEPAD) and the *African Decade of People with Disabilities* (Chalken, Swartz & Watermeyer, 2006:93). To this end, Schneider (2006:17) identifies the need to develop "policies that abide by principles of Universal Design" as a priority for ensuring inclusiveness.

1.6 Structure of the Thesis

The initial chapters of this thesis introduce the concept of Universal Design (UD), as well as related concepts and developments internationally. Chapter 1 gives an overview of the topic under investigation and covers such specific issues relating to the research focus, context of the study, delimitations, and the significance of the study. Chapter 2 looks at the expected impact of the FIFA 2010 World Cup[™] from an African developmental perspective. A discussion on the peculiar challenges experienced by different categories of vulnerable populations is presented in Chapter 3; whilst pertinent practical and theoretical responses to the needs of the identified populations are elaborated in Chapter 4. Chapter 5 examines the concept of UD, and its multi-disciplinary and cross-domain applications in diverse contexts. Chapter 6 discusses the theoretical, conceptual and analytical frameworks used in this study. Chapter 7 explains the rationale for the qualitative approach to the research questions and why the experimental

5

research method was adopted. Chapter 8 analyses how the data were interpreted using the developed thematic coding framework as well as presenting various data graphically – the chapter also discusses possible implications for a host of actors within the city of Cape Town. Finally, Chapter 9 summarises the main findings, conclusions and recommendations of the research study, and examines their significance to Cape Town's unique context as well as to other emerging economies. It also outlines the study's contribution to knowledge and finally, proposes future research topics as informed by the study.

1.7 Summary

The concept of Universal Design is a relatively recent one globally. However, there appears to be little research on the topic in an African context. Further, the anticipated FIFA 2010 World Cup[™] has provided benchmarking opportunities for Host Cities, as well as the opportunity to fast-track urban renewal and world-class infrastructural projects. The real value of event is in its catalytic capacity for building a lasting legacy of accessibility and usability for all residents and visitors alike. In alignment with the research question, this study explores the potential benefit of such long-term thinking to the concept of UD.

As stated elsewhere, this research hopes to make a significant contribution to the adoption of UD in related contexts on this continent and beyond. This is done by using Activity Theory as a lens, and in the subsequent proposed adjustable strategies for mainstreaming UD in the city of Cape Town. The fact that the soccer mega-event is being hosted by a country with such a highly publicised history of systemic oppression, discrimination and exclusion offers a set of geopolitical and socio-economic dynamics that are truly unique. These are the very themes that UD interrogates, and will be explored in subsequent chapters of this thesis.

Finally, and most importantly, the Cape Town aspires to be recognised as a world-class Design Capital. The legacy of UD inspired improvements to infrastructure and services would consolidate the city's standing internationally. Further, the resultant enhancement of the city's reputation of greater inclusiveness and accessibility would significantly improve its standing as a premier tourist destination.

CHAPTER TWO

AN AFRICAN PERSPECTIVE ON MEGA-SPORTING EVENTS

2.0 Introduction

This chapter presents an overview on the various socio-economic and political imperatives of hosting an event of the magnitude of the FIFA 2010 World Cup[™]. Whereas many of the issues discussed herein are peripheral to core design practice, the implications must be viewed in the context of the uniqueness of holding such a prestigious event in an African setting. Further, from a technology- and knowledge-transfer point of view, the issues covered are of relevance to many of the projects sponsored by key international actors whose expertise and experience is based exclusively on settings outside of our continent. The chapter concludes with a discussion on the anticipated outcomes and benefits for the Host City of Cape Town.

2.1 Mega-Sporting Events

Bovy (2002:6) states that Europe hosts about 75% of all the World Championships, and further, that there are more than 300 sports mega-events annually. This geopolitical reality formed the basis for South Africa's motivation to host events of such magnitude (Cornelissen & Swart, 2006; Cornelissen, 2009). The nation has the unique distinction of being the only country on the continent to host mega-sporting events of truly global interest. With the exception of the Olympic Games, at the end of 2010, South Africa will have held mega-events for rugby (1995), cricket (2003), and soccer (2010). The successful hosting of the rugby and cricket world cups provided opportunities for nation-building as all South Africans rallied around a common event – a fact often touted by the local organisers of the soccer event (Cornelissen, 2004; 2009).

Arguably, the cost of hosting such events is a prohibitively daunting challenge for most African countries (SAInfo, 2006b; Mafu, 2007; Dentlinger, 2008; du Plessis & Maennig, 2009; van der Merwe, 2009). According to Fick (2006:166) South Africa accounted for 38% of sub-Saharan Africa's gross domestic product in 2004 – the largest of any single country by far. Further, South Africa has the best infrastructure and enjoys the highest standards of financial, communications and information technology-related services. These are prerequisites for hosting any mega-sporting event. The fact that these types of events continue to grow in "overall size as well as in technological and organizational sophistication" (Bovy, 2002:6) impose further logistical challenges for organisers that must meet standards set by FIFA's Technical Coordinating Committee (TCC) (FIFA, 2003; 2004a; 2004b; 2006a; 2006b; 2006c; 2007a; 2007b; 2007c).

7

Additionally, the political goodwill enjoyed by the country in the post-apartheid dispensation has been tremendous (Cornelissen, 2007). Indeed the personal beneficence and support from the FIFA president, Joseph Sepp Blatter has been evident from the outset (Cornelissen, 2005). Blatter's constant reassurances on FIFA's commitment to the 2010 events has been widely publicised in the local media (Cape Argus, 2006; TygerBurger, 2008a). It is clear that the process of compliance for Host Countries is a complex one in view of FIFA's pervasive influence, economic clout, and prerogative over such decisions (SAInfo, 2004; Dentlinger, 2008). This was the case in the selection of the site of *Greenpoint* stadium in spite of the Metropolitan Council of Cape Town's distinct (and duly justified) preference for *Athlone* or *Newlands* for the semi-final game (City of Cape Town, 2007; Dentlinger, 2007b; Swart & Bob, 2009). Host Countries (or Cities) cannot however relinquish their social responsibility to another party, no matter how benign or benevolent the declared intent of the external partner is and must take cognisance of their own broader developmental agenda (Cornelissen, 2007; 2009).

2.2 "The Beautiful Game"

Of all the major sporting events, the month-long FIFA soccer event attracts the largest global audience. As reflected in Table 2.1, the actual attendance at each successive tournament has shown a steady growth over a sixty year period from 1930 to 1990, with a marked spike in the 1994 games held in the United States of America (USA) – this was followed by slight declines in 1998 and 2002. The 2006 attendance at Germany's World Cup[™] was the second highest ever.

Year	Host	Champions	Teams	Games	Goals	Sent-off	Attendance
1930	Uruguay	Uruguay	13	18	70	1	589,300
1934	Italy	Italy	16	17	70	1	361,000
1938	France	Italy	15	18	84	4	376,000
1950	Brazil	Uruguay	13	22	88	0	1,044,763
1954	Switzerland	West Germany	16	26	140	3	872,000
1958	Sweden	Brazil	16	35	126	3	819,402
1962	Chile	Brazil	16	32	89	6	892,812
1966	England	England	16	32	89	5	1,464,944
1970	Mexico	Brazil	16	32	95	0	1,690,890
1974	West Germany	West Germany	16	38	97	5	1,809,953
1978	Argentina	Argentina	16	38	102	3	1,685,602
1982	Spain	Italy	24	52	146	5	2,108,723

Table 2.1: FIFA World Cup facts and figures (source: http://home.netvigator.com)

Year	Host	Champions	Teams	Games	Goals	Sent-off	Attendance
1986	Mexico	Argentina	24	52	132	8	2,393,031
1990	Italy	West Germany	24	52	115	16	2,516,354
1994	USA	Brazil	24	52	141	15	3,587,088
1998	France	France	32	64	171	22	2,775,400
2002	Korea/Japan	Brazil	32	64	161	17	2,705,566
2006	Germany	Italy	32	64	147	28	3,367,000
Total			361	708	2,063	142	31,051,828

Unlike the 28 sports events of a typical Olympic Games, the soccer World Cup[™] tournament focuses exclusively on one sport wherein 32 teams play a total of 64 matches. Another distinction is that whereas the Olympics require dedicated accommodation at an Athletes Village, the soccer event is more relaxed on the issue of proximity with regards to accommodation for the players (Beasley, 1997). This latter factor poses more challenges for transportation logistics, not just for the spectators, but additionally for the players (Beasley & Davies, 2001; Bovy, 2002; 2004; Matheson & Baade, 2004; Davies, 2009).

2.3 A First for Africa

FIFA held its first officially sanctioned tentative competition in Uruguay in 1930 (FIFA, 2006c). In 2010, this will be the first time in the association's modern history that the African continent will play host to the games. Indeed, the 2010 event is billed as 'Africa's' World Cup[™] notwithstanding the fact that the coveted trophy has never been won by an African country. The positioning of the 2010 tournament as an African event is reflected in the campaigns, publicity and branding promoted by both the South African *Local Organising Committee* (LOC) and FIFA – a strategy central to the country's initial bid to host the 2006 games (Cornelissen, 2005; Czeglédy, 2009). In Figure 2.1 the image portrays the stylised map of Africa whilst using the colours of the South Africa flag; further, the word "Africa" is in red for additional emphasis.



Figure 2.1 Logo of FIFA 2010 World Cup™ (source: <u>www.FIFA.com</u>) Though neighbouring countries within the Southern Africa Development Community (SADC) region will provide additional training venues for teams, and are presently bolstering their infrastructure to ensure that they benefit from some tourist revenue during the World Cup™, the major economic and socio-technical gains from these games will accrue to South Africa and FIFA (SAInfo, 2006a). For example, Zimbabwe (whose economy is recovering from record hyperinflation) has intensified its own campaign to tap into the anticipated tourism windfall and is using the opportunity for public relations redress (Barnes, 2008; Tomlinson, Bass & Pillay, 2009). Granted that South African taxpayers will be paying for the infrastructural developments to successfully host the sports spectacular, it could be argued that the country is entitled to benefiting more from the event than any of its neighbours. In so doing, South Africa will consolidate its political and economic position as the pivotal state within the entire region (Ezeoha & Uche, 2005; Fick, 2006; Cornelissen & Swart, 2006; Daniel, Lutchman, & Comninos, 2007; Czeglédy, 2009; van der Merwe, 2009). In extolling the virtues of The New Partnership for Africa's Development (NEPAD), and as a major proponent of the African Renaissance and a champion of the anthropocentric ideals of ubuntu, South Africa's will be benchmarked by its neighbours for its oft publicised moral aspirations (South Africa, n.d; Mazrui, 1999; Mbigi, 2000; Ibhawoh & Dibua, 2003; OSDP, 2003; Creff, 2004; Davies, 2006; M'Rithaa, 2008). Ubuntu (and its relevance to socio-ethical aspirations) is discussed in detail in Section 6.3.3 of this thesis.

Of particular importance for the organisers is the need to keep tournament ticket prices within affordable limits for local fans and those from the rest of Africa (Tomlinson *et al.*, 2009). There are plans to ameliorate this situation by availing free tickets to boost attendance by this category of fans (Kortjass, 2007c; Swart *et al.*, 2009). Sub-Saharan Africa has the unfortunate distinction of having some of the poorest and most oppressed communities on the planet (Gibson, 2004). The region faces often daunting challenges of *poverty, instability, disease, illiteracy* and *corruption* as identified by Pascal Eze (cited in Jere-Malanda, 2008:1). This view often promotes a sense of Afro-pessimism; or "Africa bashing" as described by Frederick Cooper (1997:189, cited in Ibhawoh & Dibua, 2003:61).

The most sustainable long-term benefit for the continent would lie more in the socio-technical (and not necessarily in the economic or political) domain (Wilkinson, 2008; Tomlinson, 2009). The potential of soccer as a unifier of disparate societies cannot be over-emphasised – this despite the corporate elitist perception of the World CupTM and the fact that the event patently precludes the participation of *female* athletes (Rubin, 2009). Additionally, the mega-events are

10

situated in urban settings though some benefits in the form of increased tourism and significantly more bookings for accommodation are expected to trickle down to rural and other non-host communities (Atkinson, 2009). Significantly, the vast majority of the continent's 840 million inhabitants are categorised as youth. Notwithstanding, soccer, particularly in the professional sense has had a unique impact on a number of African societies:

- Promising African players (principally from West Africa) have immigrated to more developed countries to advance their careers there – this is one of the most striking effects of globalisation on Africa as described by Akindele, Gidado and Olaopo (2002);
- The extra-ordinarily high wages paid to professional players (particularly in Europe) has directly benefited the communities of the African players in their homelands (for example the role of George Weah with respect to his country Liberia); and
- This commercialisation of sports has created new role models (especially among male youth) – soccer could be promoted as a tool for countering social maladies amongst the youth (such as idleness, juvenile delinquency and related anti-social behaviour) that has adversely affected myriad families on the continent (Takyi & Oheneba-Sakyi, 2006).

The country is using the *FIFA Confederations Cup*TM South Africa 2009 (to be held from the 14th to the 28th of June 2009) as a dress rehearsal for the 2010 tournament (FIFA, 2009). The performance so far with regards to the marketing and ticket sales appears to be far below expectations. Ticket sales have been dismal with only about 300,000 out of a total of 640,000 (FIFA, 2009). The importance placed on the Confederations CupTM led to the high profile endorsement of *Danny Jordaan* (the Chief Executive Officer of the 2010 LOC) and *Ms Baleke Mbete* (the Deputy President of South Africa at the time) flanked by the FIFA 2010 official mascot *Zakumi* (shown in Figure 2.2) who went on a roadshow campaign to a number of soccer-loving locales – including the famous Soweto Township outside Johannesburg.



Figure 2.2 Zakumi – the official 2010 mascot (source: <u>www.FIFA.com</u>)

2.4 Implications for the Host Country

The global phenomenon of sports tourism is noted by Zauhar (2004:5) as a key economic driver which is of particular importance to a host country. Indeed, as Fridgen (cited in Zauhar, 2004:5) argues, the forces that have shaped tourism have shifted from advances in transport technology to "socio-demographic changes; electronic information and communication systems; more knowledgeable and demanding consumers; [and] de-regulatory market place". The fact that most of the Host Cities have yet to implement even a basic integrated transport system is cause for concern (Makgetla, 2007). The majority of visitors are from the developed world where sophisticated and efficient systems are almost taken for granted (Walsh, 2001; Matheson & Baade, 2004). It is imperative therefore for a Host City to cater for a range dedicated services for tourists whose primary motivation for visiting a country is their love of soccer (Davies, 2006).

The implications of hosting the 2010 sports mega-event in South Africa include:

- The high cost of building (at least 10) FIFA-standard stadia (Tomlinson, 2009);
- The need for reasonable accommodation of an international standard for the tournament teams and sports tourists (Ormerod & Newton, 2005);
- The provision of reliable and integrated public transportation between all major venues;
- The provision of security for the VIPs from the *FIFA Family* (its top officials) against the backdrop of unacceptable (high) levels of crime in the country; and
- The need to redress some of the spatial apartheid-inspired inequalities as is the case in Cape Town (Sherer, 2000; Boraine, 2004; Cornelissen, 2005; 2009; COHRE, 2007).

2.5 2010 and Cape Town

The *City of Cape Town* previously launched an unsuccessful bid to host the *2004 Olympic Games* (Cornelissen, 2005). The lessons learnt for this process informed the national bid for the soccer mega-event first for the 2006 games (in which Germany emerged the winner) and again for the 2010 event. Cape Town enjoys significant goodwill as a city not only because of its natural beauty (with a number of *UNESCO World Heritage Sites* in its vicinity – including the iconic Table Mountain), but also due to the local government leadership's commitment to good governance and issues such as environmental sustainability and socio-economic development (UNESCO, 2005). Consequently, the city was voted "one of ten cities that are most likely to become a global sustainability centre by 2020" (SA Good News, 2008).

According to Andrew Boraine (2004:4), Chairperson of both the *South African Cities Network* (SACN) as well as the *Cape Town Partnership* (CTP), the nine major municipalities of Buffalo City (East London), Cape Town, Ekhurhuleni, eThekwini (Durban), Johannesburg, Mangaung (Bloemfontein), Msunduzi (Pietermaritzburg), Nelson Mandela Metropole (Port Elizabeth) and Tshwane (Pretoria) – all members of the SACN host 37% of the country's population. It is worth noting that of the nine Host Cities for the FIFA 2010 World Cup[™] (Bloemfontein, Cape Town, Durban, Johannesburg, Nelspruit, Polokwane, Port Elizabeth, Pretoria, and Rustenburg), the majority are SACN members (see Table 2.2).



Table2.2: Match schedule for 2010 World Cup™ Host Cities (source www.FIFA.com)

Despite the significant role these cities play in the socio-economic life of the country, the legacy of the apartheid city form still persists in terms "race, space and poverty; fragmentation and urban sprawl; and poor access and mobility" (Boraine, 2004:7). The *Unicity Commission* (2001) recognised this challenge in Cape Town and proposes a comprehensive strategy for developing a sense of unity and inclusiveness for all its residents. Figure 2.3 shows the logo of Host City of Cape Town in association with the national 2010 one – note the stylised representation of the ocean (and waves), the sun, and Table Mountain.



Figure 2.3 2010 logos for Cape Town and South Africa (source: <u>www.FIFA.com</u>)

More specifically, the SACN members envision a future in which the following four key elements interact with regards to their respective city populations:

- PRODUCTIVE CITY wherein the local economy provides a majority of residents with means to earn a reasonable living;
- INCLUSIVE CITY wherein residents have the opportunities and capacities to share equitably in the social benefits of city life;
- SUSTAINABLE CITY wherein the city impacts on the envelope of natural resources that sustains the settlement and makes it liveable; and
- *WELL-GOVERNED CITY* wherein the political and institutional context is stable, open and dynamic enough to accommodate all interests (Boraine, 2004:3; South African Cities Network, 2005; 2008).

Dan Plato replaced Helen Zille as Cape Town's Executive Mayor in May 2009. During her tenure, Zille received the 2008 World Mayor Prize topping a list of 820 mayors worldwide (SouthAfrica.info, 2008). Zille emerged from the 2009 national elections in a far stronger position than in any previous elections since she became leader of the Democratic Alliance (DA) Party, a position she has held since 2007. This is a significant development as it presents opportunities for further consolidation of political power at three levels: at local government through ensuring the appointment of a dynamic successor to the office of the Executive Mayor; at regional government as the incumbent Premier of the Western Cape; and at national level as the Leader of the Official Opposition. Under Zille's leadership, Cape Town advanced a participatory approach to governance by constantly inviting the general public to engage with the City on a variety of developmental issues (Contact Newsletter, 2007b).

Mumford (1979, cited in Sjöberg, 1996:3) identifies three levels of participation. The following is adapted to fit the context of participation vis-à-vis the City of Cape Town:

- consultative participation where certain residents would serve as sources of knowledge, and are consulted on occasion;
- representative participation where residents are selected to represent the residents' views (in an ideal situation the chosen residents are empowered to take part in the guidance of the change process); and
- *consensus participation* where all residents are invited to take part in *all* stages of the change process (wherein a potentially emancipatory change process is engendered).

The participatory governance espoused by the City of Cape Town would at best approximate representative participation; a situation very different from the consensus participatory approach that an exemplary city such Curitiba (in Brazil) follows (Dobbs, 2001; Steinfeld, 2001; Campbell, 2006). The persistent resistance to plans for rezoning the *Greenpoint Common* is instructive. In this particular case, conservationists/representatives of the *Greenpoint Common Association* took the City of Cape Town to court (in early 2007) to challenge the latter's projections on the environmental sustainability of the project, and further questioned the rationale of building a new stadium (as opposed to refurbishing the existing one) in a city where issues such as adequate housing for its 3.5 million residents should have received priority funding (COHRE, 2007; Dentlinger, 2007a). The case was eventually dropped.

Criticism has also been directed at national and regional government departments for the absence of a national 'green' strategy for 2010 (Ross, 2009). To reduce the adverse impact of hosting the tournament, Cape Town launched its own *Green Goals Action Plan* (based on Germany's pioneering effort in 2006) targeting the nine specific areas: *Energy conservation and climate change; Water; Integrated waste management; Transport, mobility and access; Landscaping and biodiversity; Green building and sustainable lifestyles; Responsible tourism; Green Goal communications; and Monitoring and measurement and reporting (City of Cape Town, 2008; Zwane, 2008). Cape Town has consistently been at the vanguard of new policy and legislation, as well as in actual planning of 2010-related activities at local government level.*

More recently, there has been stiff resistance (often resulting in illegal and violent confrontation) by the commuter taxi drivers' associations who fear loss of jobs, income and control should the City proceed to implement its planned *Integrated Rapid Transit* (IRT) system (Steyl, 2008a; TygerBurger, 2008b). The roll-out of the full IRT system is planned in stages, with Phase 1 which will link the central Cape Town hub (Figure 2.4) to the airport expected to be ready in
time for the 2010 event (Steyl, 2008b). The remaining Phases 2, 3 and 4 expected to be completed in 10 to 12 years (City of Cape Town, 2007; 2009; Department of the Premier, 2008).



Figure 2.4 Central Cape Town's main IRT hub (source: City of Cape Town)

McKenzie (2008) and Fourie (2009) express optimism on an amicable solution being found wherein the taxi operators surrender their old vehicles in exchange for newer, safer and better regulated ones thereby providing a seamless link within the extended/complete IRT system (Figure 2.5). Public participation is critical for any economically viable and micro-level development strategy as is the case of the quasi-formal commuter taxi sector (Theron, 2005). Animated debate and acts of dissention on the part of the taxi operators should be viewed as 'healthy' signs from a purely democratic and participatory point of view, provided that the participants function within the law. As Sjöberg (1996:7) argues, participatory design views the groups as being heterogeneous in nature. *Heterogeneity* (herein synonymous with *variety* and *disparity*) subsequently embodies *diversity* and *creativity* within its membership (Stirling, n.d).



(source: City of Cape Town)

Similarly, scathing criticism has been levelled against the *Cape Metrorail* for poor maintenance, lax security, poor accessibility and the confusion commuters experience in the absence of an integrated effective wayfinding system (Prince, 2006; Powell, 2007; Marud, 2008; Smith, 2009). The problems appear not to be unique to Cape Town. At least 1.6 million commuters use the national Metrorail service daily which is run by the *South African Rail Commuter Corporation* (SARCC) (wa ka Ngobeni, 2009). SARCC admits that one-third of its entire rolling stock is too old for refurbishment and consequently plans to introduce 560 new coaches per year over the next decade at a cost of ZAR 7 billion a year (*ibid*). Cape Town's historic developmental emphasis focused on providing more roads for personal vehicles as opposed to mass transit rail and commuter bus options – the latter two being the more sustainable transportation options. Figure 2.6 shows the major highways linking Greenpoint Stadium to the Cape Town International Airport and beyond.



Figure 2.6 Map of Cape Town showing major highways linking the airport and the stadium (source: Donald Cupido – City of Cape Town)

Cities with challenges akin to Cape Town's include the South American cities of *Curitiba* and *Bogotá* (in Colombia) who introduced world-leading, accessible *Bus Rapid Transit* (BRT) systems (Dobbs, 2001; Steinfeld, 2001; Campbell, 2006; Wright & Hook, 2007). The experiences of the visionary Mayors *Jaime Lerner* of Curitiba and *Enrique Peñalosa* of Bogotá indicate that it would take considerable courage to tackle the "political and technical complexity"

to implement a successful and sustainable BRT system in a city such as Cape Town (Wright & Hook, 2007:639). This system is perceived to be less of a financial risk than the considerably more expensive *Gautrain* heavy rail system between Johannesburg and Pretoria – whose primary target is to reduce congestion via personal vehicles by encouraging motorists to use the high-speed rail transit instead (Cupido, 2008; Verster, 2008). An inaugural test was carried out on a completed section of the Gautrain system in February 2009 amidst much fanfare and publicity.

Bucher (2008) argues for the city to give public transport priority on all municipal roads. The overall benefit for residents in cities adopting sustainable transportation systems is in reclaiming more pedestrianised zones such as those planned for *Public Viewing Areas* (PVA) and the 2010 *Fan Mile* route in central Cape Town as highlighted in red in Figure 2.7 (Boraine, 2007). The *FanFest* venue will be dedicated to non-motorised transport (as elaborated in Figure 2.8) – for cyclists and pedestrians, as well as providing opportunities for small traders to carry out business (Design Indaba, 2007; Cape Times, 2008; Haferburg, Golka & Selter, 2009). Additionally, Cape Town currently offers the *Dial-a-Ride* service as specialised transport for disabled commuters (Swart *et al.*, 2008:51). A more inclusive approach would be to mainstream UD-inspired strategies as "[UD] especially helps those with physical, sensory, and cognitive disabilities [as well as] tourists, visitors, and first-time users" of a BRT system thus ensuring greater accessibility for all users (Wright & Hook, 2007:497).



Figure 2.7 Map of central Cape Town showing the route of the proposed 2010 Fan Mile (source: Andrew Boraine – Cape Town Partnership)



Figure 2.8 Detailed map of central Cape Town showing the main 2010 FanFest venue (source: City of Cape Town)

2.5.1 Building a Sustainable Legacy

The example of visionary actors in a variety of different settings can help consolidate socioeconomic and political gains from an event such as the 2010 one. For example, the championing role of *Rudolph Giuliani*, the former Mayor of the City of *New York*, formalised the central role of UD by engaging the local chapter of professional architects, as well as facilitating the involvement of the *Centre for Inclusive Design and Environmental Access* (IDEA) of the State University of New York (which also hosts *The World of Universal Design* website) in the process of researching and promoting UD strategies in that city (Danford & Tauke, 2001). Similarly, in the city of Curitiba, Brazil, former Mayor Jaime Lerner formed a partnership between the mayor's office, professionals (including engineers), and *Instituto* de *Pesquisa* e *Planejamento Urbano* de *Curitiba* (IPPUC) or the Institute of Research and Urban Planning to address similar challenges (Campbell, 2006).

Despite the fundamental differences in approach taken by the two cities – with New York taking a more traditional route of engaging policy makers and experts prior to disseminating the strategies in a *top-down* manner, whilst Curitiba took a *bottom-up* participatory approach – the impact in the respective cities was marked (Campbell, 2006; M'Rithaa & Futerman, 2007). Cape Town's "public participation process" (as discussed earlier) could provide an effective means of engagement with the residents, provided it pursued an inclusive consensus participatory approach (Sjöberg, 1996; Bühler, 2001; Luck, 2003; Contact, 2007:1; Platzky, 2007). Further, the local partnerships within New York and Curitiba illustrate sustainable tripartite collaboration between such key actors as government planners, professional designers, and design academics – including educators and researchers (M'Rithaa & Futerman, 2007).

The need for sustainable long-term partnerships is particularly significant in view of the argument advanced by Cornelissen and Swart (2006) that the political and corporate elite stand to gain more from the hosting of a sporting mega-event than the local residents, despite concerted attempts to change this perception via FIFA's highly publicised corporate social investment projects (Contact Newsletter, 2007e; Kortjass, 2007a; 2007b; 2007c). They cite the 2002 FIFA World Cup[™] co-hosted by the Asian countries of *Japan* and *South Korea* to show that it is a near impossibility to accurately anticipate the benefits of hosting the games stating that the tournament "notoriously under-delivered on the expectations of the respective organisers" due in part to the 'crowding out effect' – where certain regular or expected visitors opt not to visit during the duration of the tournament (Cornelissen & Swart, 2006:110). Other commentators have reported on this phenomenon (Roaf, van Deventer & Houston, 1996; Matheson & Baade, 2004; Matheson, 2006). In reality, there may well be a major shortfall in visitors, in part due to the magnitude of the current global economic recession.

Cape Town is led by a dynamic professional team of city managers and planners (Contact Newsletter, 2007b; 2007d). It is one of the most progressive cities in Africa, and is recognised as a world-class city by the international community (Gosling, 2008). It stands to gain even greater visibility and prestige once the visitors start pouring in for the soccer extravaganza.

Ultimately, the true legacy of hosting the World Cup[™] for residents of Cape Town will be in the more accessible and revamped sports facilities as in Figure 2.9 (the Mouille Point Beachfront and Promenade at the top left-hand corner was the site of the Students' Landscape Design Competition discussed in Section 5.2.3.3); a more 'legible' city with respect to wayfinding (Zmudzinska-Nowak, 2003); improved public transportation and road infrastructure such as the major Koeberg Road/N1 Highway interchange (Figures 2.9a and 2.9b); a more efficient and larger airport; faster commuting during rush hour/peak traffic via dedicated *bus-minibus-taxi* (BMT) lanes; upgraded bus ranks and rail stations; technology transfer in construction and mega-events management; logistics; micro-enterprise development and support; a greater eagerness to engage in participatory governance of the city's affairs; and potentially, the pride of having successfully hosted Africa's first World Cup[™] (Eason, 2005; City of Cape Town, 2006; 2007; CityNews, 2007a; 2007b; 2007c; M'Rithaa, 2006; M'Rithaa & Futerman, 2007).



Figure 2.9 Map showing the scale of development around Greenpoint Stadium (source: OvP Associates)



Figure 2.10a Ongoing upgrading of road infrastructure – *before construction* (source: Donald Cupido – City of Cape Town)



Figure 2.10b Ongoing upgrading of road infrastructure – *after completion* (source: Donald Cupido – City of Cape Town)

2.5.2 Anticipated Outcomes

South Africa is undoubtedly the most successful country in Africa in terms of bidding for sporting mega-events (Cornelissen & Swart, 2007). Cricket South Africa's recent selection to host the *Indian Professional League* in March 2009 (at extremely short notice) bears testimony to the perceived competence of the country to organise a successful event of such magnitude whilst meeting security and related obligations to players and spectators alike (BBC Sport, 2009). Further, the marketing of the event and exceptionally high tickets sales of for the cricket tournament is worthy of emulation by the *Africa Confederations Cup* and 2010 World Cup organisers. In so doing, the anticipated benefit for nation building via a renewed sense of unity and social cohesion might become a reality, provided that the games are linked to broader goals of economic opportunity, political participation, justice, equity, development, and social inclusion through increased accessibility to all public services (Kearney, 2005).

The potential for nation-building is partly tempered by the dismal performance record of the *Bafana Bafana* national soccer team presently ranked 77th worldwide, and 18th in Africa (Mail & Guardian, 2009). The team needs to improve rather urgently to inspire the level of confidence and national pride that the national rugby (the *Springboks*) and cricket (the *Proteas*) teams have done for the country. Cricket and rugby have arguably been historically better resourced and managed in part due to their association with the ruling white elite in the apartheid era as well as their ability to attract and retain the best players within the country (Desai & Nabbi, 2007). Soccer could potentially be harnessed to engage civil society in participatory nation-building initiatives. The impact of this sense of fraternity could well extend to the rest of Africa riding on the broader foreign policy and pan-Africanist agenda that South Africa has championed via the 2010 event (Ezeoha & Uche, 2005; Fick, 2006; Cornelissen & Swart, 2006; Boon, 2007).

Another possible outcome of the 2010 event would be a renewed and sustainable investment in soccer and related sports tourism potential, particularly as the sport's infrastructural needs will have been met largely due to the construction and upgrading of stadia across the country (Cornelissen, 2009; Swart *et al.*, 2009). Consequently, the newly appointed Operator will be expected to run the completed Greenpoint Stadium (Figure 2.11) in an economically viable and profitable manner so as to justify the significant investment in the said facilities (City of Cape Town, 2007). The benefit of compliance with the stringent and exacting guidelines set by FIFA guarantee exceptional quality of the facilities by (South) African standards (*ibid*).

24



Figure 2.11 Panoramic view of the completed Greenpoint Stadium (source: City of Cape Town)

2.6 Summary

The recent global recession may make a significant dent on the revenues expected from sports tourists visiting the country. Though the real benefits to the country can only really be determined accurately *ex post facto*, the actual gains may well lie in the legacy projects and participatory processes inherited by the residents of South Africa's Host Cities. These cities have many pressing challenges of a socio-economic and developmental nature. In positioning the 2010 projects within the broader developmental strategies, advances can be made in the provision of long overdue infrastructural upgrades, as well as the opportunity to introduce truly sustainable mass transit systems.

Further, Host Cities in general (and Cape Town in particular) should consolidate gains made from lessons learnt in the planning and implementations thus far to embrace a long-term inclusive and potentially emancipatory consensual participatory approach to matters of public interest so as to facilitate what Sjöberg (1996:3) refers to as "a well functioning dialogue" between city planners and managers on the one hand; and their residents on the other. In this latter scenario, the expression of their viewpoint in the residents' own unique 'voice' would ensure that they have a greater sense of ownership as co-planners of the cities in which they live out their lives.

CHAPTER THREE VULNERABLE POPULATIONS IN CONTEXT

3.0 Introduction

This chapter provides a general background on definitions and categories of vulnerable populations. The chapter also elaborates on available data from a variety of official sources and discusses the practical implications of emerging demographic trends from a number of contexts. The findings provide a sound basis for the argument for an urgent paradigm shift in dealing with the vulnerable groups; a shift to a unified non-exclusive approach to the patently diverse needs.

3.1 Vulnerable Populations

According to the Microsoft® Encarta® (2009), the adjective 'vulnerable' implies: *without adequate protection* – open to physical or emotional harm; *extremely susceptible* – *easily persuadable or liable to give in to temptation;* or *physically or psychologically weak* – *unable to resist illness, debility, or failure.* Though arguably some of these lexical meanings fail to elaborate any direct bearings on design or the built environment, some common features emerge – vulnerable populations typically cannot (for a host of reasons) fend for themselves and are thus open to coercive and manipulative practices by others (Benson, 1997). In reality, in applying a convenient label to such broad and heterogeneous category there is the constant risk of losing the inherent diversity through generalisation for the sake of economy (Stirling, n.d). Subsequently, for the purposes of this study, the vulnerable populations will include the disabled; the elderly; women; the young; as well as foreigners and visitors. Vulnerability resulting from natural disasters will not be considered herein.

Vulnerable populations are identified as specific categories of people who are considered to be patently disenfranchised or are likely to be excluded from participation in all the rights and opportunities that are guaranteed to the general citizenry of a particular place (Jaeger & Bowman, 2005). Membership to such categories/populations may be as a consequence of structural or systemic reasons (such as inadequate support systems that promote equitable socio-economic participation); or due to more mundane and functional reasons (such as a bodily condition that limits a person's ability to function fully within the given social context). Pheasant (1998:23) advances the principle of the 'limiting user' – a concept that seeks to

redress vulnerability within the discipline of Ergonomics/Human Factors. Other targeted responses to the needs of vulnerable populations will be discussed in Chapter Four.

3.1.1 The Disabled

As Madden and Hogan (1997) argue, disability is a complex multi-dimensional term whose working definition (and thus perceived meaning) often changes depending on which audience of actors is being addressed. Further, there is a diverse range of professionals interested in disability for a host of different reasons. These include: "providers of support services; funders and planners of broad disability programs; administrators of legislation outlining the rights of people with a disability; people responsible for income security policy, including the social security and compensation fields; clinicians; and national and international statisticians" (Madden & Hogan, 1997:2). Similarly attempts to understand disability from a variety of geopolitical perspectives is reported elsewhere (DPSA, n.d; Oliver, 1990; 1996; European Commission, 2002; Winam, 2003; DePoy & Gilson, 2004; EIDD, 2004). With respect to the populous sub-Saharan region of Africa, there is a demonstrable link between disability and poverty (United Nations, 2005; Albone, 2008).

To facilitate a common understanding of disability among the various role players, the World Health Organization (WHO) proposed the *International Classification of Functioning, Disability and Health (ICF)* for use in the collection of data, as well as in informing responses to disability among WHO member countries (WHO, 2001; 2002; 2004). The ICF makes a direct link between the issues of disability and a person's levels of functioning, and more importantly, the ICF acknowledges the impact of environmental factors on the latter (Pasha & Pasha, 2006; Schneider, 2006). Lately, the medical profession has also adopted the ICF paradigm on disability (Leonardi, Bickenbach, Ustun, Kostanjsek & Chatterji, 2006).

According to United Nations (2003) estimates, it is believed that between 10-12% of the world's population is disabled. As depicted in Table 3.1, the Census 2001 statistics for South Africa however indicate a much lower 5% (Stats SA, 2005a:1) – a figure that might conceal the actual extent of the prevalence of disability in this country in part due to under-reporting as a result of inadequate data capture instruments or due to the social stigma associated with disability (ODP, 1997; Child Health Policy Institute, 2001; WHO, 2002). The *Integrated National Disability Strategy* on the other hand acknowledges a more realistic figure of between 5 and 12% of the general population as being disabled (ODP, 1997).

In other instances, the definitions and underlying assumptions affect an accurate interpretation; as in the working definition used for the *Census 2001* as "a physical or mental handicap which has lasted for six months or more, or is expected to last at least six months, which prevents the person from carrying out daily activities independently, or from participating fully in educational, economic or social activities" (Stats SA, 2002:vi). Of the various disability types recorded, the "prevalence of sight disability was the highest (32%) followed by physical disability (30%), hearing (20%), emotional disability (16%), intellectual disability (12%) and communication disability (7%)" (Stats SA, 2005a:1). A detailed breakdown of available data based on ethnic background of South Africans with disabilities for 1996, 1998, and 2001 is provided in Table 3.1 (Health Systems Trust, 2006).

 Table 3.1 Prevalence (%) of disability in South Africa by ethnic background (source: Health Systems Trust)

	Physical		Sight Disability		Hearing Disability		Mental Illness		All Types of Disabilities		
	Disa	bility									
Ethnicity :	1996	2001	1996	2001	1996	2001	1996	2001	1996	1998	2001
African/Black	1.6	1.2	3.2	1.4	1.1	0.7	0.5	-	7.5	6.1	5.2
Coloured	0.9	1.3	1.0	0.8	0.4	0.5	0.4	-	3.6	4.5	4.2
Indian/Asian	0.9	1.1	1.5	0.8	0.5	0.4	0.4	-	4.1	4.8	3.7
White	0.6	1.3	0.7	0.7	0.6	0.8	0.3	-	3.3	5.3	4.5
Other/Unspecified	0.8	-	1.3	-	0.6	-	0.4	-	4.4	-	-
All Groups	1.4	1.2	2.7	1.3	1.0	0.7	0.5	-	6.5	5.9	5.0

South Africa had a total of 2,255,982 people with various forms of disability according to the 2001 Census (Stats SA, 2005a:1). The vast majority (1,854,376 or 82.2%) were *African*; 168,678 (or 7.5%) *Coloured*; 41,235 (or 1.8%) *Indian/Asian*; and 191,693 (or 8.5%) *White* (*ibid*). These figures closely mirror those on the ethnic composition of the general population of 79% African; 8.5% Coloured; 2.5% Indian/Asian; and 9.6% White (Stats SA, 2002:17).

The number of females affected was 1,173,939 (or 52%), whilst 1,082,043 (or 48%) were males – percentages that are identical with the general population's (*ibid*). With respect to age however, there is a marked disparity in the prevalence rates starting with a low of 2% among the young aged 0-9 years, and rising to a high of 27% among senior citizens aged 80 years and above (Stats SA, 2005a:1).

The picture in the Western Cape Province is slightly different. The total figure of over 145,000 people with disabilities suggested in another source appears conservative at best (Western Cape, 2004:2). The *Census 2001* data appears more realistic. It is believed that out of a total

of 186,850 people with disabilities, 96,549 (or 4.4%) were male, whilst 90,301 (3.9%) were female (Stats SA, 2002:12). The total figure works out to 4.1% of the general population – the second lowest disability prevalence rate of any of South Africa's nine provinces (*ibid*). Tabl 3.2 summarises some of the comparative data available for 1996, 1998, and 2001 on the prevalence rates of various disability types in the various provinces in South Africa.

	Phy	/sical	Sight D	isability	Hearing	Disability	Me	ntal	All Type	es of Disa	abilities
	Dis	ability	U	,	0	,	Illn	999	51		
	0130	ability					11111	633			
Province:	1996	2001	1996	2001	1996	2001	1996	2001	1996	1998	2001
Eastern Cape	1.9	1.5	2.6	1.3	1.1	0.8	0.7	-	7.3	8.9	5.8
Free State	1.6	1.3	5.2	2.2	1.3	1.0	5.4	-	9.8	5.8	6.8
Gauteng	1.0	1.0	2.9	1.0	0.8	0.4	0.3	-	6.2	5.2	3.8
KwaZulu-Natal	1.6	1.3	2.2	1.2	0.9	0.7	0.5	-	6.0	6.7	5.0
Limpopo	1.5	1.0	2.3	1.3	1.1	0.8	0.4	-	6.0	6.3	5.1
Mpumalanga	1.1	1.3	3.6	1.6	1.2	0.9	0.5	-	7.6	4.5	5.8
Northern Cape	1.2	1.6	2.3	1.5	0.8	0.7	0.5	-	5.6	4.5	5.7
North-West	1.8	1.4	3.4	1.7	1.6	0.7	0.7	-	8.3	3.1	5.8
Western Cape	0.9	1.2	3.8	0.8	0.5	0.6	0.4	-	3.7	3.8	4.1
South Africa	1.4	1.2	2.7	1.3	1.0	0.7	0.5	-	6.5	5.9	5.0

Table 3.2 Prevalence (%) of disability in South Africa by province (source: Health Systems Trust)

Disability when viewed within the broader sense of gross incapacitation reveals a starker picture of the debilitating effects of the HIV/Aids pandemic, and the impact this disease has had on the significant number of people living with Aids, particularly in sub-Saharan Africa (Barnett & Whiteside, 2002; Deacon, Stephney & Prosalendis, 2005; WHO, 2007). Swartz, Schneider, and Rohleder (2006) demonstrate a clear link between HIV/Aids and disability. The malaise, general fatigue and related limitations in functioning at the full-blown stages of the disease would justify this association (ODP, 1997; Barnett & Whiteside, 2002). To put this in perspective, the UNDP (2005) declares that "HIV has inflicted the *single greatest reversal in human development* in modern history".

The devastating results of high mortality rates within the economically active segment of the population has led to a unique phenomenon of the elderly becoming primary caregivers to not only their infected grown-up children, but also to their affected grandchildren – many of whom end up as Aids orphans (HelpAge, 2002; Albone, 2008). The UNAIDS (2008:13) estimates that Aids has created 12 million orphans under the age of 18 in sub-Saharan Africa – a factor that threatens the sustainability of entire societies as "the natural age distribution in many national populations in sub-Saharan Africa has been dramatically skewed by HIV, with potentially

perilous consequences for the transfer of knowledge and values from one generation to the next".

There were some 33 million people living with HIV/Aids in the world in 2007 (UNAIDS, 2008:39). Of this total, 0.8% was aged 15-49 – the group considered the most vulnerable or at risk of infection (UNAIDS, 2008:215). Sub-Saharan Africa has 22 million (or 67% of the global figure) living with HIV/Aids, of which 5.0% (the highest in the world) of this total were aged 15-49 (UNAIDS, 2008:32). In South Africa the prevalence rate for the same age cohort is estimated at between 16.7% (Stats SA, 2005b:8) and 18.1% (UNAIDS, 2008:215). It is estimated that 35% of people living with HIV/Aids are in the SADC region, with the seven countries of Botswana, Lesotho, Namibia, South Africa, Swaziland, Zambia, and Zimbabwe all having high prevalence rates of 15% (*ibid*). South Africa has the unfortunate distinction of having "the largest HIV epidemic in the world"– with a staggering caseload of 5.7 million people infected as of 2007 (UNAIDS, 2008:40).

The net effect of the HIV/Aids pandemic on health and life expectancy in sub-Saharan Africa has been devastating (Edries & Triegaardt, 2004). For example, in the 38 most-affected African countries (the majority of which are in the SADC region), nearly 10 years of life-expectancy will have been lost by 2020-2025 (UNDP, 2005). Universal access to treatment, such as through the highly regarded *Anti-Retroviral Therapy* (ART) is presently the most practical means of prolonging the lives of people living with HIV/Aids. Despite an initial response of official denialism on the link between HIV and Aids, the South African government has since committed itself "to the 2006 United Nations High Level Meeting on HIV/Aids, to scale up towards HIV prevention, treatment, care and support by 2010 [including] providing universal access to antiretroviral treatment and to services to prevent mother-to-child transmission of HIV" (UNAIDS, 2008:9).

3.1.1.1 Definitions of Disability

There are three important and distinct aspects of the disability process as defined by the WHO (2002):

- *Impairment* is any loss or abnormality of psychological, physiological, or anatomical structure or function
- *Disability* is any restriction or lack (resulting from an impairment) of ability to perform an activity in the manner or within the range considered normal for a human being

• *Handicap* is a disadvantage for a given individual, resulting from an impairment or a disability, that limits or prevents the fulfilment of a role that is normal (depending on age, sex, and social and cultural factors), for that individual.

To elaborate further, impairments and disabilities may be visible or invisible, temporary or permanent, and progressive or regressive. Further, there are two main 'models' with respect to disability; the *medical* 'model', and the *social* 'model' (Oliver, 1990; 1996; Jönsson & Certec, 2006a). By placing the problem of disablism within societal responses, the following definition by the *Disabled People South Africa* (DPSA) adopted from the British Council of Organisations of Disabled People reinforces its affinity towards the social model of disability (DPSA, 2004): *Disability is the disadvantage or restriction of activity caused by a society that takes little or no account of people who have impairments and thus excludes them from mainstream activity.*

The White Paper on an Integrated National Disability Strategy (INDS) in South Africa (in keeping with similar trends internationally) adopts the social model of disability (ODP, 1997): One of the greatest hurdles disabled people face when trying to access mainstream programs is negative attitudes. It is these attitudes that lead to the social exclusion and marginalisation of people with disabilities.

More recently, the *biopsychosocial* 'model' proposed by the WHO (and attributed to George L. Engel) seeks to reconcile the two often opposed views of the *medical* versus *social* ones (Pasha & Pasha, 2006). DePoy and Gilson (2004:3) refer to these schools of thought as the "*medical-diagnostic* phenomenon and disability as *constructed*". Further, DePoy and Gilson (2004; 2006; 2005/2006; 2007) advance the *Explanatory Legitimacy Theory* as a more unified and conciliatory (as opposed to a divisive dichotomous) view of disability. This subject will be the interrogated in more detail in Chapters Four and Six.

3.1.1.2 Categories of Disability

Disability is typically classified by its severity on an individual's capacity for independent functioning. Traditionally, the vast array of responses have emanated from the need for biomedical or social welfare interventions to facilitate the individual's need for autonomy (ODP, 1997). The medical model categorises disability into two broad categories:

- physical/sensory:
 - o locomotor (e.g. ambulatory impairment);
 - hearing (e.g. deafness, hardness of hearing);

- visual (e.g. blindness, macular degeneration, cataracts, poor sightedness);
- speech (e.g. speech disorders, language disorders); and
- mental/intellectual (e.g. dementia, autism, cerebral palsy, Down Syndrome).

Disabilities of the locomotor type (ambulatory aid/device users including wheelchair-dependent disabled persons) are the most visible and by far, the most prevalent in any given population and is occasioned by a variety of causes, such as the following examples:

- arthritis and stroke;
- affections of the cardio-respiratory system;
- injury (e.g. spinal lesions);
- acquired injury (in war or automobile accidents);
- congenital defects/deformations;
- amputations and spinal paralysis;
- slowly progressive degenerative disorders (especially due to the ageing process);
- disease (e.g. poliomyelitis);
- muscular atrophy; and
- cerebral palsy and multiple sclerosis, to mention a few.

Whilst respecting the great heterogeneity within this category, this study focuses on the tacit or experiential limitations brought about by the diverse forms of disability, and not on any specific expressions of individual disabilities (Rowland, 1984; Oliver, 1990; Benson, 1997; Goldsmith, 1997; Healey, 1994; Mace, 1998; Coan, 2003; Kahwaji, 2003; 2005; Dittmann-Kohli, 2007; Ongolo, 2007).

3.1.2 The Elderly

Hansson (2006:10) remarks that "in marketing theory consumers are defined as elderly at the age of 55 or older". This however appears to be too early a cut-off and not as realistic as the definition by Crews and Zavotka (2006:113) of an *elder* being "a person of 65 years and above". Further, the elderly are subdivided into three categories: the *young-old* (65-74 years); the *old-old* (75-84 years); and the *oldest-old* (85 years and older) (Berk, 2001:556). The needs of older users (in the Third Age of their lives) vary greatly from person to person as individuals age differently and experience sensory, agility and mobility limitations in different ratios and to different degrees (Benktzon, 1993; Bound & Coleman, 1993; Fisk, 1993; Pirkl, 1994; Koncelik, 1998; Jordan, 1999; Family Practice Notebook, 2000b; Berk, 2001; Goodman, Brewster &

Gray, 2005). The elderly should not be treated as one homogenous group but as one with an inherently rich diversity, whose psychosocial needs should be addressed in a societal transgenerational context (Gardner, Powell & Page, 1993; Norman, 2004; Soares, 2005; Design Council, 2006; Bernard & Scharf, 2007; Dittmann-Kohli, 2007; Ricability, n.d).

According to Mostert et al. (1998:39 in: Haldenwang, 2008:128) the United Nations classifies "a population [...] as 'old' in demographic terms if more than 7% of the population is 65 years and older, 'mature' if between 4 and 7% are 65 years and older, and 'young' if fewer than 4% are 65 years". South Africa is experiencing a very unique longitudinal demographic shift - a phenomenon that reflects similar trends elsewhere in the world (Futerman & M'Rithaa, 2007). Hansson (2006:11) suggests that this "demographic change is [most dramatically] reflected in the ageing populations of USA, Japan and Europe". The percentage of older people within the general population is increasing in both industrially developed as well as in developing countries (Korenjak-Černe, Kejžar & Batagelj, 2008). Koncelik (1998:117) identifies this as "the single most important change occurring to humanity on this planet". The World Assembly on Ageing (2002) predicts that by the year 2050, the number of people aged 60 and above will increase from 600 million to almost 2 billion. In developed countries the percentage of older people will rise from 10% to 21% whilst concomitant growth in developing/majority world contexts is expected to be fourfold (*ibid*). The CIA World Factbook (2009) reports that Japan is the most rapidly ageing country with 22.2% of its citizens aged 65 or older. Japan also has the highest life expectancy at an average of 82 years (Population Reference Bureau, 2008:13).

One of the most interesting characteristics of the South African context is that embedded within the general population are four distinct population trends based on the ethnicity of their members. For this reason, one finds specific trends that can be identified with other societies at various stages of socio-economic and political development. With respect to population pyramids, Korenjak-Černe *et al.* (2008:157) identify three main shapes:

- *expansive* shapes are typical for fast-growing populations where each birth cohort (a group of people born in the same year or years period) is larger than the previous one;
- constrictive shapes display lower percentages of younger populations; and
- stationary shapes present somehow similar percentages for almost all age groups.

This specific impact on the various ethnic groups will be discussed in section 3.2. The population pyramids depicted in Figure 3.1 vividly illustrate how the situation in South Africa is

expected to change successively from *expansive*, through *constrictive*, and finally to *stationary* shapes between 2000 and 2050 (US Census Bureau, 2009).



Figure 3.1 Population pyramids for South Africa – 2000; 2025; & 2050 (source: US Census Bureau, 2009)

In South Africa the number of people aged sixty and above was 7.3% of the general population (Stats SA, 2002). This is expected to increase to 10.8% (Kinsella & Ferreira, 1997). Strictly speaking, the category classified as 'elderly' begins at 65 (and not 60) as discussed earlier; (Kinsella & Velkoff, 2001). Notwithstanding (and more significantly for this study) the Western Cape Province has the highest population of people aged 80 and above, and the second highest percentage of elderly people living in urban areas at 91.14% (HSRC, 2003:46-47). Haldenwang (2008:130) classifies the total population of the Western Cape (with a median age of 26.7 years and 5.2% aged 65 years or more in 2001) as a "mature population". The Black African population (1.8%) in the Western Cape Province is classified as 'young'; the Coloured (4.0%) and Indian (4.5%) populations as 'mature'; and the White population (13.6%) as 'old' (*ibid*). The increase in life-expectancy in South Africa coupled with the fact that disability becomes more prevalent with the natural process of ageing will result in larger segments of the society experiencing some form of disability or limitation in functioning (ODP, 1997). This correlation between ageing and disability is confirmed by Stats SA (2005a:17):

There are striking differences at the younger ages (below 30 years) where 63% of nondisabled persons belong in this age range compared to 35% of disabled persons. This in itself implies that a higher proportion of disabled persons are older. Of the non-disabled population, only 12% were 50 years or older, while 37% of the disabled were in this age category.

According to estimates, there were 8,643,354 elderly people living in the SADC region in 2008, averaging 3.83% of the entire population (CIA World Factbook, 2008). Mauritius and The Seychelles (at 7.1%) have the highest proportion of elderly people, whilst Zambia (at 2.3%) has the lowest (*ibid*). South Africa (at 5.4%) has the largest number of senior citizens – with 40.8% of the total elderly population being male, compared to a more sizeable 59.2% being female

(*ibid*). The complete listing and breakdown of elderly population figures for the fifteen member SADC states is listed in Table 3.3.

	Pers	Percentage of Total Population		
Member State	Male	Female	Total	2009 Estimates
Angola	153,678	195,043	348,721	2.7%
Botswana	31,155	45,547	76,702	3.9%
Congo (DRC)	699,667	1,021,070	1,720,737	2.5%
Lesotho	42,074	63,915	105,989	5.0%
Madagascar	280,677	348,591	629,268	3.0%
Malawi	162,863	220,390	383,253	2.7%
Mauritius	36,309	54,465	90,774	7.1%
Mozambique	257,119	361,772	618,891	2.9%
Namibia	36,894	45,667	82,561	3.9%
Seychelles	2,321	3,913	6,234	7.1%
South Africa	1,075,117	1,562,860	2,637,977	5.4%
Swaziland	15,261	25,938	41,199	3.7%
Tanzania	513,959	663,233	1,177,192	2.9%
Zambia	114,477	164,199	278,676	2.3%
Zimbabwe	194,360	250,820	445,180	3.9%
SADC Total	3,615,931	5,027,423	8,643,354	Av: 3.93%

Table 3.3 Number of elderly people in the SADC region (source: CIA World Factbook)

The elderly in most developing countries are viewed as dependants (on the government and the general working population) as these elders often require support for institutional/medical care, welfare, and social security – this is a huge challenge for any country, not least developing ones (Edelson, 1991; ODP, 1997; Barnett & Whiteside, 2002; HelpAge, 2002; 2006; 2009; HSRC, 2003; Gibbs, 2005; Cohen & Menken, 2006; Albone, 2008; Box, 2008). The design focus for senior citizens should be to facilitate autonomy and accessibility via independent lifestyles and progressive ageing-in-place strategies (Blaikie, 1993; Haigh, 1993; Sandhu, 1993; Woudhuysen, 1993; Pirkl, 1994; Coleman, 1999; Demirbilek, 1999; Hitchcock, Lockyer, Cook & Quigley, 2001; Huppert, 2003; Owen & Johnston, 2003; Austad, 1997; Demirbilek & Demirkan, 2004; Baars, 2007; Futerman & M'Rithaa, 2007; Gross, 2008; Haldenwang, 2008).

3.1.3 Women

Women account for about 52% of South Africa's population (a feature not uncommon to most other countries) and have a life expectancy of between 7-11% higher than that of men (HSRC, 2003:82). This fact alone should justify greater attention to the specific accessibility needs of women. As elaborated in Figure 3.2, the ratio of female to male in the South African general

population is equal in the first two age cohorts (0-9 years). Between the ages of 10 till 64, the ratio is slightly higher for females, whilst the ratio increases steadily to 2:1 from age 80 onwards (Stats SA, 2002:30).



Figure 3.2: Distribution of total South African population by age group and sex (source: Stats SA, 2002:30)

Women (as well as the young and certain categories of the elderly) are typically smaller than men and have less physical strength (Benson, 1997; Pheasant, 1998; Kroemer, Kroemer & Kroemer-Elbert, 2001; Kroemer, 2006). Despite this fact, women play a key socio-economic role in the African family as providers (Macharia, 2006). They also act as default primary caregivers for the majority of families (Takyi & Oheneba-Sakyi, 2006). Women are often the unfortunate victims of violence typically meted out by their men folk (Vetten, 2007). Further, as Weisman (1994) argues, women are often subjected to embarrassment and inconvenience when they are forced to use inappropriate or inadequate public amenities and facilities typically designed by male designers and architects. This view is supported by Papanek (1971), Shapira (1991), Whiteley (1993), Owen (2004), and Davey, Wootton, Thomas, Cooper and Press (2005) in their respective critiques on the need for *socially responsible design* interventions. Any design response that targets women would naturally impact upon a much larger section of the communities they live in (ODP, 1997; Edries *et al.*, 2004; M'Rithaa, 2008).

3.1.4 The Young

Young people (or persons aged 14 and below) make up 29.1% of the general population (see Figure 3.4 for regional comparisons). They are considered dependants whose greatest cost to a social welfare system is in the form of educational support. The cost increases with the

special needs required for the support of children with disabilities (Batshaw, 1997; Child Health Policy Institute, 2001; Disability, 2003).

	Pers	Percentage of Total Population		
Member State	Male	Female	Total	2009 Estimates
Angola	2,812,359	2,759,047	5,571,406	43.5%
Botswana	352,399	340,058	692,457	34.8%
Congo (DRC)	16,161,301	16,038,024	32,199,325	46.9%
Lesotho	373,159	368,271	741,430	34.8%
Madagascar	4,523,033	4,460,473	8,983,506	43.5%
Malawi	3,272,790	3,258,893	6,531,683	45.8%
Mauritius	147,136	142,121	289,257	22.5%
Mozambique	4,829,272	4,773,209	9,602,481	44.3%
Namibia	381,904	375,059	756,963	35.9%
Seychelles	10,201	9,732	19,933	22.8%
South Africa	6,447,623	6,370,909	12,818,532	29.1%
Swaziland	15,261	25,938	41,199	39.4%
Tanzania	8,853,529	8,805,810	17,659,339	43.0%
Zambia	2,685,142	2,659,771	5,344,913	45.1%
Zimbabwe	2,523,119	2,473,928	4,997,047	43.9%
SADC Total	53,388,228	52,861,243	106,249,471	Av: 38.35%

Table 3.4 Number of young people in the SADC region (source: CIA World Factbook)

Education is often viewed as a long-term investment by a country and thus as being a lower economic cost than that of supporting the elderly (Illich, 1973). Barnett and Whiteside (2002:196) argue that orphans and the elderly are the two main categories of dependants brought about by HIV/Aids and that the epidemic "has altered and will progressively alter the demographic structure of many societies; [leading to many] children-led households; [with] grandparents as primary caregivers due to absence of parents; [and instances wherein] some children and elders also get infected (not just affected) by HIV/Aids".

Apart from its association with Aids, disability is also linked to violence and war (OSD, 1997). Masakhwe (2004) identifies (armed) conflict as a major contributor to disability on our continent. For example, anti-personnel landmines have orphaned and maimed many children as a result of the protracted armed struggles in Angola and Mozambique – vast sections of these two countries are yet to be declared mine-free and subsequently safe for civilian habitation. An inclusive response to the accessibility needs of the young in South Africa would

most likely be benchmarked by its immediate neighbours in the SADC region with similar socio-political challenges. Such responses must of necessity adopt a multi-generational (or transgenerational) perspective to ensure success. This is particularly true for African populations which have lost the bulk of their working population to the ravages of war, disease, poverty and related social ills.

3.1.5 Foreigners and Visitors

Rendall and Salt (2005 in: Marindo, 2008:151) define foreign-born individuals "by birthplace and not [by] nationality or ethnicity". Foreigners, non-citizens and visitors contribute significantly to population diversity and complexity (Marindo, 2008). Stats SA (2006:1) reports that 727,272 foreign travellers "from mainland Africa, overseas and unspecified countries, arriving through all ports of entry" visited South Africa in 2006 – an increase of 26.6% when compared to the 574,364 people that visited in 2005. These figures do not reflect the vast number of refugees many of whom crossed illegally into South Africa following the protracted political strife and virtual economic collapse in Zimbabwe. Further in 2006, 657,019 of these travellers departed from the country – this revealed an increase of 22.9% when compared to the 534,648 during a similar period the year before (*ibid*). Table 3.5 elaborates a number of tourism-related terms as used in this context.

Key term	Definition
Tourism	comprises the activities of persons travelling to and staying in places outside their usual environment for not more than one consecutive year for leisure, business and other purposes not related to the exercise of an activity remunerated from within the place visited. The usual environment of a person consists of a certain area around his/her place of residence plus all other places s/he frequently visits
A tourist (overnight visitor)	a visitor who stays at least one night in collective or private accommodation in the place visited. An international tourist is an international visitor who stays at least one night in collective or private accommodation in the country visited. A domestic tourist refers to a domestic visitor who stays at least one night in collective or private accommodation in the place visited
Traveller	any person on a trip between two or more localities. An international traveller refers to any person on a trip between two or more localities in different countries
Visitor	any person travelling to a place other than that of his/her usual environment for less than 12 consecutive months and whose main purpose of the trip is other than the exercise of an activity remunerated from within the place visited
An international visitor	any person travelling to a country other than the one in which s/he has his/her usual residence, but outside his/her usual environment, for less than 12 consecutive months and whose main purpose of the trip is other than the exercise of an activity remunerated from within the place visited
Overseas traveller	a foreign traveller visiting South Africa (SA), excluding travellers from mainland Africa and from 'unspecified' countries
Foreign traveller	a person who resides outside SA and visits the country temporarily
South African resident	South African citizens and/or any person with permanent residential permit

Table 3.5 Key terms used with respect to tourism (source: Stats SA, 2006:3)

During the same period, about 73.6% of these travellers (particularly from neighbouring SADC countries) chose to enter South Africa by road compared to via air (24.6%), with only 1.8% entering via "rail, sea and unspecified modes of travel" (*ibid*). The vast majority of those who flew into the country arrived through *O.R. Tambo International Airport* in Johannesburg (79.9%) whilst some 34,557 (or 19.3%) arrived through *Cape Town International Airport (ibid*). Similarly, the vast majority of continental visitors expected for the 2010 events will enter by road, and the overseas ones will arrive by air.

With eleven languages (plus South African Sign Language), South Africa is second only to India (which has 23) in the number of official languages recognised in the country. The initial excitement at inclusiveness (through embracing diversity) soon reveals a daunting challenge – that of communication and navigation via effective wayfinding systems (M'Rithaa & Futerman, 2007). In the Western Cape Province, the official languages are Afrikaans, isiXhosa and English (in rank order). The growing complexity of urbanscapes and the increasing number of foreigners visitors will demand an inclusive and inspired response to the diverse needs of visitors with varying degrees of visual as well as general literacy (Hine, Swan, Scott, Binnie & Sharp, 2000; Watermeyer, 2002; Baldwin, 2003; Zelek *et al.*, 2003; Zmudzinska-Nowak, 2003; CRC, 2004; Wyman, 2004; Fraser, 2007; Gill, 2007; Vrooman, 2007). Cape Town thus needs to become more open, 'legible', and inclusive to evolve into a truly universal city (M'Rithaa & Futerman, 2007).

3.2 Implications of Shifting Demographics

South Africa's population is variously estimated at 46.9 million (Stats SA, 2005a:8) or 48.3 million (Population Reference Bureau, 2008:14) and is projected to reach 51.5 million by 2025 and 54.8 million by 2050 – a growth rate of 13% between 2008 and 2050 (Population Reference Bureau, 2008:14). This growth is expected to be driven predominantly by the African sub-population which reveals trends typical of most industrially developing countries, including the vast majority of those found in the SADC region. The country has a relatively low fertility rate of 2.78% (Stats SA, 2005b:8), although variance between the different main ethnic groups reveals a sliding scale of 3.0% for African; 2.3% for Coloured; 1.9% for Indian/Asian; and 1.7% for White groups (Stats SA, 2005b:5). The Census 2001 population distribution is given as: African – 79.0%; Coloured – 8.9%; Indian/Asian – 2.5%; and White – 9.6% (Stats SA, 2002:13). With the exception of the Indian/Asian group, the 2005 mid-year estimates show slight differences in the population distribution as illustrated in Figure 3.3 (Stats SA, 2005b:9).



Figure 3.3 Population distribution by ethnic background in South Africa (source: Stats SA, 2005b:9)

Of the 4,524,335 people (or 10.1% of all South Africans) who live in the Western Cape, 26.7% were classified as Africans; 53.9% as Coloured; 1.0% as Indian/Asian; and 18.4% as White (Stats SA, 2002:7;8;12). This composition is radically different from the national one. Further, the Human Science Research Council (HSRC, 2003:46-47) reports that the Western Cape Province has the highest population of people aged 80 and above, and the second highest percentage (at 91.14%) of elderly people living in urban areas.

As discussed in section 3.1.2, South Africa's population is expected to progressively age. Globally, this demographic trend will mean that up to 25% of the population will comprise of senior citizens in 2014 (Futerman & M'Rithaa, 2007). In the developed countries (particularly in the USA, Canada, Australia, New Zealand and the United Kingdom), the unique demographic phenomenon of *baby-boomers* (those born from 1946 to 1964) will demand a more proactive response to the needs of an actively ageing generation of elderly (Coleman, 1993; Pirkl, 1994; Goodman *et al.*, 2005; Transgenerational Design Matters, 2006). The impact of ageing will be more profound as life expectancy rises in developing countries such as South Africa.

Korenjak-Černe *et al.* (2008:157) argue that "besides births and mortality, also other processes, depending on social or/and political policy and events (migrations, birth control policy, war, life-style, etc.) have strong influence on age-sex structure of the population, that reflect also on the shape of the population pyramid". Korenjak-Černe *et al.* (2008:158) further

identify four stages of the *demographic transition model* (DTM) and their corresponding characteristics with respect to shapes of population pyramids:

- Stage 1: expanding high birth rate; high death rate; short life expectancy;
- Stage 2: *expanding* high birth rate; fall in death rate; slightly longer life expectancy;
- Stage 3: *stationary* declining birth rate; low death rate; longer life expectancy; and
- Stage 4: contracting low birth rate; low death rate; longer life expectancy.

By analysing the shapes of South Africa's population by ethnic background (in Figures 3.4 and 3.5), one identifies a unique diversity in the shapes of their respective population pyramids (Stats SA, 2005b:12). The shapes of the four population pyramids roughly correspondence with the four stages of the DTM as elaborated by Korenjak-Černe *et al.* (2008). Broadly speaking, the African population pyramid corresponds to Stage 1 with the highest birth rate, high death rate (associated with HIV/Aids), and a relatively short life expectancy; whilst the Coloured one shows features of Stage 2 with a high birth rate, falling death rate, and slightly longer life expectancy (see Figure 3.4). The African and Coloured population pyramid being most typical in developing countries with the Western Cape's general population pyramid being most like that of the latter – who coincidentally are the most populous group there (*ibid*). On the other hand, the Indian/Asian population pyramid corresponds to Stage 3 with a marked decline in birth rate, lower death rate, and longer life expectancy; whilst the Unite population pyramid corresponds to Stage 3 with a marked decline in birth rate, lower death rate, and longer life expectancy; whilst the White population pyramid exhibits the classic contracting shape of Stage 4 with the lowest birth rate, lowest death rate, and longest life expectancy of all four population sub-groups (as shown in Figure 3.5).



Figure 3.4 Population pyramid of the African (left) and Coloured (right) populations (source: Stats SA, 2005b:12)

The Indian/Asian population pyramid is typical of more stable developing countries; whilst the White population pyramid's contracting shape is virtually identical to that found in industrially developed countries where the socio-economic impact of ageing is most pronounced (*ibid*). As Korenjak-Černe *et al.* (2008:166) observe, "the divide between the undeveloped and developed countries is increasing" – that divide is most dramatic when one compares the demographic realities of the African and White population sub-groups in South Africa.



Figure 3.5 Population pyramid of the Indian/Asian (left) and White (right) populations (source: Stats SA, 2005b:12)

The Western Cape Province's elderly population also has the highest amount of disposable income compared to similar age cohorts in other parts of the country (Stats 2002). Additionally, the relatively higher percentage of the White population living in the Province, (vis-à-vis the national distribution) means that the impact of general ageing will be more marked in the Western Cape before the phenomenon takes hold in the other provinces (Stats 2002; Stats SA, 2005b). This presents an opportunity for the Province to effectively develop and pilot more inclusive and long-term design strategies that ultimately benefits all its residents (Haldenwang, 2008). Current technological advances also allow unprecedented possibilities for innovation through benign interventions to better address the needs of the elderly and people with disabilities (Chapman & McCartney, 2002). Further, life expectancy is steadily rising due to improved access to healthcare resources and life-enhancing facilities. Additionally, the Western Cape is a popular destination for both domestic and international tourists requiring greater accessibility and usability of products, systems and built environments. Consequently,

children, pregnant women, and foreigners (with unfamiliar native languages, cultural practices and lifestyles) all stand to gain from more accommodating and inclusive provisions.

3.3 Summary

This chapter discussed the size and complexion of the various vulnerable populations in South Africa in general, and the Western Cape Province in particular. The exclusion and discrimination of people with disabilities (who make up 5% of the population) leads to *disablism*. Further, there is a moral imperative to include people living with Aids (some 11% of the population) in this discussion as they often experience stigma and discrimination from other members of society just like people with disabilities. *Ageism* is the equivalent negative reaction with respect to the elderly (who make 5.4% of the population).

Similarly, *sexism* is the exclusion of women (who make up around 52% of the population) from mainstream participation whether tacitly or otherwise. The young make up 29.1% of the total population and are often ignored in decision making. The great influx of foreigners (and countless refugees) particularly from neighbouring countries further compounds the situation. The combined vulnerable population is undeniably significant. There is a strong argument for (and merit in) identifying and pursuing a common design response to redress the three ills of stigmatisation, discrimination and exclusion from full social participation. Countering these negative attitudinal and structural responses requires courage and commitment – these are some of the sub-themes that will be further expounded upon in the next chapter.

CHAPTER FOUR

RESPONSES TO THE NEEDS OF VULNERABLE POPULATIONS

4.0 Introduction

This chapter elaborates on pertinent practical and theoretical responses to the needs of the various categories identified as vulnerable populations in the previous chapter. The chapter provides a general overview on formal responses from a variety of international settings with reference to South Africa's geopolitical realities. It concludes by aligning local and global interventions that aim to promote greater inclusiveness of all members of the human family.

4.1 Models of Disability

This section discusses disability in the light of some of the influential schools of thought in a somewhat chronological order. The use of the term 'model' in this section is used to describe such broad paradigmatic approaches (or schools of thought) and does not reflect the stricter and more conventional usage in a scholarly sense – wherein models can be described, controlled and predicted along empirical lines (Bailey, 1987; Microsoft® Encarta®, 2009).

4.1.1 Medical Model

The *medical model* is also frequently described as the *individual model* or the *medical-diagnostic phenomenon* (according to DePoy & Gilson, 2004).This is the approach taken by the vast majority of medical and related professions. Paradigmatically, this model approaches the issue of disability as being a problem that can be 'fixed' in the individual, failing which prescriptions are made on optimal rehabilitative interventions for the disabled person (Nichols, Haworth & Hopkins, 1981; McWilliams, 1984; Oliver, 1990; Barnes, Mercer & Shakespeare, 1999; Imrie *et al.*, 2001; Werner, 1998; Goldsmith, 2000; DePoy & Gilson, 2006; Gilson & DePoy, 2006; Schneider, Barron & Fonn, 2007). This approach has tacitly endorsed the removal of certain people with disabilities from mainstream society who are then placed in specialised institutions – a practice that has spawned a variety of the exclusion of such members via special education, protective workshops, and sheltered employment (Oliver, 1990; Barredo, 2006; Opperman, 2006; Deglon, 2008; South Africa, 2008d).

A criticism of the medical approach is its patent prescriptiveness in perpetuating *expertism* – wherein the principal actor (the medical specialist or 'expert') decides on behalf of the 'ignorant, passive, voiceless and powerless patient or victim' what the appropriate course of action

should be (Illich, 1973; Foucault, 1980; Imrie *et al.*, 2001; Manjra, 2005). Further, by removing and excluding people with disabilities from general society, it becomes increasingly difficult to re-integrate the former back into mainstream activities. This has the net effect of increasing stigma and discrimination towards the disabled by treating them as objects (as opposed to as subjects) in the given activity. In the context of this thesis, the medical model is viewed as being ontologically opposed to the social model which is described in the next section. This could be linked to the differing paradigms that inform the medical (objectivist/positivist) view on disability, as opposed to the social model which is informed by a subjectivist/post-modernist epistemology (Crotty, 1998; Krauss, 2005).

4.1.2 Social Model

Oliver (1990; 1996) argues that the circumstances of people with disabilities and the discrimination they face are socially created phenomena that have little to do with the impairments of disabled people. Oliver (*ibid*) argues passionately on the failings and limitations imposed on people with disabilities due to systemic use of the *individual* (or *medical*) *model* of disability in a variety of social and professional milieu. He proposes the "sociological approach" (now known as the social model that is attributed to him) to counter the resultant stigmatisation and marginalisation of the disabled (*ibid*). DePoy and Gilson (2004) refer to this model of disability as being *constructed*. It is now widely accepted in the majority of professional contexts that the *social model* supersedes the *medical model* and the main drive is presently on changing the society and environment to allow people with disabilities full participation in every possible respect (EDeAN, 2006). Internationally, this (social) model finds ample expression in a host of policies that seek to enforce equity and challenge discriminatory practices (M'Rithaa, 2006).

In view of the emancipatory rhetoric espoused by the postmodernist perspective of the social model of disability, various oppressed minority groups have thus found a collective voice to counter discrimination as well as to speak out against systemic exclusion from mainstream activities (Oliver, 1990; Weisman, 1994; Zola, 1994; Sjöberg, 1996; Barnes *et al.*, 1999; Edries *et al.*, 2004; Dittmann-Kohli, 2007). Zola (1994) draws parallels between *disablism* and *racism* whilst Healey (1994) presents a phenomenological reflection on age and *ageism*. Such voices also include those of the "disabled, elderly, infertile [women, and people who are] HIV-positive" and various groups representing other vulnerable populations (Whyte & Ingstad, 2007:14). Further, the use of language and related nomenclature to discuss and describe people with

45

disabilities has been duly addressed with emphasis on one's autonomous personhood, and discouraging the labelling of people by the disability they experience – this latter response tends to perpetuate stigmatisation, discrimination and social exclusion (DPSA, 2004; Bickenbach & Wasserman, 2006; Appendix A).

The disability rights movement believes therefore, that the 'cure' to the 'problem' of disability lies in restructuring society and redefining societal attitudes, as opposed to the *medical/individual model* that tends to isolate the individual with disability (Zola, 1994; Oliver, 1996; Benson, 1997; ODP, 1997; Sinclair, 1998; Barnes *et al.*, 1999; Jaeger & Bowman, 2005; George & Duquette, 2006; Whyte & Ingstad, 2007). To illustrate this conceptually dichotomous view of disability, Oliver (1990:7-8) rephrases and contrasts census survey questions used in the United Kingdom (originally phrased from an ostensibly medical view) with his own interpretation from a social perspective (Table 4.1).

 Table 4.1 Conceptual dichotomy between medical and social perspectives on disability (source: Oliver, 1990:7-8)

Office of Population, Censuses and Surveys	Mike Oliver
Can you tell me what is wrong with you?	Can you tell we what is wrong with society?
What complaint causes you difficulty in holding, gripping or turning things?	What defects in the design of everyday equipment like jars, bottles and tins causes you difficulty in holding, gripping or turning them?
Does your health problem or disability prevent you from going out as often or as far as you would like?	What is it about the local environment that makes it difficult for you to get about in your neighbourhood?
Does your health problem or disability make it difficult for you to travel by bus?	Are there any transport or financial problems which prevent you from going out as often or as far as you would like?
Does your present accommodation have any adaptations because of your health problem or disability?	Did the poor design of your house mean that you had to have it adapted to suit your needs?

The human rights-evoking *social model* has a strong following internationally (Buchanan, 2001; Buden, 2002). The dominant view in South Africa too is that "disability is a social construct and most of its effects are inflicted on people by the social environment" (SAHRC Report, 2002:9).

4.1.3 Architectural Model

This model is attributed to Selwyn Goldsmith (1997). Epistemologically, this model would fit the constructivist phenomenological standpoint that is greatly enriched by the fact that Goldsmith himself was an architect with a disability who relied on a wheelchair for ambulation (*ibid*). The *architectural model* builds upon the principles of *barrier-free design* and *accessible design*

(*ibid*). This model encourages architects to take responsibility for the built environment and in so doing counter various forms of "architectural discrimination" (Goldsmith, 1997; 2000).

Consequently, *product design* (practically synonymous with *industrial design* in this context) is also challenged to adopt bottom-up approach thus helping to create products with "*de-disabilizing effects*" for disabled users (Goldsmith, 1997:156; 2001:25.3) – a theme that resonates with the challenge posed by Victor Papanek (1971) to industrial designers to similarly take responsibility for the harm occasioned by their profession. Goldsmith (1997:119-120) is also critical of the *microist* perspective (that "what suits the disabled is good for everyone") and instead calls for a *new paradigm* that avoids stigmatisation by adopting a *macroist* view (that "what suits everyone is good for the disabled"). The *architectural model* is ontologically allied to the *biopsychosocial* one and aligns conceptually with "universal design [whose] aim is that buildings should be convenient for all their users, with *architectural discrimination* being avoided" (Goldsmith, 2000:5).

4.1.4 Biopsychosocial Model

The debate on the *medical* versus *social* model should not be seen as antagonistic to the practice of *adaptive design* which focuses on retrofitting products and environments "to the special needs of those with disabilities" (Pirkl, 1994:227). There are countless people with disabilities whose plight has been demonstrably ameliorated by benign medical interventions. Notwithstanding, the debate has brought into sharp focus the significance of the perspective of people with disabilities. Consequently, the quest for an inclusive 'universalist' model became urgently apparent. Whyte and Ingstad (2007:13) suggest that a "universal view of disability as something everyone can experience in the course of a life" tended to reconcile the opposing views on disability. In 1980, the World Health Organization (WHO, 2001) formally issued the *International Classification of Impairments, Disabilities and Handicaps* (ICIDH) as "a tool for the classification of the consequences of disease" – a tool steeped in the medical model.

The International Classification of Functioning, Disability, and Health (ICIDH-2 or ICF) was published by the WHO in 2001 as a response to the need for a more balanced multi-purpose classification protocol (Pasha & Pasha, 2006) – this more holistic approach is called the *biosocial model* (Imrie *et al.*, 2001) or more commonly as the *biopsychosocial model* (WHO, 2004; Pasha & Pasha, 2006). The ICF takes cognisance of this factor by accommodating the biopsychosocial and environmental factors in the classification of disability (WHO, 2002:3):

ICF puts the notions of 'health' and 'disability' in a new light. It acknowledges that every human being can experience a decrement in health and thereby experience some disability. This is not something that happens to only a minority of humanity. ICF thus 'mainstreams' the experience of disability and recognises it as a universal human experience. By shifting the focus from cause to impact it places all health conditions on an equal footing allowing them to be compared using a common metric – the ruler of health and disability.

The ICF is however criticised over the low representation and participation among developing countries (with only one informant from Africa) as well as its scant interrogation of cultural factors during the development of the instrument (Whyte & Ingstad, 2007). Notwithstanding, these limitations, the ICF offers data collectors and researchers a common rubric with which to address the multi-dimensional nature of disability thus allowing for greater cross-disciplinary application, increased accuracy in international comparative studies, and more realistic benchmarking potential amongst the 191 countries that subscribe to the ICF (WHO, 2002; Boonzaier, 2006; Leonardi *et al.*, 2006; Pasha & Pasha, 2006). Table 4.2 summarises the main features of the *medical*, *social* and *bio-(psycho)social* models.

medical model	social model	bio-social model
personal tragedy theory	social oppression theory	bio-social theory
personal problem	social problem	personal/social problems
individual treatment	social action	individual/social action
medicalisation	self-help	medical/self-help
professional dominance	individual/collective responsibility	collective responsibilities
expertise	experience	expert/lay experiences
adjustment	affirmation	
individual identity	collective identity	individual/collective identities
prejudice	discrimination	prejudice/discrimination
attitudes	behaviour	
care	rights	care combined with rights
control	choice	control combined with choice
policy	politics	political and policy change
individual adaptation	social change	individual adjustment and social change
-	-	-

Table 4.2 Contrasting medical, social and bio-social models of disability (sources: Oliver,1996:34; Barnes et al., 1999:30; Imrie et al., 2001:34)

4.2 Responses to the Needs of the Vulnerable

This section discusses the different cases that justify specific responses to the issue of disability in South Africa as well as their international precedents.

4.2.1 The Legal Case

The *Americans with Disabilities Act* (ADA) of 1990 is widely believed to be the most significant legislation of its kind internationally (Welch & Palames, 1995; Goldsmith, 1997; Koncelik, 1998; Vanderheiden & Tobias, 1998). This pivotal law prohibits discrimination and requires all private and public institutions to promote accessibility in all spheres of public life – such facilities, services and transportation (Jaeger & Bowman, 2005). Jaeger & Bowman (2005:42) report:

The ADA prohibits discrimination against persons with disabilities by various private and public institutions, including state governments, and provides a mechanism for legal protection and remedies [and] also prohibits discrimination by private organizations providing public accommodations, which traditionally has included hotels, restaurants, offices, housing and shopping centers, among many others.

Casserley and Ormerod (2003) also highlight the legal imperatives for designers and other professionals in the United Kingdom (UK) with respect to the *Disability Discrimination Act* (DDA) of 1995 and the *Special Educational Needs and Disability Act* (SENDA) of 2001 in that country. The former is almost identical in discourse, scope and spirit with the ADA, as well as that of the similarly named *Disability Discrimination Act* (DDA) enacted in 1992 in Australia (DDA, 1996; Madden *et al.*, 1997).

In South Africa, the highest law of the land is the *Constitution of the Republic of South Africa* (Act No. 108 of 1996). Section 9 of the *Bill of Rights* of The Constitution explicitly prohibits discrimination on the grounds of race, gender, age and disability among others, and instead seeks to uphold human dignity through the promotion of diversity and tolerance among all its citizens (South Africa, 1996). Adherence to this constitutional imperative is promulgated through the *Promotion of Equality and the Prevention of Unfair Discrimination Act* (Act No. 4 of 2000) which binds "the State and all persons" (South Africa, 2000:6). However, this Act exempts "any person to whom and to the extent to which the *Employment Equity Act* (Act No. 55 of 1998), applies" (*ibid*). This will be discussed further in the next sub-section.

4.2.1.1 Disability-specific Legislation

The *Promotion of Equality and the Prevention of Unfair Discrimination Act* (hereinafter referred to as the *Equality Act*) follows the spirit of the pioneering ADA of the USA, as well as the DDA of Australia and the United Kingdom (M'Rithaa, 2006). The *Equality Act* takes cognisance of two important factors (South Africa, 2000:6):

- (a) The existence of systemic discrimination and inequalities, particularly in respect of race, gender and disability in all spheres of life as a result of past and present unfair discrimination, brought about by colonialism, the apartheid system and patriarchy; and
- (b) the need to take measures at all levels to eliminate such discrimination and inequalities.

The *Equality Act* draws on recommendations of the *Integrated National Disability Strategy* to spell out the legal obligations and implications of the Act on all role players in the public and private domains (SAHRC Report, 2002). Further, the *Equality Act* provides for the establishment of 'equality *courts*' to determine 'fairness or unfairness' within the ambit of the Act (South Africa, 2000:8). The courts seek to determine culpability based on discrimination on 'prohibited grounds' (*ibid*). As discussed previously, exceptions are made if there is proof that 'the discrimination is fair' – such as in the legally sanctioned transformational measures of affirmative action to redress gender, race or disability imbalance in employment equity (South Africa, 1998; 2000). For purposes of clarity, the following definitions as provided by the *Equality Act* are reproduced in Table 4.3 (South Africa, 2000:4-5):

 Table 4.3: Definitions and Interpretations of Terms Used in the Promotion of Equality and Prevention of Unfair Discrimination Act (source: South Africa, 2000:4-5)

Term	Definition
age	includes the conditions of disadvantage and vulnerability suffered by persons on the basis of their age, especially advanced age
discrimination	 means any act or omission, including a policy, law, rule, practice, condition or situation which directly or indirectly:- (a) imposes burdens, obligations or disadvantage on; or (b) withholds benefits, opportunities or advantages from, any person on one or more of the prohibited grounds
equality	includes the full and equal enjoyment of rights and freedoms as contemplated in the Constitution and includes <i>de jure</i> and <i>de facto</i> equality and also equality in terms of outcomes
harassment	means unwanted conduct which is persistent or serious and demeans, humiliates or creates a hostile or intimidating environment or is calculated to induce submission by actual or threatened adverse consequences and which is related to:- (a) sex, gender or sexual orientation; or (b) a person's membership or presumed membership of a group identified by one or more of the prohibited grounds or a characteristic associated with such group
HIV/Aids status	includes actual or perceived presence in a person's body of the Human Immunodeficiency Virus (HIV) or symptoms of Acquired Immune Deficiency Syndrome (Aids), as well as adverse assumptions based on this status

prohibited	are:-
grounds	 (a) race, gender, sex, pregnancy, marital status, ethnic or social origin, colour, sexual orientation, age, disability, religion, conscience, belief, culture. language and birth; or (b) any other ground where discrimination based on that other ground —
	(i) causes or perpetuates systemic disadvantage; (ii) undermines human dignity: or
	(iii) adversely affects the equal enjoyment of a person's rights and freedoms in a serious manner that is comparable to discrimination on a ground in paragraph

The *Equality Act* also prohibits potentially discriminatory practices by the design professions such as (South Africa, 2000:7):

(a) denying or removing from any person who has a disability, any supporting or enabling facility necessary for their functioning in society;

(b) contravening the code of practice or regulations of the South African Bureau of Standards that govern environmental accessibility; and

(c) failing to eliminate obstacles that unfairly limit or restrict persons with disabilities from enjoying equal opportunities or failing to take steps to reasonably accommodate the needs of such persons.

The need to promote *universal access* is clearly inferred as one of the objectives of this Act (South Africa, 2000:7; SAHRC Report, 2002:8). The mandate to provide 'reasonable accommodation' through relevant *South African Bureau of Standards* codes and regulations fails given the rather generalised and unenforceable phrase that "every effort be made to provide such facilities" (South African Standard, 1990:152). Notwithstanding, there has been an attempt in *Part S: Facilities for Disabled Persons* at providing professional architects with guidelines on access provision in buildings (South African Standard, 1990; Ormerod *et al.,* 2005). Other requisite standards and building codes have been developed to complement *Part S* (South African Standard, 1993; South African National Standard, 2004). Recent revisions of ambiguous language and pertinent technical guidelines vis-à-vis national building codes and regulations appear to have removed the legal loopholes that previously frustrated the implementation of universal access (C. Thompson, 2008; P. Thompson, 2008).

To date there has been one recoded case of successful litigation through an *equality court* in South Africa in which a police station in Port Elizabeth was sued for being inaccessible to people with physical disabilities (Opperman, 2006; P. Thompson, 2006; Wetzel, 2008). There may well have been a much larger number of cases reported. The low visibility of such court cases may be attributed to complainants agreeing to settle such matters out of court as is the case in the USA and United Kingdom (Koncelik, 1998; Dong, Keates & Clarkson, 2003;

51
Casserley & Ormerod, 2003). By agreeing to settle out of court, disability and accessibility activists are denied vital ammunition to effectively counter such discriminatory practices which consequently continue unabated – further excluding disability concerns from the mainstream (Koncelik, 1998; Peters, 2000; Boonzaier, 2006; Opperman, 2006).

Koncelik (1998) asserts that emphasis on legislative compliance often leads to the adoption of the bare minimum requirement as the law is seen to be punitive in spirit – a more proactive approach would be to use incentives (as opposed to the threat of the law) to encourage voluntary compliance (Coleman, 2006). This latter approach is a more sustainable one that lends additional support for the 'business case' discussed in Section 4.2.4.

Other disability-specific legislation include the *Employment Equity Act* of 1998 which sets a minimum benchmark for designated organisations to have 4% of all their employees as being people with disabilities – *i.e.* those with "long-term or recurring physical or mental impairment[s] which substantially limits their prospects of entry into, or advancement in, employment" (South Africa, 1998:10; 2003b:8). The Act (*ibid*) and supplementary *Technical Assistance Guidelines for Employment of People with Disabilities* (South Africa, 2003b) legally enforces 'reasonable accommodation' wherein employers take "affirmative measures consistent with the purpose of this Act" – *ipso facto* in redressing equity imbalances in the workplace (South Africa, 1998:14).

Whereas the imperative for transformation in employment equity through *affirmative action* is an urgent issue for the country, the nobler objectives of the *Broad-Based Black Economic Empowerment Act* (BBBEE) of 2003 are compromised by the failure to take cognisance of long-term 'side-effects' on the national repository of intellectual and human capital (South Africa, 2003c). A more creative (and sustainable) strategy is required – a critical review of achievement of transformational goals achieved thus far, as well as an extrapolation of present trajectories by factoring in alternate scenarios – such as the potential benefits of *mentorship* and proactive *succession planning* in *performance management* systems for incumbent employees irrespective of their racial profiles (Macpherson, 2001).

The present dispensation perpetuates distrust of *positive discrimination* measures – a legacy of apartheid-era race and labour relations (Sherer, 2000; Kahn, Blankley, Maharajh, Pogue, Reddy, Cele & Du Toit, 2004). There are ethical implications in the current practice of

excluding 'white males' indefinitely whilst concomitantly seeking redress for practically every other group (Akers, 2008). Consequently, *affirmative action* policies need to be reviewed to resolve the apparent tensions arising from a desire to correct historical distortions in the country's human resource capacity, with the urgent need to promote diversity whilst recognising professional excellence as a means to forestall the debilitating effects of an unabated brain drain (Bagshaw, 2004; Kahn *et al.*, 2004; Akers, 2008).

UNESCO (1999; 2003a; 2003b) promotes the provision of *inclusive* educational modes of learning over the traditionally dichotomous practice of so-called 'mainstream' versus 'special' schools. Impetus for *inclusive education* through 'full-service' schools and colleges (to accommodate students with disabilities) is found in *The Higher Education Act* (South Africa, 1997a). Importantly, this Act encourages co-operation between public Higher Educational Institutions so as to share and benefit from best practices (South Africa, 1997:38.1). Integral to this co-operation is a critical examination of the physical environment as well as the philosophy and practice of teaching and learning (Welch & Jones, 2001; Western Cape, 2004; Boud & Falchikov, 2005; Howell, 2006; Inclusive Education Western Cape, 2006). In mainstreaming inclusive education over a twenty year period, e-learning, smart technologies, and distance learning options would need to be incorporated for greater accessibility consequently reducing the number of (excluded) students requiring *special education* interventions to learning – a system which has its roots in the *medical* model of disability (South Africa, 2001).

4.2.1.2 Political Implications

Historically, the needs and rights of previously politically marginalised, socio-economically disenfranchised and vulnerable sections of society were neglected by the apartheid regime in South Africa (Howell *et al.*, 2006). Further, the apartheid dispensation left behind "a discriminatory and weak legislative framework which has sanctioned and reinforced exclusionary barriers" (ODP, 1997). Subsequently, DPSA aligned itself with the political struggle for self-determination of blacks in the country as they shared kindred political and moral aspirations (Rowland, 1984). Upon the advent of democratic rule in 1994, the DPSA successfully lobbied for its agenda to be included in all official policy formulation – a fact that was strengthened by the *Office on the Status of Disabled Persons* (OSDP) being integrated into the *Office the Deputy President*, Thabo Mbeki (ODP, 1997; OSDP, 2003; Matsebula *et al.*, 2006). DPSA's pervading influence has led to the formation in May 2009 of the new *Department of Women, Youth, Children and People with Disabilities* headed by Ms Noluthando

Mayende-Sibiya; as well as the appointment of "the first person with a disability in Cabinet as Deputy Minister for Public Works" Honourable Henrietta Bogopane-Zulu (Matsebula, 2009). The influential *Integrated National Disability Strategy* was proposed to guide the implementation of the ideals of equity and inclusiveness as enshrined in the *Bills of Rights* by countering "the key forms of exclusion responsible for the cumulative disadvantage of people with disabilities [namely] poverty, unemployment and social isolation" (ODP, 1997). When Thabo Mbeki succeeded Nelson Mandela as president in 1999 the disability desk concurrently moved to the *Office of the State President* thereby consolidating the political power it had gained from its propinquity to the most powerful office in the land (Matsebula *et al.*, 2006).

South Africa's political commitment to the disability movement led to the establishment of the Secretariat for the *African Decade for Persons with Disabilities* (APDP) in Cape Town (South Africa, n.d; OSDP, 2003; United Nations Enable, 2004a; APDP, 2006; Chalken *et al.*, 2006). The mandate of the ADPD was recently extended by the *Africa Union* for an additional decade – till 2019 (Dube, 2008). Disability continues to enjoy growing recognition and integration into the majority of local/municipal, regional/provincial and national/state legislation and policy formulation. *The Declaration of Bamako* on Inclusive Education acknowledges the fact that the New Partnership for Africa's Development (NEPAD) omitted specific mention of the promotion of people with disabilities and seeks to remedy the situation (Inclusion Africa, 2002).

Notwithstanding, the mainstreaming of disability issues should be achievable provided there are clear, measurable goals set by the NEPAD Disability Desk within the framework of ADPD's extended tenure – given the Secretariat's track record of effective high (Ministerial) level lobbying for the successful implementation of disability-specific policies in Africa thus far (Disability, 2003; USAID, 2005; Dube, 2008). The APDP is also a key actor in a number of international initiatives, including the *New EU-Africa Strategy* that was finalised in December 2007 (DPOD, 2008). According to DPOD (*ibid*), "the strategy pays particular attention to persons with disabilities in the fields of health and education. The adopted strategy gives recognition of the need for full access to health and education services for women, children and men with disabilities as well as in the context of the Millennium Development Goals".

4.2.2 The Human Rights Case

A number of countries and regions have adopted human rights responses to issues of ageing, disability and allied social concerns (Hevi, 2004). For example, Europe adopts a human rights

and ethics-informed standpoint for its design-related interventions thereof (European Commission, 2002; EIDD, 2004; EDeAN, 2006). South Africa has a rich and well-publicised human rights tradition. The *South African Human Rights Commission* (SAHRC Report, 2002:7) adopts a stance similar to that of other progressive societies in stating rather categorically that:

Inaccessible environments deny people with disabilities their rights to equality, dignity and freedom, amongst other fundamental human rights. Lack of physical access, both to and within built environments, is a major factor contributing to the ongoing exclusion of people with disabilities from mainstream society.

The United Nations (UN) General Assembly set the moral foundation upon which subsequent human rights responses to disability (including the SAHRC one cited above) are based following celebrations to mark the end of the *Decade of Disabled Persons* in 1993 (United Nations Enable, 2004b). The UN cannot enforce rules (as in the legal case discussed previously), but it nonetheless behoves member-states to exercise good faith and a sense of moral obligation and natural justice. In the *Standard Rules for the Equalization of Opportunities for People with Disabilities* (hereinafter referred to as Standard Rules), the UN underscores the right to equal participation by all people through a comprehensive series of 22 Standard Rules categorised under three sections covering *Preconditions for Equal Participation; Target Areas for Equal Participation; and Implementation Measures (ibid)*.

The Target Areas for Equal Participation category consists of a set of 8 Standard Rules: Rule 5 – Accessibility; Rule 6 – Education; Rule 7 – Employment; Rule 8 – Income maintenance and social security; Rule 9 – Family life and personal integrity; Rule 10 – Culture; Rule 11 – Recreation and sports; and Rule 12 – Religion (ibid). These target areas all incorporate the need for persons with disabilities to enjoy the same human rights afforded to every other citizen.

Standard Rule 5 specifically covers the need for accessibility within the physical (or built) environment, as well as access to information and communication (*ibid*). In South Africa, this latter element is addressed by the principles of *Batho Pele* (literally meaning [putting] "people first" in *se*Sotho) – a "practical implementation strategy for the transformation of Public Service Delivery" that forms the basis of the public sector's service charter to its people – whom it views as customers – and is expressed as eight principles, including those of *Access* and *Information* (South Africa, 1997b; 2004b). The principles embodied in *Batho Pele* are strongly influenced by the human-centred ethos of *ubuntu* which is discussed in Chapter 6.

4.2.3 The Welfare Case

The modern industrial society is associated with the centrality of work "not simply because it produces the goods to sustain life but also because it creates particular forms of social relations" (Oliver, 1990:85). Unfortunately, for people with disabilities and other physical limitations (including gender and age-related ones), equal and gainful contribution in such social relationships was not possible due to the emphasis on productivity over participation (Illich, 1973; Oliver 1990). In South Africa, African and Coloured people (broadly classified as 'Black') were also excluded due to job reservations put in place to favour the white elite during the apartheid era (ODP, 1997; Sherer, 2000). This in effect led to the exclusion of people with disabilities, women, children, the elderly, and Blacks from accessing economic opportunities thus leading to high levels of dependency amongst these groups.

Exclusion led to dependency, which in turn resulted in a welfarist response from the State in an attempt to redress these imbalances (South Africa, 2008c). South Africa is ranked as "an upper-middle income country based on average income [yet], some of the nation's social indicators are comparable to those of the poorest countries of the world" (Samson, MacQuene & van Niekerk, 2005). Further, official unemployment in South Africa stands at 26%, and poverty presently affects close to half of the entire population of a country with "one of the most severe measures of inequality in the world" (*ibid*). This situation has been further exacerbated by the current global economic downturn/recession with the national mining industry being worst affected.

The Department of Social Development (previously named the Department of Social Services and Poverty Alleviation) offers a wide range of social grants including "old-age; disability; child-support; foster-care; care-dependency and war veterans' grants; and temporary grant-in-aid relief" (South African Government Information, 2008). Parliament provided the Department with the necessary legal backing by enacting specific legislation such as "the *Children's Act* (2005) and the *Old Persons Act* (2006) which facilitated transformation in services provided to children and older persons" (South Africa, 2008c:17).

As of September 2007, a total of 12 million people (including some 8 million children) in South Africa were receiving non-contributory social assistance – "the country's most effective poverty-alleviation programme" (*ibid*). The number of people dependent on one type of social grant or other represents a staggering 25% of all South Africans – a daunting challenge for any country (Mbeki, 2009). The greatest beneficiaries would appear to be children (particularly in rural and

peri-urban settings) – through enabled access to schooling as well as superior nutrition (Samson *et al.*, 2005). Whereas the efforts by the country to redress poverty, inequality and unemployment are laudable, the figure has "increased from ZAR 42.9 billion in 2002/03 to ZAR 74.2 billion in 2005, representing an increase of 20% a year [and] from 2.9% of gross domestic product in 2003/04 to 3.3% in 2006/07" (South African Government Information, 2008). Barnett and Whiteside (2002) support community-based (as opposed to statutory support) as the most sustainable care model for the care of orphans in South Africa. Table 4.4 compares the cost of various care models in this regard (Gow & Desmond in: Barnett & Whiteside, 2002:207-208):

Care Model	Annual cost (minimum increase standard)	Increase	Reason for increase
Community-based support structures	276	—	—
Home-based care and support	306	+30	Process for identificationProcess for placement and grant access
Informal fostering non- statutory foster care	325	+19	Higher degree of supervisionSmaller scale
Statutory adoption and foster care	410	+85	Security of accommodationQuality of accommodationMore administration
Unregistered residential care	956	+546	 High staff to child ratio Provision of emergency care Care of sick children
Statutory residential care (caring mainly for HIV positive children)	2590	+1634	 Very high staff to child ratio Care only for sick children Meet statutory requirements for a children's home High overheads On-site medical care On-site preschool education

Table 4.4 Cost of orphan care in South Africa (in Rands) (source: Gow & Desmond, in: Barnett &
Whiteside, 2002:207-208)

Further, as Nattrass (2007:179) argues, "despite this relative generous level of social assistance, pressure on the welfare system continues to grow – most notably on disability grants which rose from about 600,000 in 2000 to almost 1.3 million in 2004". Large numbers of people living with Aids and tuberculosis continue to claim disability grants "as the disability grant is the only social grant available to adults of working age" (*ibid*). Unfortunately the grant system attracts unscrupulous and opportunistic individuals and is thus not immune to abuse. Due to the high HIV/Aids prevalence and exceptionally high unemployment levels in the country, the welfare case in South Africa is ultimately unsustainable – it offers only short-term financial relief and is attracting ever more people, thereby perpetuating dependency and promoting an entitlement syndrome amongst the general populace (Mbeki, 2009).

Oliver (1996:77) also advances an argument against the welfare case in view of the fact that "providing welfare services on the basis of individual need has aided the process of excluding disabled people from society rather than facilitated their inclusion". Similarly, Samson *et al.* (2005) are critical of the means-test used to determine eligibility for specific social grants and instead propose that a more inclusive approach be taken thus eliminating the need for the means-test altogether. Further, Samson *et al.* (*ibid*) support the introduction of the *Basic Income Grant* – "a universal grant to all South Africans" introduced in a phased manner. With the *Basic Income Grant* (*ibid*):

No means test would be required: everyone in the country, rich and poor, would receive the grant, which would therefore act like a tax rebate for upper-income recipients. The universal character of the grant would, it is argued, economise substantially on the government's scarce administrative resources, while at the same time eliminating the economic costs arising from the distortionary nature of the means-test. The scheme has been advocated widely by civil society groups, and also by the Minister for Social Development.

4.2.4 The Business Case

The business case views the different categories of vulnerable populations as potential customers for more inclusive goods and services. This would appear at first to be obvious, save for the fact that a vast number of businesses in South Africa have failed to grasp this potential. Opperman (2006) approximates for example, that for every person with a disability interested in using a restaurant, at least another four 'temporarily able-bodied' (or non-disabled) people typically accompany the disabled person. An accessible restaurant would effectively be attracting the business of four 'companion customers' in addition to the disabled customer by ensuring the access needs of the latter. The same applies to businesses that are child-, elderly- or gender-sensitive and/or responsive (Mueller, 2001, 2003; Owen & Johnston, 2003; O'Herlihy, 2005).

Any inclusive business strategies that anticipates and reasonably accommodates the needs of the elderly would in effect be an enlightened response to the needs 'of our future selves' (Coleman, 2001; 2006). Further, people with disabilities offer useful and usable skills to the open labour market, and their inclusion would greatly enhance the intellectual capital or human resource capacity of progressive businesses (Oliver, 1996; Coleman, 2001; Coy, 2003; Barredo, 2006; Deglon, 2008). Coy (2003:159) reinforces this argument by stating thus:

Simply, failure to recruit people with disabilities and to provide an accessible workplace is to deprive an organisation of a pool of talent and skills that could be of great benefit. Moreover, failure to provide an accessible workplace also results in both a loss of staff who develop physical impairments during the course of their careers and long periods of absence from the workplace during recovery from illness or injury; these scenarios are where significant costs are incurred.

4.3 Global Trends

There is a growing awareness of the needs of groups considered as marginalised, vulnerable and/or minorities within societies. Attention to such groups has resulted from the very public commemoration of such significant milestones and events as:

- The UN Decade of Disabled Persons 1983-1992;
- The African Decade of People With Disabilities 1999-2009;
- The Year of the Disabled in Europe 2003;
- The International Women's Day 8th March;
- The UN Autism Awareness Day 2nd April;
- World Health Day 7th April;
- The International Day of the Family 15th May;
- The International Children's Day 1st June/20th November;
- The International Refugees Day 20th June;
- The International Day of the Left-Handed Person 13th August;
- The International Day of the Older Person 1st October;
- World Aids Day 1st December;
- The International Day of the Disabled 3rd December; and
- Human Rights Day 10th December.

Instructively, there have been spirited debates and campaigning to extend the *compulsory retirement age* – a sign that people can, and want to work for as long as they opt to. In the United Kingdom, the debate has been on pushing this age to 65. Gribben (2009) reports that robust responses such as these serve to "ensure that older British workers are judged on their skills and abilities rather than their age." Similar debates have been reported in a diverse number of other industrially developed and developing countries such as Australia, India, Japan, Kenya, and the USA. This is in part due to the dual phenomena of general life expectancy increases in tandem with lowered fertility rates – this leads to fewer members of

society being in the active labour force at any given time. Increasing the age of retirement also reduces the dependency of a large number of people on social grants and related welfare services – a welcome reality for many states grappling with ageing populations (World Assembly on Ageing, 2002; Samson *et al.*, 2005; UNFPA, 2008). Closer to home, an accessible and inclusive dedicated park for disabled children was recently opened in Johannesburg (Africa News Update, 2008).

A number of countries have pursued strategies to ensure equity in wages paid to disabled employees. For example, in Denmark the emphasis is on ensuring "spaciousness of the labour market" (Knudsen, 2009). Consequently, the move from sheltered to *supported employment* in the open labour market has raised the employment rate of disabled persons in Denmark to an impressive 57%, whereas the employment rate for the general population stands at 80% (*ibid*). Employers are offered subsidies ranging from half to two-thirds thus encouraging them to pay full (market-related) wages to their employees with disabilities (*ibid*).

Momoza (2009) reports of a similar change in the Western Cape's *Department of Social Development* (DSD) vision for protected and sheltered employment of persons with disabilities. This shift implies a departure from the traditional view of the disabled as passive recipients of grants to one of recognising the vast human capital potential that should be harnessed (Department of Social Development, 2007). Subsequently, people with disabilities should no longer be viewed as a liability or burden, but as patently creative co-producers of desirable social capital (Fransolet & M'Rithaa, 2009). The DSD initiated the Disability Workshops Development Enterprise (DWDE) to facilitate the implementation of a *co-operative model* that should ultimately convert the 42 participating protected/sheltered workshops into viable micro-enterprises able to compete in a sustainable manner in the open labour market in the province and beyond (Department of Social Development, 2006; Deglon, 2008; Momoza, 2009).

4.4 Summary

This chapter noted the various responses to the needs of *vulnerable populations* in general, but focussed on the different 'models' and cases vis-à-vis disability in particular. Globally, the *human rights* concerns for *social justice* for all categories of people is adding impetus to ensure that key actors in the public and private sectors uphold these stated principles. Further, by challenging the *medical* or *individual model*, the prevailing attitudes of the so-called experts,

specialists and professionals towards vulnerable groups is brought under scrutiny as a result of the pervasive influence of the *social model*.

Technological advancements are expected to influence requisite changes in the fields of education, employment and social services. Technology now enables people with disabilities to access learning better and perform tasks that were previously thought impossible, whilst the concept of a compulsory retirement age no longer seems ethically defensible. Finally, the law is seen to enforce the adoption of minimum standards, whereas the social welfare case perpetuates dependency – these two are high-maintenance options requiring complex fiscal, logistical and organisational support. The business case on the other hand is intrinsically more sustainable and has a self-regulating dynamic, whilst simultaneously ensuring greater diversity and enhancing practicable innovation. This latter case informs the discussion on UD in the next chapter.

CHAPTER FIVE UNIVERSAL DESIGN AND RELATED CONCEPTS

5.0 Introduction

This chapter explores the developments and scope of applications of several concepts leading up to a discussion on their relevance to the main focus of this thesis, namely, *Universal Design* (UD) with respect to various design professions. A number of commentators acknowledge the fact that the design profession has often (knowingly or otherwise) been 'part of the problem' in perpetuating exclusionary practices. Consequently, a number of cross-disciplinary specific concepts have been formulated to redress the situation from diverse geographic and professional milieu.

5.1 Design Responses to Counter Exclusion

The various cases discussed in Chapter Five resulted in a spirited effort to make 'offending' professions (and by extension, key professional actors) accountable for the systemic exclusion of various vulnerable populations from participating in mainstream social activities (Buden, 2002). Design is a powerful tool in the shaping of artefacts and should be harnessed to tackle the problem of design exclusion. As Edeholdt (2004, cited in Jönsson & Certec, 2006b) categorically asserts, "design has *direction* [and] at its core, it deals with how things *should* be, what they *should* become, rather than how they *are*". Dalcher (2006:253) acknowledges the 'wicked' nature of the discipline in that "design is neither orderly nor linear; it implies a continuous and active search to resolve trade-offs and satisfy changing constraints". Designers should aspire to be 'part of *the* solution' – "good design enables, bad design disables" (EIDD, 2004). This section addresses the responses of the design professions to this challenge.

5.1.1 Accessible Design

Accessible design attempts to make products, places and spaces "more usable by people with disabilities" (Vanderheiden, 2001:65.4). Wijk (2001) suggests that the term *accessibility* implies an interim, transitional or stop-gap measure – the antithesis of an *integral* or *integrated* approach (*ibid*; Sandhu, 2001). The limitations of accessible design have been noted by Steinfeld (1994, cited in Imrie *et al.*, 2001:14) who states that:

accessible design acknowledges that people with disabilities have a right to access and use of products and [built] environments, but it doesn't go far enough because it doesn't express social integration.

Wijk (2001) blames the educational training of designers and architects that depended on idealised and unrealistic anthropometric representations of end-users thus perpetuating the fallacy of "training in the average", as opposed to an emphasis on a more sophisticated appreciation of anatomical/physiological, psychological and anthropological diversity.

5.1.2 Barrier-free Design

Barrier-free Design (BfD) emerged as a response to what Gleeson (2001 in Imrie *et al.*, 2001:153) refers to as 'architecture of apartheid'. This is essentially a systemic problem resulting in poorly planned and designed built environments which buttressed structural and attitudinal barriers to people with disabilities (Goldsmith, 1997; Mace, 1998; Imrie *et al.*, 2001; Danford & Tauke, 2001). As expected, this concept was widely used in the architectural profession and initially focussed on the needs of wheelchair users. BfD is closely allied to *Universal Access* (UA) which is elaborated in Section 5.1.9. Additionally, BfD complements the concept of *reasonable accommodation* – whose aim is to "reduce the impact of the impairment on the person's capacity to perform the essential functions" required of them (South Africa, 2003b:13). The limitation of BfD is however very similar to that of accessible design in the treatment of disability as a *special provision*, as opposed to being a core criterion in the design ethos (Goldsmith, 1997; 2000; 2001; Imrie *et al.*, 2001).

5.1.3 Design-for-All

This concept originated in Europe and finds expression through the *European Universal Design Principles or European Concept for Accessibility* (Wijk, 2001; European Commission, 2002; EIDD, 2004; Coleman, 2006; EDeAN, 2006; Klironomos, Antona, Basdekis, & Stephanidis, 2006). In its present usage, *Design-for-All* (DfA) is virtually synonymous with *Inclusive Design* and *Universal Design* – which evolved in the United Kingdom and the USA respectively (Næss, 2003; Design Council, 2006; EDeAN; 2006; Tiresias, 2006). Whilst noting the ontological propinquity between UD and DfA, Hansson (2006:27) links the latter specifically with "discussion about accessibility and disabled people", and further that DfA "is a prioritized area within the European Union and on governmental levels in the member countries" (*ibid*). This political goodwill and support is unique to DfA and characterises its mode of diffusion and dissemination – in the case of UD (and *Inclusive Design*), consumers and professional designers are the main promoters. DfA is defined as the "means of designing, developing and marketing mainstream products, services, systems and environments to be accessible and usable by as broad a range of users as possible" (Build-for-All, 2006:9).

According to the *EIDD Stockholm Declaration* of 9th May 2004, DfA "is design for human diversity, social inclusion and equality [– a] holistic and innovative approach [which] constitutes a creative and ethical challenge for all planners, designers, entrepreneurs, administrators and political leaders" (EIDD, 2004). DfA envisages achieving these goals in three specific ways (Build-for-All, 2006:9):

- by designing products, services and applications that are readily usable by most potential users without any modifications;
- by designing products that are easily adaptable to different users (e.g. by adapting their user interface); and
- by standardising interfaces of products to be compatible with specialised equipment (e.g. technological aids for disabled persons).

The three tiers of DfA render the approach versatile and comprehensive in both scope and reach, whilst allowing for greater design flexibility (Coleman, 2006; Build-for-All, 2006).

5.1.4 Design for Disability

According to Clarkson, Coleman, Keates and Lebbon (2003:599), *Design for Disability* can also be described as *Design for Special Needs* and is related to the medical model of disability in that it "mainly focuses on aids and adaptations to everyday equipment and buildings [with] the underlying intent [being] essentially prosthetic, with origins in post-trauma rehabilitation, particularly of war veterans". This approach to design has since been superseded by the philosophy and practice of *inclusive design*, particularly in Europe (*ibid*).

5.1.5 "Extra-Ordinary" Ergonomics

According to Wilson (1990:3), the approach of *ergonomics* is about "designing for people". Much of the scope of this thesis relates to what Davis (2006:2) describes under *organisational ergonomics* as it focuses on investigating the introduction of new systems and work paradigms vis-à-vis UD. *Ergonomics* actively places people at the centre of its endeavour, and is patently *user-centred*, yet it inadvertently failed to take cognisance of *special human needs* until recently (Danford & Tauke, 2001). Indeed, Davis (2006) specifically equates the three concepts of *ergonomics*, *user-centred design* and *inclusive design*. Wells, Norman, Frazer, Laing, Cole and Kerr (2003:5) state unequivocally that "ergonomics is human-centred design". The hypothetical principle of the *limiting user* as elaborated by Pheasant (1998:23) approximates a

concession towards "extra-ordinary" individuals, but it places constraints on its own usefulness by focusing exclusively on *reach* and *clearance* (in relation to *anthropometric* characteristics).

Anthropometrics, as an indispensable branch of ergonomics, concerns itself with measurements pertaining to the human body such as body size and strength (Pheasant, 1996:7). Kroemer, Kroemer, and Kroemer-Elbert (2001) attempt to ameliorate this situation by proposing ergonomic design guidelines for special populations. Subsequently, Kroemer (2006) presents a dedicated work on *"extra-ordinary"* ergonomics with a focus on *"Small and Big Persons, the Disabled and Elderly, Expectant Mothers, and Children"*. Kroemer (2006:1) states that "the common user population is not a unitary, homogenous assemblage but rather a group of individuals with diverse characteristics and varying capabilities".

Further, in acknowledging the fallacy of designing for "regular use by normal adult" user populations, "extra-ordinary" ergonomics intrinsically embraces the great diversity of such human factors as "strength, mobility and endurance" (*ibid*). The resultant anthropometric data on "extra-ordinary" populations is particularly useful to designers with regards to changes that occur with natural growth, ageing and pregnancy – *i.e.* data pertaining to children, the elderly and women. It is however not possible to generalise the changes occurring amongst people with permanent disabilities as there is a far greater diversity than is commonly appreciated (*ibid*). Table 5.1 (Kroemer, 2006:77) summarises the diverse constituency discussed herein.

	Energetic Capabilities (metabolism, circulation, respiration)	Biomechanical Capabilities (strength, power, mobility, endurance)	Sensations, Perception, Decision Making, Controlling Actions	Capabilities to Endure and Function in Heat and Cold	Stress Tolerance
PREGNANCY	reduced, diminishes further with run of pregnancy	reduced, diminishes further with run of pregnancy	no change except in taste and smell	reduced, diminishes further with run of pregnancy	reduced, diminishes further with run of pregnancy
CHILDHOOD	much smaller but increases	much smaller but increases	not systematically known	much smaller but increases	not systematically known
AGEING	reduced, diminishes further with ageing	reduced, diminishes further with ageing	reduced, diminishes further with ageing	reduced, diminishes further with ageing	depends on individual and circumstances
WITH PERMANENT DISABILITY	no general statements possible, depends on the individual				

Table 5.1 "Extra-ordinary" trends and changes in functioning (source: Kroemer, 2006:77)

The focus of "extra-ordinary" ergonomics is on those groups that are patently disadvantaged when data for the 'average' or '50th percentile' individual is used as a reference for the entire

population of user, irrespective of patent physiological and psychological differences. These "extra-ordinary" groups "who differ from this normative adult model [include]: pregnant women, children, and ageing and old persons, as well as those with disabilities" (Kroemer, 2006:4).

5.1.6 Inclusive Design

Inclusive design has a long and well-documented relationship with the study of *ergonomics* (Coleman & Pullinger, 1993; Coleman, 2006). Indeed, in 1993 the journal *Applied Ergonomics* dedicated the 1st issue of its 24th volume to the subject following a successful conference which was "organized jointly by the Ergonomics Society and DesignAge on 6 May 1992 at the Royal College [...] entitled *Designing for our future selves* in recognition of the fact that we all age and that it is inappropriate to regard older people as anything other than ourselves in the future" (Coleman & Pullinger, 1993:3).

Tiresias (2006) defines *inclusive design* as "the design of mainstream products and/or services that are accessible to, and usable by, as many people as *reasonably* possible on a global basis, in a wide variety of situations and to the greatest extent possible, without the need for special adaptation or specialised design". Gill (2004:3) and other commentators leave out the words in parenthesis in their functional definition of inclusive design. Further, the inclusion of the word 'reasonably' in the definition could either be read as an exit clause/disclaimer, or as an emphasis on the intent to achieve this noble ideal (Tiresias, 2006). As mentioned in Section 5.3, this term is considered synonymous with DfA and UD (Næss, 2003; Design Council, 2006; EDeAN, 2006; Tiresias, 2006).

Jordan (1999:174) defends *inclusive design* for its ability to "eliminate much of the social stigma associated with products for use by the disabled". Similarly, Clarkson, Keates, Coleman, Lebbon and Johnston (2000:4) acknowledge the host of diverse terms used to describe the 'goal of non-exclusive design' and adopt the term *inclusive design* to refer to 'designing for the whole population'. Imrie *et al.* (2001) see *inclusive design* as an advancement on the principles of UD and liken the approach of the former to that of *social design* as conceived by Sommer (1983, in Imrie *et al.*, 2001:18).

For heuristic purposes, two useful tools, the *Inclusive Design Cube* (Clarkson *et al.*, 2003:99; EDeAN, 2006) and the *Universal Design Pyramid* (Goldsmith, 2000:3), demonstrate that with a change in design thinking it is indeed possible to significantly increase the number of potential

users accommodated by products and built environments (Figures 5.1 and 5.4 respectively). Keates and Clarkson (2003a; 2003b) present a three-dimensional *inclusive design cube* as an extension of the *user pyramid* proposed by Maria Benktzon (1993). The *inclusive design cube* according to Clarkson *et al.* (2000:6) is meant for use as a "diagnostic tool" in seeking to expand the market potential of products.



Figure 5.1 The Inclusive Design Cube (source: Keates & Clarkson, 2003b:100)

5.1.7 Kyoyo-Hin

The term *Kyoyo-Hin* is functionally synonymous with *Inclusive Design*, *Design-for-All*, and UD (CUD, 1997; Kyoyo-Hin, 2001; Design Council, 2006; EDEaN, 2006). Related fields include *user-centred design*, *human-centred design*, *ergonomics*, *usability studies*, *design for disability*, *rehabilitation design*, *gerontechnology*, *transgenerational design*, and *lifespan design*. Kyoyo-Hin emerged as a dedicated response to the needs of the rapidly ageing population in Japan, before adopting broader strategies to include other vulnerable groups. The significant elderly population there led *Kyoyo-Hin* and *Kyoyo Services* to integrate UD with *accessibility design*, *lifespan (transgenerational) design*, and *adaptive (adjustive design)* (Kyoyo-Hin, 2001). The accepted practice is that "Kyoyo-Hin and Kyoyo Services are designed to be used by as many people as possible, including the elderly and those with disabilities" (*ibid*). The relationship between proprietary and general products vis-à-vis *Kyoyo-Hin* is depicted in Figure 5.2.



Figure 5.2 Kyoyo-Hin and Kyoyo Services (source: Kyoyo-Hin Foundation, 2004)

Japan's contribution to UD includes the formulation of ISO Guide 71: *Guidelines for Standards Developers to Address the Needs of Older People and Persons with Disabilities* (International Organization for Standardisation, 2001). Further, Japan has embraced UD as an effective vehicle to extend its own technological and marketing advantage and currently hosts the secretariat of the *International Association of Universal Design* (IAUD) in Yokohama (IAUD, 2006). The IAUD was inaugurated on 28th November 2003 with the objective of reviving the recession-hit Japanese economy and promoting UD among key actors in industry, government and academia in the spirit of *The International Universal Design Declaration* reproduced in Appendix C (Tsutatani, 2005; IAUD, 2006). Major companies such Canon Incorporated, Fuji Photo Film Company Limited, Mitsubishi Motors Corporation, Sony Corporation, and Toyota Motor Corporation have taken UD seriously – as reflected in their active membership in the IAUD (IAUD, 2006). The IAUD has also been instrumental in propagating the convergent UD activities such as the prestigious *Ron Mace Award* (which recognises special achievement in UD), as well as the highly publicised *Design for the Twenty-First Century* competitions aimed at emerging (student) designers (Tsutatani, 2005).

5.1.8 Transgenerational Design

Koncelik (1998:127) argues that the effects of ageing is not uniform "in the onset of changes or in the continuum of change" – the elderly are not a homogenous group with identical needs. Other commentators also emphasise the diversity and variance associated with the phenomenon of ageing (Benktzon, 1993; Pirkl, 1994; Jordan, 1999; Goldsmith, 2001). Further, according to Sixsmith and Sixsmith (1993), our individual differences in functioning and aptitude become more pronounced with the natural progression of ageing.

James Pirkl is credited with formulating the concept of 'transgenerational design' (Clarkson *et al.*, 2003:600; Jordan, 2008). This is one of the "practice-based strategies' that influences design practice (Coleman *et al.*, 2003:15). The emerging discipline of *Gerontechnology* is cited as a specific response to the phenomenon of population ageing – a collaborative effort "between European and North American academics and the practice and education-based programmes supported by the Helen Hamlyn Foundation in the UK" (Coleman *et al.*, 2003:10). *Transgenerational Design* is also known as *Design for Lifespan (ibid)*. A useful biopsychosocial indicator of the capacity for independent living among the elderly is the *Katz Scale* which measures the *activities of daily living* (ADLs). The ADLs are a set of six basic activities – *inter alia*; *bathing*, *dressing*, *toileting*, *transferring*, *continence* and *feeding* (Family Practice)

Notebook, 2000a). The *instrumental activities of daily living* are a further refinement of the ADLs and include more community-focused activities such as *transportation*, *housework*, *cooking* and *shopping* (Lawton & Brody, 1969; Family Practice Notebook, 2000c).

Pirkl (1994:42) argues that products and environments are usually directed toward a single disability, "thereby failing to accommodate the combined effects of multiple losses" associated with ageing. Additionally, Koncelik (1998:117) exhorts designers to "relate the human factors of ag[e]ing" to the products that they design. Similarly, the *empathic design* approach employed by Patricia Moore (2001) seeks to sensitise designers towards the needs of the elderly (and other vulnerable populations). This expression of empathy has resulted in the development of such creative approaches as *Design by story-telling* (Moggridge, 1993) and useful interactive tools such as the *Third Age* (or *Empathy*) *Suit* (as shown in Figure 5.3). The *Third Age Suit* was created to simulate ageing-related ambulatory and sensorial limitations and was used in the design of the Ford Focus automobile (Steinfeld & Steinfeld, 2001; Mueller, 2003).



Figure 5.3 The Third Age Suit (source: Mueller, 2003:327)

5.1.9 Universal Access

Universal Access refers to the ability of all people to have equal opportunity and access to a service or product from which they can benefit, regardless of their social class, ethnicity, background or physical disabilities (Obrenović, Abascal & Starčević, 2007). It is a vision, and in some cases a legal term, that spans many fields, including education, disability, telecommunications, and healthcare. It is tied strongly to the concept of human rights (Swart *et al.*, 2008:9). One thus can argue that *universal accessibility* is the (interactive) affordance that is achieved through active and tangible expressions of UD – or in other words – the effective implementation of UD principles results in *universally accessible* places, spaces and products.

5.2 Universal Design

According to Vanderheiden (2001:65.3), UD is "the process of looking at the design of a product from a general perspective, which would include all users". Story (2001:10.3) defines UD as "the design of all products and environments to be usable by people of all ages and abilities, to the greatest extent possible, without the need for adaptation or specialised design". This is the widely quoted official definition as phrased by the Center for Universal Design at the North Carolina State University (CUD, 1997; 2006; Story, 2001). The visionary pioneering work of Tim Nugent of the University of Illinois at the Champaign Urbana campus set the foundation for what would eventually become known as *Universal Design* (UD) (Goldsmith 2000:4). Through their passion and commitment, Tim Nugent, Ron Mace (Clarkson, Coleman, Keates & Lebbon, 2003:13; McGuire, Scott & Shaw, 2006:167), and Selwyn Goldsmith (Clarkson *et al.*, 2003:13) are acknowledged as champions of the UD movement.

UD claims its roots in *barrier-free design* (Mace, 1998) which according to Pirkl (1994) is synonymous with *accessible design*. Other sources suggest that UD has its origins in *barrier-free design* as well as *universal accessibility* in the field of architecture (Kyoyo-Hin, 2001; Preiser & Ostroff, 2001; CUD, 2006; EDeAN, 2006) with the actual definition accredited to Ron Mace who first coined the term *"Universal Design"* in 1985 (Kyoyo-Hin, 2001; Preiser & Ostroff, 2002; Clarkson *et al.*, 2003; Centre for Universal Design, 2006; EDeAN, 2006). Inclusive design, on the other hand – as discussed earlier – evolved from collaboration between *design* and *ergonomics* research and symposia (Coleman, 2006). Coleman (2003:10) notes the unique orientation of the related concepts in that "in Europe the focus has been on social inclusion, in the US on individual rights, and these drivers have resulted in a range of philosophic, academic and practical approaches".

At its inception UD emerged out of more immediate practical and pragmatic concerns within "the disability and design communities" (Crews & Zavotka, 2006:116), whilst inclusive design was informed primarily by the realisation of potential implications of future trends (such as ageing) that was supported by data on human factors (Taylor, Roberts & Hall, 1999). This might explain the assertion by Koncelik (1998:149) that UD practice is informed by an "intuitive approach". Koncelik (*ibid*), while conceding that UD is a "useful philosophic device", is critical of a cavalier "one-size-fits all" approach and instead calls for product variation (as opposed to "product universality"). This latter consideration is reflected in the multi-pronged strategy of *Design-for-All* as discussed earlier (Coleman, 2006; Build-for-All, 2006). Moalosi (2007) on the

other hand argues against seeking universality as a value and suggests that such an ideal is untenable (or undesirable) in our present post-industrial era, particularly where user- and context-responsive solutions are considered to be of paramount importance.

Notwithstanding, UD has enjoyed considerable recognition and popularity internationally (Aslaksen, Bergh, Bringa & Heggem, 1997; Adaptive Environments, 2006; Coleman, 1999; Balaram, 2001; Danford & Tauke, 2001; Gomes, 2005; Hansson, 2006), whilst Pirkl (1994:228) concurs with this view as he relates the ambit of transgenerational design exclusively to the natural process of ageing but describes UD as an "unreachable design ideal" notwithstanding the fact that Pirkl's definition of transgenerational design is virtually identical to that of UD.

Goldsmith (2001) uses the *universal design pyramid* (in Figure 5.4) first elaborated as the *user pyramid* by Benktzon (1993) to encourage inclusion by expanding the base of potential beneficiaries of more accessible built environments by mainstreaming products of a *universal* (as opposed to *special*) appeal. The *universal design pyramid* as a heuristic device elaborates the concept for "design for the broader average" whilst acknowledging that certain categories of users will always require carer assistance and/or assistive devices to facilitate functioning (Coleman, Bendixen & Tahkokallio, 2003:15). The underlying assumption however is still valid – that by considering the needs of the entire population (including those individuals with severe limitations in functioning) right from the outset, it is possible to significantly increase (and profit from) the expanded constituency of potential users in so doing. DePoy and Gilson (2004:46) however caution against the danger of commodifying a sensitive issue such as disability as "reproduction of human experience removes it from community and human interaction and places it in the arena of economic consumption".



Figure 5.4 The Universal Design Pyramid (source: Kose, 2003:313)

There appears to have been a phase of divergence and competition as different proponents sought to clarify their unique perspective on the UD debate. Further, one must take cognisance of the fact that a number of these UD-related concepts developed in isolation and completely independently of one another in response to the unique localised challenges identified in those contexts. This phenomenon is not uncommon as a reading on the birth and growth of *Ergonomics/Human Factors* reveals. As Kose (2003:311) argues "'design for all' should be the basic assumption of the design profession, and everything should have been designed with that idea in mind". Common UD-related concepts are summarised and presented in Table 5.2.

Concept	Definition
Universal Design (UD)	"Universal design is an approach to the design of all products and environments to be usable by everyone, to the greatest extent possible, regardless of age, ability, or situation. It serves people who are young or old, with excellent or limited abilities, in ideal or difficult circumstances. Universal design benefits everyone by accommodating limitations" (Universal Design Education Online, 2006).
Inclusive Design: mainly in the UK	"Inclusive design is not a new genre of design, nor a separate specialism, but an approach to design in general and an element of business strategy that seeks to ensure that mainstream products, services and environments are accessible to the largest number of people" (Design Council, 2006).
Design for All: Europe; India	"Design for All is a process whereby designers, manufacturers and service providers ensure that their products and environments address users irrespective of their age or ability" (EDeAN, 2006; Design for All Institute of India, 2006).
Kyoyo-Hin and Kyoyo Services: Japan	"Kyoyo-Hin and Kyoyo services are designed to be used by as many people as possible, including the elderly and those with disabilities" (Kyoyo-Hin, 2001:3).
Design for our Future Selves: mainly in the UK	"Concept developed by DesignAge programme to encourage young designers to engage with design for ageing populations. Became the theme for many events at the RCA and of an annual competition resulting in many concept exemplars of age- friendly design" (EDeAN, 2006).
Transgenerational Design: mainly in the USA	"The practice of making products and environments compatible with those physical and sensory impairments associated with human ageing and which limit major activities of daily living" (Transgenerational Design Matters, 2006).
Gerontechnology: mainly in USA and Europe	"Concept developed at Technical University of Eindhoven, The Netherlands (NL), with US and Finnish colleagues. Combines human factors, social sciences, gerontology and engineering. Applying technology to address age-related factors. Consumer/market oriented approach" (EDeAN, 2006).
Universal Access	"Universal access/access for all: to information and communications technology (ICT). Also used in assistive technology to refer to specialist interfaces and control devices to make ICT products accessible to people with high levels of impairment" (EDeAN, 2006).

 Table 5.2 Comparison of definitions of UD and related concepts (various sources)

The experiences of the cities of New York (in the USA) and Curitiba (in Brazil) serve as bestpractice exemplars of the positive psychosocial impact that the adoption of integrated strategies of UD engenders – particularly with respect to perceived quality of life for their respective residents. Levine (2003:10) reports of the enthusiasm with which the publication of *Universal Design New York (UDNY)* was received internationally – way beyond its intended audience which in part is due to the special focus on *technical solutions* to problems identified primarily in products and in the built environment. The information is presented as *"accessible"* against *UD* considerations – attesting to 'superiority in coverage' in adopting UD principles, as compared to the implementation of traditional baseline guidelines on accessibility (*ibid*).

The implementation of effective UD strategies would involve a trans-disciplinary approach in which designers, among others, could play a significant part (Zimmerman, 2006; Kurvinen, 2007). One would expect designers to adopt this progressive approach with little resistance or opposition. Clarkson, Keates, Coleman, Lebbon, and Johnston (2000:1) report however, that there is little industry awareness of the efficacy of adopting inclusive design strategies even in more progressive industrially developed economies like the UK. The situation is exacerbated in South Africa as few professional designers are aware of the (potential) benefits of UD. Koncelik (1998:122) reports of the "alternating panic and jubilation" that designers (including architects) went through soon after the ADA was promulgated in the USA. Similar dynamics would be anticipated in this country should inclusive considerations become a professional imperative.

One of the principal reasons for reluctance to embrace UD is that of perceived high cost of implementation. Subsequently, Kroemer (2006:1) argues that applying a UD approach to the design of everyday artefacts "is often impractical and expensive". This association of UD "as a consumer market driven issue" is acknowledged by Mace (1998) as well as by Danford and Tauke (2001:19) and reiterated by Levine (2003:17). There is an attempt however to temper this fact by postulating that disability should be everyone's concern – everyone is bound to experience some form of functional limitation with the natural progression of ageing. Hansson (2006:203) cautions that "if consumers are neglected in marketer's consumer conceptualisation they experience design exclusion as they likely will be excluded from the design process as well".

5.2.1 Principles of Universal Design

The seven *Principles of Universal Design* were developed by a cross-disciplinary team at the *Centre for Universal Design* at the *North Carolina State University* and with the current *Version 2.0 (4/1/97)* being published in 1997 (Centre for Universal Design, 1997; Appendix B). They provide a practical set of guidelines for designers, planners and architects to ensure that the resulting designs are universally usable. The *Principles of UD* are widely diffused and lend themselves easily to consumer marketing applications and in the development and utilisation of

"designed artifacts" in myriad contexts (Hanson, 2006:8). Though there is a stated intent to be as inclusive as possible, the authors/originators (*ibid*) acknowledge that:

The Principles of Universal Design are not intended to constitute all criteria for good design, only universally usable design. Certainly, other factors are important, such as aesthetics, cost, safety, gender and cultural appropriateness, and these aspects must also be taken into consideration when designing.

A number of related principles and guidelines have been developed elsewhere, with the *Principles for Kyoyo-Hin and Kyoyo Services* being the closest semantically (Kyoyo-Hin, 2001). The seven *Principles of Universal Design*, and the five *Principles for Kyoyo-Hin and Kyoyo Services* are virtually identical in scope and wording. Table 5.3 offers a helpful comparison and further demonstrates the similitude of the said principles.

Table 5.3 Comparison of Principles of UD (Centre for Universal Design, 1997) and Principles forKyoyo-Hin and Kyoyo Services (Kyoyo-Hin, 2001)

Principles of Universal Design	Principles for Kyoyo-Hin and Kyoyo Services	
1. Equitable Use. The design is useful and marketable to people with diverse abilities.	1. Adaptation to various types of physical and mental disabilities for easy use by as many	
2. <i>Flexibility in Use</i> . The design accommodates a wide range of individual preferences and abilities.	people as possible.	
<i>3. Simple and Intuitive.</i> Use of the design is easy to understand, regardless of the user's experience, knowledge, language skills, or current concentration level.	 Operation and use by means that are intuitively understood and require little psychological strain 	
 Perceptible Information. The design communicates necessary information effectively to the user, regardless of ambient conditions or the user's sensory abilities. 	2. Easily comprehensible communication using plural means (visual, aural, tactile, etc.).	
5. Tolerance for Error. The design minimizes hazards and the adverse consequences of accidental or unintended actions.	5. Concern for safe use in all aspects, including material, structure, function, procedure, and environmental features	
6. Low Physical Effort. The design can be used efficiently and comfortably, and with a minimum of fatigue.	4. Easy use with little physical burden (e.g., can be handled with little effort, easy	
7. Size and Space for Approach and Use. Appropriate size and space is provided for approach, reach, manipulation, and use regardless of the user's body size, posture, or mobility.	accommodation of motion and approach, etc.)	

5.2.2 Universal Design Applications in Education

Papanek (1995:210) in his discussion on *"Design Education for All"* proposes four distinct ways in which design could benefit education that can be summarised as follows:

- offering courses of study to design students at Higher Education Institutions (HEIs);
- educating and sensitising the youth on design-related matters;
- addressing physical barriers within the built environment; and
- (re)designing educational software for student-centred learning applications.

Ormerod *et al.* (2005:293) on the other hand views the task of the 'enlightened' designers as being complex in nature, and states that:

the context within which designers operate should not be under-estimated. Lawson (2004) suggests that the difficulty in achieving good design is that designers do not solve well-formulated problems but ones which are ill-structured, open ended and often referred to as wicked. [...] overall, it can therefore be concluded that while designers are keen to ensure that buildings and their environments facilitate social inclusion, there are significant barriers to achieving this, particularly due to lack of understanding of disability and how a person with a disability interacts with a building, and how the regulations and legislation can support this interaction rather than just providing a minimum standard.

The student/learner-centred Outcome-Based Education (OBE) system is the formal educational paradigm that has been formally adopted in South Africa (Makoni, 2000; Malan, 2002; Volbrecht, 2003; Tisani 2004; Favish, 2005; Winberg, 2008). The implementation of OBE in countries such as Australia Canada, New Zealand and the USA and has led to an acceleration in the rate of globalisation and massification within the higher education landscape (Olivier, 1998). According to Malan (2000:22), OBE is "currently favoured internationally to promote educational renewal". This new dispensation requires that educators engage with and manage diversity within a dynamic and fluid multi-cultural environment that has greater interconnectivity with the rest of the world. Subsequently, OBE encourages more accessible educational praxis such as lifelong learning - this imperative introduces rather unique challenges to curriculum development and implementation (Smith & Spurling, 1999; Gough, 2000; Knapper & Cropley, 2000; Granville & Dison, 2001; Bogdan & Biklen, 2003; South Africa, 2003a; 2004a). Further, the *Higher Education Act* (South Africa, 1997a) and the *White Paper* on Special Needs Education (South Africa, 2001) recognise the non-discriminatory and destigmatising benefits of *inclusive education* (as opposed to special education) and makes provision for adoption of this measure right through the different levels of education – themes that inform the UD ethos.

The *UD Educators' Network* (2006) provides an opportunity for exchange between UD educators across the globe, as well as hosting an extensive database of useful links and resources. Similarly, though an array of options is available to UD educators including; *UD for Learning* (UDL); *UD for Instruction* (UDI); *Universal Instructional Design* (UID); and *UD for Education* (UDE), the first two are the more commonly used (McGuire, Scott & Shaw,

2006:167; 172). McGuire *et al.* (2006:172) caution against using confusing terminology which is essentially a *"befuddling of adjectives with no specific meaning"* and emphasise that although UDL emphasises an inclusive vision of UD, there is need for responsibility and recognition that some students will still require individualised special education services and supports. Kennig and Rhyl (2002) have compiled an extensive database on institutions teaching UD-related courses around the world thereby underscoring the versatility and efficacy of UD to diverse contexts globally.

One of the most effective strategies for mainstreaming any new idea is by influencing fledgling and/or future professionals in the targeted fields of endeavour. The motivation for sensitising and exposing design students to a non-exclusive design discourse is such a pre-requisite, and the following statement by the Design Council (2006) elegantly articulates this view;

Inclusive Design is important to the designers of tomorrow – and those who educate them. Future consumer markets will be more diverse than ever in terms of age and physical ability. In the past, design education looked at special needs design for special needs groups. This attitude is changing. Now the focus is on better mainstream solutions for everyone, supported by new design research techniques to make the development process more user-centred.

5.2.3 Case Studies on Universal Design in Cape Town

This section discusses three case studies from the specific vantage point of UD. The first is based on action research – in the form of an experiment during a dedicated module on *UD for Product Design Students* at the Cape Peninsula University of Technology. The second is an exploratory UD audit of the *Kirstenbosch Botanical Gardens*; whilst the final one is a participatory action research case study based on the 2010-related *Mouille Point Promenade and Beachfront Student Landscape Design Competition*. The findings and implications of these cases are also interrogated as they provide valuable insights in the formulation of adjustable UD strategies for the metropolitan area of the City of Cape Town.

5.2.3.1 Case Study One: UD for Product Design Students

This case study is based on a classroom experiment (discussed in further detail in Section 7.2) that was used to test the validity and reliability of adopted outcome-based assessment and evaluation methods that include self-, peer-, and facilitator-assessment modes in the context of a newly introduced dedicated module on *"Universal Design for Product Designers"* offered to students in their third year of study for a *Diploma in Three-Dimensional Design* at National

Qualifications Framework (NQF) Level 6 (SAQA, 2005a; 2005b). Further, implications for *design education* in general and *inclusive education* in particular, were interrogated with reference to a majority world context such as South Africa's (*ibid*). The stated exit outcome/competence for each student would be the following:

- *identify special human needs in a product, system, or practice;*
- apply The Principles of UD to actual design problems;
- use UD strategies to mediate and solve product design-related problems; and
- *justify specific UD considerations in (re)designed solutions.*

The *problem-based learning* activities as advanced by Biggs (2002) required students to:

- identify one example of a product that they had previously designed and made a 3D model of;
- make a detailed case study and critique of their chosen example using UD principles;
- identify specific ways in which they could improve the chosen example to accommodate otherwise excluded users; and
- incorporate suitable and justifiable UD considerations in their redesign solution(s).

The researcher made presentations to students on "Special Human Needs" and "UD Perspectives in Context". One of the presentations to the 32 students included a guest lecturer from the Cape Town Society for the Blind who made a presentation on "The Pathology of Blindness". Students were given access to both hard and soft copies of all presentation materials. At the end of the module, students also received feedback from two CPUT lecturers (who sat in as observers and repertoires for the duration of the module), one focusing on their presentation and communication skills, whilst the other focused on technical competency in their proffered (re)design solutions. Similarly, each student was assessed by at least three of their peers on their presentations and submissions (Luca & McLoughlin, 2002). The modes of formative assessment included:

- 5-10 minute presentations highlighting specific points of UD intervention;
- sketches with accompanying notes on A4 format detailing and justifying specific UD considerations in the redesign of the selected product;
- peer-assessment of presentations on the dates specified (at least three per presenter);
- self-assessment of their presentation and submitted projects;
- assessment by the facilitator; and
- feedback on the module submitted to the facilitator.

81% of students submitted self-assessment evaluations, whilst 94% of students received peerassessment. Students were required to submit self- and peer-assessments and assign a grade or mark corresponding to the formal grading system at the Department of Industrial Design, CPUT. Botha and du Plessis (2007) support this approach of *self-appraisal in learning*. The said grades were correlated to *Not Yet Competent: 0-49%; Competent: 50-74%;* and *Highly Competent: 75-100%*. All participating students attained competency at the end of the module with 84% being assessed as "competent, and 16% as being "highly competent". A statistical analysis of the actual marks awarded indicated an average standard deviation of 1.7 (*ibid*).

Students' responses to self-assessment evaluative questions indicated that the majority (31%) felt they did *best* on applying appropriate UD principles to their chosen products. Interestingly, the majority of students (22%) felt that the *hardest part* of the assignment in applying appropriate UD principles to their chosen products. Of great significance for UD was the finding that the majority of students (50%) responded in their self-assessment that they felt that the *most important* thing they learned in doing the assignment was the sensitivity they developed towards the needs of (potential) 'differently-abled' users of their chosen products; whilst 32% of their peers felt that the application of UD principles to selected products was the *most important* thing the student learned (*ibid*).

The case-study at CPUT vindicated the usefulness of using a triangulated means of assessment and evaluation of a UD module (Creswell, 2002). Due to the short duration in which the module was presented, students were encouraged to explore the possibility of applying UD principles in all future assignments. Indeed as Kennig and Rhyl (2002:9) propose, "ultimately, Universal Design should not be taught as a separate course in the graduate program since this would be against the principles of an integral and inclusive approach. UD should rather be part of all relevant courses and assignments". Subsequent interaction with the same group of students at CPUT confirms that they are indeed implementing UD principles in their design work. It would be instructive to follow up on the progress of the same students once they are employed in industry and align the resultant feedback with pedagogic practice.

According to available records on student enrolment at CPUT (2007) indicate that less than 0.2% of the total enrolment were individuals with disabilities. This figure on students with disabilities works out to one-thirtieth of the national average of 5% (Health Systems Trust, 2006) and about one-sixtieth that of international prevalence figures of 10-12%. Given that

CPUT is now the largest HEI in the Western Cape Province (with some 33,000 students enrolled); these statistics reveal the enormity of the challenge of increasing access to students with disabilities – a task that underscores the critical role of the institutional Disability Units presently located at the two largest campuses at Bellville and Cape Town. CPUT is committed to the establishment of Disability Units on all its campuses in the long-term (Hornea, 2008).

Students' experiential learning – as advocated by Chang, Tremblay and Dunbar (2000), as well as by Thapar, Warner, Drainoni, Williams, Ditchfield, Wierbicky and Nesathurai (2004) – was provided before, during and after the UD module via a wheelchair for their exclusive use. Students reported that they experienced the disabling effects of many barriers that they would have otherwise been unaware of. Students also wrestled with moral and ethical/philosophical dilemmas as to whether designers in majority world contexts should concern themselves with promoting 'design for need' in a prescriptive manner. There was consensus around the view postulated by Whiteley (1993:170) that "design, fun and imagination" are fundamental to the design process, and that "Form Follows Fun" (Papanek, 1995:151). Further, Rose and Strangman (2007) assert that UD for Learning is a proactive pedagogic strategy that elicits positive emotion and motivation by accommodating the three main cognitive learning networks (recognition, strategic and affective) to enhance effective learning. The products designed by participating students illustrate how UD was incorporated in products (Figures 5.5-5.6).

Figure 5.5 shows a *paraffin container* incorporating Braille. By including the needs of people with visual disabilities in the redesign of the paraffin container, the students demonstrated the efficacy of the UD ethos in an unobtrusive yet aesthetically sound manner – in this instance the end-product demonstrates the principles of *Perceptible Information* and *Tolerance for Error*.



Figure 5.5 Paraffin container (source: © K. Golin & C. Fouche)

The *universal bathtub* shown in Figure 5.6 is designed along UD principles (with particular emphasis on *Low Physical Effort*) so that the bathtub walls rise around the user thereby eliminating the strain of getting in and out of the tub. This design also allows professionals and other caregivers to assist in bathing people with disabilities who sometimes might require such assistance – this is a good example of applying ergonomics to HEI contexts (Mokdad, 2005).



Figure 5.6 Universal bathtub (source: © D.J. Coetzee & P. Kleinschmidt)

Though the case study on the UD module was of a diagnostic type, it did reveal promising examples of how UD can be effectively incorporated within design curricula. The implications from this case study are that:

- UD is relevant to design curricula in a majority world context;
- UD principles naturally complement the *Inclusive Education* paradigm;
- UD should be integrated into all design-related programmes offered to students; and
- UD requires dedicated strategies to mainstream it into other spheres of life.

5.2.3.2 Case Study Two: UD for Public Parks

This case study was carried out under the title: "Wayfinding at the Kirstenbosch Botanical Gardens: the makings of an inclusive legible space". The *Kirstenbosch National Botanical Gardens* are a unique and priceless national heritage whose benefits and pleasures should be accessible to all visitors. The case study evaluated the effectiveness of wayfinding systems at the Kirstenbosch Botanical Gardens via a *Universal Design Audit Checklist*¹ (Levine,

¹ The Checklist is not designed for use as an audit for accessibility code compliance, but rather to complete an audit of a design in process or facilities already in use. Copies of the Checklist used are available online at www.ap.buffalo.edu/idea

2003:212-225). The UD Audit Checklist (Appendix D) forms the basis of the discussion on the challenges of implementing inclusive and accessible provisions in a majority world context.

The Kirstenbosch National Botanical Gardens are set in a picturesque site on the eastern slopes of Cape Town's world-renowned *Table Mountain*. The estate is home to the *world's first* indigenous botanical garden having established been in 1913 with the specific objective, according to the *South African National Biodiversity Institute* (SANBI, n.d), "to promote and conserve the flora of southern Africa".

The expansive estate covers 528 hectares of nature reserve of which 36 hectares (or approximately 7% of the total area) forming a cultivated landscaped garden (*ibid*). Kirstenbosch is an integral part of the famed *Cape Floristic Region* (popularly known as the Garden Route). The Cape Floristic Region is one of the county's eight *World Heritage Sites*. These UNESCO-recognised sites are places on earth that are considered to be of "outstanding universal significance to humanity" (SANBI, n.d; UNESCO, 2005; 2007).

Kirstenbosch boasts several public venues for exhibitions, weddings, dining, meetings, parties, shopping and conferences as well as unique features like the *Fragrance Gardens* and an accessible *Braille Trail* to accommodate disabled visitors (SANBI, n.d). According to the Estate Manager, *Augustine Morkel* (2007), there were some 648,238 visitors during the 2006/2007 period. This figure includes visitors who enjoyed free entry but excludes concerts. Of this number, about 3% (or just under 19,500) of the visitors were people with disabilities (including physical and sensory disabilities). Approximately 5% (or just over 32,400) were elderly people (*ibid*). Kirstenbosch subscribes to the concept of *Batho Pele* (introduced in Section 4.2.2) which naturally complements the people-centred philosophy of *ubuntu* (discussed in Section 6.4.3). *Batho Pele* forms the basis of the public sector's service charter to its people whom it views as customers and is expressed as eight principles: *Consultation; Service Standards; Access; Courtesy; Information; Openness and Transparency; Redress; and Value for Money* (South Africa, 1997). Consumer advocacy is encouraged as it makes service providers more accountable and customer/user-centred as exemplified by Kirstenbosch's management.

As a response to the expressed special needs of visitors, the management at Kirstenbosch National Botanical Gardens began an inclusive process of adapting certain sections of the garden where the highest traffic of disabled and elderly visitors frequent to

make their experience more comfortable and rewarding. These adaptations are in the form of smoothed/paved walking surfaces, ramps, handrails, and innovative footpath wayfinding designs. The accessible area is essentially confined to the flatter sections in and around the main garden. Though the initial landscaping design work was done as early as 1988 (South Africa, 1998), most of the accessible projects have been carried out much more recently (over the last 5 years) on an *ad hoc* basis in consultation with representative organisations such as those representing people with various forms of disabilities (Morkel, 2007). The signage system (which is primarily in English) is complemented by printed brochures and maps highlighting areas for botanical or recreational preferences. Official printed material as well as some of the interpretational and directional signage incorporates the other two major languages (*Afrikaans* and *isiXhosa*) spoken in the *Western Cape Province* – South Africa has *eleven official languages* (or twelve if *South African Sign Language* is included).

Botanical gardens are often described as "living museums". Kirstenbosch is described as the "flagship" of the *National Botanical Institute* (NBI) as well as being the *largest* and *most prestigious* of the country's eight botanical gardens forming an extensive nation-wide network administered by the NBI (SANBI, n.d). It has in excess of 4,500 plant species in cultivation, including many that are endangered (*ibid*). There have been 9 curators of the gardens since 1913 (Morkel, 2007). As mentioned earlier, Kirstenbosch forms part of the Cape Floristic Region which is mandated as a World Heritage Site to protect its *ecological processes*, and its *biodiversity and threatened species* (SANBI, n.d; UNESCO, 2005). It is internationally acclaimed as one of the world's great botanical gardens and offers accessible specialised tours organised for visitors with disabilities from across the globe (Accessible Travel & Leisure, 2007; SANBI, n.d). Tourism is a key economic driver in the province (South Africa, 2008a; 2008b).

Consumer advocacy and a participatory approach led to an organic process of accessible provision, beginning in 1988 under the auspices of the *Department of Public Works and Land Affairs* (South Africa, 1988). The multi-disciplinary project team included professionals from the disciplines of architecture, landscape architecture, ecology, engineering and cartography (*ibid*). The project reflects concern for the ecological integrity of the gardens as well as several considerations for wheelchair users in keeping with *Part S* of the South African Standard: *Facilities for Disabled Persons* (South African Standard, 1990).

Following the demise of *apartheid* and the emergence of a more progressive democratic dispensation in 1994, previously excluded groups of people were encouraged to engage in all aspects of public life (ODP, 1997; SAHRC, 2002). Consequently, current South African legislation supports these aspirations by explicitly prohibiting unfair discrimination on grounds of *disability* or *age* (among others) (South Africa, 2000). Similarly, specific national building regulations and code of practice address accessibility needs of people with disability (and by extension a multitude of other user including women and children) (South African Standard, 1990). *The South African Museums Association* (SAMA, 2006) also has guidelines for *public programmes and visitor services* in which issues of "equitable and universal access, [...] inclusiveness through multi-cultural approaches, multi-lingual programmes and diversification of resources" listed as basic standards and transformation indicators.

In order to promote greater access, Kirstenbosch offers free entry to senior citizens on Tuesdays (SANBI, n.d). Several challenges however must be surmounted to enable the vast majority of South Africans to access and enjoy this priceless heritage right in 'their neighbourhood'. These challenges include language and literacy and the impact of these and other factors on the provision of an accessible and inclusive wayfinding system in the context of Kirstenbosch.

5.2.3.2.1 The Makings of a Legible Space

The concept of legibility is usually associated with literacy in general, and textual references in particular. The idea that places and spaces should be legible is attributed to Kevin Lynch in his influential book *"The Image of the City"* (Zmudzinska-Nowak, 2003; Fraser, 2007). Lynch is also believed to have first documented *human wayfinding* (Imrie *et al.*, 2001; Runhaar, 2002; Zmudzinska-Nowak, 2003; Fraser, 2007). His goal was to develop means by which city forms could be evaluated based on the concept of *imageability* in the practice of city design (Runhaar, 2002; Zmudzinska-Nowak, 2003). Imageability, according to Runhaar (2002) "is a term that is used to describe how individuals experience places [–] a person's experiences are defined by what they remember and understand about a particular space". A related concept is that of *affordances* (a concept from ecological psychology), which Raubal (2001; 2007) defines as, "what an object, or an assemblage of objects, or an environment enables people to do". On the other hand Norman (1998) coined the phrase *knowledge in the world* to describe external information that people depend upon for wayfinding as guidance, and further elaborates the applicability of the concept of affordance to mundane everyday objects.

Wayfinding behaviour generally involves interaction "between the wayfinder and the environment" to reach a targeted destination whereas affordances are based on the "paradigm of direct perception" (Raubal, 2007). Raubal (*ibid*) further argues that affordances are measurable aspects of the environment, though such measurements can only be done from the vantage point of the visitor. Additionally, whilst citing other studies, Raubal (*ibid*) states that an individual's "culture, social setting, experience, and intentions also determine her perception of affordances". Zmudzinska-Nowak (2003) lists legibility and comprehensibility as "humanistic aspects of place" and concurs that these attributes are perceived subjectively by different individuals. Zmudzinska-Nowak (*ibid*) further argues that the concept of the *ideal environment* is culture-specific. Read (2003) also underscores the importance of colour in such context.

Harrell, Bowlby, and Hall-Hoffarth (2000) corroborate this view and report further of older visitors having difficulty in orientation. Further, they report that men are more likely to use *cardinal indicators* (such as 'north' or 'south') when giving directions, whilst women are more likely to reference *landmarks* when giving directions (*ibid*). Yielding (2003) also notes several practical limitations associated with ageing. These assertions should inform the development of a more intuitive and multi-sensorial wayfinding system (Cornell, Sorenson & Mio, 2003).

Singer and Ritz (1996) recommend that signage should avoid using ambiguous or complicated language that is "beyond the sixth grade reading level" to accommodate visitors with lower levels of literacy. Subsequently, interactive legible spaces (that facilitate intuitive wayfinding) should intrinsically incorporate affordances and innovative of features that enhance their imageability (Popovic, 1999). The said wayfinding system would need to be culture, age, gender, literacy and context-sensitive to achieve efficacy.

5.2.3.2.2 Universal Design and Wayfinding

The ideal of UD is to pursue a 'non-exclusive' design agenda to tackle stigmatising and discriminatory attitudes such as sexism, ageism and disablism. In so doing, UD seeks to contribute to the goal of social inclusion. The seven principles of UD (CUD, 1997; Danford & Tauke, 2001) were developed by a multi-disciplinary team just over a decade ago and offer a practical tool for assessing products and built environments. The scope of UD ranges from micro (singular products) to macro (involving entire cities, such as New York) (Danford & Tauke, 2001). As discussed earlier in this chapter, related concepts of "non-exclusive design" include *Design for All* (in Europe), *Inclusive Design* (in the United Kingdom) and *Kyoyo-Hin* (in

Japan). *Transgenerational Design* focuses on the needs of the elderly with a view to expanding the consumer base to include other age cohorts. Whilst acknowledging certain differences in epistemological interpretations between these related concepts, the aforementioned terms shall be assumed to be functionally synonymous with UD in this thesis.

Specifically, UD views wayfinding as being of essentially two main forms: *architectural*; and *graphical* (Danford & Tauke, 2001). Architectural wayfinding elements include:

- paths/circulation: the key organising element of a site or building;
- markers: an object that marks locality;
- nodes: a point at which subsidiary parts originate;
- edges: determine where an area begins or ends; and
- *zones/districts:* regions with a distinguishing character that assists in the general identification of place.

The above-named elements are virtually identical to the elements of wayfinding attributed to Kevin Lynch namely: *paths*; *landmarks*; *nodes*; *edges*; and *districts* (Zmudzinska-Nowak, 2003). Figures 5.7 and 5.8 illustrate examples of architectural wayfinding elements at Kirstenbosch.



Figure 5.7 Paths (intersecting)



Figure 5.8 Landmark (mountain) and node (sculpture)

Graphic wayfinding elements include (Danford & Tauke, 2001):

- orientation;
- directional information;
- destination identification; and
- situation and object identification.

Figures 5.9, 5.10, 5.11 illustrate examples of graphical wayfinding at the gardens.



Figure 5.9 Directional signage at nodal intersection



Figure 5.10 Destination signage (in green)



Figure 5.11 Situational warning signage (in red)

The UD Checklist and Audit discussed herein covers the afore-mentioned elements, examples of which are illustrated using UD principles further on in this thesis (Danford & Tauke, 2001).
5.2.3.2.3 Wayfinding in Context

The search for the *ideal environment* or *legible city* should be at the heart of every city's strategic planning for urban regeneration (Wyman, 2004). The historic opportunities of hosting the forthcoming FIFA 2010 World CupTM is a natural catalyst for the implementation of an integrated and comprehensive accessibility strategy in participating host cities (Contact Newsletter, 2006; 2007f). Fraser gives the example of "Legible London" where the city took advantage of the opportunity of hosting the 2012 Olympics to engage in a similar process (Fraser, 2007). Fraser (*ibid*) speaks of the innovative wayfinding maps in Philadelphia that are conveniently "always the right way up" for viewers. Coleman *et al.*, (2003) present the interesting project of Marjala (a suburb of the Finnish city of Joensuu) in its quest to become a "city-for-all". The emphasis in this particular case is transgenerational and embraces the issues of ageing and disability.

Wayfinding should be user-centred. As Yielding (2003:105) points out, *"it is the consumer who bears the brunt of poor design, who may be irritated or inconvenienced by ill-thought-out products or badly designed environments every day. It is the consumer who has to adapt to poor design and negotiate a way around it. So design should, by right, be a major consumer issue."* People with disabilities and the elderly are particularly vulnerable to the disabling effects of bad or insensitive design. Yielding (*ibid*) further identifies a number of compounding factors in relation to elderly consumers such as: small print on labels/signs; poor colour contrast (such as pale grey on white); instructions in small hardly legible print; an absence of tactile feedback on hand controls; poorly lit display panels; and complicated menu systems.

For a place to be truly legible it must be easily and readily *deciphered* by everyone. In terms of wayfinding it implies a system of guides, signs, maps and directional devices which can be understood by whoever is seeking direction. Wyman (2004) states that designers should use wayfinding systems "to reference the history, culture and essence of a place" and goes on to propose that an effective system can be "a visual ambassador". The tension between cultural sensibility and universal interpretability needs to be addressed to achieve legible spaces that also reflect local design inspiration (Algotsson & Davies, 1996). Subsequently, the design of wayfinding must be appropriate to both the constraints imposed by the planned physical setting, users-specific needs, as well as being sensitive/responsive to cultural nuances (Norman, 1998; 2004; Moalosi, 2007).

5.2.3.2.4 Legible Kirstenbosch...

All the buildings constructed at Kirstenbosch since 1995 have been designed for universal access, but not explicitly with UD in mind. Most of the accessible facilities have been implemented by a number of architecture firms and were initiated upon reviewing responses from visitors' surveys (Morkel, 2007). A more professional and effective model of environmental and product assessment is proposed by Yielding (2003). This would entail having a system of checklist assessments carried out by three assessors: a *disabled person* who is experienced in the range of disabilities under review; an *ergonomist*; and a *UD specialist (ibid*).

The area of garden between the *Main Entrance* and the *Visitors Centre* is where most people (especially people of disability and the elderly) move around. Owing to the site located on the slopes of the mountain, some of the upper sections are too steep for disabled access. There is, however, a golf cart tour, which takes place at an hourly interval beginning at 09.00 am every day at a cost of ZAR 50.00 (and includes a guided tour through the gardens) (Morkel, 2007).

Language does pose a problem to certain international visitors. The fact that the wayfinding signage is essentially in English presents a challenge to other local visitors whose first language is either *Afrikaans* or *isiXhosa* – these are the languages predominantly used in the *Western Cape Province*. There has been a phased introduction of brochures and maps incorporating these two regional languages. There is an ongoing process of translation into other languages that are commonly used both nationally and internationally.

An audio system called *My Guide*TM which uses radio technology may be rented on request at the *Ticket Office*. The system is a form of *electronic travel aids* (ETAs) as described by Gill (2007). My GuideTM is presently available in two languages (namely *English* and *German*) and comprises a handset (Figure 5.12) that receives radio frequency transmissions via radio beacons (Figure 5.13) identified by red markers placed along the *Braille Trail* (Figure 5.14). The trail is designed in such a way as to return the visitor to the point from which the journey began. The system is one-way and therefore limited in terms of user interactivity needs – one can only access a pre-set menu.



Figure 5.12 My Guide™ handset in use



Figure 5.13 Beacon for My Guide™ with cardinal direction



Figure 5.14 Braille Trail

The *Braille Trail* (Figure 5.14) was specifically established to cater for people with varying visual disabilities and uses a contrasting red guide rope with information boards located along it that explain the different aspects of the trail. Each board is marked by a large red block along the route and the actual information is displayed on the front in a visual and text format and on the top in Braille (Figure 5.19). The trail itself offers a multi-sensorial experience of nature in the form of aural and tactile activities and information relating to the immediate environment of the small forest area. The start and finish of the *Braille Trail* are located very near to each other on the same main pathway and directly opposite the *Fragrance Garden* (Figure 5.15) to reduce the distances visitors need to traverse to access these amenities. The *Fragrance Garden*, with its array of smells from the indigenous *fynbos* (*Afrikaans* for 'fine bush'), has been designed to appeal to visually disabled visitors as well as the general public for an enhanced experience. It offers reasonably support via a sturdy guide and support rail and information boards, which have both text, images and Braille, posted along the path. The plant bed is also raised to (standing) waist height in order to facilitate greater interaction between visitors and the flora.



Figure 5.15 Fragrance Garden

The garden is located within a short distance of the facilities at the *Main Entrance*. Ablution facilities are well catered for enabling access for wheelchair users as well as for mothers with babies. There is adequate wheelchair access at the building sites, with handrails on all the staircases for unstable individuals. Wheelchairs are at hand at both the *Main Entrance* and the *Visitors Centre* for anyone that may need their use. The signage, facilities and specialised paths are specific to Kirstenbosch and are not found in any other garden in the country (Morkel, 2007).

As discussed previously, the seven UD principles were used to audit the gardens (*ibid*; CUD, 1997). The UD Audit Checklist covers such issues as *site*, *building*, *environment systems*, *communication systems*, and *programme spaces* (Danford & Tauke, 2001). Overall, Kirstenbosch averaged a satisfactory 70% in terms of its general level of usability and accessibility from a list of thirty criteria. The general impression is that the architectural wayfinding is well-articulated, though specific aspects of graphical wayfinding require intervention to become more effective. Certain spaces had the highest scores for usability: including *access points*; *parking*; *pathways*; *lobby*; *reception area*; *amenities* (such as drinking fountains and telephones); *work stations*; and *public assembly*. The following illustrations (Figures 5.17-5.23) serve to exemplify the Principles of UD as elaborated with brief descriptors.

1. *Equitable Use:* the design is useful and marketable to people with diverse abilities (CUD, 1997). The main entrance offers access via a ramp and steps. This allows equal access irrespective of whether one has a disability or not (Figure 5.16).



Figure 5.16 Principle 1: Equitable Use

2. *Flexibility in Use*: the design accommodates a wide range of individual preferences and *abilities (ibid)*. The turnstiles at the crowd control gates are designed to swing away to allow wheelchair access (Figure 5.17).



Figure 5.17 Principle 2: Flexibility in Use

3. **Simple and Intuitive**: use of the design is easy to understand, regardless of the user's experience, knowledge, language skills, or current concentration level (*ibid*). The garden route map shows two views of the same site by using easy to identify landmarks (Figure 5.18).



Figure 5.18 Principle 3: Simple and Intuitive Use

4. **Perceptible Information**: the design communicates necessary information effectively to the user, regardless of ambient conditions or the user's sensory abilities (*ibid*). The signs along the brail trail provide information in both text and Braille (Figure 5.19).



Figure 5.19 Principle 4: Perceptible Information

5. **Tolerance for Error**: the design minimizes hazards and the adverse consequences of accidental or unintended actions (*ibid*). The paths at this node eventually converge to the same destination irrespective of the route one chooses (Figure 5.20).



Figure 5.20 Principle 5: Tolerance for Error

6. Low Physical Effort: the design can be used efficiently and comfortably, and with a *minimum of fatigue (ibid)*. The sink taps, toilet flush handle and baby changing panel require less physical effort than standard fittings (Figure 5.21).



Figure 5.21 Principle 6: Low Physical Effort

7. **Size and Space for Approach and Use**: appropriate size and space is provided for approach, reach, manipulation, and use regardless of the user's body size, posture, or mobility (*ibid*). There is clear signage and sufficient space for approach and use at the entrance (Figure 5.22).



Figure 5.22 Principle 7: Size and Space for Approach and Use

5.2.3.2.5 Towards More Inclusive Interactivity

Imrie *et al.* (2001) criticise *tokenism* and *'expertism'* by professionals due to oversimplification of bodily diversity because such sources tend to reinforce disablism. Designers and human factors practitioners should adopt more participatory approaches in engaging with (potential) users (or visitors in this case) to create effective and sustainable experiences (Zalk, 2001; Wilson, Haines & Morris, 2005). Mamoru (2001) supports this view with respect to *'the concept of an integrated society'* or *'universal society'* goals in *Japan*. To attain these ideals, *Urban Parks Access Committees* include users with disabilities as active members (*ibid*).

Hine *et al.* (2000) describe the four major dimensions of social exclusion as: *Physical; Economic; Temporal; Spatial;* and *Psychological.* These dimensions should be factored into the design of an effective wayfinding system that is cognisant of human needs.

Miyake (2001) discusses simple cost-effective solutions such as rain shelters, using an aluminium guide pipe embedded in a route/path as a practical and less invasive alternative to more elaborate directional tactile surface tiles found in built environments (CRC, 2004). Miyake further argues that flowerbeds with strong fragrances and cascading water should be avoided as they tend to disorientate and inconvenience people with visual impairments whom he views as having the most difficulty navigating parks. This particular suggestion would appear to contradict the ideals of the Fragrance Garden at Kirstenbosch though the fundamental issue is one of degree, not direction. Further, Miyake suggests provision of companion seating on benches for those accompanying wheelchair users (Matheson, 2006). Some of the provision are more common-sense and would include rain shelters to shield visitors from the elements.

Robb (2001) discusses the importance of "nature outings" for people with disabilities and supports the business case for provision of accessible amenities in outdoor areas. He shows the impact of legislation in countries such as *Australia, Sweden, Canada, the UK,* and *the USA* in promoting and enforcing accessibility. The challenge even in these countries is the lack of 'full' accessibility. It should be noted however that the *Americans with Disabilities Act of 1990* (ADA) of the USA does allow for exceptions "where compliance would cause substantial harm to cultural, historic, religious, or significant natural features or characteristics" (Robb, 2001:20.9). The same would apply to Kirstenbosch, particularly in the natural environment section dedicated to the protection and preservation of indigenous flora.

Gill (2007) identifies three specific needs of people with visual disability which would appear to be universal in nature: *the ability to avoid obstacles on the path of travel; the ability to stay on desired route; and the ability to identify when the destination is reached.* Further, transgenerational provisions would enable elderly visitors to live fuller lives. Goodman *et al.* (2005) report that the elderly have greater difficulty maintaining *extra-personal orientation*, and identify three aspects of daily living that promote mobility among the elderly, namely: *maintaining social connectivity, accessing local facilities* and *remaining independent*. Navigation or wayfinding is an important factor in maintaining mobility and independence. Elderly people often experience declines in *cognitive, perceptual, motor abilities* and *special orientation* (*ibid*; Harrell *et al.*, 2000). This results in difficulty in memorising maps, as well as making distance and direction judgements about novel environments (Goodman *et al.*, 2005).

Pictograms are recommended as they offer specific benefits over text (Singer & Ritz, 1996; Murungi, 2003). Signs and symbols generally offer the benefits of instant communication and elimination of cross-language barriers (Wyman, 2004). However, they have the possible drawbacks of being subjective (and limited to context-specificity) and open-ended interpretation (and therefore multiple meanings) (Wyman, 2004; Singer & Ritz, 1996). There is need to use iconography judicially and to take cognisance of cultural context and content (Moalosi, 2007). Indeed as Wyman (2004) asserts, "one symbol is too much if it does not communicate effectively". Practical solutions include: increased contrast in colour; larger displays and contact points such as buttons; simple menu structures; minimise cognitive demands; and simplified hierarchy of information/tasks. This in turn will make wayfinding systems more accessible and user-friendly to all.

Wayfinding research should in the future focus more deeply on commonalities and differences between wayfinding in the real world compared to *wayfinding in electronic and virtual spaces* (Wood, 1993). Finding the particularities with regard to *human spatial cognition* will help designing more user-friendly wayfinding systems (Raubal, 2007). In proposing a *perceptual wayfinding model*, Raubal (2007) distinguishes between the human *cognizing agent* (who can perceive, decide, and act), and the inanimate *non-cognizing objects* (such as signs, counters, and gates). This is a useful distinction as it supports the assertion by Singer and Ritz (1996) that the role of *industrial designers*, *cognitive psychologists* and *ergonomists* (human factors engineers) is pivotal in making wayfinding easier as humankind progresses in to the new millennium. A truly multi-disciplinary team would need to include information/graphic designers,

interaction designers, architects, and a whole host of related professions. In order to be inclusive, such a team would engage with consumers in a participatory manner.

Baldwin (2003) argues that the focus should transcend the *product*, and instead focus on *process* and *platforms*. His argument is valid in the context of the incredible speed at which previous technologies are supplanted by newer, smarter, and more efficient ones. Further, there is an increasing degree of convergence – a phenomenon that blends smart products and technologies. Inherent in this challenge is the need to train a new kind of design specialist – one equipped with superior *environmental literacy* (*ibid*). The example of the haptic glove developed as a *tactile-sensory navigation aid* for persons with visual impairments illustrates the direction and pace at which wayfinding technologies are developing (Zelek *et al.*, 2003). In this example, the portable tactile glove works in conjunction with a wearable computer and camera system.

Traditional assistive aids (such as guide dogs and long canes) are fast becoming obsolete as computer-assisted devices become more ubiquitous subsequently affording users greater independence. Baldwin (2003) criticises the often touted *business model* and instead calls for a more enduring *service model* that offers "lifelong education and training, location-based services, technical support, upgrades and new technologies at reduced cost". This latter approach would suit Kirstenbosch well and grant it opportunities for implementing seamless wayfinding technologies that fit into a comprehensive integrated city-wide system for a 'Legible Cape Town' beyond the FIFA 2010 World Cup[™] projects (Contact Newsletter, 2007a; 2007c).

5.2.3.2.6 Implications for UD in Public Parks

South Africa is categorised as a *middle income* country with majority world challenges such as scarcity of funding and significant income disparities (SAMA, 2006). The unique challenges that the country faces include: multiple languages and concomitant multiplicity of cultures; low-levels of literacy; a large technology chasm between the privileged and the marginalised members of society; and the urgent need for empowerment of previously disadvantaged individuals. These geopolitical realities will mean that institutions must compete for scarce resources for public infrastructural improvements. On the other hand, the future could see increased inclusiveness even of the so-called *bottom-of-the-pyramid* if current design practice embraces contextual realities to offer cost-effective creative solutions for a sustainable future (Prahalad, 2005). These opportunities include: increased use of culture-sensitive (yet

universally appropriate) pictograms; audio and tactile customisation for electronic travel aids; exciting possibilities evolving from the rapid convergence of diverse technologies; falling costs of high technology options; and of course, its priceless natural and cultural heritage.

The challenge at Kirstenbosch is fairly representative of the situation in public places and spaces in South Africa. There is an urgent need to promote inclusiveness and embrace the diversity that is integral to the country's social fabric. Indeed as is argued in this thesis, interaction between people from otherwise diverse backgrounds can and should be enhanced through creative and progressive adoption of inclusive practices that also incorporate accessible web-based resources (Theofanos & Redish, 2003). A more holistic UD strategy targeting the needs of visitors with lower literacy levels, the disabled, the elderly; and non-English speakers demands a context-sensitive response to make Kirstenbosch (and by extension Cape Town) truly inclusive and legible. As noted by Gill (2007), despite the successful implementation of a proliferation of wayfinding systems, there is a lack of synchronisation or harmonisation within different wayfinding systems and services which ensure that users are not confused by different variants in every new context they encounter.

This latter consideration underscores the need for a sustainable multi-disciplinary *product-service-system* approach to the design of effective wayfinding systems (Vezzoli, 2007; Vezzoli & Manzini, 2008). The universal value of Kirstenbosch should be accessed and enjoyed by all, more so when viewed in the context of the expected increase in eco- and geo-tourism (as well as community-based tourism) in the wake of 2010. It is noteworthy that the drive for inclusiveness at the gardens was driven by a combination of consumer advocacy, and proactive management. The example of Kirstenbosch aptly illustrates the efficacy of inclusive empathic interventions over detached and exclusive knowledge.

5.2.3.3 Case Study Three: UD in a Students' Design Competition

According to Kamal Pezeshki (2008), "the Mouille Point Promenade and Beachfront Student Landscape Design Competition is most probably the first student competition in Cape Town which [specifically] refers to the forthcoming 2010 FIFA World Cup. Thus a successful implementation might be crucial for follow-up contests and further collaborations between the City and academic institutions with regard to the 2010 World Cup". The competition was planned to initially include postgraduate landscape design students of the University of Cape Town (UCT) in partnership with the Department of Environmental Resource Management of

the City of Cape Town. Other actors would include ward councillors, and officials from both local and regional (provincial) government. A key actor was a local professional landscape architect whose firm *OvP Associates* would oversee the successful implementation of the winning designs on behalf of the City of Cape Town. The author successfully negotiated with the organisers of the competition for the inclusion of undergraduate design students of the Cape Peninsula University of Technology (CPUT).

Whereas the invitation of CPUT student participation in principle was more inclusive, having students at varying levels of experience and competence from two very different institutions proved to be a challenge. The author was formally appointed on the planning and coordination team (as an academic partner) and tasked with promoting the competition at CPUT. Subsequently, a host of design-related disciplines (including architectural technology, landscaping technology, interior design, town and regional planning) at CPUT were contacted through their heads of department to foster 'transdisciplinarity' (Toft & Joubert, 2005). Advertising posters and a data package with requisite materials and resources were compiled in CD-ROMs for distribution and dissemination amongst all design sub-disciplines. Further, a series of seminars was organised to brief the teaching staff and students in which the key actors and other significant role players were also in attendance.

The following assessment criteria were proposed to guide the participants in their preparations (Pezeshki, 2008):

- Integration: a winning design illustrates well-devised and organic linkages of the promenade and beachfront with the urban park, with considerations regarding safe crossings of the Beach Road, as well as integration of the lighthouse and establishing accessibility for all age groups, including disabled persons creating a playground on site to attract children though needs consideration;
- *Sustainability:* a winning design contains elements of efficient energy use, waste separation, and responsible water use, it covers the topics of the durability and recyclability of the chosen elements as well;
- Creativity: a winning design is innovative in its intention and presents the state of the art;
- *Feasibility:* a winning design does not contain exaggerated daydreams, which won't be feasible; and
- *Design quality:* a winning design is presented in a high-value layout as well as a high grade illustration.

Due to intractable internal schedules, priorities and timetable-related challenges, only one of the design-related departments at CPUT – the *Department of Interior Design* – was able to participate in the competition. Consequently, the researcher renegotiated with the central coordination team to have two separate judging categories (Appendix G) to reflect the different competencies of the UCT and CPUT students with the former focussing on landscaping, and the latter on urban furniture considerations. The cooperation of the Head of Department and

two of his teaching colleagues was pivotal in their students' commitment, completion and submission of entries. The author was invited on two occasions during the competition period to make presentations and to give specific input with regards to UD principles. The 3rd Year Interior Design class submitted a total of 22 individual entries (including two collectively compiled analysis and site plans). An example of the entries is shown in Figure 5.23.





5.2.3.4 Summary on Case Studies

The three cases discussed in this section provide an overview on the potential role of UD in a number of settings, and in this particular case; industrial/product design education, public parks, and (student) landscaping design competitions. Further, the role of UD vis-à-vis specific design disciplines is illustrate as communities of actors include *students of industrial/product design and design educators* (in the first case); *architects, landscape designers*, and *graphic designers/visual communicators* (in the second case); and *students of interior and furniture*

design as well as landscape architecture, professional landscape designers, urban planners, and design educators (in the third case). In two of the instances, the opportunity for action research presented themselves and the role and influence of the author is clearly evident. In the case of the Kirstenbosch gardens, the role of the author is *ex post facto*, in the form of an audit on an existing design intervention. The common theme evolving is that it is possible to incorporate UD thinking at various stages of any design process though the impact is arguably greatest if such intervention/input is employed at the inception or commencement stage. A detailed analysis using an Activity Theory lens will be discussed further on in Chapter 7 to elaborate on the students' design competition to better elaborate on this assertion.

5.3 Convergence of Concepts

The terms *Inclusive Design*, *Design for All*, *Kyoyo-Hin* are for the purposes of this research treated as synonyms of UD (CUD, 1997; Design Council, 2006; EDeAN, 2006; Kyoyo-Hin, 2001:7) in their quest for non-exclusiveness. Indeed as Keates & Clarkson (2003a:440), the said design approaches "are fundamentally derived from user-centred design theory". Critics have argued about the assumption that UD seeks to offer universal solutions for everyone whilst lauding the aspiration of UD to include all users by incorporating and moving beyond the needs of disabled users (Hansson, 2006). This is not practically feasible, and is conceded by the originators of the Principles of UD (CUD, 1997; Story, 2001).

Notwithstanding, the usefulness of UD is recognised by the majority of commentators (Pirkl, 1994; Mace, 1998; Clarkson *et al.*, 2000; Keates, Clarkson, Harrison & Robinson, 2000; Coleman *et al.*, 2003; Hansson, 2006; Jönsson & Certec, 2006a). The *Oxo*[™] *Good Grip* range of household consumer products is a good example of the benefit of UD-thinking in product design. Further, Koncelik (1998) acknowledges the extreme popularity of UD within the design professions) whilst Imrie *et al.* (2001:16) note the importance of the "potentially progressive" principles of UD in the quest for transformation in the lives of all people. As a compromise, Fletcher (2002 in: Hansson, 2006:31) proposes that UD be seen as "an orientation to design in which designers strive to incorporate features that make each design more universally usable".

Dalcher (2006:254) argues that "universal access relies on formulating and promulgating the principles, methods and tools of universal design in order to develop technologies that are accessible and usable by all citizens". Dalcher (*ibid*) further highlights the role of *consilience* in the convergence of UD-related concepts:

Consilience helps to draw together hitherto unrelated streams and brings to the surface assumptions that underlie and underpin entrenched thinking processes. Linking together the theories and facts from across several disciplines provides a common ground that strengthens the overall case as each discipline continued to operate in an isolated mode (in the evolutionary sense).

The fundamental ethos of UD is informed by a desire to engage diverse consumer groups by mainstreaming products and services targeted at them (Hansson, 2006), as well as promoting the ethos of "full social membership" (Sinclair, 1998:510). There is also a growing realisation in both industrially developed and developing contexts of the need for progressive and sustainable strategies that promote social equity and cohesion (Vezzoli & Ceschin, 2008). UD forms a critical component of resultant strategies. Such collaborative and trans-disciplinary strategies in South Africa would need to focus on all key actors with academia playing an pivotal role as it equips future designers with requisite skills to formulate appropriate and sustainable solutions (Winberg, 2006; Vermeulen & M'Rithaa, 2007; Broom, 2008; Chisin, 2008; de Villiers, 2008; P. Thompson, 2008; Smith, 2008; Snaddon, 2008; Tapela, 2008).

The convergence of the different UD-related concepts has more recently enjoyed greater visibility through the ubiquity supported by internet-diffused networks such as the *International Association for Universal Design* (IAUD, 2006) based in Japan, as well as the *Global Universal Design Education Online Network* (GUDEON, n.d) and the *World of Universal Design* (WUD, n.d) – both based in the USA.

5.4 Summary

This chapter explored the evolution of design-specific concepts and responses to the issue of 'exclusion by design' and how these ideas relate to and influence the multi- and transdisciplinary field of UD from a global perspective. Subsequently, the relevance of UD within the scope of this thesis has been discussed. Whereas this chapter primarily dealt with the more pragmatic aspects of non-exclusive design practices, Chapter 6 will deal with the more theoretical underpinnings and linkages with respect to UD.

CHAPTER SIX

THEORETICAL, CONCEPTUAL & ANALYTICAL FRAMEWORKS

6.0 Introduction

This chapter presents the theoretical, conceptual and analytical frameworks underpinning this study. A synopsis of the literature review on the concept of UD in the previous chapter provided insights into the theory and concepts, which inform the study. A theory is expected to perform two major functions, namely, explanation and prediction of phenomenon. Whereas Activity Theory is the analytical lens employed in this study, it must however be emphasised that other theories and/or concepts are viewed as pertinent to this study. As discussed in the previous chapter, UD is a relatively new concept within industrially developing (or majority world) contexts and should be adopted, tried and tuned to suit a country (or region's) socio-economic priorities, industrial aspirations, and geopolitical realities.

6.1 Activity Theory

Bedny, Seglin and Meister (2000:168) describe Activity Theory (AT) as "a psychological paradigm that was a foundation for the study of work behaviour in the former Soviet Union [and that] AT assumes a distinctive human psychology defined by goal directed behaviour". Bedny *et al.* (*ibid*) further state that "in AT, goal-orientation mobilises cognitive, behavioural and motivational components into an integrated system composed of discrete cognitive-affective units that includes goal-oriented feed-forward and feedback components". The first generation of AT is also known as cultural-historical activity theory (CHAT) and originated in the Union of Soviet Socialist Republics (USSR) and was initially developed by Russian psychologist Lev Semyonovich Vygotsky (1896-1934) (Korpela, Mursu & Soriyan, 2001). Figure 6.1 illustrates the three main components of this theory.





Vygotsky makes an important contribution – particularly in the context of learning, instruction or testing – in his concept of the *zone of proximal development* (ZPD) (Bedny *et al.*, 2000). According to Vygotsky (1978:86 in: Bedny *et al.*, 2000:171) the ZPD is the gap between a child's, "actual development as determined by independent problem solving... [and] potential development as determined through problem-solving under adult guidance or in collaboration with more capable peers". Bedny *et al.* (2000:171) further state that based on this notion:

... [Vygotsky] calls for dynamic assessment, a two-step process. First, the child performs at the limits of his solitary competencies, then again assesses his difficulty limits of his problem solving with the aid of an adult or peer expert. The difference between independent and assisted solutions enables one to assess his 'learning potential' as real ability. This method affords one the opportunity to assess the ZPD.

AT was further refined by Sergei L. Rubinshtein (1889-1960), Alexei Nikolaevich Leontiev (1903-1979) and Alexander Romanovich Lurija (1902-1977) (Vygotsky, 1981; Engeström *et al.*, 1990; Engeström, 2000). Engeström *et al.* (1990) identify the key elements of Leontiev's three-level model which is based on a hierarchical conception of activity (see Table 6.1). Yrjö Engeström (1999; 2000) attributes the second generation of AT (shown in Figure 6.2) to Leont'ev and expands the 'social infrastructure of activity' to demonstrate the efficacy of AT in the visibilization, analysis and redesign of work.

Unit	Directing factor	Subject
activity	object / motive	collective
action	goal	individual or group
operation	conditions	non-conscious

Table 6.1 Leont'ev's Three-Level Model (source: Engeström et al., 1990)

The dynamic interacting (and mediating) elements of an activity system and their interrelatedness are elaborated in Table 6.2 (Engeström *et al.*, 1990).

Element	Description
Subject	the individual or group whose point of view is taken in the analysis of the activity
The object (or objective)	the target of the activity within the system
Instruments	internal or external mediating artifacts that help to achieve the outcomes of the activity
The community	comprised of one or more people who share the objective with the subject
Rules	regulate actions and interactions within the activity system
The division of labour	how tasks are divided horizontally between community members as well as referring to any vertical [or hierarchical] division of power and status

Table 6.2 Elements of an activity system (source: Engeström et al., 1990)



(source: Engeström *et al.*, 1990)

AT draws inspiration from 18th century philosopher Immanuel Kant's discourse on the centrality of phenomenology and activity in developing knowledge and experience, as well as Georg Wilhelm Friedrich Hegel (of the 19th century) who drew attention to the manner in which "reality develops by means of the reconciling (synthesis) of contradictions (between thesis and anti-thesis) (King, 2004:120). AT also follows in the philosophical tradition of dialectic materialism and power advanced by Karl Heinrich Marx and Friedrich Engels in the mid-nineteenth century (Marx, 1972; Mursu, 2002; King, 2004). AT's provides a practicable philosophical framework for the study of humans at both the individual and social levels (and the links therein) as developmental processes. Of particular significance is the cultural-historic impact of prior influence and epistemological precedents in shaping the activities that people engage in individually and collectively.

AT views conscious learning as a product of activity – this underscores the importance of the latter to understanding human motivation and consciousness. Further, as Decortis, Noirfalise and Saudelli (2000:30) attest, AT "sees the division of labour as a mediator between the community and the object in the process of transforming the object into outcome". This collective or collaborative object-orientedness of activity is critical to the appreciation of *ubuntu* which essentially relies on a voluntary, participative, co-operative and egalitarian ethos with a lateral (or horizontal) power structure to function properly. In *ubuntu* there is no power or control over other humans – only potential influence. The pertinent attributes of *ubuntu* will be further elaborated in Section 6.3.3.

Osiris (n.d) argues that AT's usefulness lies is in that it "provides a broad conceptual framework that can be used to describe the structure, development and context of tasks that are supported by a [work] system". Osiris (*ibid*) further elaborates on the five basic principles of AT as follows: *the hierarchical structure of activity*; *object-orientedness*; *internalisation/externalisation*; *mediation*; and *development*.

AT has found application in a wide range of disciplines including *design* (Lauche, 2005), *organisational ergonomics* (Bedny *et al.*, 2000; Decortis, 2000; Eason, 2005), *work study* (Engeström, Brown, Engeström & Koistinen, 1990; Engeström 1999; 2000), *information systems* (Korpela *et al.*, 2001; Collins, Shukla & Redmiles, 2002; Korpela, Mursu, Soriyan, de la Harpe & Macome, 2006; Mursu, Luukkonen, Toivanen & Korpela, 2007), *human-computer interaction* (Nardi, 1999), *co-operative education* (Engel-Hills, Garraway, Nduna, Philotheou, & Winberg, 2005; van Aalast & Hill, 2006; Garraway, 2008), *information communication technology* (Lim & Hang, 2003); as well as *leadership and professional development* (Hansson, 2002). Sjöberg (1996) applies AT in an exploration of links between the 'Scandinavian approaches' to participatory design and information systems. Korpela, Mursu, Soriyan, Hakkinen and Toivanen (2008) build upon Engeström's mediational structure to advance *Activity Analysis and Development* (ActAD) as a further refinement of AT within the information systems domain (see Figure 6.3).



Figure 6.3 Elements of a work activity using an ActAD lens (source: Korpela *et al.*, 2008)

6.2 Explanatory Legitimacy Theory

Explanatory Legitimacy Theory as posited by Elizabeth DePoy and Stephen French Gilson is a post-modernist framework that draws on various cultural-historical debates to analyse human diversity (DePoy & Gilson, 2004; 2006; Gilson & DePoy, 2006). This theory adopts a 'universal ideology' to interrogate the grand category of disability as well as to challenge the categorisation of people along nomothetic lines – a popular approach aided by the pervasive application of the Gaussian or 'normal' (bell-shaped) distribution curve in representing statistical and anthropometric (or physical anthropological) data with respect to human populations. Further, DePoy and Gilson (2007) emphasise the role of community in proffering judgement and legitimacy to group membership. This recognition of the discourse on *ubuntu* as has been argued elsewhere in this thesis.

Of particular import to this thesis, the inclusive, non-discriminatory and category-expanding outlook of the Explanatory Legitimacy Theory explicitly links universal access to UD (DePoy & Gilson, 2005/2006:130):

Universal access is based on universal design principles. Different from accommodation strategies, which provide problem specific adaptations and special responses, universally accessible resources, whether physical, sensory, or virtual, are designed from their inception in forms that all people, to the extent possible, regardless of age, disability, ethnicity, primary language, or other diversity characteristics can access without the need for special adaptation.

Gilson and DePoy (2006:700) criticise the "value based use of phrases such as *normal* [which do] not provide the conceptual clarity sufficient for distinguishing description from axiology". Rather, the authors (*ibid*) prefer non-stigmatising value-free terms such as *typical* and *atypical* as these suggest "magnitude rather than desirables".

Explanatory Legitimacy Theory is comprised of three interactive elements; *description, explanation,* and *legitimacy* (DePoy & Gilson, 2004; Gilson & DePoy, 2006):

 The first element, *description*, encompasses the full range of human activity (what people do and do not do, how they do what they do), appearance and experience. Two intersecting dimensions of description: *typical/atypical* and *observable/reportable*, are particularly germane to disability;

- The second, element of explanatory legitimacy is the set of *explanations* for doing, appearing, and experiencing; and
- Third and most important in defining and determining the value of groups and their members is *legitimacy*. Legitimacy is comprised of two sub-elements: *judgement* (about acceptability and worth) and *response* (by an individual, community, social group or policy formulating body).

The key contribution of the Explanatory Legitimacy Theory (as summarised in Table 6.3) is the elegant manner in which it offers a sound theoretical underpinning for UD by reconciling the ontological and epistemological conflicts of the *medical/individual* versus the *social* model debate. This point is particularly germane to the practice of industrial design – a discipline that borrows almost in equal measure from the objective/technological as well as the subjective/social domains. Further, the Explanatory Legitimacy Theory accommodates the descriptive components of *activity* – the primary focus of Activity Theory.

Description - typical/atypical - observable/reportable	Explanation - for doing; appearing; and experiencing	Legitimacy - value of groups and their members
Activity	Medical-diagnostic (medical or individual model)	Judgement
Appearance	Constructed (social model)	Response
Experience	_	_

Table 6.3 Elements of Explanatory Legitimacy Theory (adapted from: DePoy & Gilson, 2004)

Explanatory Legitimacy Theory offers a sound theoretical argument for the provision of universally accessible features, irrespective of the explanation for the user's special needs. For example, in the case of an elevated ramp in a building, a woman pushing a baby's pram, or a deliveryman with a trolley full of wares, or a person in a wheelchair, or an elderly person using an ambulatory aid would all equally benefit from this singular infrastructural provision – the legitimacy of each of these descriptive categories of users (in terms of judgement and response) leads to the self-same inclusive solution – in this case, the ramp.

6.3 Other Concepts Informing this Study

Three other pertinent concepts inform this study. The first two (*Participatory Design* and *Design for Sustainability*) are drawn from design-related fields, whilst the last one (*ubuntu*) is of geopolitical and socio-ethical relevance to the context of study. Whereas the first two concepts have not been integral to the actual research process followed in this study, they are included here for consideration in the development of context-responsive UD strategies.

6.3.1 Participatory Design

Participatory Design (or PD) as the name suggests acknowledges the need to engage (potential) users, consumers or other key actors in the design process. PD is premised on the fact that such actors should be viewed and accepted as 'co-designers' or 'co-producers' of desirable design solutions (Ehn, 1988; Ehn *et al.*, 2002; Buxton, 2007; Moalosi, 2007; Vezzoli & Manzini, 2008). *Participatory Ergonomics* pursues an analogous co-operative focus within its own community of practice (Haines, Wilson, Vink & Koningsveld, 2002; Kogi, Kawakami, Ujita & Khai, 2005). Within the context of participation, inclusion and democratisation, PD is congruent with an inclusive and collaborative process of 'design by all'.

Sjöberg (1996:26) reports that PD evolved from the *Participative Design* approach in Australia with an aim at "developing self-managed design groups to create an evolutionary process involving democratization of the working situation". Demirbilek (1999:38) however states that PD "was first used for the development of communal areas, mainly in architecture and urban planning". Sjöberg (1996:29) tracks the development of 'participation in information systems development' (the precursor to PD) and identifies a number of similar approaches that go by such names as: *user-centred design*; *joint application design*; *co-operative design*; *work-oriented design*; and collective resource approach.

Futerman and M'Rithaa (2007) distinguish between 'participation as means' and 'participation as an end' – the latter is inherently more rewarding and emancipatory for participants. *Emancipation* resulting from participation is the ultimate goal in this context in keeping with the post-modernist paradigm of moving beyond mere *empowerment* – the latter is essentially a renegotiation of existing power imbalances (Balcazar, Keys, Kaplan, & Suarez-Balcazar, 1998; Wanger & Burnette, 2003). Moreover, PD is now a widely accepted effective tool for resolving the needs of previously excluded groups, including the disabled and the elderly (Demirbilek, 1999; Demirbilek & Demirkan, 2004). Within the practice of *ubuntu*, participation is considered a core response to action by all community members (M'Rithaa, 2008; 2009). Shutte (1993:47) argues that participation has deeper meaning in intra- and inter-personal developmental dynamics, and reinforces this view with the statement that "I participate, therefore I am".

6.3.2 Design for Sustainability

Design for Sustainability (DfS) is a branch of design that addresses social, ethical, environmental and economic equity with respect to developmental and technological

imperatives (Findeli, 2008). *Sustainable development* seeks to promote advancements that benefit humanity without negatively impacting the potential livelihoods of future generations on our planet (Margolin, 1996; Khumomotse, M'Rithaa & Moalosi, 2002; Findeli, 2008). Vezzoli (2007:39) defines DfS as "a design practice, education and research that, in one way or another, contributes to sustainable development". This objective of DfS has been extended to address design for social equity and cohesion in line with guiding *United Nations* principles (Vezzoli, 2007; Vezzoli & Manzini, 2008). The *International Council of Societies of Industrial Design* (ICSID, 2002) incorporates this holistic notion of activity in its definition of design:

Design is a creative activity whose aim is to establish the multi-faceted qualities of objects, processes, services and their systems in whole life cycles. Therefore, design is the central factor of innovative humanisation of technologies and the crucial factor of cultural and economic exchange. [...] Design concerns products, services and systems conceived with tools, organisations and logic introduced by industrialisation – not just when produced by serial processes.

Design as a creative activity is not the purview of professional practice exclusively. Indeed there is growing evidence of bottom-up grassroots expressions of social innovation by *creative communities* (Manzini, 2007). Manzini (2009:49) describes these 'creative communities' as "emerging number[s] of creative people who invent and enhance new and more sustainable ways of living and producing". Importantly, these creative communities differ in composition from traditional communities in that within the former, *members elect to belong* based on shared goals and objectives in order that the entire community functions and lives more sustainability. Traditional communities on the other hand *find themselves linked* by culture, customs or blood ties – factors that could pose a threat if they fail to take cognisance of individual liberty and responsibility (Shutte, 1993; Bhengu, 1996). This thesis supports the view that promoting emerging creative communities would orientate emerging elective communities in Africa thus enabling them to leapfrog into more sustainable ways of being and doing.

An encouraging development in DfS is the emerging cross-disciplinary field known as *Service Design*. Service Design moves beyond the traditional paradigm of three-dimensional artifact production and consumption to a more progressive one of interrogating ways in which design can address more subtle tacit needs such as that of 'satisfaction' as well as other social and relational needs (Cipolla, 2009; Manzini, 2009). Cipolla (2009:233) describes emergent 'relational services' as those that are "deeply and profoundly based on the quality of interpersonal relations between and among participants".

For the vast majority of African communities, the currency of exchange is trust. People come first! In this manner the relational interdependence of community members is reaffirmed and perpetuated. Further, this mutual interdependence transcends a materialist agenda (Shutte, 1993). In Africa, women form the relational superstructure through which convivial social interactions happen (Tisani, 2008). As caregivers and key economic contributors, African women render indispensable services to their families and to their communities at large – an investment in women leads to concomitant benefits for the entire community (M'Rithaa, 2008). As Cipolla (2009:233) elaborates, in an interpersonal service model, "no participant can be easily replaced, because together they produce [...] community, a common story".

There is ample evidence of the pivotal socio-economic role played by women, particularly in the informal economic sectors known as *jua kali* (which in kiSwahili means '[in the] hot sun') in Kenya (Macharia, 2006, Orwa, 2007); and the *second economy* in South Africa (du Toit & Neves, 2007). A strategy for enhancing the quality and efficacy of relational services would of necessity need to engage with women. There are a vast number of (mainly women-driven) *ad hoc* cooperatives facilitating micro-lending clubs and group savings/purchasing schemes on the continent wherein typically, a handful of members enjoying deeply interpersonal relationships function essentially on mutual trust. Such a voluntary group is known variously as *motshelo* in Botswana; *chama* in Kenya; and *stokvel* in South Africa. Many have gone on to formalise their membership into *bona fide* long-term investment companies (M'Rithaa, 2009). As stated earlier on though, the goal of community membership is not driven merely by economic interest, but the spirit of conviviality and trust that facilitates and perpetuates such social interaction (Illich, 1973; Cipolla, 2009; M'Rithaa, 2009).

Further, as du Toit and Neves (2007) argue, the aim of policy makers should not be to absorb the (informal) *second economy* into the so-called (formal) *first economy*. Eliminating the former would be counterproductive in the long term. Rather, progressive policy should strengthen the *second economy* by providing for greater resilience and reduced associated risk factors that act as barriers for entry therein. In view of the extra-ordinarily high unemployment rates in most Sub-Saharan economies, this latter approach would aid in the absorption of a higher proportion of the available labour force (Orwa, 2007). Developmental policies that place emphasis on the potential contribution of service systems (and not just products) would ultimately promote more sustainable communities and consequently, facilitate greater social equity and cohesion (Vezzoli, 2007).

6.3.3 Ubuntu

van Niekerk and M'Rithaa (2008; 2009) argue that *ubuntu* could be viewed as the African variant of Aristotelian *virtue ethics*. The essence of this universally applicable ideal is variously transmitted via folklore and wise sayings or proverbs – a popular medium not very dissimilar to the practice in the East (Asian cultures). An example from the Zulu language in southern Africa illustrates this concept: *"umuntu ngumuntu ngabantu"*. This can be literally interpreted as *"a person is a person through other persons"*, or in other words, *"I am because we are"* (Mbigi, 1997; Creff, 2004; Bhengu, 2006) – or to put a twist on Cartesian logic; "we are, therefore I am", and "we think, therefore I can". The important point here is that there is an inextricable link between the individual and his or her community. The benefit of voluntarily relinquishing self-serving pursuits and self-indulgence is that doing so allows the individual enjoyment of a more wholesome and meaningful life – an ideal that Hursthouse (2007) relates to the Aristotelian concept of *eudaimonia*. Such quality of life is experienced within a nurturing, supportive and mutually affirming community setting utilising local socio-ethical know-how (Varela, 1999).

Strictly speaking though, *ubuntu* does not fit the Western model of formalised knowledge but is flexible as well as being context-, and content-dependent. It is negotiated, adjustable, and thus by extension, versatile (van Niekerk & M'Rithaa, 2008). As Boon (2007:26) points out:

Ubuntu is not empirical. It does not exist unless there is interaction between people in a community. It manifests itself through the actions of people, through truly good things that people unthinkingly do for each other and for the community. One's humanity can, therefore, only be defined through interaction with others... It is believed that the group is as important as the individual, and a person's most effective behaviour is in the group. All efforts working towards this common good are lauded and encouraged, as are all acts of kindness, compassion and care, and the great need for human dignity, self-respect and integrity.

Aristotelian virtue ethics posits three different forms of knowledge, namely; *episteme*; *techne*; and *phronesis* (Jönsson & Certec, 2006c:179). Whereas *episteme* (from which epistemology or the theory of knowledge is derived) and *techne* (technology or technique) have found greater acceptance (and indeed attained some degree of prestige), the same can not be said of *phronesis*. Jönsson and Certec (*ibid*) argue that *phronesis* has remained generally underappreciated as "there is no active, contemporary equivalent" meaning assigned to its understanding. Hursthouse (2007:4) approximates the closest definition of *phronesis* (with respect to *ubuntu*) as "moral or practical wisdom". As designers trained in an essentially neo-

Bauhaus model, the articulation and practice of *episteme* and *techne* modes of knowledge happen essentially by default, in part due to the inherent philosophical bias towards Western thinking. This thesis takes the view that *phronesis* and *ubuntu* are not as dissimilar as would appear initially – both are "about values and reality, about people and their actions" (*ibid*).

Further, as Jönsson and Certec (2006c:180) assert, *phronesis* "is not scientific in the epistemological sense, since epistemology is primarily concerned with scientific knowledge that is universal, constant in time and space, context-independent and based entirely on analytical rationality. The knowledge relativism that is an integral part of *phronesis* is thus almost unforgivable in an epistemological approach". This view is supported by Ehn and Badham (2002:6) who challenge designers to re-interrogate their present notions of *phronesis* by going back to a time when the "virtue of *phronesis* had not yet been suppressed". They argue that *phronesis* lost out in part due to "the fragile and unpredictable nature of human action" (*ibid*). Notwithstanding, Ehn and Badham (*ibid*) have shown the efficacy of such reasoning to interaction and participatory design wherein they describe *phronesis* as an "Aristotelian vision of ethical life [and] practical wisdom". Jönsson and Certec (2006c:181) ascribe the renewed interest in *phronesis* to the fact that "the epistemological and the technological alone are not able to stand for all that is relevant in [...] design". From the view point of *ubuntu*, the defence of *phronesis* would be just as valid for the former. Ehn and Badham (2002:6) present the following rationale:

In phronesis, wisdom and artistry as well as art and politics are one. Phronesis concerns the competence to know how to exercise judgement in particular cases. It is oriented towards analysis of values and interests in practice, based on a practical value rationality, which is pragmatic, and context dependent. Phronesis is experience-based ethics oriented towards action.

van Niekerk and M'Rithaa (2008; 2009) suggest that *ubuntu* as an African form of '*ethics by consensus*' is ontologically allied to the concept of *phronesis*. Further, *ubuntu* is a pragmatic concept that bridges the Western and Eastern concepts of ethics whilst simultaneously offering a dynamic platform for debate and engagement of individuals and their communities (elective or otherwise). As members of local communities and societies, no one can stand outside of the same. The responsibility of (design) educators extends beyond interpreting paradigmatic changes and necessitates that we "integrate them into the education system so that they become meaningful, and take root in the consciousness of the people of South Africa" (Tisani

2004:174). Tisani (*ibid*) places a greater responsibility on higher education practitioners as the onus on production of new knowledge "falls directly on their shoulders". Tisani (2004:175) emphasises the importance of engaging African *indigenous knowledge systems* as a transformational tool. Other knowledge systems should not be discarded, but similarly critically engaged with where there is proven efficacy of their value. Higgs (2007:669) concurs by placing emphasis on *reason* (or rational thinking) as a universal human phenomenon.

To put this discourse in perspective, traditional societies in sub-Saharan African have relied on various expressions of *ubuntu* (which is an isiZulu word) since time immemorial (Forster, 2006). *Ubuntu* is also known as *utu* in kiSwahili (spoken in Central and Eastern Africa) and *botho* in both seTswana and seSotho (spoken in Southern Africa). *Ubuntu* is evident through various forms of self-reliance and mutual assistance such as *bataka kwegaita* (communal solidarity) among the Banyakore people of Uganda, *boipelogo* (self-reliance) in Botswana, *harambee* (pulling together) in Kenya, and *ujamaa* (familyhood) in Tanzania (M'Rithaa, 2008). Similarly, in Ghana, *nobwa* (reciprocal assistance) is an invitation to family, friends or neighbours to join in the activity of ploughing the host's farm in the knowledge that should any of the participants require assistance in future, the host would answer the call to action accordingly (M'Rithaa, 2009). These values are still alive on the continent and in The Diaspora wherever an authentic African presence is detectable. Bhengu (1996:5) describes *ubuntu* as "a way of life that contributes positively towards the well-being of a people, community or society". As Desmond Tutu (1999:34-35) elaborates:

Ubuntu is very difficult to render into a Western language. It speaks to the very essence of being human. When you want to give high praise to someone we say, "Yu, u nobuntu"; he or she has ubuntu. This means that they are generous, hospitable, friendly, caring and compassionate. They share what they have. It also means that my humanity is caught up, is inextricably bound up, in theirs. We belong in a bundle of life. We say, "a person is a person through other people". [...] I am human because I belong, I participate, I share.

As Tutu (*ibid*) alludes, response to the call of *ubuntu* is through participation – everyone is invited, and everyone can contribute towards the goals and common wealth of the community. A person reaffirms their personhood through interaction with, and contribution into the lives of others (Shutte, 1993; Bhengu, 1996; Tambulasi & Kayuni, 2005). This people-centred, inclusive and participative spirit of *ubuntu is* what Africa shares with the rest of humanity (M'Rithaa, 2008). It infuses the network of relationships that provide resilience against the ever-changing socio-economic and geopolitical landscapes. It is this spirit that guided South

Africa's emergence as a united democratic country in 1994 after forty years of apartheid rule.

In contemporary Pan-African politics, *ubuntu* is the catalyst for the much touted *African Renaissance* – a principal objective of the *New Partnership for Africa's Development* (NEPAD) initiative under the aegis of the Africa Union (Maloka, 2001). The stated long-term objective of NEPAD is the eradication of "poverty in Africa and to place African countries, both individually and collectively, on the path of sustainable growth and development and thus halt the marginalisation of Africa in the globalisation arena" (Ezeoha & Uche, 2005:6). Closer to home, the philosophy of servant leadership and people-centredness guides *Batho Pele* (elaborated in Sections 4.2.2 and 5.2.3.2). *Batho Pele* as an operational strategy of *ubuntu* underpins the national and provincial governments' commitment to transparency, accountability and service delivery (South Africa, 1997a). As Creff (2004:8) states: "the extent and importance attributed to values shared by *ubuntu* and servant leadership are significant".

Characteristic	Description
Participation in networks and groups	 interlocking networks of relationships between individuals and groups social and interpersonal engagement through a variety of formal and informal associations that are both voluntary and equal
Reciprocity and exchange	 combination of short-term altruism and long-term self-interest the individual provides a service to others, or acts for the benefit of others at a personal cost, but in the general expectation that this kindness will be returned at some undefined time in the future in a community where reciprocity id strong, people look after each other's interests reciprocity and exchanges increase trust and contributes to the development of long term obligations between people
Trust and confidence	 trust entails willingness to take risks in a social context based on a sense of confidence that others will respond as expected and will act in mutually supportive ways, or at least that others do not intend harm trust eases interpersonal relationships and reduces the transaction costs between people and so liberates resources it can also create a social obligation – trusting someone engenders reciprocal trust trust reduces the uncertainties present in social life, ensuring social equilibrium subsequently becoming an indicator of social sustainability
Social norms	 common rules and social norms are the mutually agreed norms of behaviour that place group interests above those of individuals individuals can take responsibility and ensure their rights are not infringed social norms are usually unwritten but commonly understood formulae for determining what patterns of behaviour are expected in a given social context – they also define what forms of behaviour are valued or socially approved
The commons	 the combined effect of trust, networks, norms and reciprocity creates a good community, with shared ownership over resources known as 'the commons' to maintain the commons the presence of s a sense of personal and collective efficacy is needed

 Table 6.4
 Characteristics of social capital (adapted from: Padovan, 2008:273-274)

The value of *ubuntu* in the development of *social capital* has received wide acceptance in the business/socio-economic (Mbigi, 1997; 2000; Boon, 2007) and political/socio-ethical spheres of life in (South) Africa (Bhengu, 1996; 2006; Mazrui, 1999; Tutu, 1999; Maloka 2001; Creff, 2004; Tambulasi & Kayuni, 2005). Participation is a universal imperative in the development of personal and social capital (Chapman, 2002). Padovan (2008:272) asserts that "social bonds, social norms, trust and other social features play an important part in sustainable livelihoods". Padovan (2008:273) further identifies five characteristics of social capital: *participation in networks and groups*; *reciprocity and exchange*; *trust and confidence*; and *social norms*. These characteristics are further elaborated in Table 6.4 and are coincidentally the self-same core elements of the *ubuntu* ethos (Bhengu, 2007; M'Rithaa, 2008).

6.4 Summary

UD evolved as a response to practical (as opposed to theoretical) imperatives. This meant that for a while no singular theoretical framework appeared to adequately elaborate the discipline – it could be argued that practice preceded theory. Recently though, the Explanatory Legitimacy Theory (discussed in Section 6.2) offers the most elegant theoretical framework for explaining UD. Activity Theory on the other is as an appropriate analytical lens for interrogating the way in which individuals and groups negotiate new tools for dealing with specific socio-technical challenges – in this case – in the adoption and dissemination of novel practical tools such as UD Principles. Additionally, ubuntu offers a unique socio-ethical perspective on a contextspecific application of the aforementioned – more space was dedicated to the discussion on *ubuntu* vis-à-vis design as this perspective is potentially an original contribution of this thesis. Further, this thesis argues that incorporating the *ubuntu* ethos in appropriate UD strategies would inherently result in the mobilisation and development of sustainable social capital. Finally, some pertinent concepts provide lateral and vertical dimensions to the understanding and application of UD to ensure a more participatory process of engagement with all key actors in a quest to evolve context-responsive and truly sustainable strategies in a majority world setting.

CHAPTER SEVEN RESEARCH METHODOLOGY

7.0 Introduction

This chapter discusses the basis for the chosen epistemological orientation and describes the research methodology employed in achieving the objectives outlined in Chapter One of this thesis. The chapter provides a description of the sources of data and the methods used in collecting the required data sources. Additionally, it also provides a brief description of the analytical methods employed in analysing the information gathered during the various phases of the study.

7.1 Contextualising Research Perspectives

Hitchcock and Hughes (in Maree and van der Westhuizen, 2007:31) suggest the three following lenses to examine the practice of research:

- Ontological assumptions (which give rise to);
- Epistemological assumptions (which give rise to); and
- *Methodological* considerations (which in turn give rise to e.g., [methods of] instrumentation and data collection.

A realist ontological assumption assumes "that social reality can be understood from an *external* point of view (the *realist* position that abstract objects have an objective existence: reality is of an objective nature, out there)" (Maree & van der Westhuizen, 2007:31). Alternatively, reality can be understood "merely through words and names created in the mind and within levels of individual consciousness (the *nominalist* or "in name only" position) (*ibid*). This thesis adopts the latter nominalist viewpoint as elaborated in Table 7.1.

From an epistemological standpoint, this study adopts a *qualitative* (rather than *quantitative*) vantage. Schwandt (2001:213) describes qualitative inquiry as that which "aims at understanding the *meaning* of human *action*" and further distinguishes qualitative data (as being non-numeric data in the form of words) from quantitative data (which is essentially numeric data). Further, this study follows the *applied research* (as opposed to *pure/basic* research) route. Pure (or basic research involves the formulation and testing of hypotheses and theories that need not be linked to actual applications (Fielding & Lee, 1998). Applied research on the other hand is typically problem-based often locating its focus in real-life social

challenges – action research mobilises available tools and resources to address the same (Badenhorst, 2007). This study is a response to perceived systemic exclusion of certain categories of potential users and consumers as a result of typically apathetic design praxis.

The *nominalist* (interpretivist) epistemological stance adopted in this study assumes that "the meaning of human action is inherent in that action, and that the task of the inquirer is to unearth that meaning" (Schwandt, 2001:134). Interpretivism "foregrounds the meaning that individuals or communities assign to their experiences. [...] the social context, conventions, norms and standards of the particular person or community are crucial elements in assessing and understanding human behaviour" (Jansen, 2007:21). The study is subsequently influenced by the *postmodernist* theoretical perspective. *Postmodernism* in this context "assigns value to multiple meanings rather than the single, authoritative voice of the expert/scientist. [...] It values 'voice', the subjective and multiple voices of individuals and communities rather than predetermined rules for action" (Jansen, 2007:22).

Bailey (1987:24) defines a *paradigm* as "a perspective or frame of reference for viewing the social world, consisting of a set of concepts and assumptions". This study assumes that it is not only desirable, but a matter of human dignity that all people should be able to enjoy access to all products, systems and public built environments of their choice. It further recognises the discrimination, stigmatisation and exclusion of people not fitting the standard definition of 'average' or 'normal'. Consequently, this view places the onus on designers to ensure that *universal access* is guaranteed through the mainstreaming of the concept of UD. The different research premises are summarised in Table 7.1 (Maree & van der Westhuizen, 2007:33-34).

Ontological dimensions	Epistemological dimensions; paradigm/perspective	Nature of relationship between researcher and what is being studied
Realist stance: external reality is stable; general laws govern universe	Positivist (modern)	Researcher is an objective, detached observer
Nominalist stance: informants' internal and subjective experiences are important	Interpretivist (postmodern)	Researcher is empathetically and (inter-) subjectively immersed in the research
Reality is constructed by persons; researchers need to analyse the informants' discourses	Constructivist (postmodern)	Researcher is suspicious of object of study; political undertones can be identified; constructs own version of events

Table 7.1Delineation of paradigmatic assumptions and perspectives (source: Maree & van der
Westhuizen, 2007:33-34)

Bailey (1987:32-33) differentiates between *method* and *methodology* as follows:

By "method" we simply mean the research technique or tool used to gather data. [...] Furthermore, some of the differences in the method between the physical and social sciences are differences of degree rather than kind. [...] By "methodology" we mean the philosophy of the research process. This includes the assumptions and values that serve as a rationale for research and the standards or criteria the researcher uses for interpreting data and reaching conclusions.

The researcher chose to follow an action research methodological approach as the subject of UD was new to many of the informants in this study (Leedy & Ormrod, 2001). Schwandt (2001:3) states that Kurt Lewin (1890-1947) coined the term 'action research' to describe "a particular kind of research that united the experimental approach of social science with programs of social action to address social problems". In this context, "social problems (as opposed to a scientist's own theoretical interests) should serve as the impulse for social research" (*ibid*). Nieuwenhuis (2007:74) refers to the "typically cyclical [nature of action research] in terms of data collection and analysis"; whilst Schwandt (2001:3) notes the use of "a spiral of interlocking cycles of planning, acting, observing, and reflecting" – a useful device ideally suited to this study (as the researcher could investigate a number of related contexts in an iterative manner). Further, the FIFA 2010 World CupTM is an historic event from an African perspective that presents a unique opportunity to contextualise this study with respect to UD.

Notwithstanding, the merit of employing action research has often met with challenges (and strong opposition) within the research domain (Goulding, 1998). For example, Nieuwenhuis (2007:74) reports that action research "was viewed with caution, partly because of its lack of focus on the positivist standards and partly because of its limitation in the number of sites used to conduct the research". Schwandt (2001) notes similar criticism within the social scientific community, especially on the grounds that action research "blurs an important distinction between theory and practice". To reduce the potential for researcher bias and to ensure the validity of the study, mixed/multi- methods (triangulation) were employed.

This study utilises an assortment of qualitative methods to aid in the investigation of the issues at hand and includes the following (Bailey, 1987; Wilson, 1990; Crotty, 1998; Schwandt, 2001):

- experimentation;
- observation;
- case studies;

- survey using questionnaires (with appropriate rating and ranking scales);
- workshop/focus group discussions;
- interviews (structured and semi-structured); and
- desk review of available literature.

7.2 Experimental Design and Sampling Method

Bailey (1987:214) identifies two approaches to social research concerned with establishing relationships between variables: *correlational analysis* and *experimentation*. The fundamental difference in the two approaches is that in correlational studies researchers "typically have very little control over the research environment. [...] In such a case the researcher's control is limited almost entirely to statistical data manipulation". In contrast, the researcher "is present on the scene when the data are collected and exercises considerable control over the experimental environment. This control over the research process allows the experimenter to attempt to establish causation rather than mere correlation, and thus the establishment of causation is the goal of the experimenter" (*ibid*).

Bailey (1987) further distinguishes between 'pure' experimentation and quasi-experimentation – the latter utilise 'partial' experiments lacking one or more of the factors/characteristics of pure experiments (*ibid*). The steps taken in a typical experimental process include the following (Bailey, 1987:221):

- Select subjects;
- Select experimental environment;
- Pre-test;
- Administer experimental stimulus (test factor); and
- Post-test.

Of the various forms described, the *Universal Design for Product Designers* module (described in Section 5.2.3.1) followed a 'one-group pre-test post-test design'. Such a design has all the steps listed above but has no control group (Bailey, 1987). The choice of this variant of experimentation was due to tight time constraints within the academic timetable (a one-week dedicated week). The UD module was designed to determine whether students/learners would be interested and/or motivated in utilising Principles of UD in their projects once they were exposed to the same. Table 7.2 gives a detailed breakdown of the project activity plan followed by the researcher. Two fellow lecturers (who were in charge of the experimental group of

students) played the role of observers and repertoires to reduce the possibility of bias as well as to enhance the validity of the findings from the study. This experimental study informed a larger multidisciplinary survey to establish whether the findings could be triangulated (and ultimately generalised).

		PROJECT ACTIVITY PLAN: Universal Design for Product Designers module			
DAY	STAGE	In core lecture during the dedicated module in the Third Year Design Studio – done in a group setting	Design exercise during the dedicated module (individual)	Homework for the following day – done individually	
One	Pre-test Introduction, discussion of module objectives, and handouts on Principles of Universal Design Researcher as an active participant observer	 08.45-9.00 – responses by learners on knowledge of and/or use of UD 09.00-09.20 – PowerPoint presentation: <i>Special</i> <i>Human Needs</i> (Researcher) 09.2009.30 – Discussion on the exclusion of "extra-ordinary" populations and elaboration of terms and concept relevant to UD 09.30-10.15 – PowerPoint presentation: <i>Principles of UD</i> (Researcher) 10.15-10.30 – Interactive discussion on learner's personal experiences with respect to vulnerable populations 10.30-11.00 – 30 minute break 12.00-12.45 – PowerPoint presentation: <i>The</i> <i>Pathology of Blindness</i> (Guest Lecturer) 10.15-10.30 – Interactive discussion and questions relating to guest lecture populations 13.00-13.45 – Lunch break 13.45-14.30 – Small group (of up to 6 members per group discussions) with facilitator and two lecturers as observer and discussants 14.30-14.40 – 'Health' break 14.40-15.30 – Continuation of small group discussions 13.15.30 – End of formal contact for Day 1 	Sketches and notes Identification of 3D project (done previously) for analysis and (re)design using Principles of UD	Identification of various vulnerable populations – this exercise may also be done in a small group setting Self-initiated visits to see first-hand the challenges experienced by vulnerable groups with special reference to the elderly and people with disabilities Learners to take turns to use provided wheelchair during the course of the day to access all facilities in the Design Building to experience limitations and barriers within the built environment	
Тwo	Self-directed learning and group discussions by learners <i>Researcher</i> <i>as a non-</i> <i>participant</i> <i>observer</i>	 09.00-09.45 – Implementation of Principles of UD in furniture design project (facilitators: Lecturers 1&2) 09.45-10.45 – learners to participate in and complete institutional student Satisfaction Survey Breaks and schedule for remainder of the day (to be decided by Lecturer 1) 	Sketching, drawing and dimensioning of various proposal under the guidance of the 3 rd Year coordinators (Lecturers 1&2)	Discussion of problems facing vulnerable populations Learners to continue the use provided wheelchair during the course of the day to access the built environment	
Three	Learners present their (re)design interventions Researcher as a participant observer	 09.00-110.30 - First Group of learners .present, discuss and answer questions for 5-10 minutes each (facilitator: Researcher; observers: Lecturers 1&2) 10.30-11.00 - Tea break 11.00-13.00 - Continuation of presentations by learners 13.00-13.45 - Lunch break 13.45-15.30 - Continuation of presentations by learners 15.30 - End of formal contact for Day 3 	Sketches and notes	Formulation of design concepts to resolve identified problems by integrating Principles of UD Learners to continue the use provided wheelchair during the course of the day to access the built environment	

Four	Learners present their (re)design interventions <i>Researcher</i> <i>as a</i> <i>participant</i> <i>observer</i>	 09.00-110.30 - Second Group of learners present, discuss and answer questions for 5-10 minutes each (facilitator: Researcher; observers: Lecturers 1&2) 10.30-11.00 - Tea break 11.00-13.00 - Continuation of presentations by learners 13.00-13.45 - Lunch break 13.45-15.00 - Feedback Session and Comments on UD Module and Assignment by Lecturer 2 at the end of the final presentation. 15.00 - Wrap up and concluding remarks at the end of formal contact for Day 4 	Sketches and notes	Learners to continue the use provided wheelchair during the course of the day to access the built environment Learners to submit evaluation forms on UD module for quality assurance purposes Learners to make appointments for feedback on summative assessment for following week Complete Self-Assessment of UD module for submission on Day 10
Ten	Learners submit Self- Assessment forms(Post-test	 09.00 – Deadline for submission of Self- Assessment of UD Assignment write-up Commence grading and documentation (Researcher) Learners finalise appointments for feedback interview with researcher 	Learners encouraged to apply UD in their future projects	Final submission on UD module and related assignments
Twelve-	Learners	1. Researcher completes documentation and		
Fourteen	receive	analysis of assignments and notes key outcomes		
	from	purposes		
	Researcher			

7.3 Role of Researcher

Consistent with action research practice, the role of the researcher was that of an active facilitator/participant observer in the case of the one-week classroom experiment and focus group discussions. During the 2010 student competition, the initial and final contact sessions with students, as well as briefing presentation to adjudication panellists is in a similar capacity as a participant observer. However, the researcher assumes a non-participant observer status during stages within the various research studies that require only observation and recording/documentation of observations and informant responses. The administration of questionnaires for the student survey was carried out with research assistants and thus required minimal involvement of the researcher (save for the pre-testing/piloting stage).

7.4 Research Activities

This research primarily employs qualitative methodological approaches drawn mainly from the fields of ergonomics and social sciences and utilises both deductive and inductive processes as relationships between certain variables are not always explicit (Gray, 2004). As the emphasis of this study is on promoting change within design praxis, an action research orientation provides such an opportunity to carry out pertinent case studies, engage with various actors involved in the FIFA 2010 World Cup[™] projects in Cape Town, interview design educators and professionals, as well as carry out experimentation using student projects at
CPUT in alignment with the theme of UD and/or the mega event. The said study and student projects focus on trans-disciplinarity and *collaborative design* teamwork across various design-related sub-disciplines (such as architecture; graphic/information design; industrial design; informatics; interior design; and town and regional planning) in the Greenpoint Stadium and urban park projects. The potential role of leaders or champions in promoting change (in this case the adoption and mainstreaming of UD strategies) was also investigated.

The empirical research is positioned within an Activity Theory (AT) framework (Engeström, 1999; 2000; Decortis, 2000; Collins *et al.*, 2002; Korpela *et al.*, 2006). As discussed earlier on in Chapter 6, AT provides a practical analytical lens to view the activities of multiple actors in developmental contexts. Apart from the on-going literature and databases search (with specific reference to UD-related activities globally), the following research methods (as shown in Figure 7.1) were used:





7.4.1 Observational Methods

Direct observation of an unobtrusive nature coupled with expert analysis (involving walkthroughs of drawings and models, guidelines, audits and checklists) was used to complement analysis of available archival data and published information (Wilson, 1990). Further, similar observation and documentation of students' PowerPoint[™] presentations during

the classroom experimentation were used to determine the levels of UD adoption and interrogation by participants.

7.4.2 Database Methods

This process involved consulting books, journals, newspapers, internet-based resources, and other published information available from the case study (such as drawings, plans, and strategic documents). Advice from experts was also obtained to validate available raw data.

7.4.3 Subjective Methods

This included the use of scaling, rating, and ranking tools to measure such cognitive outcomes as acceptability, knowledge and practicability of UD strategies by participating informants/respondents (Sinclair, 1990:62-63). Nominal scales – those used to capture only one specific variable or dimension (such as gender) were used for participant demographic data in the student survey (Bailey, 1987). The Likert scale was adopted for use measuring more subjective responses during the classroom experimentation and questionnaire survey (Oravec, 2002). Bailey (1987:466) notes that Likert scaling is "a type of summated rating scale designed to aid in elimination of questionable items from the scale" (see Section 7.6 for more elaboration).

7.4.3.1 Personal Interviews

Semi-structured questionnaires were administered to key informants in the expert/reference group for primary interview purposes to compliment questionnaires, as well as to identify potential gaps in information. The responses were open-ended and based on informants' experience. The decision Delphi technique was employed in those instances where consensus on key issues was desired.

7.4.3.2 Questionnaires

The questionnaires used for interviews with experts consisted of both closed and open-ended questions, which were used to gather qualitative information with respect to the study (Denzin & Lincoln, 2005). Structured and semi-structured questionnaires were administered to design practitioners (including student designers, professional designers, and design educators) to determine their level of engagement with UD-related concepts. Gray (2004) proposes the use of quota sampling techniques to complement general convenience samples to ensure the inclusion of typically under-represented categories of subjects such as female students or

students with disabilities. Table 7.3 provides a breakdown of the sample population and selection for the student survey (see Appendix F for student enrolment figures).

Study Discipline	Total student population	Anticipated number of informants to be sampled	Actual number of informants sampled for survey
Architectural Technology	99	20 (20.2%)	29 (29.3%)
Graphic Design	239	48 (20.1%)	44 (18.4%)
Industrial Design	157	32 (20.4%)	38 (24.2%)
Interior Design	106	22 (20.8%)	21 (19.8%)
Surface Design	80	16 (20.0%)	0 (0%)
Town & Regional Planning	167	34 (20.4%)	15 (8.98%)
Total	848	172 (20.3%)	147 (17.3%)

 Table 7.3 Sample population and selection (author's construct)

7.5 Workshops and Focus Groups

Workshops and focus group discussions were held during strategic phases of the research in collaboration with key actors drawn from local and provincial government, industry (professional designers and manufacturers), the disability sector, and collaborating academia (from CPUT, and interested individuals from other higher education institutions in the Western Cape Province). Boonzaier & M'Rithaa (2007) ran a participatory workshop that served to stimulate debate on different perspectives on UD-related concerns and to promote broad-based participation (see Appendix O for a summary of issues arising at the workshop at the 2007 DEFSA conference). The researcher also attended two conferences, one national and another international to share emerging outcomes of the study.

7.6 Data Collection Methods

With the product analysis survey method, photographic capture, voice recording, and questionnaires were used as the main tools or instruments of data collection to generate quantifiable data (Denzin & Lincoln, 1998). Questionnaires primarily employed Likert scales due to the relative unfamiliarity of UD in the context being researched. Further, Maree and van der Westhuizen (2007:167) concur that a Likert scale "is very convenient when the researcher wants to measure a construct" as well as allowing for the "ordinal measure of an informant's attitude" in terms of the "strength of feeling or attitude".

7.6.1 Coding Framework

Interviews and resultant responses were recorded, transcribed, coded and analysed using appropriate conventions (Sapsford & Jupp, 1996; Durrheim, 1999). Table 7.4 shows the main thematic coding and analytical protocols used in this research, as well as their properties, codes, descriptions, interpretation parameters, and examples from interview transcripts. The related data is represented graphically in the final chapter of this thesis. Three research assistants operated independently as coders to ensure inter-rater reliability and validity of the data (Strauss & Corbin, 1997; Boyatzis, 1998).

Table 7.4 Thematic coding framework (author's construct)

Theme	Property	Code	Example	Description	Interpretation
					Parameters
Identification with Vulnerable Populations	Personal Factors	PER	<i>"I believe it is the right thing to do: I call it 'vulnerable'</i> ".	Personal attitudes towards vulnerable members of society	Personal operations or actions that influence an individual's attitude
			This sector of community [senior citizens] has made an invaluable contribution to society.		towards vulnerable populations
	Social Factors	SOC	"DAWN: Democratic Alliance Women's Network with a specific focus and emphasis on women and gender issues. They look at empowerment programmes, network opportunities amongst women and the like. The same applies to children [] We also have	Past or prevailing practices by society towards its vulnerable members	Group activities that influence an individual's attitude towards vulnerable populations
			the DA Youth ".		
	Political Factors	POL	"the Democratic Party gave room for the individual to express your views, your opinions, and as we know, liberal democratic fielded are	Factors related to the way in which formal	Structural responses to issues relating to the
			underpinned by the rights of the individual, rather than using a group as means to gain	structures are formed and organised	Vulliciane
			power but rather focusing on the individual".		
	Technical Factors	TEC	"And when I speak of short-term, they will look at building a new road (and we've seen that in our own area where we live which I've now done) but they will look at particular issues such as new roads, or a new community hall – things that are very important – because that contributes to the functionality of your particular area where you live, work and play".	Technical factors that focus on design interventions for vulnerable populations	Factors dealing with the application of scientific knowledge to solve practical design problems
Familiarity with UD- related concepts	Identification		"Destination identification! The people at the City Health Department have made Bothasig Clinic a pilot project for me (as Chair of the Health Portfolio).	Specific association of symbols with approximate meaning	Accurate identification of Principles of UD symbols
	Definition	DFN	"But what is vulnerable, or who are deemed to be vulnerable? [These are] the youth , senior citizens , and the disabled . Those are the three categories which I describe as being the most vulnerable sectors of our society "	Familiarity with the inherent meaning(s) of UD-related concepts	Descriptive statement of understanding of UD- related concept(s)
1	Knowledge	KNW	"That is why I say this is what inspires me and	Formal understanding	Declaration of

			when you look at each of those individually, their needs are almost the same, or rather, let me say, the outcome-based approach of theses three are almost the same because they need something from society to keep them going".	of UD-related concept(s)	knowledge of UD- related concept(s)
	Application	APL	"They [City Health Department] have used a user-friendly integrated colour-coding [at the new Bothasig Clinic] as a way-finding system to make it easier for visiting (out) patients find the venue more accessible".	Specific Principles of UD used	Declaration of actual use of UD principles
Participation/Interest in 2010 Projects	Events	EVT	"2010 is maybe a vehicle (or catalyst) , while we appreciate that 2010 will come and go. We need to see if 2010 is outcome-based, whether we would be able to see a lasting benefit for the people of Cape Town – all people of Cape Town	Specific or anticipated operations, actions, and activities	Declaration of interest or involvement in 2010- related activities
	Logistics/ Planning	LOG	"The DA also has a Disability Desk . We also have a dedicated DA Member of Parliament tasked specifically to deal with issues relating to People with Disabilities, because when one introduces legislation, one must never ignore the right of the disabled".	Specific operations, actions, and activities	Planning that impacts directly on 2010 projects
	Policy	PCY	"something which we are pushing from the DA will be the Extended Family Tax Allowance ".	Actual policies that relate to related issues	Formulation of policies leading up to and/or beyond 2010
	Projects	PJT	"So what we've done is now we've introduced a pre-packed system [for chronic medication]".	Specific or anticipated operations, actions, and activities	Declaration of interest or involvement in 2010- related projects
Integration of UD Principles	Individual	IDV	"Universal Design is an exciting and creative concept which aims to achieve more-or-less what we have set out ourselves to achieve".	Interest in using UD Principles	Declaration of personal commitment to adopting/using UD Principles
	Professional	PRO	"in the month of October and November we [in the DA] have 'Senior Citizens' Month' and other awareness events, and our proposals to influence the powers that be ".	Professional body's interest in using UD Principles	Declaration of professional body's commitment to adopting UD Principles
	Group	GRP	"Where the DA shares your views on how to influence legislation, how to influence policy in our country. To speak in favour and to the benefit	Group's interest in using UD Principles	Declaration of group's commitment to adopting/using UD

			of these people (these sectors of society) then the DA can benefit".		Principles
	Community	COM	"And if the DA can benefit, people ultimately also benefit".	Community's interest in using UD Principles	Declaration of communal commitment to adopting UD Principles
Promotion of UD concept	Self	SLF	"I'm also saying that we have a duty, and a responsibility (I'm going to give this to you) to take special care of our senior citizens (we are only taking about seniors now, but this can apply to the disabled and the youth)"	Personal interest in promoting UD	Declaration of personal commitment to promoting UD adoption
	Organisations/ Institutions	ORG	 "This could be a long-term relationship (with UD) and one can only benefit and the people [of my Ward] will ultimately benefit". "I am going to set up an appointment with our [DA] policy advisors and possibly the [Executive] Mayor [of Cape Town]". 	Strategic interest in promoting UD	Declaration of organisational / institutional commitment to promoting UD adoption
	Champions	CHP	"The Executive Mayor , Helen Zille (as leader of the DA) leads by example by reaching out to refugees. She was instrumental in organising a summit on refugees in Cape Town in 2007 involving all stakeholders".	Names of key actors to champion UD at various levels of society	Identification of potential candidates for championing UD concept in Cape Town
	Media	MED	"This is a copy of a speech I delivered on Friday [18 th April] morning. Today I'm speaking at the Tygerberg Service Centre for the Aged, Friday the 18 th of April [2008]"	Specific or anticipated operations, actions, and activities	Employing strategies for promoting UD adoption

Key

PER – Personal FactorsSOC – Social FactorsPOL – Political FactorsIDT – IdentificationDFN – DefinitionKNW – KnowledgeEVT – EventsLOG – Logistics/PlanningPCY – PolicyIDV – IndividualPRO – ProfessionalGRP – GroupSLF – SelfORG – Organisations/InstitutionsCHP – Champions

TEC – Technical Factors APL – Application PJT – Projects COM – Community MED –Media

7.7 Data Evaluation Methods

The data analysis was conducted side-by-side with data collection, data interpretation and narrative report writing. Information on the number of returns and non-returns of the survey was concurrently reported. This information was presented in tabular format with special attention to the number of informants and non-informants (see Table 7.2 as an example). *Summarising, explicating* and *structuring* types of content analysis as described by Gray (2004:328-329) was done on available documents (such as draft policies, plans and legislation) and used in conjunction with other typical desk review methods on relevant categories, labels and words such as "accessibility", "disability", "inclusive/universal design" and "usability". Similarly, the above data informed the pre-coding and thematic analysis of data drawn from additional loci of the study.

Simulation, systematic observation and discourse (conversational) analysis were used in the experimentation aspect of the research as they are complimentary to the action research setting in the classroom (Krippendorff, 1980). Foster (1996:59) notes that observation offers advantages over interviews and questionnaires in that it allows for recording of human behaviour directly "without having to rely on the retrospective and anticipatory" responses of informants. Bailey (1987) however cautions against the danger of "reactivity". Bailey (1987:31) describes a reactive research as "one whose application causes a reaction on the part of the persons being studied in such a way that the data are affected".

Within practical limits, different sources and methods of data collection and evaluation were employed to permit triangulation, especially with "why" questions thereby enhancing interpretability (Silverman, 1993). This is particularly prudent in the case of work system analysis and specific task analysis (Stammers, Carey & Astley, 1990). The keeping of an updated "audit trail' by means of a *reflexive journal* aided in the process of evaluation (Kelly, 1999).

The data analysis was based on 'reduction' and 'interpretation' – that is, voluminous amounts of information was reduced to certain patterns, categories or themes prior to interpreting this information by using specific schema (Sapsford *et al.*, 1996). During data analysis, information was organised categorically and chronologically, reviewed repeatedly and continually coded; then analysed with *Microsoft Excel*[™]; and version 13 of the *Statistical Package for Social Sciences* (SPSS)[™] software. The research necessitated an iterative approach as many of the studies overlapped with time. An open relationship also permitted key role-players to feed into the process at their convenience throughout the entire research process.

131

7.7.1 Testing of Pictograms

Pictograms for seven *Principles of Universal Design* developed by Beth Tauke (Danford & Tauke, 2001) were presented in Question 6 in the students' survey questionnaire (see Appendix L). This was a supplementary component to the study as the researcher sought to know whether indeed the pictograms were as 'universally applicable' to contexts outside of their original locus of development. As such, informants were required to simply assign a semantic label to the pictogram that best reflected (or approximated) the theme of the said label/principle.

7.7.2 Case Studies and Exemplars

The case study of Kirstenbosch Botanical Gardens (discussed in Section 5.2.3.2) and data base of UD exemplars were documented photographically. Additionally, the *Universal Design Audit Checklist* (Appendix D) was used to carry out an exploratory accessibility audit in Kirstenbosch. The said images are presented in CD-ROM format in folders appropriately labelled under the seven *Principles of Universal Design* appended to this thesis.

7.7.3 Students' Design Competition

The visual and text-based submissions of proposal for landscaping and furniture design solutions from the 2010 Students' Design Competition (discussed in Section 5.2.3.3) are also presented in CD-ROM format and can be made available by the researcher.

7.8 Ethical Considerations

There are many role players and interested parties in a research of this magnitude. It was thus essential to consider ethical issues relating to each of them. These key actors include:

- the participant, informant or subject for the purposes of this study, the preceding terms as well as actor, collaborator, interviewee, respondent, and the observed are represented by the single term *informant* as "used to identify the individuals that a researcher studies" (Schwandt, 2001:126);
- the researcher; and
- the sponsoring body.

7.8.1 Ethical Issues Relating to Informants

Any research must be ethical especially when dealing with issues of subjects involved in the study, whether they are human beings or animals. The following measures were taken to ensure that the study is ethical:

- The informants were not compelled into participating in the study. A signed consent form (Appendix E) was one of the pre-requisites for participation.
- Any informant who chose to withdraw from the study and revoke the information she/he supplied to the investigator was granted their wish without prejudice.
- Information collected is confidential so as to protect the informants from any (potential) psychological harm.
- Information on personal details by informants was optional and treated with due confidentiality.
- Names of the informants (or their organisations) were not used in the findings if doing so could harm their reputation or jeopardise their work. Explicit requests were made to informants to allow their names to be included in this thesis.
- The informants were not deceived but were told the truth about the intent of the study. Informants were not seduced into the research by giving them any incentives to participate in the study. Instead the importance of the study was emphasised to them.
- After analysing data and finishing the report, the informants were provided with the findings so as to clear any misconceptions arising during the data collection phase.

7.8.2 Ethical Issues Relating to the Researcher

The researcher was ethical when dealing with the research and the following issues were taken into account:

- The investigator endeavoured to ensure that the data collection procedure and interpretation was not biased.
- Appropriate methodology was used when conducting the study.
- The results of the report were communicated correctly without bias.
- There was no information obtained from the subject in any adverse manner to them.
- The investigator endeavoured to uphold accepted and expected code of ethics principles incorporating among others, the guidelines of beneficence, respect for human dignity and justice.

7.8.3 Ethical Issues Relating to the Sponsoring Organisation

The primary source of the funding for this study came from the Cape Peninsula University of Technology (CPUT) from conception to completion. There was no chance of manipulation of the study by the sponsor because the research was not commissioned by nor conducted for the

sponsor but for academic purposes. Advice was sought from the Doctoral Supervisors concerning unforeseen ethical implications.

7.9 Summary

This study benefited from an action research approach as it allowed access to students at the Cape Peninsula University of Technology, Cape Town to participate in quasi-experiments and design competitions addressing UD. Further, thematic codes emerging from the qualitative data generated during the various stages of this research greatly enriched the understanding of prevailing attitudes towards UD, and adoption of the same in planning and general design practice. The next chapter will elaborate on the specific implications of the findings from this study, not only in the context of the anticipated FIFA 2010 Soccer World Cup[™], but future aspirations for an 'Inclusive Cape Town'.

CHAPTER EIGHT DATA ANALYSIS AND DISCUSSION OF FINDINGS

8.0 Introduction

This chapter introduces the detailed data analysis protocol and discusses the dynamics between various actors involved in activities relating to the FIFA 2010 Soccer World CupTM in Cape Town. The actors involved in this study include design students, design educators, project planners, professional designers, as well as experts drawn from the fields of politics and indigenous knowledge systems (IKS). In the case of the student survey, the data from structured questions is recorded statistically, whereas semi-structured and open-ended responses were thematically coded using sentences as the main units of analysis, before being broken down further into statements (which are viewed as specific thought units) – in turn made up of contextual use of phrases, clauses or words (Weber, 1990). Similarly, all interviews were coded and analysed thematically.

8.1 Identifying Universal Design Exemplars

The criteria for identification of specific typologies or exemplars was aligned with the seven *Principles of Universal Design* advanced by Story (2001) and the Center for Universal Design (1997), and elaborated in the UD poster (Appendix B). Initial visits to various parts of the city of Cape Town included Metrorail platforms and stations; bus and multi-passenger taxi (minibus) stations; Kirstenbosch Botanical Gardens; and public buildings (including entrances, lobbies, and toilets). Photographic documentation focussed on functional accessibility and usability. The case study (in Figures 5.8-5.23) shows some of the exemplars, whilst the CD-ROM appended at the end of this thesis contains a more comprehensive database of collected exemplars.

8.2 Roles of the Principal Actors

As summarised in Section 6.1, the three-level AT model proposed by Leont'ev utilises operations (directed by conditions at the non-conscious level of the subject), actions (directed by goals at the individual or group level of the subject), and activities (directed by objects/motives at the collective level of the subject) – this study primarily focuses on the last two levels as they deal with group level activities (Engeström *et al.*, 1990). Further, Engeström (*ibid*) defines the subject as the individual or group whose point of view is taken in the analysis of the activity; and the object (or objective) as the target of the activity within the system. This thesis adopts designers as the keystone subjects (though other principal actors' points of view is acknowledged), and identifies the object (or objective) as that of mainstreaming UD in Cape Town.

8.2.1 Activity Analysis: UD in Academic Settings

The quasi-experiment was carried out in a classroom/design studio for third year product design students. The UD module was designed to articulate into a dedicated week within the 'Technology 3' course. The objective was to engage with students to ascertain their willingness to incorporate UD in the (re)design of three-dimensional artifacts/products. The key actors within this design education setting (as illustrated in Figure 8.1) included the following:

- Industrial design lecturers;
- Product design students; and
- Members of the disability sector.



Figure 8.1: Actors within the educational environment

The specific group level actions of the various actors (shown in Figure 8.2) were:

- Industrial design lecturers: *planning and coordination of teaching, learning and research activities;*
- Product design students: class participation and submission of assignments; and

• Members of the disability sector: *advocacy on disability-related issues, as well as* offering expert advice – in this instance with specific reference to the needs of people with visual and sight disabilities.

Typically, members of the disability sector operate independently of educational settings (except in the *special* and *inclusive education* milieu). In the case of CPUT, the disability sector's expertise was not formally incorporated into the curricula as guest lecturers and content specialists prior to the commencement of this dedicated module. A manager from *The Cape Town Society for the Blind* (CTSB) was invited as a guest lecturer for the UD module.



Figure 8.2: Group activities within the educational environment

At the organisational level, the interaction of these actors happens in the Cape Peninsula University of Technology – specifically at the Department of Industrial Design at the Cape Town Campus. The following illustration (Figure 8.3) shows the broadly defined activities.

Cape Peninsula University of Technology

Design Students		Design Educators
Design assignment	15	Planning & Coordination
	Disability Sector	
	Advocacy and advice	

Figure 8.3: Organisational level activities within the educational environment

Similarly, the societal level activities are located in Cape Town. The various groups are illustrated in Figure 8.4.

Cape Town

	Cape Peninsula University of Technol	ogy
Design assignment	nts	Planning & Coordination
	Disability Sector	
	Advocacy and advice	
	Civil Society Members	



An overview of the operations and actions of various actors shows the linkages inherent in such activities (Figure 8.5). The dashed red lines indicate communication flows that need to be strengthened so that future collaboration can be sustained.



Figure 8.5: The educational environment: an overview

The following activity analysis (Figure 8.6) elaborates on the abovementioned. Students typically rely on their lecturers almost exclusively for all content input relating to a design brief. The contradictions or tensions arising here are that the failure on the part of the student to consult more widely with experts results in design apathy – the resultant design solutions lack empathy and authenticity. Further, the disability sector relies principally on governmental and non-governmental policy documents for direction. The disability sector on its part rarely engages academia in their quest for pertinent data and related information. Such traditional sources are usually dated and lack currency and authority.



Cape Town

Figure 8.6: Activity analysis of the educational environment

The activity analysis reveals the following implications:

- Design activities focusing on people with special needs should incorporate the perspective of the end-users, preferably as advisors or content experts;
- Lecturers should facilitate (non-prescriptive) contact between students and industry;
- Greater collaborative (action) research should be encouraged between the disability sector and higher education institutions. Such collaboration could also be extended to student projects to sensitise and expose the learners to emerging trends; and
- Mainstreaming of UD was a unifying object amongst community members.

8.2.2 Activity Analysis: UD in Public Parks

Kirstenbosch Botanical Gardens (as discussed earlier in Section 5.2.3.2) is the most prestigious of a large network of national botanical gardens and public parks under the National Botanical Institute administered by the South African Biodiversity Institute. The UD Audit performed by CPUT was not solicited by the management of Kirstenbosch, but was rather inspired by the quest to validate the efficacy of UD in public parks. The research focussed on the accessibility and usability of the features incorporated in the gardens to address UD-related concerns raised by specific categories or groups of visitors. These groups included:

- foreign visitors (speaking languages other than one of the 11 official ones);
- local visitors from every walk of life; and
- people with disabilities (particularly those experiencing physical and visual limitations in functioning).

A number of actors playing a variety of roles impact upon the Kirstenbosch Gardens setting. The actors in this particular environment (as illustrated in Figure 8.7) include:

- Kirstenbosch Botanical Gardens management represented by the Estate Manager;
- the South African National Biodiversity Institute (SANBI);
- the educational community comprising of design educators, researchers and students;
- professional landscapers as well as architects, ecologists, engineers and cartographers;
- contractors, suppliers and other service providers; and
- members of the disability sector especially those specialising in physical and visual disability.



Figure 8.7: Actors within the botanical gardens environment

At the group level, the actions of the various actors (as shown in Figure 8.8) include:

- Kirstenbosch Botanical Gardens management: planning, negotiating and coordinating accessibility projects;
- the South African National Biodiversity Institute (SANBI): strategic planning, direction/oversight and funding of projects;
- the educational community: planning, research and execution of UD audit;
- professional landscapers: responsibility for design-related activities and formulation of detailed design briefs;
- contractors, suppliers and other service providers: supply, implementation and commissioning of accessible technologies and infrastructure; and
- members of the disability sector: advocacy on disability-specific needs and expert advice on user needs.



Figure 8.8: Group activities within the botanical gardens environment

Each of the groups organises its own rather specialised activities which in turn are coordinated by the Kirstenbosch Gardens management team. The organisational level activities are set in Cape Town as illustrated in Figure 8.9. The key actors include CPUT, Kirstenbosch Gardens, SANBI,

professional landscapers, suppliers/service providers, and civil society members from the disability sector.

Cape Town



South African National Biodiversity Institute



Contractors/Suppliers/Service Providers



Figure 8.9: Organisational level activities within the botanical gardens environment

The societal level activities are set nationally (in South Africa). The various groups are shown in Figure 8.10. The national coverage of the South African Biodiversity Institute provides an ideal

platform for coordinating strategic and logistical planning activities that mainstream more inclusive and accessible public parks and botanical gardens.



Figure 8.10: Societal level activities within the botanical gardens environment

Figure 8.11 presents an overview of the pertinent operations and actions of the key actors in the Kirstenbosch Gardens accessibility enhancement development. The role of CPUT was in an *ex post facto* capacity in performing an audit of the infrastructural and technological developments at the botanical gardens. The dashed red line indicates a lack of a synergetic link between CPUT and suppliers, particularly those involved in the provision of inclusive and accessible technical products and systems. The engagement of a technically-orientated programme such as the disciplines of architectural technology, interior design, industrial design and graphic design would

potentially lead to the development of locally sourced and context-responsive design solutions as opposed to the over-reliance on (often prohibitively expensive) imported solutions.



Figure 8.11: The botanical gardens environment: an overview

The activity analysis in Figure 8.12 reveals a number of issues with implications for mainstreaming UD in public parks and botanical gardens. The role of the management teams operating the public parks is pivotal as this is where all the energies are focussed – they are the custodians of the object of the activity. In the case of the Kirstenbosch Gardens, an open-minded and responsive management engaged various actors is focusing on the task of revamping the gardens to improve usability and accessibility for foreigners, the elderly, and people with

disabilities. Certain contradictions/tensions were identified around three key areas: participation; consultation; and communication. An elaboration follows:

- *participation:* there needs to be greater interaction between CPUT and design professionals for the latters' input to be incorporated in academic projects;
- *consultation:* professional designers rely on technical guidelines as opposed to direct user-expert input. Further, CPUT needs to maintain stronger links with suppliers; and
- *communication:* service providers and the disability sector should actively exchange information in a participatory manner.



Figure 8.12: Activity analysis of the botanical gardens environment

8.2.3 Activity Analysis: UD in the Public Domain

The 2010 Green Goal Mouille Point Promenade and Beachfront Student Landscape Design Competition was conceived by the Department of Environmental Resource Management (DERM) of the City of Cape Town and coordinated by local government officials from the City, as well as foreign consultants with experience of the Green Goals initiative first unveiled during the planning for the FIFA 2006 World Cup[™] hosted by Germany (Pezeshki, 2008). Abigail Joustra (2008) reported that the competition was envisaged to elicit creative design proposals from students in landscaping design and allied fields.

The project was highly complex in that it involved the coordinated operations and actions of a relatively large number of primary and secondary actors (Figure 8.13). These include members of the DERM technical team, members of the educational community represented by the *University of Cape Town* (UCT) and CPUT; and professional landscape designers. Further, other actors indirectly involved in this domain include the *Department of Education* (DoE) and allied institutions such as the *South African Qualifications Authority* (SAQA) and the *National Research Fund* (NRF); professional bodies representing various design disciplines; members of the disability sector and other industry partners; as well as independent contractors and allied service providers.



Figure 8.13: Actors within the students' competition domain

The group level activities of the project (depicted in Figure 8.14) also reveal many actions and activities that are similar in nature irrespective of the community-of-practice involved. For example, the DoE, SAQA, and the NRF in the higher education landscape compliment the role of professional bodies. The former accredit academic programmes and finance research activities in HEIs, whilst the latter oversee continuing professional development and regulation of professional practice/standards.



Figure 8.14: Group activities within the students' competition domain

Organisational level activities were set in the City of Cape Town/Western Cape Province as the management of 2010-related projects typically involves different levels of government (local and regional). Figure 8.15 shows a grouping of the three main communities of actors: DERM/City of Cape Town; HEIs (CPUT and UCT); and professional landscape designers. Additionally, the other communities are depicted with respect to the student competition domain.









The societal level activities are a further development of the abovementioned in the context of South Africa as illustrated in Figure 8.16.



Figure 8.16: Societal level activities within the students' competition domain

The overview details some of the specific actions and activities performed in the planning, coordination and implementation of the students' competition (see Figure 8.17). Whereas the initial planning envisaged the participation of landscaping architecture and architectural students at both CPUT and UCT, the quest for broader involvement met with unforeseen challenges. Timetabling and other logistical issues limited the participants to *third year undergraduate* students of *interior design* at CPUT and *first year postgraduate* students of *landscape architecture* at UCT. The personal commitment and dedicated promotion by the coordinators for

the two HEIs was pivotal in ensuring the participation of the said students. Consequently, the judging criteria had to be revised to reflect this dichotomous complexity to avoid disadvantaging any of the students. (Appendix G).



Figure 8.17: The students' competition: an overview

As mentioned earlier, the coordination of such a diverse number of communities and actors required constant flows of communication and information (Figure 8.18). Notwithstanding, several unforeseen challenges necessitated the re-interrogation of the process of the engagement. Further, some contradictions/tensions had to be resolved:

- the lack of commitment from design-related disciplines (other than interior design) necessitated a rethink on the multi/cross-disciplinary expectations in terms of group dynamics, the scaling down of the absolute number of expected student submissions, and the reformulation of judging criteria to reflect the student competency profiles;
- the DERM technical team did not engage the disability sector for specialist input though the interaction with industry partners resulted in additional potential prizes being offered for winning entries; and
- contractors and service provider did not invite the participation of civil society members representing special needs (such as the disabled).



Western Cape Province/South Africa

Figure 8.18: Activity analysis: the students' competition

The activity analyses in Sections 8.2.1, 8.2.2 and 8.2.3 provide insight into the group and collective aspects of the activities discussed. Activity Theory (AT) provides a practical analytical lens to view the dynamics of various actors and communities of practice which though very different in composition in each of the three cases discussed, show how the shared objective (of mainstreaming UD) potentially unites such disparate groups. The following section (8.2.4) is more detailed as it addresses operations and actions at the individual level. Further, the section presents results of analyses of responses in graphical form to elaborate emerging trends.

8.2.4 Thematic Analysis: Institutional Architect's Perspective

The five main themes utilised in the coding framework formulated are derived from the thematic analysis process (as elaborated in Table 7.4). The emerging twenty codes are summarised in Table 8.1.

Theme	Code	Abbreviation
Identification with Vulnerable Populations	Personal Factors	PER
	Social Factors	SOC
	Political Factors	POL
	Technical Factors	TEC
Familiarity with UD-related concepts	Identification	IDT
	Definition	DFN
	Knowledge	KNW
	Application	APL
Participation/Interest in 2010 Projects	Events	EVT
	Logistics/Planning	LOG
	Policy	PCY
	Projects	PJT
Integration of UD Principles	Individual	IDV
	Professional	PRO
	Group	GRP
	Community	COM
Promotion of UD concept	Self	SLF
	Organisations/Institutions	ORG
	Champions	CHP
	Media	MED

Table 8.1 Summary of emerging themes based on coding framework

In the analysis of the institutional architect's responses (Figure 8.19), the three main codes identified (in order of frequency) were those relating to *policy* on UD (20%), *logistics/planning* of 2010 projects (15%), and *identification* of UD-related concepts (12%). These are followed by *social factors* (7%), *technical factors* (7%), and *organisational/institutional* imperatives relating to the promotion of UD (7%) – these seemed to be of equal importance in this respect. The critical value of guiding policy cannot be over-emphasised. According to Adriana Hornea (2008):

The executive management of each university should define via its Disability Policy, the range, as well as degrees of disabilities which are to be accommodated by its facilities and means of instruction. There is no tertiary education institution in the world, which could accommodate all disabilities, out of practical/economic considerations (budgets are always limited, never sufficient).



Figure 8.19: Thematic analysis: an architect's perspective

As Hornea (*ibid*) points out, other institutional priorities tend to dominate the agenda, thus relegating UD-related concerns to the periphery. This challenge is particularly significant for a diversity-embracing institution such as CPUT should it seek to adopt UD as an official policy for addressing issues of inclusion and accessibility – UD must be viewed as a long-term investment and should be phased in gradually. The most effective way to demonstrate the efficacy of UD in an institutional context is in incorporating UD principles in new developments. Retrofitting is more expensive and often offers only superficial or temporary redress. Political commitment is required for "the consistent implementation of the UD principles in all decisions related to the university facilities are *a must*, and *should be a consistent priority* at top management level" (*ibid*).

Hornea (*ibid*) suggests that the Disability Units should be the most vocal and visible champions for UD within the CPUT community. The obstacles identified in the adoption and implementation of UD include: general human resistance to change; mindset (including older institutional cultures); budgetary constraints; and [lack of] in-depth understanding of the UD principles (*ibid*).

8.2.5 Thematic Analysis: Design Students' Perspective

The thematic analysis of responses by the third year interior design students who participated in the landscaping competition reveals a different emphasis (see Appendix I). In this instance, the issue of *identification* of UD-related concepts (17%) is clearly the most important theme (shown in Figure 8.20). The next significant theme was the expressed desire for *group* participation in projects (9%). The next three themes are the identification of *social factors* (8%), *definition* of UD as a concept (8%), and *logistics/planning* for active participation in 2010-related projects (8%).



Figure 8.20: Thematic analysis: a design students' perspective

The value of group/team work was emphasised by the students. The following are quotes from informants regarding what they enjoyed most about the 2010 landscaping competition:

- What was enjoyed about the brief was that it was a group project and we got to interact with each other (Student 1).
- What was nice about the project was that we all had to work together with each other's positive and negative, and we had to compromise, even if we didn't agree, but in the end we produced good work through working together. We enjoyed the group dynamics (Student 2).

The challenges that the students encountered included *time constraints*, *lack of sufficient engagement* with the concept of UD prior to the competition, and *lack of confidence* when they learnt that postgraduate landscape architecture students from UCT would be participating in the

competition as well. Subsequently, the students' courage/confidence increased when the dual criteria (Appendix G) was elaborated and they learnt that they "were not going to be judged at the same level as landscape architect students" (Student 6).

The informants were unanimous in their support for the *involvement of students from other design disciplines* as an integral part of their teams. The various rationales provided include the following:

- I think the students are aware that in industry they will be working in multi-disciplinary teams, and they understand that each discipline brings something different to the team and are kind of conditioned to this idea. It would be wonderful if we could get the opportunity to work in a multi-disciplinary team (Lecturer 1).
- Because where we lack knowledge they can build us up (Student 5).
- Didn't we do that already? Last year's [interior design] third years were going to have to work with the surface designers (Student 7).
- So we have done it before, it was a very successful project, but was also very strenuous and difficult in the beginning because people have different priorities, and we want to be individuals and we like our differences but we still need to work together (Lecturer 2).

The disciplines most likely to be co-opted for membership in teamwork included *landscape architecture*, *urban design*, *tourism* and *graphic design*. In response to the question of whether they talked to (or consulted with) people outside of their own discipline, students reported that they spoke to *graphic designers*, *architects*, and *landscape architects*. One student visited Metrorail to consult technical experts on a train carriage theme that they were exploring.

In response to the question on whether the informants considered competitions to be an effective platform for promoting new design concepts/ideas, the following responses are sampled:

- I think it is an opportunity for anybody to prove themselves and show what they can do in terms of 'green design', and all these things (Student 3).
- I also like to think, and like to promote this, that having competitions geared to young developing designers, you are at the nexus of innovation, because they haven't been tainted by all the other things in the world. So I think competition at this level is probably the best learning curve that students can gain (Lecturer 2).
- It gives us feedback and a sense of achievement when you design something and someone else [in industry] likes it and compliments you for it (Student 4).

- If I design something at home in my room, nobody's going to know about it, but if it's for a competition, there's a big panel of people who knows where it's actually moving to so they can see potential in your work (Student 6).
- Also I think it gives you more freedom to design what you want to design, because at the end of the day you're not going to get a mark for it, whether you win a prize for it or not. Because you can do something you've always wanted to do before but you were too scared because of [the concern over] marks (Student 2).

The students felt that *effective communication* and *allocation of sufficient time* for participation would make future competitions more effective in promoting novel concepts or ideas. With respect to UD, informants stated that what worked best for them was that:

- I found it quite a challenge. I had the really exciting topic of doing the paving, which wasn't like the most design-award-winning topic to have, but to try and incorporate everyone; someone with a blind cane, someone with a wheelchair, people running, etc. It is really tough and it is a big challenge to try and think of all the different ways of doing that. I have not had the chance to apply it to another project since that, but I know I found it really tough (Student 8).
- Incorporating it without making it obvious, because in the bench that I did, I looked at the seating and realised, that if I took one of them away, wheelchairs could roll up to it so that it provides the facility, without putting up a big sign saying "here it is for you to use", you just make it usable, and I think that made quite a big impact (Student 9).

Asked what did not work for them vis-à-vis UD, the informants stated:

- You know what it was? It was: "okay it's an awesome idea, great concept, it includes everyone, then you get down to the practical nature of doing it, and you try incorporating those ideas, but then you have to try get your ideas past the person in charge of the project who has their own ideas, and believes you should have certain barriers. It was an interesting challenge (Student 10).
- We had to constantly remind each other about how someone in a wheelchair was going to get there or access that. So we had to keep it in the back of our minds (Student 11).
- It's like knocking away at a block to create a statue. It becomes very difficult to accommodate everybody. To find the ways, and when you have the ways, it places limitations on the design, and takes away from the aesthetic. To find a design that doesn't scream "I'm here for disabled people" or "I'm here for people with wheelchairs." Every time I designed something I had to go back because I came to

the realisation, that vandals could rip this up, or that there was no place for the disabled to sit. So I had to go back to the drawing board and adapt it (Student 12).

Informants found it difficult not to associate UD with disability-specific needs. Some however found that "it was actually easier to say that [we were] thinking of other people because a child could also use it because it is lower, so [we're] also fulfilling two purposes, kind of. So it wasn't too much of a bad thing that we had to think of more people" (Student 13).

Regarding the level at which UD should be formally introduced to them, there was general consensus that it should begin in first year. Here are some responses:

- It would need to be taught in first year, and in 'Design' but also in 'Services', but definitely in 'Design', because its no just about the technicality of it but also how to integrate it (Student 14).
- I think that the theory should be done in first year, and the practicality should be done in second year. Because in second year we start dealing with wheelchair [users], but in first year we don't have to think about that (Student 15).
- I think first would be the theoretical component, where we look at functions and human occupants and how we make human occupancy viable and comfortable and I think ultimately, all things considered, UD actually extends design to a totally different platform. The more we include it in our answers the bigger the design becomes [and] the more inclusive and democratic it becomes. We should be conditioned from the beginning. So I do agree [with the students] that we should incorporate it at the theoretical level in first year and at a more practical level in second year (Lecturer 2).
- I think for too long we have just been looking at wheelchair people only and the other people get lost along the way (Student 7).
- It is impressed on us that we need to incorporate people in wheelchairs, but in this project we had to look at blind people and deaf people, all kinds of people, children and elderly. Usually we would just throw in a wheelchair line, put a button lower, that kind of thing (Student 11).
- That runner [Oscar Pistorius] that has no feet, that is also something that we will eventually have to start looking at, because that's almost like a new technology, so people who have lost their limbs are not even going to sit in a wheelchair, so we need to look at what the faults are with that and what's going to be (Student 16).

Informants agreed that the principle of *Low Physical Effort* was the one they would most like to incorporate in future. Regarding the way in which UD could best be mainstreamed in interior
design, the informants suggested that engagement with people of diverse abilities would help them challenge some of the stereotypes they acquired regarding diverse users. There was general agreement that the students certainly felt more confident (with respect to professional competence) after participating in the competition. Additionally, all informants stated their intention to continue using UD principles in their future interior design projects.

The only individual (or organisation, or department) offered for consideration as a possible champion for UD in Cape Town was (former President) Nelson Mandela. There was however concern that in attempting to be politically correct in every respect, the impact of the final message could be compromised and potentially lost altogether. To address this ethical dilemma, informants discussed the possibility of using a "green-coloured" fictitious (animated) character that would be neutral and therefore more acceptable to all parties concerned (Student 17).

The analysis of the focus group's responses validates the Vygotskian concept of *internalization* (in Activity Theory) as a transformational process which results in measurable changes in the structure of activity. Internalization "involves social processes and semiotic mechanisms – particularly language [...] – this transforming process depends upon cooperative labour and social interactions" (Bedny *et al.*, 2000:171). The mediating cooperative strategies/tools (as advocated by Chmela-Jones, Buys and Gaede, 2007) were the principles of UD (as semiotic mechanisms) through a social process that invited the students' participation in the landscaping design competition and was simultaneously open to participants from another institution

8.2.6 Thematic Analysis: Indigenous Knowledge Expert's Perspective

The one theme (based on the questionnaire in Appendix J) that stands out most prominently from the analysis of the indigenous knowledge expert's responses is that of *identification* of UD-related concepts (26%). The next significant theme was the need for engagement with UD concepts at the *community* level (17%). UD appears to be an alien concept in the discourse on *indigenous knowledge systems* (IKS). A third theme relating to the promotion of UD suggests that emphasis should be on *self*-driven operations and actions as opposed to a group level activities at the collective level (11%). Further, the *social factors* relating to the identification of vulnerable populations (7%), and the lack of a clear identification with UD-related concepts (7%) suggest the need for clarity in the grand narrative of disability (and other "extra-ordinary" human conditions). The other themes (as shown in Figure 8.21) appear to be of minor significance in the context of IKS.



Figure 8.21: Thematic analysis: an IKS expert's perspective

Nomathamsanga Tisani (2008) links the discussion on UD to 'universality' and 'university' in arguing that these three ideas should counter the notion of homogeneity and class belongingness, and instead "[are] supposed to denote *openness* and *inclusion*". As such, UD strategies (that are cognisant of IKS) should adopt a postmodernist stance and guard against being associated with "a homogenous group, a particular group that speaks a particular language; with a particular background; with particular thought-forms; with specific thought-forms" (*ibid*). Further, disability is arguably "a very difficult concept in an African context" if one posits disability within "the strong binary philosophy [or] dichotomous logic" linked to Western thinking (*ibid*).

Tisani (*ibid*) argues that with respect to apparent physical differences, people considered 'disabled' in modern parlance were traditionally "seen as being *special* because of the difference and actually as being *selected* – as one *elected* because of being special". Subsequently, the focus with respect to "extra-ordinary" individuals was on their *ability*. This view was consistent with the principles engendered in *ubuntu* which humanises the condition by acknowledging the humanity of the person with a disability – *ubuntu* intrinsically respects physical and intellectual diversity – "uniqueness is actually celebrated and recognised" (*ibid*).

Additionally, elective socio-cultural practices including various forms of initiation (such as facial markings and related rites-of-passage) were celebrated for reaffirming the body-image, identity

and (inter)connectedness of individuals and their respective communities and were collective/communal bonding experiences. As such, if children were born with unique features, a conscious effort (usually championed by female relatives) was made to link the said appearance with that of another relative (whether living or deceased). In this sense "the notion of *being different* and *being special*" were strongly correlated (*ibid*). Facial markings that were traditionally a source of adoration are often presently referred to as 'scarring' (or scarification) – a fact that reveals negatory attitudes towards the practice. Further, individuals exhibiting atypical appearance or behaviour are branded as being 'scarred' and elicit strong responses of revulsion or denial outside one's community, possibly due to social ignorance or cultural chauvinism (*ibid*).

A number of people with disabilities in traditional African societies achieved great success in various spheres of life and were consequently honoured by their communities. Tisani (*ibid*) gives the example of a famous traditional leader known as Sandile who "had a limp". Sandile was born in the 1820s was the brother of Maqoma, the renowned military leader of the *ama*Xhosa who fought against the British in the nineteenth century. Both brothers were equally revered for their legendary courage and military prowess.

Tisani (*ibid*) suggests that negative, stigmatising or discriminatory connotations of atypicality have Biblical associations and illustrates this point as follows:

... because also now you get these strange things in the Biblical teachings – the Bible does not allow difference and this albinism is always associated with curse. The notion of the curse has very strong Biblical roots or steerings and [as such] albinism is very difficult. All I know is that around albinism a lot of myths [are] built – one of the things that is said is that they are described in animal-like language such as 'inkawu' (which is a monkey). So they [people with albinism] are referred to as monkeys, and I don't know whether we were referring to them as monkeys because of the vervet monkey. There is a lot of mystery and ignorance about them, but they are embraced by family and friends. So again there is that linear [dichotomy] – that line of revulsion and complete acceptance. [...] So this rejection of something that is different we carry it along because if those children are born, we actually love them. That actually is a condensed African philosophy where you've got what would be seen as antithetical co-existing because they are not exclusive. A day will not be if there is not going to be a night because they are complementary; they are mutually reaffirming. That is our identity. So even that rejection is not finite as [one finds] with a way of thinking that is exclusive...

Ubuntu should not be invoked selectively to buttress responses to people with disabilities as this would perpetuate the stigma associated with observable differences – it should instead engage all members of society regardless of their (dis)abilities in alignment with the UD ideal. *Ubuntu* is informed more by *empathy*, not *sympathy* in response to other people. As Tisani (*ibid*) argues:

Ubuntu is [...] being put into books rather than being a way of life [and is consequently] becoming quite difficult. Because we are not going to begin to have ubuntu when engaging with people who are disabled – that will immediately negate ubuntu – it will be just being either paternalistic or maternalistic. I would prefer a broader definition of ubuntu and even in the 'first culture' you will see [reference to] the relationship of ubuntu between the normative institutions and those seen to be different, because once it is given to them, as [a] special dispensation, then it will no longer be ubuntu...

According to Tisani (*ibid*) women "are pace-setters in African communities in their roles and interaction with everybody be it the young or the elderly; be it the sick or well; be it the different – because it is in rare instances where the mother rejects a child, and will struggle even in the present society". As primary caregivers, women would naturally champion UD-related issues in their own families and communities. With regards to ways in which UD can incorporate IKS for a more sustainable future for all, Tisani (*ibid*) argues:

Sustainability is more important for systems, rather than for concrete examples. People want to give concrete examples and want to give that as an African thought-form. But it is in understanding the underlying systems that hold reality together. And when you think very hard on sustainability it does come out, so we have to understand indigenous systems by sometimes debunking them from the factors that have influenced us. And if you think hard enough you can actually fish out indigenous systems which are sometimes more enriching, more enlightening and [by extension] more sustainable...

8.2.7 Thematic Analysis: Politician's Perspective

The thematic analysis of codes emerging from the semi-structured interview (in Appendix K) with the politician from the *Democratic Alliance* party indicates that *group* activities (10%) and *organisational/institutional* imperatives (9%) with reference to integration of UD principles are most significant. As a category, *participation/interest* in 2010-related projects are important concerns with *logistics/planning*, *policy* matters, and specific *projects* (all with 8%). *Personal* factors (8%) and *social* factors (6%) in identification with vulnerable populations is also significant (at 8%), whereas *knowledge* and familiarity with UD-related concepts (6%) follows

closely. The remaining codes are not as significant statistically, though a comparison with Figures 8.19, 8.20 and 8.21 shows the politician's responses to be the most evenly distributed.



Figure 8.22: Thematic analysis: a politician's perspective

According to James Vos (2008), those categorised as 'vulnerable' include "the youth, senior citizens (or the elderly), and the disabled – those are the three categories which [are] the most vulnerable sectors of our society". As a councillor and politician in the Democratic Alliance (DA) party (and in keeping with the DA's vision of *An Equal Opportunity Society for All*), Vos (*ibid*) views politics as "a vehicle to have a much more longer-term benefit and impact" on the social development agenda in the Cape Town UniCity, with the "objective of bringing focus, relief, and giving more emphasis on these three sectors of society". Whilst acknowledging the usefulness of short-term infrastructural developments, Vos (*ibid*) emphasises the need to adopt an *outcome-based approach* to leverage large-scale projects such as those associated with the hosting of the FIFA 2010 World Cup™ to benefit micro-entrepreneurs.

Vos (*ibid*) sees UD as a useful tool to address the needs of the vulnerable sectors of the society "because their needs though different, at the end will all require the same provisions". The DA is associated with "*liberal democratic values* which are underpinned by the rights of the individual" (*ibid*). Further, the DA has a special focus on women through the *Democratic Alliance Women's Network* (DAWN) as well as a number of initiatives of benefit to vulnerable populations (*ibid*):

• holding the regular Senior Citizens' Month in October and November;

- maintaining an active *Disability Desk* in the national parliament;
- supporting activities for the *Tygerberg Service Centre for the Aged* through the *network of care* concept;
- sponsoring the *Extended Family Tax Allowance* in the national parliament so as to offer tax relief to those families opting to stay with their elderly relatives instead of sending the latter to institutional facilities (such as old-age homes);
- sponsoring *The Funding of Homes for the Aged* proposal in collaboration with CommuniCare to develop ageing-in-place *gap/social housing* for the elderly;
- supporting the call for more *subsidised meal schemes* known as *Meals on Wheels* (without promoting unnecessary dependency by recipients);
- promoting better access to proper health care through the City of Cape Town Health Portfolio Committee (which Vos chairs);
- supporting the roll-out of more *mobile clinics* through *private-public partnerships*;
- the improvement of distribution and collection of *chronic medication* (especially for senior citizens) through the introduction of a *pre-packed system*;
- developing a user-friendly integrated colour-coding as a *wayfinding system* Bothasig Clinic as a pilot project to make the clinic more accessible; and
- hosting *senior citizens* through a regular programme called *Breakfast in Bed with James Vos* which is publicised through the media to raise awareness on the needs of the elderly.

Vos (*ibid*) believes that care for and showing respect for the elderly is a reflection of society's values. He states thus:

I say no community, no country, can really talk about improving the quality of life if it does not show respect for its elderly, if it does not show respect for its youth, and if it does not show respect for the disabled. No country, no city, and no community can talk about improving the quality of life if it does not show respect for the vulnerable. [...] I'm also saying that we have a duty, and a responsibility to take special care of our senior citizens (we are only taking about seniors now, but this can apply to the disabled and the youth). This sector of the community has made an invaluable contribution to society. [...] I say to them; "you know, you must not be put in an old age home, we must keep you here amongst us. We must build communities and build families. We must recognise your valuable contributions because you have the wisdom, you have the persistence, and you have the know-how... With regards to the possibility of incorporating UD within the DA's overall response to vulnerable populations, Vos (*ibid*) states the following:

UD is an exciting and creative concept which aims to achieve more-or-less what we have set out ourselves to achieve. [...] As a political party the DA can benefit from case-studies [on UD] as the DA believes in taking extra-special care of our vulnerable citizens within society. Where the DA shares your views on how to influence legislation, how to influence policy in our country. To speak in favour and to the benefit of these people (these sectors of society) then the DA can benefit. And if the DA can benefit, people ultimately also benefit. So I think most definitely at the level where I am involved (and) I would also like to propose that we put you into contact with our DA policy advisers and researchers so that one can pick up on that from the national level.

Vos (*ibid*) believes that the individuals and/or institutions that would best champion UD in different spheres of life should be "organisations looking after people with disabilities". In his opinion, he believes a number of "well-structured civil societies of active rate-payers, active sports clubs, churches, Lions International, and a number of other organisations [...] would definitely be able to benefit, and to be sustained by engaging with UD" within the ward he represents (*ibid*). The potential legacy of the FIFA 2010 World CupTM for residents (and visitors) in Cape Town should be on the small entrepreneurs who ply their trade within the city of Cape Town – the event should be used as a catalyst to ensure a legacy of sustainable economic development (*ibid*).

8.2.8 Thematic Analysis: Consolidated Perspective

The single most significant theme (with a group average of 14.25%) emerging from the consolidated/comparative analysis shown in Figure 8.23 is that of *identification* with respect to familiarity with UD-related concepts. This places importance on the need to create an accessible database of exemplars that clearly identifies UD and its related principles for ease of recognition. Other significant themes include that of *logistics/planning* (7.75%) and *policy* imperatives (7.25%) with regards to participation/interest in 2010-related projects; *social factors* (7.00%) in identification with vulnerable populations; *group* (6.75%) and *community* level dynamics (5.75%) in the integration of UD Principles in various activities; *self*-driven (6.25%) reasons for promoting the concept of UD; and *knowledge* (5.25%) or familiarity with UD-related concepts. The remaining codes and themes are not a significant and there is a more representative spread in coverage by the design students as compared to the architect and politician's responses. Analysis of the IKS expert's responses indicates that a number of codes are under-represented.



Figure 8.23: Thematic analysis: a comparative view

Consequently, context-responsive UD strategies would need to allow for prior planning and mapping out of key actors within society to be included in the community-of-practice. Further, the coordinators/drivers of the project should engage (potential) participants individually to establish their levels of expertise and commitment. Champions would then be brought on board once some initial research and groundwork has been done to chart out the planned project.

8.2.9 Findings from Student Survey

A survey (see Appendix L) on the levels of engagement with UD by design students was carried out at the Cape Town Campus, CPUT. Whereas six design sub-disciplines within the Faculty of Informatics and Design (FID) were approached at the commencement of the survey, surface design students at the time felt that their design process did not align with the problem-based learning model the others followed. Jewellery Design and Manufacture students were not approached for the self-same reason. Notwithstanding, a senior lecturer in the Department of Fashion and Surface Design participated in the semi-structured interviews administered to design teaching staff. Altogether, 147 students from the Departments of Architectural

Technology, *Graphic Design*, *Industrial Design*, *Interior Design*, and *Town and Regional Planning* participated as shown in Table 7.3.

The top four categories of "extra-ordinary" population groups (in rank order) that the majority of students were most familiar with are: *elderly people*, *young people*, *foreigners*, and *women* who were either pregnant or had toddlers (as depicted in Figure 8.24).



Figure 8.24: Familiarity with "extra-ordinary" population groups (symbols: © David Christer, 2009)

People with *physical* disabilities (including those using wheelchairs on a regular basis) were the form of disability that most students were familiar with. This point is confirmed in the focus group discussion with interior design wherein the most frequently cited representative typology for disability were wheelchair users (see transcribed verbatim responses in Section 8.2.5). Consequently, strategies for introducing UD at CPUT/FID (particularly in the first year of study) could start with discussions around challenges brought about by ambulatory limitations with regards to products, systems and built environments. The experiential exercises facilitated through the use of a wheelchair (as described in *UD for Product Designers* module in Section 5.2.3.1) could be readily extended to other sub-disciplines (Benktzon, 1993; Moore, 2001). Other forms of disabling limitations in functioning could also be simulated to promote empathic design

responses by the students whilst simultaneously advancing their *zone of proximal development* (Vygotsky, 1981; Bedny *et al.*, 2000; Coleman, 2001).

Interestingly, many informants also stated familiarity with persons with *learning* disabilities (such as dyslexia) – a fact attributable to the mandatory requirement at the time that as a prerequisite to entry into design programmes, students had to undergo a battery of *psychometric tests* organised by the CPUT *Student Counselling Unit*. The majority of informants (74%) reported that they had not received any UD-related training during the previous six months leading up to the survey (Figure 8.25). Of those that had received some UD-related training, Architectural technology students had the highest rates of attendance. A UD module targeting the design disciplines of FID could be formulated and coordinated by the Departments of *Architectural Technology* and *Industrial Design* as lead partners/collaborators.



Figure 8.25: Whether the informants had UD-related training during the previous six months

Of the different UD-related concepts, *Design for Disability* and *Design for All* (with 25% each) were the ones that students were most familiar with (see Figure 8.26). A significant number (16%) were unfamiliar with any of the concepts. Indeed, this number was greater than those who were familiar with UD (15%). Further, those who knew of *Barrier-free Design* (10%) were more

than those who were familiar with *Inclusive Design* (9%). *Graphic design* students formed the majority of those who claimed *not to know* any of the concepts. Notwithstanding, the combined total who linked UD to its more inclusive synonyms (that is, *Design for All* and *Inclusive Design*) totalled 49% – which works out to about half of the total number of responses.



Figure 8.26: UD-related concepts that were most familiar

Of the UD-related concepts discussed above, the ones that were best understood followed more or less the same trend (Figure 8.27). *Design for Disability* (30%) is the best understood concept, whilst *Inclusive Design* (3%) is the least understood. *Industrial design* students form the significant majority of those who claim to understand UD, whilst virtually no *architectural technology students* claimed to understand either *Barrier-free Design* or *Inclusive Design*. As such, a course targeting this community of students should clearly articulate the links between UD and allied concepts to facilitate greater understanding and uptake of the former.



Figure 8.27: UD-related concepts that were best understood

In response to the question requesting informants to provide *a brief definition* (in their own words) of the concept they selected in the previous question, informants submitted a variety of definitions as sampled below:

- design for people with disabilities, to aid in society;
- recognisable design;
- keeping in mind all the different shapes, sizes, cultures, etc of people and designing something everyone can use and understand;
- designing intelligent solutions for the disabled; user-friendly designs for people with disabilities;
- designing for everyone even if they can't design;
- design that reaches a large audience, that is easily understood by all;
- design where all people are taken into account, but with a bias towards those who have difficulty using some object or facility, catering for them, but in such a way that the largest group of users can still use the facility, usually better than before;
- a set of 7 design guidelines which, if applied, helps to make products/surroundings easier to use for all – but mostly those with special needs. If it is easy for individuals with special needs it will make the use by regular individuals much easier;
- UD is 'good' well thought out design. UD comes naturally if the designer takes into account all the possible users and designed accordingly to suit their specific needs;
- UD is well, proportioned, and flexible design, that speaks to the user including all people from all backgrounds, and all abilities;
- design allowing for all variations in human physiology and anthropometry;
- design for use of a multitude of people, creating environments which can be used by more than the general public;
- designing a product which targets your specific target market to the best of your ability;
- UD is when objects go beyond specific areas of 'acceptance' i.e. something that is universal can fit or adapt to any related object. A universal designed object is inclusive;
- design that relates to all and includes as many people as possible;
- designing for all people, including the elderly and persons with disabilities;
- to design for disabled people to allow for comfort and freedom in their environment;
- designing in ways that fulfil the human need mentally, physically, etc;
- answering all human needs; design for all people, disabled or not;
- designing appropriate spaces for the individual and the environment;
- making things accessible for people with disability;
- taking wheelchairs, etc into account when designing entrances, spaces, etc;
- design that provides safety, comfort and practical solutions for disabled people;

- innovative design that recognises and meets the needs of those with disabilities and helps make their lives easier especially in those tasks we find easy to do, but is very difficult for them;
- Design for Disability would be designing spaces that are comfortable for people with disabilities as well as how they would use the space;
- accommodating for disabled people and enhancing their mobility;
- design that caters for a diverse group of people, not a specific race or culture;
- designing for the disabled using ergonomics, to make their quality of life better; and
- allocations and adaptations to improve design, so that it helps the disabled.

The vast majority of informants correlate UD-related concepts predominantly with designing for people with disability. A number of responses acknowledge the need to accommodate diversity and heterogeneity in the quest for inclusiveness. Arguably, disability is a strong recurring theme in the discourse on UD and would need to be incorporated as one of several key constituencies within the purview of UD. The aforementioned definitions are vital for accurate semantic and semiotic interpretative descriptors for Principles of UD and their corresponding pictograms (Figure 8.28).



Figure 8.28: Recognition of Principles of UD pictograms

(source: © Beth Tauke, 2001)

The pictograms that have the least ambiguous associations appear to be *Simple and Intuitive Use* and *Flexibility in Use*. The four most ambiguous pictograms appear to be those of *Equitable Use*, *Tolerance for Error*, *Perceptible Information* and *Size and Space for Approach and Use*. Generally speaking, the more explicit pictograms are easier to recognise as opposed to the more abstract ones. Further, one notes that a clearer correlation emerges between responses from *graphic design* students with the intention of the designer of the pictograms. This could be as a result of the affinity towards higher levels of abstraction and visual literacy associated with the discipline. Ultimately such symbols should be accessible to everyone. As Dalcher (2006:254) argues "design should [...] take into account the needs of the entire community of potential users and their varied levels of abilities to ensure that the results are widely available for the entire community". Consequently, further research needs to be done to evolve a clearer set of culturally-appropriate pictograms for the South African context. A detailed statistical analysis using SPSS[™] for comparison within and across disciplines is provided in Appendix P.

When asked whether they used UD extensively in their design process, just over half (51%) answered in the affirmative (Figure 8.29). The significant majority of those informants utilising UD were *industrial design* students (possibly as result of greater exposure from their lecturers). Further analysis reveals that there were more students of *graphic design* and *town and regional planning* who did not use UD than those who did. A special focus on UD-related applications for these two disciplines could potentially increase use of the concept.



Figure 8.29: Whether UD is used extensively in the informant's design process

Informants stated that the Principles of UD (elaborated in Appendix B) they most frequently used were: *Simple and Intuitive Use* (23%); *Flexibility in Use* (21%); *Size and Space for Approach and Use* (18%); *Perceptible Information* (13%); *Low Physical Effort* (9%); *Equitable Use* (8%); and *Tolerance for Error* (8%). As shown in Figure 30, *architectural technology, industrial design* and *interior design* students claimed to use the first three principles (in rank order) most frequently. *Equitable Use* was the principle most often used by *town and regional planning* students used. All seven Principles of UD were used in varying degrees of frequency by all groups of informants, except for Low Physical Effort which was not used at all by *town and regional planning* students.



Figure 8.30: Principles of UD that are most often used

When asked why they did not use Principles of UD in their work, the majority of informants (44%) claimed it was because they *had not heard of the Principles of UD before*. Responses from the majority of *graphic design* and *interior design* students fell into this category (see Figure 8.31). Equal numbers of *town and regional planning* students picked this response and that which stated that they *had to choose from a different range of prescribed principles*. 27% of informants stated that they *could use any principles they wanted to*; whilst 22% chose from *a different range of prescribed principles*. 4% responded that they *were not interested in applying such principles in their work*; whilst 3% indicated that *these principles did not apply to their work*. The implications from these responses is that a dedicated UD course should promote awareness of Principles of UD in general, whilst clearly articulating links between UD and similar principles that are

prescribed in discipline-specific domains. Additionally, focus groups as well as joint interdisciplinary seminars and workshops should engage students who are reluctant to embrace UD – the aim should be on effectively addressing concerns students gave for their initial reservations.



Figure 8.31: Reasons for non-use of Principles of UD

28% of informants stated that they were very likely to use information on UD should it be made available, whilst 48% responded that they would be *quite likely* to do so (see Figure 8.32). 15% were non-committal as they were *neither likely nor unlikely* to use information on UD if it were availed. 7% were *quite unlikely* to do so, whilst 2% responded that they were very unlikely to use information on UD. Of those that stated that they were *neither likely nor unlikely* to use the information, 52% were *graphic design* students. Consequently, a concerted effort should be made to demonstrate the relevance and efficacy of UD thinking within the practice of *graphic design*. The benefits of such a strategy could potentially include better designed (and more accessible) signage, symbols, pictograms and wayfinding systems, among others. Further, in adopting a cross-domain approach to UD, concilience through associations with *industrial design*, *human-computer interface design* and *interaction design* could potentially reinforce the value of UD in *graphic design* applications (Dalcher, 2006).



Figure 8.32: Likelihood of using information on UD should the same be availed

With regards to the preferred format for availing information on UD, the most popular modes include: *exemplars/samples*; *useful internet links*; *workshops/seminars*; and *training/short courses* (all with 9% of responses). Other significant options include: *printed reference guidelines*; *posters*; and *projects* (all with 8% of responses). The full range of responses is provided in Figure 8.33 below.



Figure 8.33: Preferred format for availing information on UD

Provision was made for formats not covered in the above question. Other preferred formats for availing information on UD suggested by informants include: *media coverage (radio and*

television) coverage; student magazines and *blogs; site/field visits; books;* and *advertisements in magazines.*

Informants identified *design educators* as the actors who would be the most effective champions of UD (43%). This response was consistently the highest for all disciplines except *interior design* (see Figure 8.34). *Individual designers* were proposed as champions by 28% of the informants (who did not include any from *town and regional planning*). 20% of the informants chose *team/project leaders*, whilst 9% proposed *political leaders* – interestingly, no *interior design* students opted for the latter. Section 8.4 discusses this issue further.



Figure 8.34: Actors perceived to be the most effective champions of UD

In response to the question on their *degree of satisfaction with current levels of accessibility within built environments, products, and/or systems used most often,* the majority of informants (43%) felt that they were *somewhat satisfied*; 28% felt that were *neither satisfied nor dissatisfied*; 21% felt *somewhat dissatisfied*; 5% were *very satisfied*; and 3% felt *very dissatisfied* with current levels of accessibility. No informants from *industrial design* reported any of the extremes (either *very satisfied* or *very dissatisfied*), whilst no *architectural technology* students reported to be *very satisfied* with current accessibility levels. A further breakdown is provided in Figure 8.35.



Figure 8.35: Degree of satisfaction with current levels of accessibility within built environments, products, and/or systems used most often

Regarding their *perception of the likelihood of UD as adding to the total cost of a design process/project*; the breakdown of responses in rank order is as follows: *quite likely* (39%); *neither likely nor unlikely* (35%); *quite unlikely* (17%); *very likely* (8%); and *very unlikely* (1%). *Architectural technology* students were the only ones who felt that UD would be *very unlikely* to increase the total cost of projects, whilst almost equal (but not significant) numbers from *all design disciplines* felt that incorporating UD was *very likely* to increase costs (Figure 8.36).



Figure 8.36: Perception of the likelihood of UD as adding to the total cost of a design process/project

A significant majority of informants (41%) reported that they would be *quite likely* to participate in a multi-disciplinary team on a UD project (see Figure 8.37). Equal numbers (25%) stated they would be *very likely*; or *neither likely nor unlikely* to participate. A less significant number (8%) reported that they were *quite unlikely*, whilst only 1% were *very unlikely* to participate. All the design groups were represented in the response categories except for those who responded that they were *very unlikely* to participate in a multi-disciplinary team on a UD project – only students of *town and regional planning* and *industrial design* gave responses in this category.



Figure 8.37: Likelihood of participation in a multi-disciplinary team on a UD project

An overwhelming majority of informants (84%) had positive perceptions about the *importance of designers to engage in socially responsible design to counter various forms of discrimination* (such as ageism, disablism, racism and sexism) – 63% felt that it was *very important*, whilst 21% felt that it was *somewhat important* for designers to do so. 10% of the informants were non-committal and felt that it was *neither important nor unimportant* for designers to engage in such responses, 4% felt it was *somewhat unimportant*, and 2% felt that it was *not important at all* for designers to engage in socially responsible design as a means to counter various forms of discrimination. The complete distribution of responses is shown in Figure 8.38.



Figure 8.38: Perception on the importance of designers engaging in socially responsible design to counter various forms of discrimination

The majority of informants (63%) felt confident (or sure) that they would be practicing design in South Africa beyond the FIFA 2010 World Cup^{TM} – with 33% being somewhat confident/sure and 30% being very confident/sure (see Figure 8.39). 20% were neither confident nor sure; 10% reported being somewhat unsure, whilst 7% were very unsure of their professional prospects.



Figure 8.39: Degree of confidence that informant will be practicing design in South Africa beyond the 2010 FIFA World Cup™

The question on the usefulness of *ubuntu* to design practise in South Africa produced interesting results (see Figure 8.40). The majority (52%) were felt that *ubuntu* was useful – with 28% *slightly agreeing* and 24% *strongly agreeing*. A significant number (15%) stated that they *did not know*. 13% *disagreed* on the usefulness of *ubuntu* – with 8% *slightly disagreeing* and 5% *strongly disagreeing*; whilst 20% were non-committal as they *neither agreed nor disagreed*. The strategic implication is that the philosophy of *ubuntu* would need to be clearly and robustly discoursed for greater appreciation of its potential usefulness to various design disciplines.



Figure 8.40: Degree of agreement that ubuntu is useful to design practice in South Africa

The question on the *likelihood of informants increasing adoption of UD if the value of UD was showcased in 2010 projects* elicited encouraging responses (see Figure 8.41). 69% of the responses were positive – with 48% replying that they were *quite likely*, whilst 21% were *very likely* to adopt UD to a greater degree. Notwithstanding, a significant number (26%) responded that they were undecided – *neither likely nor unlikely*; 5% were *quite unlikely*; and no one reported being *very unlikely* to adopt UD if its value was showcased in 2010-related projects.



Figure 8.41: Likelihood of greater adoption of UD should its value be showcased in 2010 projects

The majority of responses (63%) indicate a likelihood of compliance in implementing UD in projects should the law mandate the same (see Figure 8.42). 45% of informants stated they would be *quite likely* to comply, whilst an additional 18% would be *very likely* to comply with such mandatory legislation. A significant number (26%) were *neither likely nor unlikely* to comply; 9% would be *quite unlikely*; and 2% were *very unlikely* to comply with such a law. Of these groups, no *graphic design* or *industrial design* students expressed strong objections by indicating that they would be *very unlikely* to comply.



Figure 8.42: Likelihood of compliance should implementation of UD in projects be mandated by law

There was general agreement (70%) that UD courses should be integrated into all designrelated disciplines at university level. 36% of informants slightly agreed, whilst 34% strongly agreed with this suggestion (see Figure 8.43). 14% neither agreed nor disagreed; 10% slightly agreed; 1% strongly disagreed with this proposition. A small number (5%) stated that they did not know whether or not UD courses should be integrated into all design-related disciplines at university level. Only graphic design students expressed strong disagreement; whilst there were no informants from architectural technology and industrial design who claimed that they did not know whether UD courses should be integrated into such disciplines. Subsequently, concilience should inform the development of a multi-disciplinary and cross-domain UD course that is available (at least as an elective) to all design students at university level (Dalcher, 2006).



Figure 8.43: Degree of agreement that UD courses should be integrated into *all* design-related disciplines at university level

Responses to the question on the *likelihood of implementing UD in projects if incentives were available to reward designers* (shown in Figure 8.44) are comparable to a similar question on whether mandatory compliance would promote implementation of UD (shown in the previous page in Figure 8.42). A markedly higher number of informants (77%) were favourably disposed towards voluntary implementation – with 44% being *quite likely*, and 33% stating that they would be *very likely* to implement UD in projects. 18% were *neither likely nor unlikely* to implement UD in projects; 3% said they were *quite unlikely*; and 2% reported that they were *very unlikely* to do so. Only *architectural technology* and *town and regional planning* students indicated that they would be *very unlikely* to use UD in projects if incentives were available to reward designers.



Figure 8.44: Likelihood of implementing UD in projects if incentives were available to reward designers

Informants also made the following (self-explanatory) suggestions of possible incentives for rewarding UD-compliant designers:

- bursaries and scholarships to study UD locally and abroad;
- prizes, internships, mentorship, recognition from industry;
- cash prizes, contracts, rewards, vouchers, money, recognition, software;
- higher marks; opportunities for jobs; travel opportunities;
- media coverage and more public visibility of good design
- recognition of UD products in design;
- bypassing [Broad-based Black Economic Empowerment] BBBEE requirements, civil projects awarded to UD companies;
- *tax rebates to UD-compliant firms;*
- tickets to attend UD-related expos;
- certification, endorsements from reputable organisations;
- international competitions and awards;
- global recognition for designers to showcase their use of UD; and
- public humiliation for designer if not compliant with UD.

Further, informants offered some additional comments when invited to do so on UD-related concerns:

- UD must become an essential part of Industrial design;
- Ergonomics is an extremely interesting subject, as well as important and relevant. It should be a main subject in the 3D design course;
- Should be implemented and taught academically;
- UD is a great tool, but cannot be used as the only tool;
- I would like to know more about UD but I am concerned it might limit the designer creatively;
- more discussions, brainstorming and practice sessions on UD for design students;
- interesting way of trying to understand the way design influences us;
- engaging survey UD sounds very interesting. Should have been a subject as part of my course;
- it is a great initiative done in letting us know about design solutions; and
- UD would have enriched my work process.

The aforementioned suggestions and comments would be factored in when formulating responsive strategies that seek to promote the adoption and mainstreaming of UD within academic and professional design domains. The overall impression is that UD is an effective (and more importantly in this context, desirable) tool that would suitable equip designers to respond to the diverse needs presented by various categories of vulnerable populations.

8.3 Implications for Principal Actors

The following section analyses responses to specific questions posed to various principal actors. The said responses are reproduced verbatim in some instances for clarity and authenticity.

8.3.1 Implications for the City of Cape Town

The interviews with city planners sought to establish their perceptions with regards to 2010related projects. There was general agreement on *the value of enlisting the participation of student in 2010 projects*. Pezeshki (2008) felt that the greatest benefit accruing from involving students lay in harnessing their creativity – "they contribute their ideas, their visions to the city and the planning team". Joustra (2008) concurs by adding that:

It is important also for students to be aware of what is happening around them and be open to comment or criticism on the developments around them applying their knowledge and ideas of this developing UD approach. I would advise students to get involved where they can through public participation processes.

Pezeshki (2008) and Joustra (2008) both agree on the need to involve design educators in the planning and execution of projects within the city. Joustra (*ibid*) places the onus on educators to be more proactive in engaging with city planners and puts it thus:

It would be useful for educators to be able to have access to departments in the City and relevant persons that are aware of the processes and progress being made in the City in mainstreaming UD. Therefore educators can inform their students of everyday challenges and opportunities relevant to their studies and their city the most probable their workplace to be in the future.

The design disciplines most likely to be co-opted to work with city planners include "urban planners, architects, transport planners, landscape architects, and fire consultants" (Griffith, 2008). There was also consensus on the perceived *value of competitions as effective platforms for promoting new design concepts or ideas*. Key elements identified for making competitions more effective in promoting novel concepts include: "multi-disciplinarity" (Griffith, 2008); "broad-based appeal and cross-disciplinarity" (Pezeshki, 2008); and "multi-lateral, awareness raising [potential]" (Joustra, 2008). All informants agreed that the 2010-related projects would leave a lasting legacy to the residents of Cape Town. According to Clive Griffith (2008) the principal infrastructural legacy would be "a world class multi-purpose stadium with an urban park, sporting facilities, eco-centre and golf course for relaxation and entertainment".

Some of the informants stated that they were not familiar with the concept of UD, whilst Griffith (2008), the City's *project architect* responsible for the design and construction of the new Greenpoint stadium, stated that he was familiar with the concept "but not named as such". Griffith (*ibid*) came to know of the UD-related concept through his association with "a deaf and dumb *architect* in Zimbabwe who had his own practice and was obviously sensitive to this subject". Griffith (*ibid*) felt that 2010 project architects had been sensitive to UD in their work though he felt that UD was "not at present" relevant to any of the projects or functions he was personally involved in.

With respect to Principles of UD (elaborated in Appendix B), *Flexibility of Use* and *Low Physical Effort* were the two that were most relevant to the 2010-related projects (*ibid*). The self-same UD principles were the two that Griffith (*ibid*) felt were most neglected in terms of implementation in projects at present. Griffith (*ibid*) affirmed his personal commitment to continue utilising UD, and further suggests that the best way to mainstream UD within his specific department (as well as to the general public) was by making people "aware of such issues through *education*".

According to Griffith (*ibid*) the *Council of Architects* was proposed as a suitable champion, and as an effective means for promoting UD as the media had not been effective thus far in promoting UD nationally. Within the context of Cape Town, the *Institute of Architects* was identified as an ideal champion for UD (*ibid*). The main actors with regards to UD should include "designers such as industrial, architects – urban and landscape, engineers, [and related] disciplines" (*ibid*). The government policies that would best align with (and accommodate) UD would be the *National Building Regulations* and *environmental policies* (*ibid*). Others include the *Batho Pele* principles; the new *Consumer Protection Act*; the *Information Act*; and the *Promotion of Equality and Prevention of Unfair Discrimination Act*. The role of official policy formulation and implementation within the city of Cape Town is critical to effective mainstreaming of UD in the city.

In summary, the following strategic implications apply to city planners and policy formulators in the context of the city of Cape Town:

- formulate comprehensive integrated UD/Universal Access (UA)-specific policies that articulate across various operational departments and other administrative units;
- create a community-of-practice with other key actors as well as strengthen collaborations with other *Triple Helix* partners in academia and industry/civil society to build capacity around applied research and project development;

- support (and possibly co-sponsor) multi-disciplinary, cross-domain competitions that are problem-based and responsive to real-life challenges through *public-private* initiatives that showcase the value of integrated UD strategies to all residents;
- engage residents and other key actors through a *consensus participatory process* on UDrelated issues potentially affecting their welfare or livelihood within Cape Town;
- highlight and recognise UD-compliance through popular media and official publications;
- invite the involvement of professional bodies that represent designers, architects, town planners, and engineers as key co-actors in project oversight and implementation;
- identify and support internal champions to fast-track the adoption and diffusion of UD; and
- link proposed UD strategies to local, regional and national policies, acts, and other legislative imperatives.

8.3.2 Implications for Professional Designers

The responses from professional designers who were interviewed are based on questions provided in Appendix M. A number of informants reported that they were specifically involved in 2010-related projects. Therina Wetzel (2008), the Manager for Resource Mobilization with the South African National Council for Persons with Physical Disabilities (NCPPD) reported that her organisation was an accessibility consultant in a number of 2010-related projects including stadia, the Gautrain heavy rail system in Gauteng Province, as well as a number of bus transit systems nationally. Bernard Smith (2008) of the Industrial Designers Association South Africa (IdeaSA) stated that he participated in meetings with the Department of Science and Technology to explore the potential contribution of *industrial design* to 2010-related projects, whilst Mel Hagen (2008), a design activist, educator and researcher, reported that she had "conduct[ed] consumer research for the provincial government of the Western Cape with a view to developing a craft development strategy, using 2010 as a catalyst for development beyond 2010". The importance of anticipating the impact of such activities from 2011 onwards is supported by other commentators (du Plessis & Maennig, 2009; Swart et al., 2009; Tomlinson, 2009; Tomlinson et al., 2009). Craig Thompson (2008) of the IDC Consultants reported that their firm had "approached the LOC and had some discussions on mainstreaming universal access and UD in a number of projects as opposed to [merely] incorporating industrial design or signage design".

All the informants said they would welcome the involvement of design students and independent professional designers in such projects. The professional designers felt that they would most likely co-opt the following disciplines/professionals in 2010-related projects: *accessibility auditors*,

engineers, architects, and designers (Wetzel, 2008); industrial designers; engineers and graphic designers (Smith, 2008); product/industrial designers, graphic designers, and surface designers (Hagen, 2008); industrial designers, IT [information technology] professionals, public policy makers, and educators (C.Thompson, 2008). Product/industrial design appears to be a key discipline as it is the one most frequently cited by the informants – the reason given for selecting industrial design lies in its capacity the cross-disciplinary collaboration with a diverse number of other profession(al)s.

When asked whether they considered competitions as effective platforms for promoting new design concepts, the general consensus amongst informants was that competitions were indeed a useful and effective tool. C.Thompson (2008) felt that UD-specific competitions should focus on real-life challenges as is the case in Denmark, and the USA. Hagen (2008) concurred but cautioned against the potential of competitions to sometimes be narrowly-defined and rather prescriptive. These sentiments echo those of Biggs (2002) who emphasises *problem-based* (as opposed to *solution-based*) *learning*, as well as Kruger and Cross (2006) who support *problem-driven* (and not *solution-driven*) strategies. Hagen (2008) argues thus:

Frequently competitions are launched by organisations that have their own agendas that might work against the development of totally new design concepts. However, they may provide an opportunity for engaging with real-life, implementable problems which are aspects that is often lacking in the academic environment. I personally believe that students need to be much more exposed to the real world, and engaging with major problems that beset our society, rather than producing the n^{th} generation of an existing project. Development of new concepts is much more likely when the problem is framed in a way which precludes approaching it as a redesign of an existing product. I like [the Integral Design] method of framing problems in a way that does not predetermine the product, e.g. students being asked to provide a body support system rather than a chair this immediately opens the way for looking at all sorts of different solutions. Or providing students with a specific problem, e.g. the reality of most shack dwellers is that they are situated on the Cape Flats where the water table is just beneath the surface. [...] What can be done to ameliorate this particular problem – where can design play a role? [...] What I am trying to say is that most competitions are not framed in this open way, and thus do not necessarily invite new concepts/solutions. What they tend to do is expand on existing products and often this merely involves restyling. I also believe that most competitions are framed around a narrow design area which does not necessarily encourage students to engage in methodologies that exist in the real world, i.e. multidisciplinary teams, etc.

The key elements identified for making competitions more effective in promoting novel concepts include: *presentations on real projects and not just ideas* (Wetzel, 2008); *well-defined briefs, good organisation and good incentive prizes* (Smith, 2008); *competitions that specifically explore the usage of IT with regard to interior environments* (C.Thompson, 2008); *a UD-specific focus* (Mulder, 2008); and an emphasis on *real-life, implementable problems* (Hagen, 2008).

The legacy that informants felt the FIFA 2010 World Cup[™] would leave for residents (and visitors) in Cape Town once the mega-event was over include the following:

- Transport and some other facilities will be accessible to all (Wetzel, 2008);
- The 2010 WC should leave a legacy that South Africa and Africa have a strong identity of who we are as a nation and continent, a legacy that should convey confidence and a sense of determination for hope and prosperity for all the people of Africa (Smith, 2008);
- The event certainly provides a strong focus and catalyst. [..] A major impact has been that the World Cup provides a focus which pushes us into developing vital infrastructure which would normally take its course depending on availability of funds. In particular, in Cape Town, the development of the urban park around the stadium complex, improvements in road systems, and the phasing in of a public rapid transport system, the development of public open spaces (the Grand Parade and the station deck). And if Universal Design principles can be employed in all infrastructure for the World Cup, then we could have a major showcase for how design can provide accessibility and inclusivity to all useful for advocacy. The other plus of the World Cup is the opportunity to present South Africa to the world. This could have major impacts on not only tourism (which is the immediate beneficiary) but on the world developing a positive outlook about South Africa's infrastructure, economy, etc (Hagen, 2008); and
- My impression of the 2010 World Cup is that ideally that it be used a platform for the civil society to engage in. To engage with government and with stake-holders. To try and raise a lot of awareness and do advocacy work. To use the larger-scale products like the stadiums and to integrate the World Cup with the tourism grading process that's being run. To try and use this as a catalyst to try and engage with hotels and tourism venues and with DEAT [Department of Environmental Affairs and Tourism] as well. To try and make their facilities, their services and their tourist products as successful as possible (C.Thompson, 2008).

When asked if they were familiar with UD, all the informants responded in the affirmative. C. Thompson (2008) elaborates thus:

Yes [we] have also been involved in the Saudi [Arabian] government, specifically the Prince Salman Center in Riyadh. What they are busy doing at the moment is they are actually busy framing legislation policies to implement UD across the board. In tourism,

transportation, communication, the built environment, and that is a program they are running over the next 5 or 6 years. The aim of which is to make Saudi Arabia completely accessible.

In response to the question on how they first came to be involved with UD, informants made the following statements:

- Working in the disability sector for a long time [as a disability activist for over 20 years]. This is what we lobby for. [Learnt about UD from people like Philip Thompson (architect), Joan Seirlis (architect), and David Boonzaier (medical doctor) some ten years ago] (Wetzel, 2008);
- As an industrial designer having worked on hundreds of commercial design projects, I have become of UD when considering the use and the 'user' of the product to be designed – this may be as simple as 'left or right-hand use' to more complex issues around disabilities. The work of Henry Dreyfuss on ergonomics and Leonardo da Vinci's "Vitruvian Man' have influenced my perception towards UD, specifically concerning the relation between the human form (abled and disabled) and the environments/products we create to live in. Technology has also influenced my thinking towards UD. During my employment at CSIR in 2005 I was involved in a collaborative research project concerning UD with the Meraka Institute and Built Environment (Smith, 2008);
- I have been a strong champion of design in all its forms over a number of years. And while I was aware of design for those who are challenged in some way, the concept of Universal Design, which is an inclusive concept, was strongly brought to my attention a number of years ago by Dr David Boonzaier, who is a strong proponent. So I have read some of the literature on this subject and have been interested in particular design projects, e.g. the Royal College of Art's programme for 'Designing for the Third Age' (Hagen, 2008);
- Through projects and research on the topic (C. Thompson, 2008); and
- It goes back to... it would have been about 1985 or 1986. That was the first time I started reading up on it. Interestingly enough as I was associated with some work we were doing on education; what we then called 'special skills' in the Transkei region of the Eastern Cape. The interesting part of that was that it was before I had a developed a spinal injury. So that was my first contact with UD (Philip Thompson, 2008).

As for the reasons why UD had not found wider acceptance within their spheres of influence, informants offered the following reasons:

- Ignorance and politics (Wetzel, 2008);
- For the reason that the world was developed mainly 'by abled persons for abled persons' (Smith, 2008);

- I do not believe that there has been sufficient advocacy out there. Most people understand the notion of designing for the disabled, but the concept of providing design solutions that would meet most if not all physical/kinetic constraints is not something they have ever thought of. Currently certain aspects of designing for the disabled are being dealt with in the architectural profession, but the way they are being implemented shows a clear misunderstanding of the importance to all. (Hagen, 2008); and
- My personal impression on the matter is that there is just a lack of political will. You know there just doesn't seem to be enough buy-in at the highest level. There also seems to be a lack of understanding. Most people's interpretation of UD is that it is "designing for paraplegics", which is quite incorrect. I think that tends to be a big problem, that there isn't any understanding that it relates to a whole series of user groups and user requirements (C.Thompson, 2008).

Informants reported that UD was relevant to some of the projects or functions they were currently involved in. The Principles of UD (in rank order) found to be most relevant to the projects the informants were involved in include:

- Low Physical Effort, and Size and Space for Approach and Use (Wetzel, 2008);
- Size and Space and Approach and Use; Perceptible Information; and Flexibility in Use (Smith, 2008);
- Flexibility of Use, Size and Space for Approach and Use, Low Physical Effort, and Perceptible Information (Hagen, 2008); and
- Size and Space for Approach and Use, Perceptible Information, and Low Physical Effort (P. Thompson, 2008).

The most relevant Principles of UD (elaborated in Appendix B) from the above responses are: *Size and Space and Approach and Use, Low Physical Effort, Perceptible Information,* and *Flexibility in Use.* The aspects of UD considered most neglected include the fact that UD is treated as an *add-on* (as opposed to being an *integral part*) of the design process; lack of incorporation into mobility provisions as well as *Information and Communication Technology* (ICT); lack of sensitivity towards the needs of people with visual disabilities; and the ill-consideration of certain Principles of UD (such as *Perceptible Information,* and *Tolerance for Error*). *Education* and *awareness promotion* were the main strategies suggested for the mainstreaming of UD in the informants work(places) – all the informants acknowledged the fact that very little promotion of UD was evident presently.

Informants proposed the following as the main actors with respect to UD: designers, developers, architects, engineers, human, social, physical sciences, and government officials (at local,

regional and national levels). The local, regional, and national government departments that were proposed as being best placed to accommodate UD include: Departments of *Science and Technology, Transport, Public Works, Labour, Tourism and Economic Development*; whereas the *Council of Architects, Council of Engineers*, and *professional associations for designers,* were the professional bodies (in the private or public sectors) considered to be the ideal champions for UD. Wetzel (2008) proposed *Jim Stanburry* (a civil engineer) as a possible champion for UD in Cape Town. Smith (2008) on the other hand advanced the names of two popular Paralympics athletes (*Natalie du Toit* and/or *Oscar Pistorius*) and justified his choice by arguing that "UD is about equality both in theory and practice and would require a champion that has displayed exceptional perseverance living in a 'disabled' world" – the suitability of these two candidates/nominees is interrogated in greater detail in Section 8.4.

When asked what strategies they felt would mainstream UD at local, regional and national levels in an effective and sustainable manner, informants suggested engaging *Triple Helix* partners in joint efforts and initiatives; linking national strategies to measurable outcomes; and creating clear, comprehensive strategic plans involving all key actors at various levels (local, regional and national). In response to the question on whether they intended to promote the use of UD principles in future, informants were unanimous in stating that they would do so. Further, when asked how they would do so specifically, informants proposed the use of exhibitions, workshops, seminars, conferences, applied research and consultancy as effective tools. Two further comments were offered by informants:

- Good to see that somebody is taking us seriously (Wetzel, 2008); and
- UD is an extremely relevant subject to pursue, particularly in Africa where there is a great need for UD but due to a 'third-world' mentality, the subject remains largely unknown, left for more developed nations to research due to Africa having more serious developmental problems. This scenario should be reversed where UD could be a means for development. [That is] if Africa could use the unique challenges to research and develop UD, then this could not only benefit the continent but be exported to other developing countries where similar development and social challenges occur such as poverty – in countries such as India, China and Brazil (Smith, 2008).

In summary, responses from professional designers inform the following strategic imperatives:

 ensure the inclusion and participation of professional bodies representing designers, architects and engineers in any strategic planning around UD-related strategies;

- support the development of SAQA-aligned credit-bearing *continuing professional development* training programmes on UD-related topics for practicing professionals;
- encourage collaborative work through multi-disciplinary, cross-domain teamwork in UDrelated project planning and implementation;
- convene regular symposia and related public *consultative fora* (such as exhibitions/expos, workshops, seminars and conferences) to sustain robust debate and dialogue on UD;
- identify support mechanisms (such as *tax incentives*, as well as peer-recognised *awards* and *prizes*) to promote the adoption of UD the *Design Excellence Awards* programme run by the SABS Design Institute serves as a good reference for a possible model;
- identify and support suitable *champions* (particularly those who enjoy broad public appeal);
- maintain accessible and expandable databases of UD-compliant professionals and projects – this could be done in collaboration with other *Triple Helix* partners to ensure greater inclusiveness and reach;
- engage in *socially responsible design* projects that are *participatory* in approach to demonstrate the value of UD to civil society/the general public;
- participate in and support competitions that promote the *sustainable* diffusion of novel concepts such as UD in *majority world/industrially developing* contexts;
- proactively incorporate and promote the adoption of Principles of UD within their own communities-of-practice by benchmarking international best practices; and
- provide the popular media and press with UD-compliant exemplars for dissemination to the general public.

8.3.3 Implications for Design Educators

Design educators from the Departments of *Architectural Technology*, *Design History and Theory*, *Fashion and Surface Design*; *Graphic Design*, *Interior Design*, and *Town and Regional Planning* (all based at the Cape Town Campus, CPUT) answered semi-structured questions listed in Appendix N. *Industrial Design* educators were not interviewed – they were already cognisant of, and committed to research, teaching and learning support with regards to UD.

In response to the question of who (students and/or staff) were carrying out 2010-related projects, Andrea Broom (2008) and Bruce Snaddon (2008) both *graphic design* educators reported that students were involved. Roelof Oelofsen (2008), the Head of the Department of *Interior Design* also reported that two lecturers and the entire class of third year students were

involved in the students' landscaping competition (discussed in Section 5.2.3.3). Alettia Chisin (2008) of the *Department of Fashion and Surface Design* also identified one of their students involved in a project to design furnishing and lighting for the VIP lounge at the Greenpoint Stadium. Chisin (*ibid*) however clarified that the said project was "not an official collaboration with FIFA". When asked to describe the projects that were carried out in their departments, informants stated the following:

- With Cape Town Tourism, I've got a 4th year (BTech Graphic Design) who was very interested in looking at tourism after 2010, and how he could capitalise on the 2010 World Cup to communicate to that particular target audience. He has done interviews with Cape Town Tourism and they were very interested to see his project. He conducted readability research and wrote a report in collaboration with [the Department of] Sport Management. So far he has come up with a loyalty card concept, which has been well accepted. We had it moderated by industry. [...] So he is now working on how to activate that card, which is a huge incentive for overseas visitors to return, and it looks like Cape Town Tourism is quite interested in the concept. Even though they already have a loyalty card, apparently, we did discover, but it is not along the same principles. It is not 2010-related, I think that's the point of difference, it's actually for soccer fanatics and introducing them to the city and the region (Broom, 2008);
- The project took the form of a competition which involved the upgrading of the Mouille Point beach promenade area as part of the new 2010 sport stadium development in Greenpoint. Students submitted design proposals for elements such as bus shelters, seating, paving, site maintenance sheds, balustrading along the seafront, etc (Oelofsen, 2008); and
- [The student] focused on the essence of Swedish and African (particularly Zambian) design, to
 originate a visual design language which will find expression in surface designs for upholstery,
 furnishing and lighting. It is envisaged that this design language will render the particular surface
 applications accessible to the international and local soccer supporters making use of the VIP
 lounges (Chisin, 2008).

When asked if they would support the participation of their students in 2010 projects, design educators replied as follows:

- In principle, yes, depending on the interior design content of the specific project (Oelofsen, 2008);
- Absolutely. Obviously 2010 can be seen as a huge opportunity to use design, and communication design, to do something good for the city. Not just a commercial venture. I would try to keep it out of the commercial world and make it more for social projects (Broom, 2008); and
- Yes. The type of exposure that they will gain in designing/participating for and in such a highprofile event is invaluable. The participation must be properly structured to ensure maximum student and CPUT benefit (Chisin, 2008).

Informants were unanimous that they would support the involvement of students from other departments in such projects. Broom (2008) felt that "collaboration with other disciplines is probably the best way [to go] about projects of this nature". Jake de Villiers (2008), Head of the Department of *Architectural Technology* included "other departments of the *Faculty of Informatics and Design*", CPUT. Johann J. van der Merwe (2008), Head of Department, *Design History and Theory* selected "graphic design and informatics"; whilst Nigel Tapela (2008), Head of Department, *Town and Regional Planning* identified "graphic, industrial design, photography, civil engineering, architecture, and art". Oelofsen (2008) picked "Industrial Design and Architecture"; while Broom (2008) responded that "the first one is usually *Industrial Design*. Then also *Interior,* definitely *Fashion*, we often work with *Surface Design* – even *Jewellery*. I think *Communication Design* actually works with all of them".

Informants offered the following elaborations for selecting the aforementioned disciplines:

- The fits between Industrial and Graphic [design] are the most obvious, whereas graphic, or communication design, works for some of the other disciplines in more of a communication of selling their design, it complements. With industrial [design] there is a definite crossover (Broom, 2008);
- Whereas Graphic Designers might very well produce things, like packaging, as part of the communication needs. But that is only one aspect of what it is that we do. So we interface with Industrial Design, Interior, Surface, Fashion, if and when the need arises driven by the communication needs (Snaddon, 2008);
- A certain synergy exists between Interior Design and the above mentioned disciplines, inasmuch as Interior Design can be seen as the interface between buildings and occupants, and furniture design is a shared activity between the three (Oelofsen, 2008);
- The current proposal to initiate a project around Cape Town as a legible city could be an example. This would entail the involvement of a range of design disciplines (de Villiers, 2008);
- We have had successful collaborations in the past. We have built up a good understanding with Graphic Design and Industrial Design with the inter-departmental crits, and I believe that Interior Design and Surface Design are well-positioned to complement each other in projects (Chisin, 2008);
- Internationally these three departments (industrial, graphic and informatics) are being seen as forming a "natural" combination that can provide solutions to what are hybrid problem situations (J.J. van der Merwe, 2008); and
- Collaboration in project conception, design, development and packaging would benefit from interdisciplinary team for impact (Tapela, 2008).
When asked whether they considered competitions to be effective platforms for promoting new design concepts/ideas, the responses were varied. Broom (2008), Snaddon (2008), de Villiers (2008), and Tapela (2008) all felt that competitions were an effective platform for such purposes. Oelofsen (2008) and Chisin (2008) were more cautious in observing that the value of competitions could not be generalised as they were outcome-dependent. J.J. van der Merwe (2008) was not convinced of their value and replied that "competitions have their place, but *problem-based learning* (learning about and with the real world) is much wider than any competition" – this need to link learning activities to real-life problems is an approach endorsed by other educational commentators (Biggs, 2002; Kruger *et al.*, 2006; Hagen, 2008).

In response to the question on how competitions could be made more effective in promoting novel concepts or ideas, informants said the following:

- I must say that we've found with anything that has a sustainable focus, our students are getting to be very aware of sustainability issues and it's getting more and more intense every year, but naturally I find that Graphic Design students are very aware of their role. So anything that will motivate them will be the fact that they are making a difference. I find it so interesting, it used to be money and prestige, but now (also with the community-based projects that we have) we see that the internal motivation is very much about making a difference (Broom, 2008);
- I think when it comes to a competition brief you've got to be absolutely up-front with what the agenda is, and if it is UD then that's fine. The way we operate as communication designers is we use the brief as the starting point, the framework; it's the well from which we draw everything. [...] ... because when you work for a client, your client is going to give you the parameters and our job as the designer is to actually further develop those liberating constraints. The concept of 'liberating constraints' is a nice one, even though it sounds contradictory, I think in terms of how competitions can be more effective in promoting new concepts and ideas inherent in the nature of the work that we do is the necessity for a brief to be accurate and to be inclusive of every aspect of what it is that you are trying to achieve. [...] Another thing is the recognition of the power that a designer wields, especially in the world of communication design, and that people need to be responsible with that power (Snaddon, 2008);
- *Promotion and publicity in a range of both technical and popular media will be a start* (Oelofsen, 2008);
- Wide publicity, "tightly" written briefs, effective workshops for competitors in which the organisers/instigators disseminate relative guidelines information and advice and attractive prizes (de Villiers, 2008);
- The media coverage that the competition generates is important. Consistency of the competition and a national format helps, such as the Mr Price and "No Kak Environmental" competitions.

However, comparing those with the "New Signatures Art competition" is problematic since the latter has status in the academic/cultural sphere while the former, as a competitions focusing on consumer products, lack that status. This is changing though...The value of design as a catalyst for change and a facilitator for system and social design should be promoted more aggressively in competitions (Chisin, 2008);

- Use [competitions] as mere platforms/springboards for design learning, by allowing students to break out of the rules and guidelines of the sponsor of the competition. A straightforward competition is just another way of making students follow his master's voice (J.J. van der Merwe, 2008); and
- Competitions tend to release energies that invoke innovation (Tapela, 2008).

Based on the above responses, the design educators should be involved in the conceptualisation, implementation and adjudication of competitions to enhance the effectiveness of competitions in the promotion of new concepts and ideas such as UD. With specific reference to UD, the majority of informants stated that they were familiar with the concept of UD and wished to interrogate the subject further. About half of the total number of informants had taught UD to some of their students – representing the Departments of *Architectural Technology*, *Design Theory and History, Interior Design*, and *Town and Regional Planning*. J.J. van der Merwe (2008) stated that "UD has formed a part of [the Department of Design History and Theory's] first module for years". Of those who offered UD to students, the majority offered the concept in an unstructured way – that is, not as a dedicated module or course, but as an integral part of other design, technology and theory-related course offerings. Those who had not yet begun teaching their students about UD (as understood and elaborated in this thesis) were from the Departments of *Graphic Design*, and *Fashion and Surface Design*. Informants from the Department of Graphic Design said the following:

- Not formally no. Not as a genre or area... (Broom, 2008); and
- No, though I feel that one maybe needs to qualify this and say that UD in that terminology structure has not been taught. I do believe that we do teach UD, it just has not necessarily been called UD (Snaddon, 2008).

When asked whether UD was relevant to any of the courses or modules they currently offered in their programmes of study, all informants replied in the affirmative. Some of their responses are sampled below:

 In short, yes – Communication Design, History and Contemporary Design studies offered from levels 1-4 (Snaddon, 2008);

- Yes relates to the Design component (Oelofsen, 2008);
- Yes, particularly the module "Language, Culture and Difference" that I do with the BTechs, since UD speaks to aspects of difference, culture and language (Chisin, 2008);
- Yes the relevance to [Design] Theory is to make students aware of the wider focus they need to take when dealing with potential users of theory designs (J.J. van der Merwe, 2008); and
- Yes most of our planning design courses involve UD principles both in terms as understanding human behaviour and use of space (Tapela, 2008).

The Principles of UD (elaborated in Appendix B) that informants found most relevant to their disciplines included the following:

- Mostly Simple and Intuitive Use, and Perceptible Information (Broom, 2008; Snaddon, 2008);
- Mostly Equitable Use, Simple and Intuitive Use, and Perceptible Information in public spaces; Low Physical Effort – especially [with reference to] children and the aged; and most importantly Size and Space for Approach and Use (Oelofsen, 2008);
- All the seven Principles of UD (de Villiers, 2008);
- Flexibility in Use, Size and Space for Approach and Use, and Tolerance for Error, but very much dependent on the curriculum and the students (Chisin, 2008);
- All these Principles are necessary for theory (as a way of seeing the world) the combined aspects of UD serve as a type of design thinking constraint – which fits in very well with Donald Norman's rules of [1] designing with constraints and [2] make use of both knowledge in the head and knowledge in the world (J.J. van der Merwe, 2008); and
- All 7 Principles are important in human perception and use of space, oversight on one Principle impacts on overall design functionality, usability and aesthetic (Tapela, 2008).

The comments on what aspects of UD that informants wished to incorporate in future are sampled below:

- I think we'll appropriate all. We don't go to a category of user, we go to a niche within the category (Broom, 2008);
- Now see, 'Equitable Use' is the major problem, because a lot of the time we're actually having to niche-market. I think it really acknowledges difference. You know, our approach, the old analogy is the shotgun shot versus the marksman shot. The shotgun blast is more 'universal' and whatever you're going to do is going to work for everybody, but in marketing terms, in advertising terms, in budgetary terms you can't do this. You know the flyer that you get through your mailbox is the shotgun blast and how often does that end up missing the mark. Whereas something that is more marksmanship, aimed at you, you might even get something with your name on it. Now they've managed to get that information from somewhere, from some database. Then you are more likely

to read that, and they know, so that is a critical factor. [...] When you think of Coca Cola – if you were to hand a blind person a Coke bottle they would know exactly what it is if they hold it, even though they can't see the logo. That logo would never transform into Braille. I think that maybe cuts quite nicely to the issue. When we teach 'Corporate Identity', 'Logo Design' is a facet of 'Corporate Identity' [which in turn] is a facet of 'Branding'. So corporate identity (the way we teach it) is very holistically [done] – beyond just a logo. Branding is the most universal of those three categories. In simple terms a logo is a trigger. I like to use words that are more universal, a trigger, because graphic design also has to be a visual trigger. It can most certainly be a trigger in many other ways. There's smell, there's tactile, sound, there's tone of voice. These are all triggers that can relate a brand to a person who might be sight-impaired or hearing-impaired or whatever. Or just of a culturally different calling (Snaddon, 2008);

- It is important for design students to be familiar with all aspects of UD (Chisin, 2008); and
- ALL, but would like to develop a more nuanced analytical/interpretative model that explores intersections between these (Tapela, 2008).

The responses from Broom (2008) and Snaddon (2008) inspire a potentially new area of multidisciplinary and cross-domain exploration for UD with respect to two- and three-dimensional interfaces. The possibility of developing *tactile branding* would increase the accessibility and usability of such branded products to categories of end-users/consumers who are typically unable to interact with brands in their present forms.

Informants offered some suggestions on how UD could be mainstreamed into their respective/specific disciplines:

- UD should be integrated into every single brief. I don't think it should be separated. Communication design has to communicate on so many different levels that you couldn't segregate it to study it separately as a specialisation. I think they do deal with psychographics and demographics. Sometimes discoveries that you make when you're aiming at a general or normal target audience are wonderful and they actually are duplicated universally – this can specifically apply to UD (Broom, 2008);
- If you think about it fits really well in the research phase. As far as the design process is concerned it is in the research phase. Increasingly what we are trying to do is trying to think holistically. Were trying to start with the macro and then we come down to the micro. And when it comes to UD, and I love all the Principles of UD, because what it is effectively doing, is talking to the concept of the web of life – the connectivity of everything (Snaddon, 2008);
- UD can be mainstreamed as an integrated module within the subject 'Design' (Oelofsen, 2008);

- Through pre-qualification education and through post-qualification education, in cooperation with our profession, making use of the compulsory Continuing Professional Development (CPD) programme of South African Council for the Architectural Profession (SACAP) (de Villiers, 2008);
- It can be brought into the BTech curriculum, as part of the above mentioned module, in order for students to start incorporating the principles into their 'design thinking' (Chisin, 2008);
- I can imagine UD being introduced to first year students in [Design] History and Theory, to "take them out of themselves" in terms of thinking about the 'whys' and 'hows' of designed artifacts through the ages (J.J. van der Merwe); and
- Develop curriculum where theoretical exposition of these principles and the scientific, aesthetic/cultural and economic rationale underpinning them; at the same time using planning studio work and projects to demonstrate relevance of these [same principles] (Tapela, 2008).

When asked whether they would be interested in having UD offered to their students in a formal model, all the informants answered in the affirmative. Snaddon (2008) puts it thus:

We would be very interested in teaching UD, but we would have to customise the current wording and terminology to make it relevant to the discipline. 'To find the correct fit' – as we say – it is almost like a blueprint that one uses to evaluate best practice and the right way to move forward. I think we can definitely factor it in. It is about making something that is so tacit a lot more explicit with regard to the way that we test our students.

The subjects that informants considered to be best aligned with UD include the following:

- Communication Design, History and Contemporary Design studies (Broom, 2008);
- As a integrated module within the subject Design (Oelofsen, 2008);
- Principles of Architectural Design, Applied Design and Studio Work 1 to 4 (de Villiers, 2008);
- In the first 3 years with Contemporary Studies, and in BTech with the [Design] Theory module (Chisin, 2008); and
- Planning Graphics 1; Planning Design 2, 3 & 4, Environmental Planning & Computer Applications (Tapela, 2008).

Informants were also unanimous in their choice of the *first year* of study as the best time to first introduce UD to their students. Broom (2008) believes that the first year is ideal as this is the stage at which "students are actually being made aware of communicating to a target audience [and exploring] notions of how to talk to a specific niche". Tapela (2008) proposes that students could engage with UD "right from the beginning in Year 1, if not earlier". Subsequently, an investigation of the feasibility and practicality of introducing design students of the Faculty to UD at the multidisciplinary *Foundation Course* or *Extended Curriculum Programme* phase is merited.

The informants also confirmed their commitment in continuing the use of UD principles in their future discipline-specific projects.

When asked who they felt would be the best champion for UD in the context of Cape Town, informants there was a diverse range of possibilities put forward as sampled below:

- When you're working with a commercial sponsor there needs to be quantifiable returns that can be shown. So you need to find a brand that has Universal Design promise. You need to align it with a brand promise with what it is you would like to promote. So I'm thinking and here it is again "Coke", their whole attitude is diversity, when you look at their advertising, usability, inclusivity, when you look at the way the product is viewed – it is viewed as a universal product (Broom, 2008);
- The Cape Town Metropolitan Council in respect of the town planning scheme and building regulations, as well as professional bodies (Oelofsen, 2008);
- In our profession, it would probably be the Cape Institute for Architecture and the local branch of the South African Institute of Architectural Technologists (de Villiers, 2008);
- Users and design specialists (Chisin, 2008);
- *Natalie du Toit* (J.J. van der Merwe, 2008);
- I think where collaboration between disciplines and professions either in project conception and/or implementation is often the cutting-edge of innovation and therefore the site of any champion of UD design (Tapela, 2008).

The subject of champions is discussed further in the Section 8.4. Informants offered the following additional comments with regards to the object of the study:

- I think that any brand that doesn't include UD into their corporate responsibility program now, as well as just into their general brand values is going to be missing out. Yes, because with this view on inclusivity and non-discrimination worldwide being topical at this particular time, any brand that discriminates in any way will be missing out (Broom, 2008);
- It is becoming increasingly important to integrate UD into the Interior Design (and other) programmes. However, success will depend, to a great degree, on the availability of qualified staff to present the modules in an integrated and meaningful manner (Oelofsen, 2008);
- We have been left behind with even basic UD Principles and implementation at CPUT. Wheelchair toilets have been malfunctioning, wheelchair access is non-existent or malfunctioning on campus, to name but a few. Working with hearing-impaired students is problematic, and so the list goes on. This points to the fact that a sensibility of inclusion is still lacking, and by introducing UD from 1st year on, this situation will be helped (Chisin, 2008); and
- I would like to see more discussion between staff and students regarding the connections between UD and systems thinking, and the benefits to both (J.J. van der Merwe, 2008).

The above comments point towards a need to engage both students and staff in a robust discourse and discussion prior to the implementation of UD at CPUT. Chisin (2008) and J.J. van der Merwe (2008) propose that linking the concept of UD to 'design thinking' and 'systems thinking' would foreground the teaching, learning and research imperatives more concretely. The potential for the creation of *transdisciplinary* or *Mode 2* knowledge can further be enhanced through exploring the overlapping fields that UD facilitates (Winberg, 2006; Garraway, 2008).

In summary, the responses from the design educators have the following implications with regards to the formulation of strategies for UD-related activities at CPUT:

- explore the possibilities of introducing students to UD concepts when students are in their *First Year* of study (or even earlier at *foundation* level if practicable);
- provide students with *theoretical foregrounding* in UD through the cross-disciplinary course offerings of the *Department of Design History and Theory*, and integrate the *practical components* of UD into discipline-specific *studio-based* projects;
- develop *unit standards* for UD to be offered as *core* and *elective* courses through recurriculated undergraduate programmes – this could eventually evolve into *major* and *minor* subject offerings available via appropriate *web-based learning* platforms;
- facilitate the development of SAQA-aligned credit-bearing (multi-disciplinary and crossdomain teamwork) continuing professional development training programmes on UDrelated topics for practicing professionals;
- promote *contract research* (and other *third-stream* income) outputs to industry and other *Triple Helix* partners this can be done in collaboration with other HEIs ;
- collaborate with professional designers, architects and engineers in co-facilitating cooperative learning opportunities for students around UD-related projects;
- promote UD-related collaborative teaching, learning and research activities at CPUT;
- participate in institutional projects to make CPUT a UD-compliant model through the development of effective wayfinding, accessible environments and inclusive educational teaching and learning resources;
- convene regular inter-departmental discussion fora involving teaching staff and students to sustain robust debate and dialogue on UD;
- identify support mechanisms (such as teaching and learning materials, equipment and other resources) to aid in the promotion of UD adoption in various academic projects;
- identify and support suitable internal UD champions with a transdisciplinary appeal;

- maintain an accessible and expandable (electronic) database of UD-related academic publications and research outputs (with particular reference to activities in CPUT);
- identify and secure sustainable sources of funding for research activities (possibly through an NRF niche research focus area) to build institutional capacity in UD. This could also be done by linking such strategies with *Centres of Excellence* and other institutions already working in UD-related activities such as the *Intelligent Environments for Independent Living* (IE4IL) research project being carried out at the *Meraka Institute: African Advanced Institute for Information and Communication Technology* (Coetzee, 2008);
- participate in and support competitions that align with a problem-based learning ethos in promoting novel concepts such as UD in *majority world/industrially developing* contexts;
- encourage staff and students to engage in *participatory* and *socially responsible design* activities to demonstrate the value of UD to local communities;
- proactively incorporate and promote the adoption of Principles of UD within their own communities-of-practice by constantly exchanging best practices; and
- provide the popular media and press with UD-specific exemplars for dissemination to the general public.

8.4 Possible Champions for Universal Design

Based on findings from this research, a number of individuals were proposed as possible champions for UD in Cape Town. The interior design students also included a 'neutral' cartoon character as a potential candidate. The following is the shortlist of public figures (with a brief qualifier) – two sports figures, and two politicians as suggested by informants in rank order – the athletes proved to be the most popular choice:

- Natalie du Toit world record-holding Paralympian swimmer (Bo, 2008);
- Oscar Pistorius world record-holding Paralympian sprinter (du Swardt, 2008);
- Nelson Mandela internationally renowned statesman and former President of South Africa; and
- *Helen Zille* Premier of the Western Cape Province and former Executive Mayor of Cape Town.

Additionally, a number of professional bodies, corporate organisations and public service agencies were proposed though none emerged as a clear favourite. These include: the *Council of Architects*; the *Institute of Architects*; *Cape Institute for Architecture*; *Coca-Cola*; and the *Cape*

Town Metropolitan Council. The professionals identified as key champions for UD in rank order are: *architects*; *designers*; *engineers*; *planners*; and *consumer advocates*. A collaborative approach "between disciplines and professions either in project conception and/or implementation" was also proposed as an effective strategy for championing UD (Tapela, 2008).

The choice of *Natalie du Toit* and *Oscar Pistorius* who are both members of the *South African Sports Association for the Physically Disabled* (SASAPD), and the fact that they enjoyed such popular support is instructive. As disabled athletes, they captured the public imagination by transcending their own physical limitations to achieve world acclaim (Bernard, 2008). Bo (2008) reports of Natalie du Toit's popularity in the recent *Beijing 2008 Summer Olympics* (also known as the *Games of the XXIX Olympiad* with *'One World, One Dream'* as its motto) in part due to the fact that despite her disability, Natalie managed to qualify for the 'non-disabled' swimming heats without the use of a prosthetic device – she made history by being the first disabled athlete to qualify for the Olympics. Natalie (shown in Figure 8.45) finished 16th overall but won tremendous respect for challenging the status quo (McCallum, 2008; Schroeder, 2009).



Figure 8.45 Paralympian swimmer Natalie du Toit (source: © <u>http://legendsrevealed.com</u>, 2009)

Oscar Pistorius nicknamed '*The Blade Runner*' (as he requires prosthesis for running) made similar headlines when he challenged the South African national Olympics committee's rules that bar disabled athletes from participating in the non-disabled version of the games (McCallum, 2008). The case sparked off an ethical debate on the role that technology (potentially) plays in an athlete's performance (du Swardt, 2008). Oscar (shown in Figure 8.46) was eventually allowed to participate in the qualifying heats. Oscar missed the Olympics qualifying time narrowly and as such could not secure a place on the country's national team of non-disabled sprinters – he did however participate in the *Beijing 2008 Paralympics Games* where he lived up to his reputation as 'the world's fastest man without legs' (ibid).



Figure 8.46 Paralympian sprinter Oscar Pistorius (source: © <u>http://rawartint.files.wordpress.com</u>, 2008)

The feat of the South African Paralympics athletes is made all the more impressive when compared to the dismal performance of the Olympics team. The 136 (non-disabled) athletes only managed a single medal overall – a silver medal by *Godfrey Khotso Mokoena* in the long jump event and were ranked out 70th out of the 204 nations that participated in Beijing – a low ranking that echoed that of *Bafana Bafana* in the soccer milieu. The Paralympics team on the other hand, finished in a respectable *sixth overall* position with an impressive tally of *30 medals*, 21 of which were gold – including Natalie's (five), Oscar's (three) and Hilton Langenhoven's (three) personal gold medal tallies (Morgan, 2008). Further, Natalie was "named the female winner of the *Whang Youn Dai Achievement Award* for the 2008 Games" (*ibid*).

An additional factor that lends support to the choice of these two sports figures as possible champions for UD is the significant influence and coverage that sports endorsements attract through sponsorship by the corporate/private sector. This would provide sponsoring agencies opportunities to position themselves strategically with respect to corporate social investment provided that ethical concerns are duly addressed (Manzini, 2006). Internationally, South Africa enjoys a good reputation in certain sports (as discussed in Chapter 2) – sports in this context allows otherwise disparate communities to transcend the potentially divisive nature of politics and instead form a more collective response to issues affecting all citizens and residents alike.

8.5 Mainstreaming of Universal Design in Cape Town

For UD to be effectively mainstreamed in Cape Town, the coordination local, regional and national arms of government would need a collaborative process of logistical and operational articulation across the various departments. The absence of such coordinated action would result in duplication of scarce resources or worse still, confusion regarding the individual level operations and group level actions of key actors on the one hand, and the collective or societal level activities of the emerging community.

Further, some level of inter-departmental coordination would be required in a democratic, transparent, and yet flexible manner. The recently promulgated *Department of Women, Youth, Children and People with Disabilities* is uniquely placed to act as in such a capacity, provided that such a role is legitimised through a popular mandate by all participating actors. Once such a structure is established, a suitable champion should be elected to drive the process that is integrated at all levels of policy and planning. Other departments that could play such a coordinating role include the *Department of Economic Development and Tourism, Department of Social Development*, and the *Department of Public Service and Administration*.

Other important actors based on the *Triple Helix* partnership model (of government, industry and academia) would include the *Tourism Grading Council*; the *Cape Town Partnership*; *Higher Education Institutions* (including all the major universities within the greater Cape Town region); professional councils/institutes representing various architectural, design and engineering disciplines; consumer advocacy groups; members of civil society representing the disability sector; as well as representatives from media and the arts, among others. For CPUT, engagement with other key actors around UD-related activities would present an opportunity to realise the institutional mission: "to develop and sustain an empowering environment where,

through teaching, learning, research and scholarship our students and staff, in partnership with the community and industry, are able to create and apply knowledge that contributes to development". Further, UD should become a core strategic component in future efforts to position Cape Town internationally, for example in the city's bid to become the *World Design Capital* (WDC) in 2014 – an influential programme managed by ICSID.

8.6 Summary

This chapter focussed on the analysis of data generated from multiple sites and at various phases of this study. Activity analysis of various operations, actions, and activities shed light on the complex and multi-sectoral imperatives in promoting a novel concept like UD in an industrially developing context. With reference to education, linking *inclusive education* policies and strategies to education-specific applications of UD (such *Universal Design for Learning*) could offer immediate practical benefits to the two related fields. Aligning UD with institutions that are already involved in the provision of more accessible and usable products, services, systems and built environments would potentially eliminate wasteful duplication of effort and instead channel the collective social capital towards ensuring greater sustainability of such activities within Cape Town. Further, as has been argued in this thesis, the interpretivist stance facilitated the voice of various actors in addressing UD-related concerns from their own unique perspectives.

Additionally, the ethos of *ubuntu* cannot (and should not) be applied selectively to inform responses to issues such as discrimination, systemic exclusion or discrimination. By its very nature, *ubuntu* is a unifying philosophy that applies to *all* human beings without exception. Thus *ubuntu* can be invoked in alignment with the ideals of UD, but not narrowly in the context of specific constituencies (such as in response to the needs of people with disabilities or the elderly) whose members exhibit observable or demonstrable differences in appearance and functioning – such actions would only perpetuate the stigma associated with atypicality. Instead *ubuntu* and UD should be viewed as being complementary in their celebration of human diversity in its entirety. Finally, the role of credible champions to promote adoption and mainstreaming of novel concepts is critical to the success of such an enterprise. Informants indicated a general preference for apolitical public figures to play this key role. Notwithstanding, the selection of such champions should be done in an open and participatory manner – short-listed candidates/nominees should enjoy broad-based cross-sectoral appeal so as to ensure acceptance within the communities that require UD promotion and adoption.

207

CHAPTER NINE CONCLUSIONS AND RECOMMENDATIONS

9.0 Introduction

This final chapter reviews the research objectives and presents pertinent conclusions from this exploratory study. Further, the chapter attempts a justification of the study in terms of its contribution to knowledge in a number of inter-related fields. Though every effort has been made to address as many issues pertaining to UD in a developing/majority world context, no single study can claim to adequately address all questions arising from the research process – consequently, limitations of the study and implications for further research are presented for due consideration.

9.1 Revisiting the Research Objectives

The stated overall objective of this study was to propose *strategies for effectively mainstreaming UD considerations amongst design practitioners in an emerging economy context.* To achieve this purpose, the specific objectives have been addressed by doing the following:

- Evaluating the current level of UD awareness among selected designers and planners in Cape Town, especially in the light of FIFA 2010 World Cup[™]-related activities within the city;
- Assessing the applicability of UD strategies that have evolved outside the context of this study;
- Proposing appropriate sustainable and adjustable strategies for mainstreaming UD applications in Cape Town beyond the 2010 sporting mega-event; and
- Contributing to the strategic goals of the African Decade for Persons with Disabilities (1999-2009).

The specific conclusions linked to the abovementioned research objectives are elaborated in the following section.

9.2 Conclusions Drawn

With reference to evaluating the current levels of UD awareness in the context of Cape Town, this study found mixed responses. Planners and professional designers appear to have heard of UD but linked it predominantly to disability-specific applications. Further, *Universal Access* is supported by agencies such as the United Nations and thus more widely known than UD (P.

Thompson, 2008). With the exception of the *Mouille Point Promenade and Beachfront Student Landscape Design Competition*, UD had not been explicitly engaged with nor addressed in policy, planning and projects within the city of Cape Town. Notwithstanding, *IDC Consultants*, a leading architectural design consultancy has been using UD in its professional practice for more than a decade (*ibid*). Further, the *Disability Workshops Development Enterprise* (DWDE), *National Council for Persons with Physical Disabilities*, and *Quadriplegic Association of South Africa* (QASA), all actively promote UD in their work (Deglon, 2008; Wetzel, 2008). The student survey also revealed that a negligible number of students outside the *Department of Industrial Design* (where the researcher is based) were exposed to UD concepts. The students and their lecturers however expressed interest in further engagement with UD in a more formalised and structured manner in future.

In assessing the applicability of UD strategies outside of Europe, Japan and North America, this study found that UD is indeed relevant and adaptable to local contexts. However, the study noted the need to address the central issue of participation (and by extension, ownership) by potential end-users of products, services, systems and built environments. The UD model as practiced elsewhere appears to follow the traditional approach wherein the designers are viewed as professional experts who only engage potential end-users in an *ad hoc* and *post hoc* manner in the design process. The proposed UD strategies should be flexible and allow for a *distributed network* of exchange (Johansson, Kisch & Mirata, 2005). They should also accommodate *diversity* as a means to ensure *resilience* and *sustainability* of resultant strategies (Stirling, n.d).

This thesis proposes sustainable and adjustable strategies for mainstreaming UD applications in Cape Town beyond the 2010 imperative that would enhance participation and ownership by all key actors. Such strategies can only be deemed appropriate if they align with emerging fields as *Participatory Design* (as a methodology for engaging participants), *Design for Sustainability* and *Service Design* (with their allied didactic tools for promoting social equity and cohesion) (Ehn *et al.*, 2002; Manzini, 2006; Vezzoli, 2007; Cipolla, 2009). The said strategies should also engender *consensus* participation – as opposed to *consultative*; and *representative* modes of participation (Sjöberg, 1996). Additionally, the strategies should embrace elements of local inspiration that are easily recognised for their authentic design language (Algotsson & Davies, 1996).

During the compilation of this thesis, the mandate of the Secretariat on the *African Decade for Persons with Disabilities* (ADPD) was extended by another decade. The ADPD was initially set to

run from 1999 to 2009, and will now run all the way to 2019 (Dube, 2008). As noted elsewhere in this thesis, this study addresses certain strategic goals of the ADPD by paying "particular attention to [the needs of] persons with disabilities in the fields of health and education" (DPOD, 2008). The issue of education is the specific purview of the concept of *UD for Learning* – a field allied to *Inclusive Education* (see Section 5.2.2).

9.3 General Conclusions and Proposed Strategies

The UD-related activities discussed in Sections 8.2.1, 8.2.2 and 8.2.3 revealed a number of contradictions and tensions from an Activity Theoretical perspective. Whereas the *object* (or target of the activities) was consistently that of mainstreaming UD adoption within the city of Cape Town, the *units* of analysis, *directing factors* and *subject* changed according to the specific *communities* involved (Engeström *et al.*, 1990). Further, the *instruments* (Principles of UD and related tools used internally and externally) remained more or less constant throughout the diverse contexts. The *rules* and the *division of labour* on the other hand were content- and context-responsive – these elements are elaborated in Section 6.1. The general conclusions and proposed adjustable strategies for mainstreaming UD in various contexts within the city of Cape Town are summarised using the different activity systems of the three cases covered in this thesis and depicted in Figures 9.1-9.3 for reference.



Figure 9.1: Summary of activities: the educational environment

Figure 9.1 presents an overview of the educational environment in which the *UD for Product Designers* module was presented. The following adjustable strategies can be employed:

• CPUT needs to facilitate stronger links with the disability sector. Ensuing benefits could include:

- o giving students opportunities to engage experts on disability-related issues;
- supporting collaborative projects with a focus on such topics as reasonable accommodation in the workplace, inclusive education, and the design and development of assistive technology and devices;
- o engaging members of the disability sector as guest lecturers;
- o facilitating experiential learning opportunities for students; and
- o advancing UD-specific teaching, learning and research activities.



Figure 9.2: Summary of activities: the botanical gardens environment

Figure 9.2 presents a summary of the activities relating to the UD audit at the *Kirstenbosch National Botanical Gardens*. The following adjustable strategies can be employed:

- CPUT should negotiate involvement in public projects in its capacity as a leading HEI in the Western Cape Province. Ensuing benefits would include:
 - o engaging staff and students in real-life *problem-based* projects;
 - o co-developing relevant policies and strategies within *research niche* areas;
 - o generating third-stream income through *contract research* outputs;
 - o strengthening of *Triple Helix* partnerships; and
 - o providing expert input on disability-related issues.
- CPUT should strengthen its links with design professionals (particularly its own alumni).
 Such a strategy would hopefully result in:

- o providing opportunities for *internships* and *cooperative learning* for students;
- engaging alumni as external examiners, guest lecturers, and as industry advisory panel members for quality assurance purposes; and
- co-developing course content for upgrading skills of in-service designers through continuing professional development.



Figure 9.3: Summary of activities: the students' competition

Figure 9.3 presents a summary of the activities relating to the 2010 *Mouille Point Promenade and Beachfront Student Landscape Design Competition*. The following adjustable strategies can be employed:

- Staff within CPUT in general, and the *Faculty of Informatics and Design* in particular, should identify areas of potential collaboration around UD and strengthen interdepartmental and cross-disciplinary links therein. Ensuing benefits would include:
 - building capacity around UD-related clusters and leverage the same in advancing CPUT's vision: "to be at the heart of technology education and innovation in Africa";
 - developing multi-disciplinary and cross-domain research projects that invite the broad participation of staff and students within the Faculty;
 - developing links with other disciplines and departments within the CPUT community that could potentially enrich teaching, learning and research capabilities; and
 - participating in UD-related activities within operational units where the efficacy of UD-thinking could be demonstrated these should include institutional *Disability Units* and the *Fundani Teaching and Learning Centre*, among others.

9.4 Contributions to Knowledge

The main objective of this research was to evolve and propose strategies for mainstreaming UD applications in design projects in the context of an emerging economy. The proposed strategies are sufficiently adjustable and context-responsive to allow for application in similar industrially developing or majority world settings. It is anticipated that the process of *social change* would be "informed by the potent principles of *ubuntu* [...] as the primary organising principle of African morality" (Siyabulela & Duncan, 2006:306).

The research makes a specific contribution towards the call of the African Decade of People with Disabilities for the "formulation and implementation of technical guidelines and legislation to promote access by disabled persons to buildings, public facilities, transport and communications systems, information, education training, and technical aids as well as empowerment of persons with disabilities" (ADPD, 2006; Chalken *et al.*, 2006:93). As argued throughout this thesis, these concerns would be adequately addressed by a UD approach.

The study contributes to the development of multi-disciplinary and cross-domain UD modules for teaching and learning at interested *higher education institutions* in the Western Cape Province in particular, and South Africa in general (Lorenzo, ka Toni & Priestly, 2006:189). This follows the lines of the *Universal Design Education Project (UDEP)* as described by Ostroff (2003:348; Universal Design Education Online, 2006), the global examples for teaching UD/Design for All presented by Kennig and Rhyl (2002), and UD applications in educational environments (McGuire *et al.*, 2006). This is in alignment with the recommendations of the *Special Needs Education* White Paper (South Africa, 2001) and *Higher Education Act* (South Africa, 1997b) for synergetic collaboration on matters of common interest (such as inclusive education policies and strategies) among higher education institutions (Howell, 2006).

The research offers a database of best practice exemplars for reference and benchmarking purposes by design practitioners. These UD exemplars from Cape Town are found in the supplementary CD-ROM appended to this thesis. This expandable database could be regularly updated and uploaded onto the Internet and offered as a web-based learning resource for greater accessibility, as well as hyperlinked to related hosts (such as national and continental organisations for ageing and disability). Cronjé, Bligaut and Bothma (2002) support the use of *web-based resources* for information, communication and educational applications of this nature.

It is believed that issues raised in this study will help facilitate a vibrant and robust collaborative UD forum/network for all interested parties to sustain debate around pressing social issues; coordinate research by making UD an NRF-recognised *research niche area* at CPUT (South Africa, 2002); facilitate capacity-building of locally available UD-related expertise; provide strategic direction; and act as a one-stop resource base for UD-related queries where as Thomas (2006:65) proposes, "value-driven designers [...] can share their ideas and work more effectively in support of the ideals in which they believe". Keating (2008) reports of the potential commercial value of innovative research to CPUT – this is the kind of value that UD embodies. Further, *ubuntu* is one of eight core institutional values of CPUT as articulated in its vision and mission statements (CPUT, 2006). This thesis argues that the principles engendered in both the UD ideal and the *ubuntu* ethos are not only perceived to be progressive, but that they are vital to the formulation of context-sensitive and sustainable responses to counter social exclusion.

9.5 Limitations of Research

From a purely quantitative point of view, the limited number of informants (averaging just over 17% of the total student population) could impact upon the generalisability of the findings from the survey of design students. However, with regard to the research intent and focus, the findings reveal rich multi-dimensional aspects that offer a reasonable degree of transferability in related concerns (Dolan & Hall, 2001). The study also provides a firm basis for introducing UD across the different design disciplines, particularly earlier on in course programmes – this would promote enhanced uptake of UD-thinking in latter years of study.

Further, though the initial sample population included all students within the design disciplines, *Fashion and Surface Design* as well as *Jewellery Design and Manufacture* students did not participate in the survey due to the perception by their lecturers that their typical approach to the design process is one of situated learning (as opposed to problem-based learning in the other sub-disciplines). A reasonable degree of representativeness in the survey was ensured by the somewhat higher than anticipated response rates in the departments of *Architectural Technology* (29.3%) and *Industrial Design* (24.2%) – two key disciplines in the community-of-practice of UD.

Finally, whereas the initial intent for the 2010 students' design competition was to promote multidisciplinarity through the voluntary participation of all design students (including those of the *Department of Landscape Technology*); only students from the *Department of Interior Design* participated. Those departments that did not do so cited reasons such as inflexibility in timetabling, lack of content knowledge, and perceived irrelevance of the competition's theme to the core focus of their study discipline. This somewhat disappointing outcome consequently limited the possibility of exploring the cross-disciplinary dynamics of collaborative design tasks.

9.6 Implications for Further Research

The findings provide interesting and useful information on various attitudes towards UD. Further, the study revealed the inherent confusion between UD and *Universal Access* (UA) by many professional designers. The majority of informants from professional design backgrounds suggested that UD should be offered as a [SAQA-aligned] credit-earning (unit standard) course for *Continuing Professional Development* via their respective professional bodies. Though etymologically linked, a useful introduction to UD would need to interrogate the link with UA, as well as a discussion on the unique methodology of the UD paradigm. There is thus scope for research into multi-disciplinary and cross-domain collaborative design work involving mixed architectural and design teams.

There is also a need to document newly-completed and ongoing infrastructural developments that incorporate UD. Emerging exemplars of UD could subsequently be disseminated and diffused through accessible physical and virtual platforms so as to facilitate best practice benchmarking. The said exemplars would form a valuable basis for service learning opportunities for design students. From a pedagogical point of view, a study on the potential synergies between national ministerial policies on *Inclusive Education* with UD (such as *Universal Design for Learning*) should be initiated to consolidate the benefits of UD to the Department of Education and other relevant or related ministries. Subsequently, the newly established *Department of Women, Youth, Children and People with Disabilities* should explore the possibility of coordinating trans-sectoral strategies/activities through an *inter-ministerial committee* that enables government departments/ministries to leverage potential benefits of UD concepts as a means to address such diverse imperatives as the promotion of social equity and cohesion, inclusion, access and participation for all South Africans.

The findings from the analysis on UD pictograms reveals the fact that the more explicit/transparent pictograms lent themselves better to interpretation than the more abstract/opaque ones. This is corroborated by a number of other studies on semiotics and semantics (Singer & Ritz, 1996; Harrell *et al.*, 2000; Murungi, 2003; Yielding, 2003). A dedicated research and design study should be conducted to establish a context-responsive set of

215

pictograms in the vein of the *Adinkra Symbols* (2007) from West Africa for utilisation as wayfinding signage and related symbols specifically for use in South Africa. As argued elsewhere in this thesis, South Africa's large number of official languages presents certain practical, logistical and economic challenges that could be ameliorated by the development and adoption of an accessible and inclusive visual language employing culturally pertinent signs and symbols. This can be done through a national collaborative research effort involving tertiary/higher education institutions offering Information, Interaction, Visual and/or Graphic Design employing a participatory design methodology across the various unique cultural/linguistic groups.

This study addresses the issue of mainstreaming UD in products, services, systems and built environments. Admittedly, the focus was on an urban setting with world-class infrastructure. Further research would need to interrogate the potential applications of UD to rural and periurban, resource-poor settings. The rural areas in this country for example account for a significant population of people categorised as "extra-ordinary" or 'vulnerable' – these include people living with HIV/Aids (of all generations), women (usually related to migrant workers in urban areas), children (including large numbers of orphans), seniors citizens, people with disabilities, and certain categories of foreign nationals (such as refugees). Further, the role of *ubuntu* in developing participatory, culturally sensitive, and sustainable models for promoting social inclusion should be interrogated further. Indeed, *ubuntu* would buttress any community-based care model to potentially ameliorate the plight of people living with HIV/Aids in particular, and also inform a robust and comprehensive response to the stigmatising and discriminatory practices experienced by other vulnerable groups. This issue is of particular urgency in the South African context and should be the focus of a dedicated study (Barnett & Whiteside, 2002).

9.7 Summary

Universal Design is a relatively new concept in industrially developing or majority world contexts. The hosting of the FIFA 2010 World Cup[™] in South Africa in general, and in Cape Town in particular, presented an historic opportunity for action research into means by which UD could be mainstreamed in the said milieu. The study revealed certain challenges that would need to be addressed in the adoption of UD by a number of actors and communities-of-practice. Notwithstanding, the general impression is that UD is considered worthy of further engagement and application, and that with minor tuning, similar strategies could be employed in related contexts elsewhere in South Africa, the African continent, and beyond. The findings also inform the context-responsive strategies that will be disseminated to a wider audience via an interactive,

collaborative, inclusive and participatory ethos. The emphasis is on maintaining diversity through a battery of strategies, as opposed to amalgamating the diverse proposals into a unified or standardised approach that ignores or negates *in situ* realities. It is critical that for any proposed strategies to ultimately find acceptance within the intended constituencies, as many potential beneficiaries as possible should engage freely and continually with the change process.

Subsequently, the findings in this study suggest that UD should be presented in a distributed and non-prescriptive manner that respects and embraces cultural and philosophical diversity. These trans-cultural and transgenerational imperatives will become more pressing as the double impact of accelerated *international migration*, and the global phenomenon of *population ageing* become increasingly apparent in developing or emerging economies. Additionally, the study challenges prevailing welfarist models that are patently unsustainable and typically serve to perpetuate stigmatisation and discrimination, as well as reinforcing dependency (and in many instances a retroactive entitlement syndrome) in people with special needs.

Finally, the popular notion that the responsibility for UD implementation be left solely in the hands of 'experts' or professionals such as planners, architects and designers is deemed untenable in that every citizen has an inalienable (and constitutionally enshrined) right to social inclusion and participation in all matters that touch on their wellbeing (and quality of life). The notion of 'ownership through participation' is a theme that resonates throughout this thesis. Consequently, any future UD strategies must be informed by a potentially *emancipatory* social change process wherein the end-users are co-opted as co-designers through a *consensus participatory process* – all residents should be *invited and expected* to take part in *all* stages of the design process as far as is practically (and logistically) possible. Such a participatory approach fundamentally fosters democracy and good governance, as well as contributing positively to the development of appropriate and desirable social capital. In so doing, the inclusive character of UD-thinking would facilitate the quest for enhanced accessibility, usability and sustainability in products, systems, services and built environments *for everyone*.

REFERENCES

Accessible Travel & Leisure. 2007. *Helping the less mobile access the world.* www.accessibletravel.co.uk [12 July 2007].

Adaptive Environments. 2006. *Universal Design: what is universal design?* <u>http://www.adatenv.org/index.php?option=Content&Itemid=3</u> [5 February 2007].

Adinkra Symbols ©. 2007. *West African Wisdom: Adinkra Symbols and their Meanings.* <u>http://www.welltempered.net/adinkra/</u> [4 July 2008].

Africa News Update. 2008. *Park for Disabled Children Opens*. Johannesburg: Africa News Update. http://www.africafiles.org/database/www.afrika.no [26 August 2008].

African Decade for Disabled Persons (ADPD). 2006. Declaration of the African Decade. <u>http://www.secretariat.disabilityafrica.org/documents</u> [20 October 2006].

Akers, L. 2008. *Ethical dilemmas: individual human rights versus sustainable development excellence, migration and equality policy: managing unintended consequences?* ResIST Thematic Paper: December 2008. <u>www.resist-research.net/cms/.../Deliverable%2011_Thematic%20Paper%203_Excellence%20Migration%20and%20Equality%20Policy.pdf</u> [12 May 2009].

Akindele, S. T., Gidado, T. O. & Olaopo, O. R. 2002. Globalisation, its impact and consequences for Africa. *Globalisation*. **2**(1): 3-22.

Aslaksen, F., Bergh, S., Bringa, O.R. & Heggem, E.K. 1997. *Universal Design: Planning and Design for All.* The Norwegian State Council on Disability. <u>http://home.online.no/~bringa/universal.htm</u> [7th December 2008].

Albone, R. 2008. *Creating a Clearer Picture.* pp. 6-7 in: Ageways Issue 71: June 2008. London: HelpAge International.

Algotsson, S. & Davies, D. 1996. *The Spirit of African Design.* New York: Clarkson N. Potter.

Atkinson, D. 2009. The 2010 and the rural hinterland: maximising advantage from megaevents. pp. 153-173 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Austad, S.N. 1997. *Why We Age: What Science is Discovering About the Body's Journey Through Life.* New York: John Wiley & Sons, Inc.

Baars, J. 2007. Introduction: Chronological time and chronological age: problems of temporal diversity. pp. 1-13 in: J. Baars & H. Visser (eds), *Aging and Time: multidisciplinary perspectives. Amityville*, New York: Baywood Publishing Company, Inc.

Baber, C. 2005. Evaluation in Human-Computer Interaction. pp. 354-387 in: J.R. Wilson & N. Corlett (eds), *Evaluation of Human Work.* 3rd ed. Boca Raton: Taylor & Francis.

Badenhorst, C. 2007. *Research Writing: breaking the barriers.* Pretoria: Van Schaik Publishers.

Bagshaw, M. 2004. Is diversity divisive? A positive training approach. *Industrial and Commercial Training*. **36**(4): 153-157.

Bailey, K.D. 1987. *Methods of Social Research*. 3rd ed. New York: The Free Press.

Balaram, S. 2001. Universal Design and the Majority World. pp. 55.20 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Balcazar, F.E., Keys, C.B., Kaplan, D.L. & Suarez-Balcazar, Y. 1998. Participatory Action Research and People with Disabilities: principles and challenges. *Canadian Journal of Rehabilitation*. **12**: 105-112.

Baldwin, D. 2003. Wayfinding technology: road map to the future. *Journal of Visual Impairment & Blindness*. October 2003: 612-620.

Barredo, M.S. 2006. Semi-structured interview with author at Cape Town Society for the Blind. Cape Town: 13 April 2006. [Notes in possession of author].

Barnes, C. 2008. *Zim still hopes for 2010 tourism.* Cape Argus: Monday, 22 December 2008: p. 3.

Barnes, C. Mercer, G. & Shakespeare, T. 1999. *Exploring Disability: a Sociological Introduction*. Oxford: Polity.

Barnett, T. & Whiteside, A. 2002. *AIDS in the Twenty-First Century: Disease and Globalization*. New York: Palgrave Macmillan.

Batshaw, M.L. 1997. *Children with Disabilities.* 4th ed. Maryland: Paul H. Brookes Publishing Co.

BBC Sport. 2009. *IPL confirms South Africa switch: the 2009 Indian Premier League will now take place in South Africa.* BBC Sport: 24 March 2009. <u>http://news.bbc.co.uk/sport1/hi/cricket/7958664.stm</u> [1 April 2009].

Beasley, K.A. 1997. *The Paralympic Village: a barrier-free city.* Barcelona: Centre d'Estudis Olímpics UAB. <u>http://olympicstudies.uab.es/pdf/wp088_eng.pdf</u> [10 March 2006].

Beasley, K.A. & Davies, T.D. 2001. Access to Sports and Entertainment. pp. 47.1-47.16 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc

Bedny, G.Z., Seglin, M.H. & Meister, D. 2000. Activity theory: history, research and application. *Theoretical Issues in Ergonomics Science*. **1**(2): 168-206.

Benktzon, M. 1993. Designing for Our Future Selves: the Swedish experience. *Applied Ergonomics.* **24**(1): 19-27.

Benson, S. 1997. The Body, Health and Eating Disorders. pp. 122-181 in: K. Woodward (ed), *Identity and Difference*. London: Sage Publications.

Berk, L.E. 2001. *Development through the Lifespan.* 2nd ed. Needham Heights, MA: Allyn& Bacon.

Bernard, M. & Scharf, T. (eds). 2007. *Critical Perspectives on Ageing Societies*. Bristol: The Policy Press.

Bhengu, M. J. 1996. Ubuntu: the essence of democracy. Cape Town: Novalis Press.

Bhengu, M.J. 2006. *Ubuntu: the global philosophy for mankind.* Cape Town: Lotsha Publications.

Bhengu, M.J. 2007. Semi-structured interview with author at Canal Walk. Century City: 13 June 2007. [Notes, voice recording and transcribed responses in possession of author].

Bickenbach, J.E. & Wasserman, D. 2006. Ethics. pp. 618-624 in: G.L. Albrecht (ed), *Encyclopedia of Disability*. Thousand Oaks, CA: Sage Publications.

Biggs, J. 2002. *Guide to Curriculum Design: Personal Development Planning.* Learning and Teaching Support Network (LTSN) Generic Centre, 1st draft: 22/07/02.

Blaikie, A. 1993. Images of Age: a reflexive process. Applied Ergonomics. 24(1): 51-57.

Bo, W. 2008. *High five for du Toit.* China Daily/The Paralympian: <u>http://www.chinadaily.com.cn/paralympics/2008-09/15/content_7027813.htm</u> [15 September 2008].

Bogdan, R.C. & Biklen, S.K. 2003. *Qualitative Research for Education: an introduction to theory and methods.* 4th ed. Allyn and Bacon, Boston.

Boon, M. 2007. *The African Way: The Power of Interactive Leadership*. Cape Town: Zebra Press.

Boonzaier, D.A.V. 2006. Semi-structured interview with author at Cape Peninsula University of Technology. Cape Town: 26 January 2006. [Notes, voice recording and transcribed responses in possession of author].

Boonzaier, D.A. & M'Rithaa, M.K. 2007. Universal Design workshop. *Discursive workshop at the Design Education Forum of Southern Africa (DEFSA) International Conference*, Cape Peninsula University of Technology, 3-5 October 2007. Cape Town.

Boraine, A. 2004. *The State of the Cities Report 2004 and implications for city development strategies.* Presentation to Cities Alliance Public Policy Forum 1-3 November 2004, Durban.

Boraine, A. 2007. *Central City Fan-mile: concepts and considerations.* PowerPoint presentation: 15 October 2007.

Botha, H.L. & du Plessis, S.A. 2007. An investigation of self-appraised cognition versus measured cognition. *South African Journal of Higher Education, SAJHE*. **21**(4): 608-627.

Boud, D. & Falchikov, N. 2005. Redesigning assessment for learning beyond higher education, In *Research and Development in Higher Education 28*. Brew, A. & Asmar, C. (eds), Sydney: HERDSA, 34-41.

Bound, J. & Coleman, R. 1993. Activities and products for the Third Age. *Applied Ergonomics.* **24**(1): 58-62.

Bovy, P. 2002. *Transport and Exceptional Public Events: mega sports event transportation and main mobility management issues.* Lausanne: Economic Research Centre.

Bovy, P.H. 2004. *AsiaRail2004 / Hong Kong: rail developments for Beijing 2008 Olympic Games and Shanghai 2010 World Expo.* PowerPoint presentation: 14 October 2004. <u>www.mobility-bovy.ch</u> [16 June 2007].

Box, H. 2008. Focus on dementia. The TygerBurger: Wednesday, 08 October 2008: p. 10.

Boyatzis, R. E. 1998. *Transforming Qualitative Information: thematic analysis and code development.* California: SAGE Publications.

Broom, A. 2008. Semi-structured interview with author at Department of Graphic Design, CPUT. Cape Town: 15 August 2008. [Notes, voice recording and transcribed responses in possession of author].

Buchanan, R. 2001. Human Dignity and Human Rights: thoughts on the principles. *Design Issues*, **17**(3): 35-40.

Buden, B. 2002. *The Art of Being Guilty is the Politics of Resistance Depoliticizing Transgression and Emancipatory Hybridization*. <u>http://www.republicart.net</u> [31 August 2008].

Bucher, N.R. 2008. *City's roads should belong to public transport – expert.* Cape Argus: Monday, 25 August 2008: p. 4.

Bühler, C. 2001. Empowered participation of users with disabilities in universal design. *Universal Access Information Society*. **1**: 85–90.

Build-for-All. 2006. *The Build-for-All handbook: good intentions are not good enough.* www.build-for-all.net [12 October 2007].

Buxton, B. 2007. *Sketching User Experiences: getting the design right and the right design.* San Francisco: Morgan Kaufmann Publishers.

Campbell, T. 2006. *IPPUC: The Untold Secret of Curitiba*. <u>www.UrbanAge.org</u> [13 March 2007].

Cape Argus, 2006. *FIFA chief: SA can do it: Blatter dismisses claims of 'back-up plan' to relocate World Cup.* Cape Argus (Late Final). Wednesday, 25 October 2006: p. 1.

Cape Peninsula University of Technology (CPUT).

Cape Times. 2008. *City busy registering Green Point's informal traders market.* Cape Times: Wednesday, 08 October 2008: p. 4.

Casserley, C. & Ormerod, M. 2003. The Legal Argument for Inclusive Design. pp. 142-154 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Center for Universal Design (CUD).

Chalken, S., Swartz, L. & Watermeyer, B. 2006. Establishing the Secretariat for the African Decade of Persons with Disabilities. pp. 93-98 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: a South African Agenda.* Cape Town: HSRC Press.

Chang, B.V., Tremblay, K.R. & Dunbar, B.H. 2000. An experiential approach to teaching Universal Design. *Education.* **121**(1): 153-158.

Chapman, A. 2002. *Hierarchy of Needs (1990's eight-stage model based on Maslow).* <u>www.businessballs.com</u> [8 November 2007].

Chapman, K. & McCartney, K. 2002. Smart homes for people with restricted mobility. *Property Management*. **20**(2): 153-166.

Child Health Policy Institute. 2001. *Summary of Literature: disability and chronic illness prevalence in children in South Africa.* Rondebosch: Child Health Policy Institute.

Chisin, A. 2008. Response to semi-structured questions from the author. Cape Town: 18 August 2008. [Notes and email correspondence in possession of author].

Chmela-Jones, K.A., Buys, C. & Gaede, R.J. 2007. Visual learning and graphic design: a cooperative strategy. *South African Journal of Higher Education, SAJHE.* **21**(4): 628-639.

The CIA World Factbook. 2008. <u>https://www.cia.gov/library/publications/download/</u> [10 April 2009].

Cipolla, C. 2009. Relational Services and Conviviality. pp. 232-245 in: S. Miettinen and M. Koivisto (eds), *Designing Services with Innovative Methods*. Helsinki: Kuopio Academy of Design.

City of Cape Town. 2006. 2010 Football World Cup™: Cape Town and the Western Cape Business Plan. 31 October 2006.

City of Cape Town. 2007. 2010 Football World Cup™: Strategic Plan for the Provincial Government and the City of Cape Town. 22 January 2007.

City of Cape Town. 2008. The 2010 FIFA World Cup™ Host City Cape Town Green Goal Action Plan.

www.capetown.gov.za/en/GreenGoal/Documents/2010 GREEN GOAL ACTION PLAN .pdf [20 October 2008].

City of Cape Town. 2009. *Integrated Rapid Transit: expanding the system in phases*. <u>http://www.capetown.gov.za/en/irt/Pages/Expandingthesysteminphases.aspx</u> [16 April 2009].

CityNews. 2007a. Excellent return on investment in 2010. CityNews (June 2007) 3(2): 4.

CityNews. 2007b. 2010: shift to legacy projects: R10-bn for transport, security, infrastructure and services. CityNews (June 2007) **3**(5): 7.

CityNews. 2007c. Transport on track for 2010. CityNews (September 2007) 3(6): 7.

Clarkson, J., Coleman, R., Keates, S. & Lebbon, C. 2003. *Inclusive Design: design for the whole population*. London: Springer.

Clarkson, P.J., Keates, S., Coleman, R., Lebbon, C. & Johnston, M. 2000. *A Model for Inclusive Design*. <u>http://rehab-www.eng.cam.ac.uk/papers/lsk12/edc2000/</u> [30 August 2006].

Coan, S. 2003. *'I'm a quadriplegic but that doesn't mean I am disabled'*. Pietermaritzburg: The Witness Features: 29 May 2003.

Coetzee, L. 2008. *Intelligent Environments for Independent Living (IE4IL)*. Presentation to the Heinreich-Hertz Institute, Germany. Meraka Institute. PowerPoint presentation: 13 February 2008.

Cohen, B. & Menken, J. (eds). 2006. *Aging in Sub-Saharan Africa: Recommendations for Furthering Research.* Washington, D.C.: National Academic Press. http://www.nap.edu/catalog/11708.html [10 March 2008].

COHRE. 2007. *Mega-Events, Olympic Games and Housing Rights.* Geneva: Centre on Housing Rights and Evictions. <u>http://www.cohre.org</u> [13 July 2007].

Coleman, R. 1993. A demographic overview of the ageing of First World populations. *Applied Ergonomics.* **24**(1): 5-8.

Coleman, R. 1999. Inclusive Design – Design for All. pp. 158-170 in: W.S. Green & P.W. Jordan (eds), Human *Factors in Product Design: current practice and future trends*. London: Taylor & Francis.

Coleman, R. 2001. Designing for Our Future Selves. pp. 4.1-4.25 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Coleman, R. 2006. *From Margins to Mainstream: why inclusive design is better design.* <u>www.erqonomics.org.uk/espdfs/ErqSocLecture.pdf</u> [2 February 2007].

Coleman, R. & Pullinger, D.J. 1993. Designing for our future selves. *Applied Ergonomics*. **24**(1): 3-4.

Coleman, R., Bendixen, K. & Tahkokallio, P. 2003. A European Perspective. pp. 288-307 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Collins, P., Shukla, S. & Redmiles, D. 2002. Activity Theory and Systems Design: A View from the Trenches. *Computer Supported Cooperative Work*. **11**: 55-80.

Contact Newsletter.2006. *We Are Accessible*. City of Cape Town: No. 23 March/April 2006: 14.

Contact Newsletter. 2007a. *The stadium is set to go!* City of Cape Town: No. 24 March/April 2007: 1.

Contact Newsletter. 2007b. *Profile – Helen Zille: leading by energetic example.* City of Cape Town: No. 24 March/April 2007: 3.

Contact Newsletter. 2007c. *A stadium for 2010 – and beyond.* City of Cape Town: No. 24 March/April 2007: 10-13.

Contact Newsletter. 2007d. *Meet the City's 'home' team…* City of Cape Town: No. 24 March/April 2007: 11.

Contact Newsletter. 2007e. *Highlights of the game*. City of Cape Town: No. 24 March/April 2007: 11.

Contact Newsletter. 2007f. *The bigger picture- improving the facilities around the stadium.* City of Cape Town: No. 24 March/April 2007: 12.

Cornelissen, S. 2004. Sport Mega-events in Africa: processes, impacts and prospects. *Tourism and Hospitality Planning & Development.* **1**(1): 39-55.

Cornelissen, S. 2005. *The Global Tourism System: governance, development and lessons from South Africa.* London: Ashgate Publishing.

Cornelissen, S. 2007. Of proprietors and stakeholders: the political economy of planning and legacies of the 2010 FIFA World Cup[™]. *World Journal of Managing Events*. **1**(1): 34-42.

Cornelissen, S. 2009. Sport, mega-events and urban tourism: exploring the patterns, constraints and prospects of the 2010 World Cup. pp. 131-152 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Cornelissen, S. & Swart, K. 2006. The 2010 Football World cup as a political construct: the challenge of making good on an African promise. *The Sociological Review*. **54**(2): 108-123.

Cornell, E.H., Sorenson, A. & Mio, T. 2003. Human sense of direction and wayfinding. *Annals of the Association of American Geographers*. **93**(2): 299-425.

Coy, J. 2003. Inclusion – a commercial perspective. pp. 156-171 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: design for the whole population*. London: Springer.

CPUT see Cape Peninsula University of Technology.

CPUT. 2006. *Institutional Vision, Mission and Core Values*. Cape Town: CPUT. <u>http://www.cput.ac.za/institution/mission.php</u> [6 June 2006].

CPUT. 2007. *Institutional Research and Planning: enrolments for disabled students 2006.* Cape Town: CPUT.

CRC. 2004. *Way-finding in the built environment.* November 2004. <u>www.construction-innovation.info/images/pdfs/Way_finding.pdf</u> [27 July 2007].

Creff, K. 2004. *Exploring Ubuntu and the African Renaissance: a conceptual study of servant leadership from an African perspective.* School of Leadership Studies, Regent University.

<u>http://www.regent.edu/acad/sls/publications/conference_proceedings/servant_leadership_r</u> <u>oundtable/2004pdf/cerff_exploring_ubuntu.pdf</u> [1 March 2007].

Creswell, J.W. 2002. *Educational Research - planning, conducting and evaluating quantitative and qualitative research.* London: Prentice Hall.

Crews, D.E. & Zavotka, S. 2006. Aging, Disability and Frailty: implications for Universal Design. *Journal of Physiological Anthropology*. **25**: 113-118.

Cronjé, J.C., Blignaut, A.S. & Bothma, T.J.D. 2002. Meeting adult learning needs during Web-supported Web-skills training. *South African Journal of Higher Education, SAJHE/SATHO.* **16**(3): 207-217.

Crotty, M. 1998. *The foundations of social research meaning and perspective in the research process*. London: Sage Publication.

CUD see Center for Universal Design.

CUD. 1997. *Principles of Universal Design*. Raleigh: North Carolina State University. <u>http://www.design.ncsu.edu/cud/about_ud/docs/English.pdf</u> [1 September 2006].

CUD. 2006. http://www.design.ncsu.edu/cud/index.htm [1 September 2006].

Czeglédy, A.P. 2009. A World Cup and the construction of African reality. pp. 281-295 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Dalcher, D. 2006. Consilience for universal design: the emergence of a third culture. *Universal Access Information Society*. **5**: 253-268.

Danford, G.S. & Tauke, B. (eds). 2001. *Universal Design: New York*. New York: Center for Inclusive Design & Environmental Access, University at Buffalo, The State University of New York.

Daniel, J., Lutchman, J. & Comninos, A. 2007. South Africa in Africa: trends and forecasts in a changing African political economy. pp. 508-532 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation.* Cape Town: HSRC Press.

Davey, C.L., Wootton, A.B., Thomas A., Cooper R. & Press, M. 2005. Design for the Surreal World? A New Model of Socially Responsible Design. *Proceedings of the 6th International Conference of the European Academy of Design, EAD06,* University of the Arts Bremen, pp 29–31 March 2005. Bremen.

Davies, G. 2009. Managing the alchemy of the 2010 Football World Cup. pp. 33-52 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Davies, S.E.H. 2006. Semi-structured interview with author at Cape Peninsula University of Technology. Cape Town: 01 December 2006. [Notes, voice recording and transcribed responses in possession of author].

Davis, G. 2006. *About: Ergonomics*. London: Design Council. <u>http://www.design-</u> <u>council.org.uk/Documents/About%20design/Design%20techniques/Ergonomics/Ergonomics</u> <u>s%20by%20Gary%20Davis.pdf</u> [3 May 2007].

DDA see The Disability Discrimination Act.

Deacon, H., Stephney, I & Prosalendis, S. 2005. *Understanding HIV/AIDS stigma: a theoretical and methodological analysis.* Cape Town: Human Sciences Research Council (HSRC) Press. <u>www.hsrcpress.ac.za</u> [27 September 2007].

Decortis, F., Noirfalise, S. & Saudelli, B. 2000. Activity Theory, Cognitive Ergonomics and Distributed Cognition: Three Views of a Transport Company. *International Journal of Human-Computer Studies.* **53**: 5-33.

Deglon, U. 2008. Semi-structured interview with author at Cape Peninsula University of Technology. Cape Town: 24 November 2008. [Notes in possession of author].

Demirbilek, O. 1999. *Involving the Elderly in the Design Process: a participatory design model for Usability, Safety and Attractiveness.* PhD thesis. Ankara: Bilkent University.

Demirbilek, O. & Demirkan, H. 2004. Universal Product Design Involving Elderly Users: A Participatory Design Model. *Applied Ergonomics.* **35**: 361-370.

Dentlinger, L. 2007a. *Stadium court action fails: CEPA to seek new urgent interdict on Monday.* Weekend Argus: Saturday, 14 April 2007: p 1.

Dentlinger, L. 2007b. *Green Point naming rights to be up for grabs.* Cape Argus: Wednesday, 28 November 2007: p 4.

Dentlinger, L. 2008. *City warns FIFA bullies: costs overshoot budget by R2bn.* Weekend Argus: Saturday, 8 November 2008: p 1.

Denzin, N.K. & Lincoln, Y.S. (eds). 1998. *Collecting and Interpreting Qualitative Materials.* London: SAGE Publications.

Denzin, N.K. & Lincoln, Y.S. (eds). 2005. *Qualitative Research*. London: SAGE Publications.

Department of the Premier. 2008. *Western Cape Progress Report: 2010 FIFA World Cup*™. Cape Town: Provincial Government of the Western Cape.

Department of Social Development. 2006. *Operational Manual for Protective Workshops.* Cape Town: Western Cape Department of Social Development.

Department of Social Development. 2007. *Overview of 10 year integrated service delivery plan (2007-2017): deepening democracy and empowerment.* May 2007. Cape Town: Western Cape Department of Social Development.

Design Council. 2006. Inclusive design.

http://www.designcouncil.org.uk/webdav/harmonise?Page/@id=40&Section/@id=1354 [22 August 2006]

DePoy, E. & Gilson, S. 2004. *Rethinking Disability: principles for professional and social change.* Belmont: Brooks/Cole-Thomson Learning.

DePoy, E. & Gilson, S. 2005/2006. Universal Web Access. *International Journal of Technology, Knowledge and Society.* **1**(1): 129-132.

DePoy, E. & Gilson, S. 2006. Moving beyond nomothetic category. *International Journal of the Humanities*. **4**(2): 23-29.

DePoy, E. & Gilson, S. 2007. The Bell-Shaped Curve: alive, well and living in diversity rhetoric. *International Journal of Diversity in Organisations, communities and Nations.* **7**(3): 253-259.

Desai, A. & Nabbi, Z. 2007. 'Truck and Trailer': rugby and transformation in South Africa. pp. 402-424 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation.* Cape Town: HSRC Press.

Design for All Institute of India. 2006. http://www.designforall.in/ [30 August 2006].

Design Indaba. 2007. *Fan Your Passion 2010*. Design Indaba magazine 2nd Quarter 2007. Cape Town: Interactive Africa. pp 33-71.

de Swardt, W. 2008. *Oscar simply unstoppable.* <u>http://www.sport24.co.za/Content/OtherSport/Paralympics/346/0ce3b6a6cd4f4931a544574</u> <u>ec300d3d2//Oscar simply unstoppable</u> [17 September 2008].

de Villiers, J. 2008. Response to semi-structured questions from the author. Cape Town: 03 September 2008. [Notes and email correspondence in possession of author].

Disabled Peoples Organisation – Denmark (DPOD). 2008. *New EU-Africa Strategy*. <u>http://www.disability.dk/dsi-s-international-development-cooperation/new-eu-africa-strategy</u> [14 April 2009].

Disability. 2003. *NEPAD and the Disability Sector*. <u>http://www.disability.dk/site/viewdoc.php?doc_id=763</u> [28 March 2007]. The Disability Discrimination Act (DDA) – United Kingdom. 1996. London: Department for Education and Employment.

Dittmann-Kohli, F. 2007. Temporal references in the construction of self-identity: A lifespan approach. pp. 83- 119 in: J. Baars & H. Visser (eds), *Aging and Time: multidisciplinary perspectives. Amityville*, New York: Baywood Publishing Company, Inc.

Dobbs, D. 2001. *Curitiba's "Bus Rapid Transit" Operation: a critical look relative to actual American transit experience*. <u>http://www.lightrailnow.org/facts/fa_cur02.htm</u> [28 February 2007].

DoE see South Africa. Department of Education.

Dolan, R.P. & Hall, T.E. 2001. Universal Design for Learning: implications for large-scale assessment, *IDA Perspectives.* **27**(4): 22-25.

Dong, H., Keates, S. & Clarkson, P.J. 2003. UK and US industrial perspectives on inclusive design, in *International Conference on Inclusive Design and Communications (INCLUDE 2003),* The Royal College of Art, London, UK, 10:406-10:409

DPOD see Disabled Peoples Organisation – Denmark.

DPSA (Disabled People South Africa). 2004. *Disability Definitions, Models and Terminology*. <u>http://www.capegateway.gov.za/eng/pubs/public_info/D/91149</u> [22 April 2009].

DST. n.d. *DST-NRF Centres of Excellence*. Pretoria: Department of Science and Technology/National Research Foundation. <u>http://www.nrf.ac.za/centres/index.html</u> [6 November 2008].

DST. 2003. *The NRF Framework for Centres of Excellence in the South African System of Research and Innovation*. DST-NRF Centres of Excellence: May 2003. <u>http://www.nrf.ac.za/centres/documents.html</u> [6 November 2008].

DST. 2004. Handbook to assist with the operation of a Centre of Excellence. DST-NRF Centres of Excellence: April 2004. <u>http://www.nrf.ac.za/centres/documents.html</u> [6 November 2008].

Dube, A.K. 2008. Semi-structured interview with author at the Secretariat of the African Decade of People with Disabilities. Cape Town: 24 November 2008. [Notes in possession of author].

du Plessis, S. & Maennig, W. 2009. South Africa 2010: initial dreams and sobering economic perspectives. pp. 55-75 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Durrheim, K. 1999. Quantitative Analysis. pp. 96-122 in: M. Terre Blanche & K. Durrheim (eds), *Research in Practice: Applied Methods for the Social Sciences.* Cape Town: University of Cape Town Press.

du Toit, A. & Neves, D. 2007. *In search of South Africa's Second Economy: chronic poverty, economic marginalisation and adverse incorporation in Mt Frere and Khayalitsha.* Chronic Poverty Research Centre. CPRC Working Paper 102. www.chronicpoverty.org/pdfs/102duToit_Neves.pdf [19 April 2008].

Eason, K. 2005. Ergonomic interventions in the implementation of new technical systems. pp. 919-931 in: Wilson & N. Corlett (eds), *Evaluation of Human Work .3rd ed.* Boca Raton: Taylor & Francis.

EDeAN see European Design for All e-Accessibility Network.

Edelson, E. 1991. Aging. New York: Chelsea House Publishers.

Edries, S. & Triegaardt, M. 2004. The feminisation of HIV/AIDS. *African Renaissance*. **1**(2): 121-124.

Ehn, P. 1988. Work-Oriented Design of Computer Artifacts. Stockholm: Arbetslivscentrum.

Ehn, P. & Badham, R. 2002. Participatory design and the collective designer. pp. 1-10 in: T. Binder, J. Gregory & I. Wagner (eds). *Proceedings of the PDC 2002 Participatory Design Conference*, 23-25 June 2002. Malmö.

EIDD. 2004. *The EIDD Stockholm Declaration.* The European Institute for Design and Disability (EIDD). <u>www.eidd.org</u> [8 November 2007].

Engel-Hills, P., Garraway, J., Nduna, J., Philotheou, G. & Winberg, C. 2005. Reflections on life as a student from the position of employment. *South African Journal of Higher Education*, *SAJHE*. **1**(2): 292-305.

Engeström, Y. 1999. Expansive Visibilization of Work: an Activity Theory perspective. *Computer Supported Cooperative Work.* **8**: 63-93.

Engeström, Y. 2000. Activity Theory as a framework for analyzing and redesigning work. *Ergonomics.* **43**(7): 960-974.

Engeström, Y., Brown, K., Engeström, R. & Koistinen, K. 1990. Organizational forgetting: an activity theoretical perspective. pp. 139-168 in: D. Middleton & D. Edwards (eds), *Collective Remembering*. London: Sage Publications.

European Commission. 2002. *Definitions of disability in Europe: a comparative analysis*. European Commission: Employment and Social Affairs (September 2002). <u>http://ec.europa.eu/employment_social/index/complete_report_en.pdf</u> [10 March 2007].

European Design for All e-Accessibility Network (EDeAN). 2006. *Design for all: glossary of terms*. <u>http://www.education.edean.org/index.php?filters=f37</u> [5 August 2006].

Ezeoha, A. & Uche, C. 2005. *South Africa, NEPAD and the African Renaissance.* African Studies Centre. ASC Working Paper 64/2005. <u>www.ascleiden.nl/Pdf/workingpaper64.pdf</u> [14 May 2008].

Family Practice Notebook. 2000a. *Activities of Daily Living Scale*. <u>http://www.fpnotebook.com/GER2.htm</u> [21 June 2006].

Family Practice Notebook. 2000b. *Exercise in the Elderly*. <u>http://www.fpnotebook.com/GER2.htm</u> [21 June 2006].

Family Practice Notebook. 2000c. *Instrumental Activities of Daily Living.* <u>http://www.fpnotebook.com/GER3.htm</u> [21 June 2006].

Favish, J. 2005. Equity in changing patterns of enrolment, in learner retention and success at the Cape Technikon. *South African Journal of Higher Education, SAJHE.* **19**(2): 274-291.

Fick, D. 2006. Africa: continent of economic activities. Johannesburg: STE Publishers.

Foucault, M. 1980. *Power-Knowledge: Selected Interviews and Other Writings, 1972-1977.* Brighton: Harvester Press.

Fielding, N. G. & Lee, R. M. 1998. *Computer Analysis and Qualitative Research*. London, Sage Publications.

FIFA. 2003. *FIFA Safety Guidelines.* FIFA: 24 December 2003. <u>www.FIFA.com</u> [22 November 2006].

FIFA. 2004a. *Equipment Regulations: regulations governing the sports equipment at FIFA competitions.* FIFA: April 2004. <u>www.FIFA.com</u> [15 March 2007]. FIFA. 2004b. *Laws of the Game 2004.* FIFA: July 2004. <u>www.FIFA.com</u> [7 April 2007].

FIFA. 2006a. *FIFA Quality Concept: handbook of requirements for football turf.* FIFA: March 2006 Edition. <u>www.FIFA.com</u> [5 February 2008].

FIFA. 2006b. *FIFA World Cup Stadiums South Africa 2010.* FIFA: April 2006. www.FIFA.com [27 August 2007].

FIFA. 2006c. Overview of the FIFA World Cup™ Host Countries 1930-2010. FIFA: March 2006. www.FIFA.com [27 August 2007].

FIFA. 2007a. 2010 FIFA World Cup South Africa™: FIFA public information sheet (a guide to FIFA's Official marks). FIFA: Issue #3; September 2007. www.FIFA.com [10 June 2008].

FIFA. 2007b. *Football Stadiums: technical recommendations and requirements.* 4th edition. FIFA: July 2007. <u>www.FIFA.com</u> [10 June 2008].

FIFA. 2007c. *Regulations 2010 FIFA World Cup South Africa*™. FIFA: July 2007. www.FIFA.com [14 July 2008].

FIFA. 2009. *Lively Soweto welcomes FIFA Confederations Cup roadshow.* FIFA: Tuesday, 14 April 2009.

<u>http://www.fifa.com/confederationscup/organisation/news/newsid=1048113.html#lively+so</u> weto+welcomes+fifa+confederations+cup+roadshoww [16 April 2009].

Findeli, A. 2008. Sustainable Design: a critique of the current tripolar model. *The Design Journal*. **11**(3): 301-322.

Fisk, J. 1993. Design for the elderly: a biological perspective. *Applied Ergonomics.* **24**(1): 47-50.

Forster, D. 2006. Identity in relationship: The ethics of *ubuntu* as an answer to the impasse of individual consciousness.

www.spirituality.org.za/files/ubuntu%20and%20identity%20D%20Forster%202006.doc [13 March 2008].

Foster, P. 1996. Observational Research. pp. 57-93 in: R. Sapsford & V. Jupp (eds), *Data Collection and Analysis.* London: Sage Publications Ltd.

Fourie, S. 2009. *New IRT bus system takes shape; taxis have important role to play.* The TygerBurger: Wednesday, 15 April 2009: p. 14.

Fransolet, C. & M'Rithaa, M. 2009. Design as a strategic tool for Business. *Paper presentation at the DWDE International Small Business Conference 2009,* Cape Town international Conference Centre, 6-7 April 2009. Cape Town.

Fraser, N. 2007. *Wayfinding: plotting your city route*. 19 March 2007. <u>http://www.joburg.org.za/citichat/2007/mar19_citichat11.stm</u> [24 July 2007].

Futerman, R. & M'Rithaa, M.K. 2007. Engaging with the Elderly: a participatory approach to compatibility. *Proceedings of the 10th Ergonomics Society of South Africa (ESSA) Conference on Applied Ergonomics*, International Convention Centre, 21-25 June 2007. Durban.

Gardner, L., Powell, L. & Page, M. 1993. An appraisal of a selection of products currently available to older consumers. *Applied Ergonomics*. **24**(1): 35-39.

Garraway, J. 2008. *Knowledge boundaries and boundary-crossing in the design of work-responsive university curricula.* Cape Peninsula University of Technology. Paper for DLLs Seminar: 31/03/08.

George, A.L. & Duquette, C. 2006. The psychosocial experiences of a student with low vision. *Journal of Visual Impairment & Blindness*. March 2006: 152-163.

Gibbs, S. 2005. An ageing population – a complex polemic paradigm. In: A. Thatcher, J. James & A. Todd (eds). *Proceedings of CybErg 2005. The Fourth International Cyberspace Conference on Ergonomics.* Johannesburg: International Ergonomics Association Press.

Gibson, N. 2004. Africa's global futures. African Renaissance. 1(2): 108-114.

Gill, J. 2004. *Access-ability: making technology more usable by people with disabilities.* <u>www.tiresias.org.guidelines</u> [20 October 2006].

Gill, J. 2007. Accessibility for Visitors. London: RNIB.

Gilson, S. & DePoy E. 2006. Explanatory Legitimacy Theory. pp. 699-701 in: G.L. Albrecht (ed), *Encyclopedia of Disability*. Thousand Oaks, CA: Sage Publications.

Gladwell, M. 2008. Outliers: the story of success. London: Penguin Books.

Global Universal Design Online Network (GUDEON). 2007. <u>www.universaldesign.net</u> [2 April 2007].

Goldsmith, S. 1997. *Designing for the Disabled – the new paradigm*. Oxford: Architectural Press.

Goldsmith, S. 2000. *Universal Design: a manual of practical guidance for architects*. Oxford: Architectural Press.

Goldsmith, S. 2001. The Bottom-Up Methodology of Universal Design. pp. 25.1-25.16 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Gomes, R. 2005. *Designing to Live: the value of inclusive design in the future society.* <u>www.generativeart.com/papers2005/35.RicardoGomes1-Inclusive%20Design.htm</u> [19 June 2006].

Goodman, J., Brewster, S.A. & Gray, P. 2005. How can we best use landmarks to support older people in navigation? *Behaviour & Information Technology*. **24**(1): 3-20.

Gosling, M. 2008. Foreshore switch-station will light up 2010 stadium. Cape Times: Friday, 10 October 2008: p. 7.

Goulding, C. 1998. Grounded theory: the missing methodology on the interpretivist agenda. *Qualitative Market Research: An International Journal.* **1**(1): 50-57.

Gough, D. 2000. Discourse and students experience of higher education. pp. 43-58 in: B. Leibowitz & Y. Mohamed (eds), *Routes to Writing in South Africa.* Cape Town: Silk Road.

Granville, S. & Dison, L. 2001. *Thinking about thinking: integrating self-reflection into an academic literacy course.* In the Proceedings of the International Literacy Conference in Cape Town, 13-17 November 2001. <u>http://www.ched.uct.ac.za/literacy/StellaGranville.html</u>

Gray, D.E. 2004. Doing Research in the Real World. London: Sage Publications Ltd.

Gribben, R. 2009. *EU ruling on UK compulsory retirement gives older workers hope.* The Telegraph: 5 Mar 2009. <u>www.telegraph.co.uk/finance/4944422/EU-ruling-on-UK-</u> <u>compulsory-retirement-gives-older-workers-hope.html</u> [10 May 2009].

Griffith, C. 2008. Telephonic and electronic responses to semi-structured questions from the author. Cape Town: 15 September 2008. [Notes and email correspondence in possession of author].

Gross, J. 2008. *For the Elderly, Being Heard About Life's End.* The New York Times: May 5, 2008. <u>http://www.nytimes.com/2008/05/05/health/05slow.html</u> [5 May 2008].

GUDEON see Global Universal Design Online Network.

Haferburg, C., Golka, T. & Selter, S. 2009. Public viewing areas: interventions in the context of mega-events. pp. 174-199 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Hagen, M. 2008. Telephonic and electronic responses to semi-structured questions from the author. Cape Town: 19 September 2008. [Notes and email correspondence in possession of author].

Haigh, R. 1993. The Ageing Process: a challenge for design. *Applied Ergonomics.* **24**(1): 9-14.

Haines, H., Wilson, J.R., Vink, P. & Koningsveld, E. 2002. Validating a Framework for Participatory Ergonomics (the PEF). *Ergonomics.* **45**(4): 309-327.

Haldenwang, B. 2008. Population ageing in the Western Cape province. pp. 127-150 in: R. Marindo, C. Groenewald & S. Gaisie (eds), *The state of the Population in the Western Cape Province*. Cape Town: Human Sciences Research Council (HSRC) Press.

Hansson, L. 2006. *Universal Design - a marketable or utopian concept?* Göteborg: Centre for Consumer Science.

Hansson, T. 2002. Leadership by Activity Theory and Professional Development by Social Construction. *Systematic Practice and Action Research*. **15**(5): 411-436.

Harrell, W. A., Bowlby, J.W. & Hall-Hoffarth, D. 2000. Directing Wayfinding with Maps: the effects of gender, age, route complexity, and familiarity with the environment. *The Journal of Social Psychology.* **140**(2): 169-178.

Healey, S. 1994. Growing to be an Old Woman: age and ageism. pp. 175-184 in: R. Schmitt & T.E. Moody (eds), *Key Concepts in Critical Theory: alienation and social criticism.* New Jersey: Humanities Press.
Health Systems Trust. 2006. *Health Statistics: prevalence of disability.* <u>http://www.hst.org.za/healthstats/48/data</u> [8 August 2006].

HelpAge. 2002. State of the World's Older People 2002. London: HelpAge International.

HelpAge. 2006. *Madrid International Plan of Action on Ageing in an Ageing World*. <u>http://www.helpage.org/Researchandpolicy/MadridInterntionalPlanofActiononAgeing...</u> [04 June 2007].

HelpAge. 2009. *Africa Moves on Social Policy: the first ever Social Policy Framework for Africa recognises the depth of older people's poverty and promotes their right to social protection.* London: Ageing and Development. Issue **25** (March 2009): p.2.

Hevi, J. 2004. Values and Human Rights: identity, uniqueness, dignity and communion. *African Renaissance*. **1**(2): 125-130.

Higgs, P. 2007. Towards an indigenous African epistemology of community in higher education research. *South African Journal of Higher Education, SAJHE.* **21**(4): 668-679.

Hine, J., Swan, D., Scott, J., Binnie, D. & Sharp, J. 2000. Using Technology to Overcome the Tyranny of Space: information provision and wayfinding. *Urban Studies*. **37**(10): 1757–1770.

Hitchcock, D.R., Lockyer, S., Cook, S. & Quigley, C. 2001. Third Age Usability and Safety - an ergonomics contribution to design. *International Journal of Human-Computer Studies*. **55**: 635-643.

Hornea, A. 2008. Electronic response to semi-structured questions from the author. Cape Town: 01 September 2008. [Notes and email correspondence in possession of author].

Howell, C. 2006. Disabled Students and Higher Education in South Africa. pp. 164-178 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: A South African Agenda.* Cape Town: HSRC Press.

Howell, C., Chalken, S. & Alberts, A. 2006. A History of the disability rights movement in South Africa. pp. 46-83 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: A South African Agenda.* Cape Town: HSRC Press.

HSRC see Human Science Research Council.

Human Science Research Council. 2003. *Ageing in South Africa: report on the minimum set on ageing.* March 2002. Cape Town: Department of Health.

Huppert, F. 2003. Designing for Older Users. pp. 30-49 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: design for the whole population.* London: Springer.

Hursthouse, R. 2007. *Virtue Ethics.* Stanford Encyclopedia of Philosophy. <u>http://plato.stanford.edu/entries/ethics-virtue/</u> [25 June 2008].

IAUD see International Association of Universal Design.

Ibhawoh, B. & Dibua, J.I. 2003. Deconstructing Ujamaa: The Legacy of Julius Nyerere in the Quest for Social and Economic Development in Africa. *African Journal of Political Science*. **8**(1): 59-83.

ICSID see International Council for Societies of Industrial Design.

Illich, I. 1973. *Tools for Conviviality.* New York: Harper & Row. Imrie, R. & Hall, P. 2001. *Inclusive Design: designing and developing accessible environments.* London: Spon Press.

Inclusion Africa. 2002. The Declaration of Bamako. <u>www.inclusion-international.org/</u> [20 August 2006].

Inclusive Education Western Cape, 2006. *What is inclusive education?* <u>http://www.included.org.za/page2.html</u> [3 August 2006].

International Association for Universal Design (IAUD). 2006. *News.* <u>http://www.iaud.net/en/news/index.html</u> [21 August 2006].

International Council of Societies of Industrial Design (ICSID). 2002. *Facts about ICSID*. International Council of Societies of Industrial Design. <u>http://www.icsid.org/</u> [1 November 2004].

International Organization for Standardisation. 2001. *Guidelines for Standards Developers to Address the Needs of Older People and Persons with Disabilities* (ISO Guide 71). Geneva: International Organization for Standardisation.

ISO see International Organization for Standardisation.

Jaeger, P.T. & Bowman, C.A. 2005. *Understanding Disability: inclusion, access, diversity, and civil rights.* Westport: Praeger Publishing.

Jansen, J.D. 2007. The language of research. pp. 14-22 in: K. Maree (ed), *First Steps in Research*. Pretoria: Van Schaik Publishers.

Jere-Malanda, R. 2008. *And now. Positive image of Africa.* AfricaFiles No. 18393. <u>http://www.africafiles.org</u> [3 July 2008].

Johansson, A., Kisch, P. & Mirata, M. 2005. Distributed Economies- a new engine for innovation. *Journal of Cleaner Production.* **13**: 971-979.

Jönsson, B. & Certec. 2006a. Rehabilitation technology, design and pedagogy. pp 121-144 in: B. Jönsson (ed). *Design Side by Side*. Lund: Studentlitteratur (Certec/Lund University).

Jönsson, B. & Certec. 2006b. Design. pp 145- 156 in: B. Jönsson (ed). *Design Side by Side.* Lund: Studentlitteratur (Certec/Lund University).

Jönsson, B. & Certec. 2006c. Scientific positioning. pp 177-188 in: B. Jönsson (ed). *Design Side by Side.* Lund: Studentlitteratur (Certec/Lund University).

Jordan, P.W. 1999. Inclusive Design. pp. 171-181 in: W.S. Green & P.W. Jordan (eds). *Human Factors in Product Design: Current Practice and Future Trends*. London: Taylor & Francis.

Jordan, W.A. 2008. Universal Design: great-looking, great-living design for all ages, abilities, and circumstances. Beverly, Massachusetts: Quarry Publishing.

Joustra, A. 2008. Response to semi-structured questions from the author. Cape Town: 16 September 2008. [Notes and email correspondence in possession of author].

Kahn, M., Blankley, W., Maharajh, R., Pogue, T.E., Reddy, V., Cele, G. & Du Toit, M. 2004. *Flight of the flamingos: a study on the mobility of R&D workers*. Cape Town: HSRC Publishers.

Kahwaji, M.A. 2003. Case study: using technology in assisting disabled students - a disabled designer's use of technology to facilitate the design process. *Proceedings of the* 2003 *IDSA National Education Conference*, Pratt Institute, 10-12 August 2003. New York.

Kahwaji, M.A. 2005. *Portfolio for Michael Kahwaji: Don't cry over spilt milk*. September 10 2005. <u>http://coroflot.com/public/individual_details.asp?individual_id=2579</u> [15 April 2006].

Kearney, A.T. 2005. *Building a Legacy*. Chicago: A.T. Kearney Inc. <u>www.atkearney.com</u> [13 July 2007].

Keates, S. & Clarkson, J. 2003a. Countering design exclusion. pp. 438-453 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: design for the whole population.* London: Springer-Verlag.

Keates, S. & Clarkson, J. 2003b. Design exclusion. pp. 88-102 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: design for the whole population*. London: Springer-Verlag.

Keates, S., Clarkson, P.J., Harrison, L.A. & Robinson, P. 2000. Towards a practical inclusive design approach. *ACM 2000*: 45-52.

Keating, C. 2008. *Innovation Board Fund to help bring commercial value to the products of research*. Cape Peninsula University of Technology News: 5 November 2008. <u>http://www.cput.ac.za/news/featuredstory/index.php</u> [7 November 2008].

Kelly, K. 1999. Calling it a Day: reaching conclusions in qualitative research. pp. 421-437 in: M. Terre Blanche & K. Durrheim (eds), *Research in Practice: Applied Methods for the Social Sciences.* Cape Town: University of Cape Town Press.

Kennig, B. & Rhyl, C. 2002. *Teaching Universal Design.* AAOutils 2002 Project. <u>http://anlh.be/aaoutils/</u> [4 May 2006].

Khumomotse, S.M., M'Rithaa, M.K. & Moalosi, R. 2002. Botswana's quest for sustainability: scope for design intervention. *Proceedings of the Development by Design (dyd02) Conference*, 1-2 December 2002. Bangalore.

King, P.J. 2004. *One Hundred Philosophers: a guide to the world's greatest thinkers.* Cape Town: Zebra Press.

Kinsella, K. & Ferreira, M. 1997. *Ageing Trends: South Africa*. Washington, DC: Department of Commerce, Bureau of the Census.

Kinsella, K. & Velkoff, V. A. 2001. *An Aging World*: 2001. http://www.census.gov/prod/2001pubs/p95-01-1.pdf [5 November 2004].

Klironomos, I., Antona, M., Basdekis, I. & Stephanidis, C. 2006. White Paper: promoting Design for All and e-Accessibility in Europe. *Universal Access Information Society*. **5**: 105–119.

Knapper, C.K. & Cropley, A.J. 2000. *Lifelong Learning in Higher Education*. 3rd edition. Kogan Page Limited, London.

Knudsen, K.I.W. 2009. Sustainable employment for people with disabilities – inspiration from the Danish labour market. *Plenary presentation at the DWDE International Small Business Conference 2009*, Cape Town international Conference Centre, 6-7 April 2009. Cape Town.

Kogi, K., Kawakami, T., Ujita, Y. & Khai, T.T. 2005. Roles of Farmer Trainers for Participatory Ergonomics in Agriculture. pp. 626-633: in A. Thatcher, J. James & A. Todd (eds), Proceedings of CybErg 2005. *The Fourth International Cyberspace Conference on Ergonomics*. Johannesburg: International Ergonomics Association Press.

Koncelik, J.A. 1998. Design, Aging, Ethics and the Law. pp. 113-150 in: R. Roth & S.K. Roth (eds), *Beauty is Nowhere: Ethical Issues in Art and Design*. Amsterdam: Overseas Publishers Association.

Korenjak-Černe, S., Kejžar, N. & Batagelj, V. 2008. Clustering of Population Pyramids. *Informatica.* **32** (2008): 157–167.

Korpela, M., Mursu, A. & Soriyan, H.A. 2001. *Information Systems for Development as an Activity.* Conference Papers and Proceedings. <u>www.uku.fi/~korpela</u> [6 November 2006].

Korpela, M., Mursu, A., Soriyan, H.A., de la Harpe, R. & Macome, E. 2006. *Information Systems for Development in Africa: Results from INDEHELA*. Conference Papers and Proceedings. <u>www.uku.fi/~korpela</u> [6 November 2006].

Kortjass, B.B. 2007a. *Football for hope: FIFA aims to raise \$10m for centres to uplift the continent's young people.* Sunday Times. 25 November 2007 (Edition Two): p.1.

Kortjass, B.B. 2007b. '*Time to give back to Africa.*' Sunday Times. 25 November 2007 (Edition Two): p.12.

Kortjass, B.B. 2007c. *Free-ticket windfall for 2010.* Sunday Times (SoccerLife & Sport). 25 November 2007 (Edition Two): p.32.

Kose, S. 2003. The Japanese experience. pp. 309-316 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Kowal, P.R., Wolfson, L.J. & Dowd, J.E. 2000. Creating a Minimum Data Set on ageing in sub-Saharan Africa. Southern *African* Journal of Gerontology. **9**(2): 18-23. <u>www.who.int/entity/healthinfo/survey/**ageing** mds pub02.pdf</u> [9 September 2003].

Krauss, S. E. 2005. Research paradigms and meaning making: A primer. *The Qualitative Report.* **10**(4): 758-770. <u>http://www.nova.edu/ssss/QR/QR10-4/krauss.pdf</u> [10 March 2009].

Krippendorff, K. 1980. *Content Analysis: An Introduction to its Methodology*, Beverly Hills: Sage Publications.

Kroemer, K.H.E. 2006. *"Extra-Ordinary" Ergonomics: How to Accommodate Small and Big Persons, the Disabled and Elderly, Expectant Mothers, and Children.* Boca Raton: Taylor & Francis.

Kroemer, K.H.E., Kroemer, H.B. & Kroemer-Elbert, K.E. 2001. *Ergonomics: How to Design for Ease and Efficiency*. Upper Saddle River: Prentice Hall.

Kruger, C. & Cross, N. 2006. Solution driven versus problem driven design: strategies and outcomes. *Design Studies.* **27**: 527-548.

Kurvinen, E. 2007. *Prototyping social action*. Helsinki: University of Arts and Design, Helsinki.

Kyoyo-Hin. 2001. *The Kyoyo-Hin White Paper 2001.* The Kyoyo-Hin Foundation. <u>www.kyoyohin.org/</u> [4 July 2005]. Kyoyo-Hin. 2004. The Kyoyo-Hin Foundation. <u>www.kyoyohin.org/</u> [4 July 2005].

Lauche, K. 2005. Collaboration among Designers: Analysing an Activity for Systems Development. *Computer Supported Cooperative Work.* **14**: 253-282.

Lawton, M.P. & Brody, E.M. 1969. Assessment of older people: self-maintaining and instrumental activities of daily living. *Gerontologist.* **9**: 179-186.

Leedy, P. D. & Ormrod, J. E. 2001. *Practical Research Planning and Design*. New Jersey: Prentice-Hall.

Leonardi, M., Bickenbach, J., Ustun, T.B., Kostanjsek, N. & Chatterji, S. 2006. The Definition of Disability: what is in a name? *Lancet.* **368**: 1219-1220.

Levine, D. (ed). 2003. *Universal Design New York 2.* New York: Center for Inclusive Design & Environmental Access, University at Buffalo, The State University of New York.

Lewis, I. 2000. Access for all: helping to make participatory processes accessible to everyone. London: Save the Children.

Lim, C.P. & Hang, D. 2003. An activity theory approach to research of ICT integration in Singapore schools. Computers & Education. **42**: 49-63.

Lorenzo, T., ka Toni, M. & Priestley, M. 2006. Developing a Disability Studies Programme: engaging activism and academia. pp. 179-191 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider, and M. Priestly (eds), *Disability and Social Change: A South African Agenda*. Cape Town: HSRC Press.

Luca, J. & McLoughlin, C. 2002. *A Question of Balance: using self and peer assessment effectively in teamwork.* www.ascilite.org.au/conferences/auckland02/proceedings/papers/072.pdf [4 July 2006].

Luck, R. 2003. Dialogue in participatory design. *Design Studies*. 24(6): 523-535.

M'Rithaa, M.K., Moalosi, R. & Khumomotse, S.M. 2002. Communication in rural Africa. *Paper presented at the 9th Burg Wildenstein International Summer Design Workshop*, 17-27 July 2002. Leibertingen.

M'Rithaa, M.K. 2006. Redesigning Education for Inclusiveness: Universal Design in context. *Proceedings of the Design Education Forum of Southern Africa (DEFSA) Conference*, Nelson Mandela Metropolitan University, 21-22 September 2006. Port Elizabeth.

M'Rithaa, M.K. & Futerman, R. 2007. A City in Flux: Cape Town's search for an inclusive future. *Proceedings of the Design Education Forum of Southern Africa (DEFSA) International Conference*, Cape Peninsula University of Technology, 3-5 October 2007. Cape Town.

M'Rithaa, M.K. 2008. Engaging Change: an African perspective on designing for sustainability. *Proceedings of the Changing the Change (CtC) International Conference*, 10-12 July 2008. Turin.

M'Rithaa, M.K., Verveckken, B. & Futerman, R. 2008. Shifting Trajectories: advancing cosmopolitan localism through participatory innovation. *Proceedings of the Changing the Change (CtC) International Conference*, 10-12 July 2008. Turin.

M'Rithaa, M.K. 2009. Embracing sustainability: revisiting the authenticity of 'event' time... *Proceedings of the 2nd International Symposium on Sustainable Design (ISSD),* 5-6 November 2009. Sao Paolo.

Mace, R.L. 1998. A Perspective on Universal Design. *Proceedings of the Design for the 21st Century: An International Conference on Universal Design, Pratt Institute, 19 June 1998. New York.*

Macharia, S. 2006. *The urban 'informal economy' in the Global South: a feminist postempiricist study of policy discourse in sub-Saharan Africa.* www.yorku.ca/ishd/SM.Kenya.Report.field.pdf [9 May 2008].

Macpherson, M. 2001. Performance measurement in not-for-profit and public-sector organisations. *Measuring Business Excellence*. **5**(2): 13-17.

Madden, R. & Hogan, T. 1997. *The definition of disability in Australia: moving towards national consistency.* Canberra: Australian Institute of Health and Welfare (AIHW Cat. No. DIS 5).

Mafu, T. 2007. *Growth fuels inequality – Manuel: gap between rich and poor will expand over next decade.* Business Report. Monday, 23 April 2007: p. 1.

Mail & Guardian. 2009. *Bafana nose-dive in FIFA world rankings.* 09 April 2009. <u>http://www.mg.co.za/article/2009-04-08-bafana-nosedive-in-fifa-world-rankings</u> [15 April 2009].

Makgetla, N.S. 2007. Local government budgets and development: a tale of two towns. pp. 146-167 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation.* Cape Town: HSRC Press.

Makoni, S. 2000. Great Expectations: the professional and institutional demands of higher education in Southern Africa. pp. 1-9 in: S. Makoni (ed), *Improving Teaching and Learning in Higher Education: A Handbook for Southern Africa.* Johannesburg: Witwatersrand University Press.

Malan, S.P.T. 2000. The 'new paradigm' of outcomes-based education in perspective. *Journal of Family Ecology and Consumer Sciences*, **28**: 22-28.

Maloka, E.T. 2001. *The South African "African Renaissance" Debate: A Critique.* www.polis.sciencespobordeaux.fr/vol8ns/maloka.pdf [06.05.08].

Mamoru, M. 2001. Japanese Guidelines for Universal Design in Parks: harmony between nature and people. pp. 21.1-21.12 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Manjra, S. 2005. Essay: Manufacturing categories – the case of disabled athletes. *The Lancet.* **366**(Supplement 1): S8-S9.

Manzini, E. 2006. *Design, Ethics and Sustainability: guidelines for a transition phase*. Ezio Manzini's Blog (August 2006): <u>http://sustainable-everyday.net/manzini/?p=14</u> [13 May 2007].

Manzini, E. 2007. The Scenario of a Multi-local Society: creative communities, active networks and enabling solutions. pp. 78-93 in: J. Chapman & N. Gant (eds), *Designers, Visionaries and Other Stories: A Collection of Sustainable Design Essays*. London: Earthscan.

Manzini, E. 2009. Service Design in the Age of Networks and Sustainability. pp. 44-57 in: S. Miettinen and M. Koivisto (eds), *Designing Services with Innovative Methods*. Helsinki: Kuopio Academy of Design.

Maree, K. & van der Westhuizen, C. 2007. Planning a research proposal. pp. 23-45 in: K. Maree (ed), *First Steps in Research*. Pretoria: Van Schaik Publishers.

Margolin, V. 1996. Global expansion or global equilibrium? Design and the world situation. *Design Issues.* **12(**2): 22-23.

Margolin, V. 2007. Design for Development: towards a history. *Design Studies.* **28**: 111-115.

Marindo, R. 2008. Foreign-born and non-citizen populations in the Western Cape province (1996-2006): a demographic overview. pp. 151-173 in: R. Marindo, C. Groenewald & S. Gaisie (eds), *The state of the Population in the Western Cape Province.* Cape Town: Human Sciences Research Council (HSRC) Press.

Marud, M. 2008. 'Our trains will confuse tourists': challenge to find platforms or know which station you are at. Cape Argus Classifieds. Monday, 22 December 2008: p.1.

Marx, K. 1972. Machinery and Modern Industry. pp.151-170 in: B.R. Cosin (ed). *Education: Structure and Society*. Middlesex: Penguin/The Open University.

Masakhwe, P.W. 2004 African Conflicts and the Disability Toll. *Disability World*. **24**: June-August 2004. <u>http://www.disabilityworld.org/06-08_04/gov/conflicts.shtml</u> [30 June 2006].

Matheson, V.A. 2006. *Mega-Events: the effects of the world's biggest sporting events on local, regional, and national economies.* Worcester: College of the Holy Cross. <u>http://www.holycross.edu/departments/economics/website</u> [12 July 2007].

Matheson, V.A. & Baade, R.A. 2004. *Mega-Sporting Events in Developing Nations: playing the way to prosperity?* College of the Holy Cross: Working Paper No. 04-04. <u>http://www.holycross.edu/departments/economics/website</u> [12 July 2007].

Matsebula, S., Schneider, M. & Watermeyer, B. 2006. Integrating Disability within Government: the Office on the Status of Disabled Persons. pp. 85-92 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: A South African Agenda.* Cape Town: HSRC Press.

Matsebula, S. 2009. Email correspondence with author. 20 May 2009. [Email correspondence in possession of author].

Mazrui, A. A. 1999. The African Renaissance: a triple legacy of skills, values and gender. *Proceedings of the African Renaissance: From Vision to Reality Conference*, 23 November 1999, The Barbican Centre, London.

Mbeki, M. 2009. *Architects of Poverty: why African capitalism needs changing.* Johannesburg: Picador Africa.

Mbigi, L. 1997. *Ubuntu: the African dream in management.* Randburg: Knowledge Resources.

Mbigi, L. 2000. *In Search of the African Business Renaissance: an African cultural perspective.* Randburg: Knowledge Resources.

McCallum, K. 2008. *SA celebrates a gold rush.* Cape Argus Sport: Thursday, 11 September 2008: p. 24.

McGuire, J.M., Scott, S.S. & Shaw, S.F. 2006. Universal Design and its Applications in Educational Environments. *Remedial and Special Education*. **27**(3):166-175, May/June 2006.

McKenzie, K. 2008. Momentum grows for Bus Rapid Transit. *Mobility Magazine*: June/August 2008. pp. 14-16.

McWilliams, P.A. 1984. *Personal Computers and the Disabled*. USA: Quantum Press/Doubleday.

Microsoft® Encarta® 2009. English Dictionary © 1993-2008 Microsoft Corporation.

Miyake, Y. 2001. Landscape Design. pp. 48.1-48.24 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Moalosi, R. 2007. *The Impact of Socio-cultural Factors upon Human-centred Design in Botswana.* Queensland University of Technology: unpublished doctoral thesis.

Moggridge, B. 1993. Design by Story-Telling. *Applied Ergonomics.* 24(1): 15-18.

Mokdad, M. 2005. Educational ergonomics: applying ergonomics to higher education institutions. In: A. Thatcher, J. James & A. Todd (eds). *Proceedings of CybErg 2005. The Fourth International Cyberspace Conference on Ergonomics.* Johannesburg: International Ergonomics Association Press.

Momoza, P. 2009. Impact of relevant acts, policies, and programmes governing protective workshops and small businesses in South Africa. *Plenary presentation at the DWDE International Small Business Conference 2009,* Cape Town international Conference Centre, 6-7 April 2009. Cape Town.

Moore, P.A. 2001. Experiencing Universal Design. pp. 2.1-2.12 in: W.P.E. Preiser and E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Morgan, B. 2008. *Paralympic Team South Africa rocks!* SouthAfrica.info. 17th September 2008. <u>www.southafrica.info/news/sport/paralympics2008.htm</u> [01 October 2008].

Morkel, A.T. 2007. Telephonic interviews and email response to author's questions. 17 July- 7 August 2007, Cape Town. [Notes and email correspondence in possession of author].

Mueller, J. 2003 Universal products in the US. pp. 318-335 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Mueller, J.L. 2001. Office and Workplace Design. pp. 45.1-45.11 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Mulder, R. 2008. Semi-structured interview with author at Karee Bloem Lodge. Pretoria: 29 July 2008. [Notes in possession of author].

Mursu, A. 2002. Information systems development in developing countries. Risk management and sustainability analysis in Nigeria software companies. University of Jyvsäkylä.

Mursu, A., Luukkonen, I., Toivanen, M. & Korpela, M. 2007. Activity theory in information systems research and practice theoretical underpinnings for an information systems development model. *Information Research*. **12**(3) paper 311. <u>http://InformationR.net/ir/123/paper311.html</u> [11 January 2008].

Murungi, H. 2003. *Investigating visual communication in relation to graphical form using symbols for the International Red Cross and Red Crescent movement.* De Montfort University: unpublished doctoral thesis.

Næss, I.R. 2003. *Designing inclusive, satisfactory products: a holistic approach to inclusive design.* <u>http://design2.maskin.ntnu.no/fag/PD9/2003/artikkel/Naess.pdf</u> [7 July 2006].

Nardi, B. A. 1999. Activity Theory and Human – Computer Interaction. In Nardi, B. A. (ed), *Context and Consciousness.* Cambridge: MIT Press.

Nattrass, N. 2007. Disability and welfare in South Africa's era of unemployment and AIDS. pp. 179-200 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation.* Cape Town: HSRC Press.

Nieuwenhuis, J. 2007. Introducing Qualitative Research. pp. 46-68 in: K. Maree (ed), *First Steps in Research*. Pretoria: Van Schaik Publishers.

Nichols, P., Haworth, R. & Hopkins, J. 1981. *Disabled: an illustrated manual of help and self-help.* London: David & Charles Publishers Ltd.

Norman, D.A. 1998. The Design of everyday things. New York: Doubleday.

Norman, D. A. 2004. *Emotional Design: why we love (or hate) everyday things.* New York: Basic Books.

Obrenović, Z., Abascal, J. & Starčević, D. 2007. Universal Accessibility as a Multimodal Design Issue: creating adaptable solutions to address diverse users and situations. *Communications of the ACM.* **50**(5): 83-88.

ODP (Office of the Deputy President). 1997. White Paper on an Integrated National Disability Strategy.

http://www.polity.org.za/html/govdocs/white_papers/disability1,html?rebookmark=1 [09 June 2004].

Oelofsen, R. A. 2008. Response to semi-structured questions from the author. Cape Town: 05 September 2008. [Notes and email correspondence in possession of author].

O'Herlihy, E. 2005. *Access Handbook Template: a tool to help manage accessibility of the built environment.* Dublin: National Disability Authority. <u>www.nda.ie</u> [10 March 2008].

Oliver, M. 1990. The Politics of Disablement. New York: Palgrave Macmillan.

Oliver, M. 1996. Understanding Disability: from theory to practice. New York: Palgrave.

Olivier, C. 1998. *How to Educate and Train Outcomes-Based.* J.L. van Schaik Publishers, Pretoria.

Ongolo, T. 2007. Semi-structured interview with author at African Decade of People with Disabilities (ADPD) Secretariat. Cape Town: 28 March 2007. [Notes, voice recording and transcribed responses in possession of author].

Opperman, J. 2006. Semi-structured interview with author at Claremont. Cape Town: 2 February 2006. [Notes in possession of author].

Oravec, J.O. 2002. Virtually accessible: empowering students to advocate for accessibility and support universal design. *Library Hi Tech.* **20**(4): 452-461.

Ormerod, M.G. & Newton, R.A. 2005. Briefing for accessibility in design. *Facilities*. **23**(7/8): 285-294.

OSDP. 2003. *Discussion Paper on NEPAD and the Disability Sector*. Pretoria: Office on the Status of Disabled Persons (April 2003).

Osiris. n.d. *Activity Theory – and introduction.* www.osiris.sund.ac.uk/~cs0car/hci/3 con at.htm [9 August 2009].

Ostroff, E. 2003. International Design Education Strategies. pp. 336-355 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population*. London: Springer.

Owen, C. 2004. *Responsible Design. Achieving living excellence: implications, warnings and a call to action.* Institute of Design, Illinois Institute of Technology. <u>www.id.iit.edu</u> [10 March 2007].

Owen, K. & Johnston, M. 2003. Lifestyle, Design and Disability. pp. 59-69 in: J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Padovan, D. 2008. Social capital, lifestyles and consumption patterns. pp. 271-287 in: A. Tukker, M. Charter, C. Vezzoli, E. Stø & M.M. Andersen (eds), *System Innovation for Sustainability: perspectives on radical changes to sustainable consumption and production.* Sheffield (UK): Greenleaf Publishing Ltd.

Papanek, V. 1971. *Design for the Real World: human ecology and social change*. Norwich: Thames and Hudson Ltd.

Papanek, V. 1995. *The Green Imperative: ecology and ethics in design and architecture.* London: Thames and Hudson.

Parker, K.J. 2001. Developing Economies: a reality check. pp. 32.1-32.11 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Pasha, S. & Pasha, M.A. 2006. *WHO-ICF and the Current State of Affairs*. Center of Assistive & Rehabilitative Technologies, Southampton, UK: CARTUK Research Report Series.

Peters, S. 2000. Is there a disability culture? A syncretisation of three possible world views. *Disability & Society*. **15**(4): 583-601.

Pezeshki, K. 2008. Response to semi-structured questions from the author. Cape Town: 1 January 2008. [Notes and email correspondence in possession of author].

Pheasant, S. 1998. *Bodyspace: anthropometry, ergonomics and the design of work.* 2nd ed. London: Taylor & Francis.

Phiri, A.M. 2006. *African Decade (2000-2009) …the challenge of the decade.* SAFOD (South African Federation of the Disabled). http://safod.org/African%20Decade/african_decade.html [30 June 2006].

Pirkl, J. J. 1994. *Transgenerational Design: Products for an Aging Population.* New York: Van Nostrand Reinhold.

Platzky, L. 2007. From white elephant to roaring success: the city can learn from other countries how to market the 2010 stadium. Cape Argus: Tuesday, 20 November 2007: p 15.

Popovic, V. 1999. Product evaluation methods and their importance in designing interactive artefacts. pp. 26-35 in: W.S. Green & P.W. Jordan (eds). *Human Factors in Product Design: Current Practice and Future Trends*. London: Taylor & Francis.

Population Reference Bureau. 2008. 2008 World Population Data Sheet. <u>www.prb.org</u> [21 April 2009].

Powell, A. 2007. *We're Still SA's Murder Capital.* Cape Times. Wednesday, 4 July 2007: p.1.

Prahalad, C.K. 2005. *The Fortune at the Bottom of the Pyramid.* Upper Saddle River, NJ: Wharton School Publishing.

Preiser, W.P.E. & Ostroff, E. (eds). 2001. *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Prince, N. 2006. *Call to clean up graffiti on trains before 2010.* Cape Argus. Wednesday, 25 October 2006: p. 3.

Raubal, M. 2001. Human wayfinding in unfamiliar buildings: a simulation with cognizing agents. *Cognitive Processing*. (2/3): 363-388.

Raubal, M. 2007. *Wayfinding: affordances and agent simulation.* <u>www.geog.ucsb.edu/~raubal/Publications/RevBookSections/Wayfinding_Affordances_Agents_Raubal_rev.pdf</u> [27July 2007].

Read, M.A. 2003. Use of color in child care environments: application of color for wayfinding and space definition in Alabama child care environments. *Early Childhood Education Journal*. **30**(4): 233-239.

Ricability. n.d. *Meeting the needs of older and disabled consumers: guidelines for product design and testing*. <u>http://www.ricability.org.uk/reports/report-</u><u>design/guidelinesforproductdesign/forward</u> [2 June 2006].

Roaf, V., van Deventer, K. & Houston, C. 1996. *The Olympics and Development: lessons and suggestions.* Observatory: Development Action Group.

Robb, G.M. 2001. Guidelines for Outdoor Areas. pp. 20.1-20.22 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Rose, D.H. & Strangman, A.N. 2007. Universal Design for Learning: meeting the challenge of individual learning differences through a neurocognitive perspective. *Universal Access Information Society.* **5**: 381-391.

Ross, C. 2009. *2010 lacks 'green' strategy: SA could miss out on funding and support.* Weekend Argus. Saturday, 7 February 2009: p.4.

Rowland, W. P. 1984. *A Phenomenological Analysis of Blindness and a Formulation of New Objectives in Rehabilitation.* University of Cape Town. Unpublished PhD thesis.

Rubin, M. 2009. The offside rule: women's bodies in masculinised spaces. pp. 266-280 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Runhaar, J. 2002. *Imageability*. <u>www.urban.uiuc.edu/courses/up426/Fall02/Reports/Files/1326Team3imgr.pdf</u> [17 July 2007].

SABS (South Africa Bureau of Standards) Design Institute. www.designinstitute.org.za

SACN see South African Cities Network.

SA Good News see South Africa: The Good News.

SAHRC Report. 2002. *Towards a Barrier-Free Society: A Report on Accessibility and Built Environments.* Pretoria: South African Human Rights Commission: November 2002.

SAInfo. 2004. *What FIFA had to say about SA.* 5 May 2006. http://www.southafrica.info/2010/.htm [27 June 2006].

SAInfo. 2006a. *2010 a Record Earner for FIFA*. 18 July 2006. http://www.southafrica.info/2010/2010-revenue-130706.htm [27 June 2006].

SAInfo. 2006b. SA Airports Gear Up for *2010.* 21 July 2006. http://www.southafrica.info/2010/airports-210706.htm [27 June 2006].

SAMA. 2006. *Professional Standards and Transformational Indicators*. Port Alfred: South African Museums Association.

Samson, M., MacQuene, K. & van Niekerk, I. 2005. *Social grants: South Africa.* Overseas Development Institute: Policy Brief #1 of the Inter-Regional Inequality Facility. <u>www.odi.org.uk/inter-regional_inequality</u> [22 April 2008].

SANBI. n.d. *Kirstenbosch National Botanical Garden, South Africa*. South African National Biodiversity Institute, Cape Town. <u>http://www.sanbi.org/kirstenbosch/mainpage.htm</u> [23 July 2007].

Sandhu, J. 1993. Design for the elderly: user-based evaluation studies involving elderly users with special needs. *Applied Ergonomics.* **24**(1): 30-34.

Sandhu, J.S. 2001. An integrated approach to Universal Design: towards the inclusion of all ages, cultures, and diversity. pp. 3.1-3.14 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Sandhu, J.S. 2002. *Some Contextual Considerations When Designing for the Majority World*. <u>http://www.udeducation.org/teach/course_mods/sandhu/index.asp</u> [3 June 2006].

SAQA. 2005a. South African Qualifications Authority, Registered Qualification: National Certificate: Three Dimensional Design. http://reggs.saga.org.za/viewQualification.php?id=2083 [17 November 2005].

SAQA. 2005b. South African Qualifications Authority, Registered Qualification: National Higher Certificate: Three Dimensional Design. <u>http://reggs.saqa.org.za/viewQualification.php?id=2087</u> [17 November 2005]. Schneider, H., Barron, P. & Fonn, S. 2007. The promise and practice of transformation in South Africa's health system. pp. 289-311 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation*. Cape Town: HSRC Press.

Schneider, M. 2006. Disability in the Environment. pp. 8-18 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: a South African Agenda*. Cape Town: HSRC Press.

Schroeder, F. 2009. Breaking records – again. Cape Argus. Monday, 27 April 2009: p.3.

Schumacher, E. F. 1973. *Small Is Beautiful: economics as if people mattered.* London: Blond & Briggs.

Schwandt, T.A. 2001. *Dictionary of Qualitative Inquiry.* 2nd ed. Thousand Oaks: Sage Publications, Inc.

Shapira, N.H. 1991. Design or Decline: America facing the 21st century. *Art Education*. **44**(5): 20-21.

Sherer, G.2000. Intergroup economic inequality in South Africa: the post-apartheid era. *The American Economic Review*. **90**(2): 317-321.

Shutte, A. 1993. Philosophy for Africa. Cape Town: UCT Press.

Silverman, D. 1993. *Interpreting Qualitative Data: Methods for Analysing Talk, Text, and Interaction.* London: Sage Publications Ltd.

Sinclair, M.A. 1990. Subjective Assessment. pp. 58-88 in: J.R. Wilson & E.N. Corlett (eds), *Evaluation of Human Work: A Practical Ergonomics Methodology.* London: Taylor & Francis.

Sinclair, N.R. 1998. Disabling environments: a review of Oliver's Understanding Disability: From Theory to Practice. *Journal of Applied Behavior Analysis.* **31**(3): 509-510.

Singer, L. & Ritz, J.M. 1996. Wayfinding: symbols and signs that send us our way. *The Technology Teacher.* November 1996: 16-20.

Sixsmith, J. & Sixsmith, A. 1993. Older people, driving and new technology. *Applied Ergonomics.* **24**(1): 40-43.

Siyabulela, K. & Duncan, M. 2006. Psychiatric Disability and Social Change: An Insider Perspective. pp. 291-310 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: A South African Agenda*. Cape Town: HSRC Press.

Sjöberg, C. 1996. *Activities, Voices, and Arenas: Participatory Design in Practice.* Linköping: Linköping Studies in Science and Technology Dissertation No. 49.

Smith, B. 2008. Telephonic and electronic responses to semi-structured questions from the author. Cape Town: 19 September 2008. [Correspondence in possession of author].

Smith, J. 2009. *Fear travels with SA's rail and taxi commuters: a daily gauntlet of robberies, assaults and lethal accidents.* Weekend Argus. Saturday, 7 February 2009: p.13.

Smith, J. & Spurling, A. 1999. Lifelong Learning: Riding the Tiger. Cassell: London.

Snaddon, B. 2008. Semi-structured interview with author at Department of Graphic Design offices, CPUT. Cape Town: 15 August 2008. [Notes, voice recording and transcribed responses in possession of author].

Soares, M.M. 2005. Ergonomics and design: a user-centred design method. *Journal of the Ergonomics Society of South Africa*. **17**(2): 31-41.

South Africa. 1988. *Development Guide Plan: Kirstenbosch Botanic Gardens, November 1988.* Pretoria: Department of Public Works and Land Affairs.

South Africa. 1996. *Constitution of the Republic of South Africa, Act No. 108 of 1996*. Pretoria: Government Gazette.

South Africa. 1997a. Department of Education. *Higher Education Act, No. 101 of 1997*. Pretoria: Government Gazette.

South Africa. 1997b. Department of Public Service and Administration. *The White Paper on Transforming Public Service Delivery (Batho Pele), No. 1459 of 1997.* Pretoria: Government Gazette.

South Africa.1998. Department of Labour. *Employment Equity Act, no. 55 of 1998.* Pretoria: Government Gazette.

South Africa. 2000. *Promotion of Equality and the Prevention of Unfair Discrimination Act, No. 4 of 2000.* Pretoria: Government Gazette.

South Africa. 2001. Department of Education. *White Paper No. 6 Special Needs Education: Building an Inclusive Education and Training System.* Pretoria: Department of Education.

South Africa. 2002. Department of Science and Technology. *South Africa's National Research and Development Strategy.* Pretoria: Department of Science and Technology.

South Africa. 2003a. Draft White Paper on e-Education. Pretoria: Department of Education.

South Africa. 2003b. *Technical Assistance Guidelines on the Employment of People with Disabilities.* Pretoria: Department of Labour.

South Africa. 2003c. *Broad-Based Economic Empowerment Act, no 53 of 2003.* Pretoria: Department of Labour.

South Africa. 2004a. *The Higher Education Qualifications Framework*. Policy issued under the Higher Education Act, No. 101 of 1997. Draft for discussion (July 2004). Pretoria: Department of Education.

South Africa. 2004b. Department of Public Service and Administration. *Batho Pele Handbook: a service delivery improvement guide.* Pretoria: Department of Public Service and Administration.

South Africa. 2008a. *An introduction to the Department of Environmental Affairs and Tourism.* Pretoria: Department of Environmental Affairs and Tourism.

South Africa. 2008b. *Strategic Plan: 01 April 2008 to 31 March 2011.* Pretoria: Department of Environmental Affairs and Tourism.

South Africa. 2008c. *Strategic Plan: 2008 to 2011.* Pretoria: Department of Social Development.

South Africa. 2008d. Policy on Disability. Pretoria: Department of Social Development.

South Africa. n.d. *The Operationalisation of the AU and the Implementation of NEPAD Programmes*. Pretoria: Department of Foreign Affairs.

South Africa: The Good News. 2008. *Cape Town named as one of world's sustainable cities*.

http://www.sagoodnews.co.za/environment/cape_town_named_as_one_of_worlds_sustain able_cities.html [16 September 2008].

SouthAfrica.info. 2008. *Helen Zille named world's best mayor*.15 October 2008. www.southafrica.info/about/democracy/**zille-mayor**2008.htm [22 October 2008].

South African Cities Network. 2005. *The Implications of Hosting International Big Events on Municipal Governance and City Economic Footprint.* South African Cities Network. www.sacities.net/2005/world_cup_seminar.doc [13 July 2007].

South African Cities Network. 2008. *Inclusive Cities 2008.* Braamfontein: South African Cities Network.

South African Government Information. 2008. *Social Development*. 06 November 2008. <u>http://www.info.gov.za/aboutsa/socialdev.htm#Act</u> [7 May 2009].

South African National Standard: SANS 500081-70:2004: Safety Rules for the Construction and Installation of Lifts- Particular Applications for Passenger and Goods Lifts; Part 70: Accessibility to Lifts for Persons Including Persons with Disability. Pretoria: Standards South Africa.

South African Standard: SABS 0246:1993: *Accessibility of Buildings to Disabled Persons*. Pretoria: South African Bureau of Standards.

South African Standard: SABS 0400:1990: *Part S: Facilities for Disabled Persons*. Pretoria: South African Bureau of Standards.

Stammers, R.B., Carey, M.S. & Astley, J.A. 1990. Task Analysis. pp. 134-160 in: J.R. Wilson and E.N. Corlett (eds), *Evaluation of Human Work: A Practical Ergonomics Methodology.* London: Taylor & Francis.

Stats SA (Statistics South Africa). 2002. *Census in Brief, 2001.* Pretoria: Stats SA. <u>http://www.census.gov/ipc/prod</u> [17 September 2006].

Stats SA (Statistics South Africa). 2005a. *Census 2001: Prevalence of disability in South Africa*. Pretoria: Stats SA.

Stats SA (Statistics South Africa). 2005b. *Mid-year population estimates, South Africa.* Pretoria: Stats SA.

Stats SA (Statistics South Africa). 2006. *Tourism and migration, April 2006.* Pretoria: Stats SA.

Steinfeld, A. & Steinfeld, E. 2001. Universal design in automobile design. pp. 50.1-50.13 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Steinfeld, E. 2001. Universal Design in mass transportation. pp. 24.3-24.25 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Steyl, L. 2008a. *IRT on hold for now.* Milnerton TygerBurger: Wednesday, 5 November 2008: p. 1.

Steyl, L. 2008b. *Traffic nerve centre ready for 2010 and beyond*. Milnerton TygerBurger: Wednesday, 5 November 2008: p. 9.

Stirling, A. n.d. *On the Economics and Analysis of Diversity.* Brighton: Science Policy Research Unit (Electronic Working Papers Series Paper No. 28).

Story, M.F. 2001. Principles of Universal Design. pp. 10.3-10.19 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Strauss, A. & Corbin, J. 1997. Grounded Theory in Practice. London: Sage Publications.

Swart, K. & Bob, U. 2009. Venue selection and the 2010 World Cup: a case study of Cape Town. pp. 114-130 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Swart, K., Wang, Y., Turco, D.M., Vollmer, R., Knott, B., Parish, C. & McCrae, D. 2008. *Universal Accessibility (UA) Strategy for the Western Cape 2010 and Beyond* (Final Draft). The Provincial Government of the Western Cape Department of Economic Development and Tourism: March 2008.

Swartz, L., Schneider, M. & Rohleder, P. 2006. HIV/AIDS and Disability: New Challenges. pp. 108-115 in: B. Watermeyer, L. Swartz, T. Lorenzo, M. Schneider & M. Priestly (eds), *Disability and Social Change: A South African Agenda.* Cape Town: HSRC Press.

Takyi, B. K. & Oheneba-Sakyi, Y. 2006. The Study of African Families: Concluding Remarks. pp. 273-278 in Y. Oheneba-Sakyi & B.K. Takyi (eds), *African Families at the Turn of the 21st Century*. Praeger Publishers, Westport, CT.

Tambulasi, R. & Kayuni, H. 2005. Can African Feet Divorce Western Shoes? The Case of 'Ubuntu' and Democratic Good Governance in Malawi. *Nordic Journal of African Studies.* **14**(2): 147-161.

Tapela, N. 2008. Response to semi-structured questions from the author. Cape Town: 12 September 2008. [Notes and email correspondence in possession of author].

Taylor, A.J., Roberts, P.H. & Hall, M.J.D. 1999. Understanding person product relationships – a design perspective. pp. 218-228 in: W.S. Green & P.W. Jordan (eds). *Human Factors in Product Design: Current Practice and Future Trends*. London: Taylor & Francis.

Thackara, J. 2005. *In the bubble: designing in a complex world.* Cambridge, MA: MIT Press.

Thapar, N., Warner, G., Drainoni, M., Williams, S.R., Ditchfield, H., Wierbicky, J. & Nesathurai, S. 2004. A pilot study of functional access to public buildings and facilities for persons with impairments. *Disability and Rehabilitation*. **26**(5): 280-289.

Theofanos, M.F. & Redish, J. 2003. Guidelines for accessible – and usable – web sites: observing users who work with screenreaders. *Interactions.* **10**(6): 38-51.

Theron, T. 2005. Public participation as a micro-level development strategy. pp. 111-131 in: I. Davids, F. Theron & K. Maphunye (eds), *Participatory development in South Africa*. Pretoria: Van Schaik Publishers.

Thomas, A. 2006. Design, Poverty and Sustainable Development. *Design Issues.* **22**(4): 54-65.

Thompson, C. 2008. Semi-structured interview with author at Integrated Design Consultants (IDC) offices. Claremont: 20 October 2008. [Notes, voice recording and transcribed responses in possession of author].

Thompson, P. 2006. Semi-structured interview with author at Victoria Junction Protea Hotel lobby. Greenpoint: 2 May 2006. [Notes in possession of author].

Thompson, P. 2008. Semi-structured interview with author at Integrated Design Consultants (IDC) offices. Claremont: 20 October 2008. [Notes, voice recording and transcribed responses in possession of author].

Tomlinson, R. 2009. Anticipating 2011. pp. 96-113 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Tomlinson, R., Bass, O. & Pillay, U. 2009. Introduction. pp. 3-17 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

Tiresias. 2006. *Guidelines: Inclusive Design.* <u>http://www.tiresias.org/guidelines/inclusive.htm [19</u> February 2007].

Tisani, N. 2004. African indigenous knowledge systems (AIKSs): Another challenge for curriculum development in higher education? *South African Journal of Higher Education, SAJHE*. **18**(3): 174-184.

Tisani, N. 2008. Semi-structured interview with author at the Registrar's office, CPUT. Belllville: 20 March 2008. [Notes, voice recording and transcribed responses in possession of author].

Toft, Y. & Joubert, D. 2005. 'Transdisciplinarity' a new opportunity for team problem solving between, across and beyond the disciplines? – A case study. In A. Thatcher, J. James & A. Todd (eds), Proceedings of CybErg 2005. *The Fourth International Cyberspace Conference on Ergonomics*. Johannesburg: International Ergonomics Association Press. pp. 609-616.

Transgenerational Design Matters. 2006. *Transgenerational Design*. <u>http://www.transgenerational.org/</u> [7 August 2006].

Tsutatani, K. 2005. Introduction to International Association for Universal Design, *Fujitsu Science and Technology Journal.* **41**(1): 47-53.

Tutu, D. 1999. No Future Without Forgiveness. London: Rider.

TygerBurger. 2008a. *Stadium to meet deadline.* The TygerBurger: Wednesday, 2 April 2008: p10.

TygerBurger. 2008b. *City to hold IRT meetings.* The TygerBurger: Wednesday, 22 October 2008: p16.

UN see United Nations.

UNAIDS. 2008. *AIDS Epidemic Update: August 2008.* <u>http://www.unaids.org/en/KnowledgeCentre/HIVData/GlobalReport/2008/2008 Global rep</u> <u>ort.asp</u> [23 January 2009].

UNDP. 2005. Human development report. New York: UNDP.

UNESCO. 1999. *Provision for Students with Disabilities in Higher Education: a survey.* Paris: UNESCO.

UNESCO. 2003a. Open File on Inclusive Education: support materials for managers and administrators. Paris: UNESCO.

UNESCO. 2003b. Overcoming Exclusion through Inclusive Approaches in Education: a challenge and a vision. Conceptual Paper. Paris: UNESCO.

UNESCO. 2005. Operational Guidelines for the Implementation of the World Heritage Convention. World Heritage Centre, Paris, 2 Feb. 2005. http://whc.unesco.org/eng/guidelines (English) [25 July 2007].

UNESCO. 2007. *Guidelines and Criteria for National Geoparks seeking UNESCO's assistance to join the Global Geoparks network*. UNESCO, January 2007. www.unesco.org/science/earth/geoparks/2007guidelinesJanuary.pdf [25 July 2007].

UNFPA. 2008. International Day of Older Persons. United Nations Population Fund (UNFPA): <u>http://www.unfpa.org/news/news.cfm?ID=1191</u> [3 October 2008].

Unicity Commission. 2001. *Building a Unified City for the 21st Century: A summary of the proposed service delivery and institutional change proposals for the term of office of the new City of Cape Town.* Cape Town: Unicity Commission. <u>www.unicomcapetown.gov.za</u> [7 July 2007].

United Nations. 2003. The UN and Persons with Disabilities. New York: United nations

United Nations. 2005. *The Millennium Development Goals Report.* New York: United Nations.

United Nations Enable. 2004a. *African Decade of Disabled Persons (2000-2009)*. <u>http://www.un.org/esa/socdev/enable/disafricadecade.htm</u> [30 June 2006].

United Nations Enable. 2004b. *Standard Rules on the Equalization of Opportunities for People with Disabilities*. <u>http://www.un.org/esa/socdev/enable/dissre04.htm</u> [9 May 2007].

Universal Design Education Online. 2006. <u>http://www.udeducation.org/learn/index.asp</u> [2 July 2006].

USAID. 2005. *Living with Disabilities in Africa*. <u>http://www.usaid.gov/locations/sub-saharan_africa/features/disabilities.html</u> [30 June 2008].

US Census Bureau. 2009. *International Data Base*. <u>http://www.census.gov/cgi-bin/ipc/idbpyrs.pl?cty=SF&out=s&ymax=300&submit=Submit+Query</u> [12 April 2009].

van Aalast, J. & Hill, C.M. 2006. Activity Theory as a framework for analysing knowledge building. *Learning Environment Research*. **9**: 23-44.

Vanderheiden, G. & Tobias, J. 1998. *Barriers, incentives and facilitators for adoption of universal design practices by consumer product manufacturers.* <u>http://trace.wisc.edu/docs/hfes98 barriers/barriers incentives facilitators.htm</u> [22August 2007].

Vanderheiden, G. 2001. Fundamentals and priorities for design of information and telecommunication technologies. pp. 65.3-65.15 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

van der Merwe, J. 2009. The road to Africa: South Africa's hosting of the 'African' World Cup. pp. 18-32 in U. Pillay, R. Tomlinson & O. Bass (eds), *Development and Dreams: the urban legacy of the 2010 Football World Cup*. Cape Town: HSRC Press.

van der Merwe, J.J. 2008. Response to semi-structured questions from the author. Cape Town: 08 September 2008. [Notes and email correspondence in possession of author].

van Niekerk, J. & M'Rithaa, M.K. 2008. The Rhyme and Reason of Ethical Design Practices: steering designers toward sustainable solutions. *Proceedings of the CUMULUS Conference on Design and Research*, 20-23 November 2008. Saint-Etienne.

van Niekerk, J. & M'Rithaa, M.K. 2009. The Ethical Dilemma of a Rapidly Receding Watering Hole: implications for design education. *Proceedings of the Design Education Forum of Southern Africa (DEFSA)* 12th National Conference, 4-5 November 2009, Graaff Reinet.

Varela, F. J. 1999. Ethical Know-How. California: Stanford University Press.

Vermeulen, J. & M'Rithaa, M.K. 2007. From the Grave to the Cradle: the eco-design case for the re-evaluation of hemp. *Proceedings of the Design Education Forum of Southern Africa (DEFSA) International Conference*, Cape Peninsula University of Technology, 3-5 October 2007. Cape Town.

Verster, B. 2008. Email correspondence with author. Cape Town: 12 September 2008. [Email correspondence in possession of author].

Vetten, L. 2007. Violence against women in South Africa. pp. 425-447 in: S. Buhlungu, J. Daniel, R. Southall & J. Lutchman (eds), *State of the Nation.* Cape Town: HSRC Press.

Vezzoli, C. 2007. System Design for Sustainability. Theory, methods and tools for a sustainable "satisfaction-system" design. Rimini: Maggioli Editore.

Vezzoli, C. & Ceschin, F. 2008. Designing sustainable system innovation transition for lowindustrialised contexts: a transition path towards local-based and long lasting sustainable mobility solutions in African contexts. *Proceedings of the Sustainable Consumption and Production: Framework for Action, Conference of the Sustainable Consumption Research Exchange (SCORE!) Network*, supported by the EU's 6th Framework Programme, 10-11 March 2008. Brussels.

Vezzoli, C. & Manzini, E. 2008. *Design for Environmental Sustainability*. London: Springer-Verlag.

Volbrecht, T. 2003. The story of South African academic development in international perspective: have we lost the plot? *South African Journal of Higher Education, SAJHE/SATHO*. **17**(2): 110-117.

Vos, J. 2008. Semi-structured interview with author at Councillor Vos' office. Goodwood: 22 August 2008. [Notes, voice recording and transcribed responses in possession of author].

Vrooman, R.J. 2007. Regional wayfinding systems provides seamless signage for Historic Triangle visitors. *Virginia Town & City.* **42**(2): 8-10.

Vuk'uzenzele. 2006. South Africa's promise: 'we will deliver!' GCIS. 1(6): June 2006. [01 September 2006].

wa ka Ngobeni, W. 2009. *New wheels could be on track for Metrorail.* Sunday Times (Business Times). Saturday, 25 January 2009: p.3.

Walsh, C.J. 2001. Sustainable human and social development: an examination of contextual factors. pp. 33.1-33.16 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Wanger, C. & Burnette, J. 2003. *Applications of Participatory Action Research with Students who Have Disabilities*. <u>http://ericec.org/digests/e646.html</u> [25 May 2006].

Watermeyer, C. 2002. *Physical access and the UCT built environment.* UCT Disability Unit. Cape Town: University of Cape Town.

Weber, R.P. 1990. Basic Content Analysis. London: Sage Publications.

Weisman, L.K. 1994. *Discrimination by Design: a feminist critique of the man-made environment.* Urbana & Chicago: University of Illinois Press.

Welch, P. & Palames, C. 1995. *A brief history of disability rights legislation in the United States.* Universal Design Online. <u>http://www.udeducation.org/resources/readings/welch.asp</u> [26 June 2006].

Welch, P. & Jones, S. 2001. Advances in Universal Design Education in the United States. pp. 51.3-51.24 in W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Wells, R., Norman, R., Frazer, M., Laing, A., Cole, D. & Kerr, M. 2003. *Participative ergonomic blueprint.* Toronto: Institute of Work & Health. <u>www.iwh.on.ca</u> [4 May 2007].

Werner, D. 1998. Nothing about us without us: developing innovative technologies for, by and with disabled persons. Palo Alto, CA: Healthwrights.

Western Cape. 2004. *Celebrating disability in our decade of democracy: 1994-2004*. Cape Town: Provincial Administration Western Cape.

Wetzel, T. 2008. Telephonic and electronic responses to semi-structured questions from the author. Cape Town: 01 October 2008. [Notes and email correspondence in possession of author].

Whiteley, N. 1993. Design for Society. London: Reaktion Books Ltd.

WHO see World Health Organization.

Wilkinson, V. 2008. *Stadium architects watch their dream coming true.* Weekend Argus: 08 March 2008: p. 21.

Wright, L. & Hook, W. (eds). 2007. *Bus rapid transit: Planning Guide*. 3rd ed. New York: Institute for Transportation and Development Policy.

WHO see World Health Organization.

Whyte, S.R. & Ingstad, B. 2007. Introduction: Disability Connections. pp. 1-29 in: B. Ingstad & S.R. Whyte, *Disability in Local and Global Worlds*. Berkeley: University of California Press.

Wijk, M. 2001. The Dutch Struggle for Accessibility Awareness. pp. 28.1-28.17 in: W.P.E. Preiser & E. Ostroff (eds), *The Universal Design Handbook*. New York: McGraw-Hill Inc.

Wilson, J., Haines, H. & Morris, W. 2005. Participatory Ergonomics. pp. 933-962 in: J.R. Wilson & N. Corlett (eds), *Evaluation of Human Work.* 3rd ed. Boca Raton: Taylor & Francis.

Wilson, J.R. 1990. A Framework and a Context for Ergonomics Methodology. pp. 1-30 in: J.R. Wilson & E.N. Corlett (eds), *Evaluation of Human Work: A Practical Ergonomics Methodology*. London: Taylor & Francis.

Winam, R. (ed). 2003. *Disability dimension in development action: manual on inclusive planning.* Helsinki: Stakes (published for The United Nations).

Winberg, C. 2005. Continuities and discontinuities in the journey from technikon to university of technology. *South African Journal of Higher Education, SAJHE.* **19**(2): 189-200.

Winberg, C. 2006. Undisciplining knowledge production: Development driven higher education in South Africa. *Higher Education.* **51**: 159-172.

Winberg, C. 2008. *Introducing OBE*. PowerPoint presentation: Curriculum_04_OBE.ppt

Wood, J. 1993. Simplifying the interface for everyone. Applied Ergonomics. 24(1): 28-29.

World Assembly on Ageing. 2002. New York: United Nations Publications.

World Health Organization. 2001. *International classification of functioning, disability and health: report by the Secretariat.* Provisional agenda item 13.9: 9 April 2001. Geneva: WHO Press.

World Health Organization. 2002. *Towards a Common Language for Functioning, Disability and Health: ICF The International Classification of Functioning, Disability and Health.* Geneva: WHO Press.

World Health Organization. 2004. *Family of International Classifications: definition, scope and purpose.* August 2004. Geneva: WHO Press.

World Health Organization. 2007. World Health Statistics 2007. Geneva: WHO Press.

The World of Universal Design. http://www.ap.buffalo.edu/idea/wud/ [4 September 2006].

Woudhuysen, J. 1993. A call for transgenerational design. *Applied Ergonomics.* **24**(1): 44-46.

WUD see The World of Universal Design.

Wyman, L. 2004. Wayfinding systems. *Webesteem Art & Design magazine*. **9**. <u>http://art.webesteem.pl/9/wyman_en.php</u> [24 July 2007].

Yielding, D. 2003. Power to the People. pp. 104-117: in J. Clarkson, R. Coleman, S. Keates & C. Lebbon (eds), *Inclusive Design: Design for the Whole Population.* London: Springer.

Zalk, D.M. 2001. Grassroots Ergonomics: Initiating an Ergonomics Program Utilizing Participatory Techniques. *Annals of Occupational Hygiene*. **45**(4): 283-289.

Zauhar, J. 2004. Historical perspectives of sports tourism. *Journal of Sport Tourism* **9**(1): 5–101.

Zelek, J. S., Bromley, S., Amsar, D. & Thompson, D. 2003. A haptic glove as a tactilevision sensory substitution for wayfinding. *Journal of Visual Impairment & Blindness*. October 2003: 621-632.

Zimmerman, R. 2006. Accessibility for one and all: with ease of use becoming the market standard, universal design can help facility executives meet everyone's needs. *Building Operating Management.* **53**(11): 16-18.

Zmudzinska-Nowak, M. 2003. Searching for legible city form: Kevin Lynch's theory in contemporary perspective. *Journal of Urban Technology*. **10**(3): 19-39.

Zola, I.K. 1994. Communication Barriers between the Worlds of 'Able-Bodiedness' and 'Disability'. pp. 159-164 in: R. Schmitt & T.E. Moody (eds), *Key Concepts in Critical Theory: Alienation and Social Criticism.* New Jersey: Humanities Press.

Zwane, S. 2008. 2010 Green Goal action plan launched. http://www.sa2010.gov.za/node/1105 [17 October 2008].

Images:

Natalie du Toit: http://legendsrevealed.com/sports/wp-content/uploads/2009/07/natalie2.jpg

Oscar Pistorius: http://rawartint.files.wordpress.com/2008/05/pistorius.jpg

APPENDICES

APPENDIX A: Disability Definitions, Models and Terminology (source: DPSA, 2004)

SUMMARY

These guidelines on disability terminology and definitions are taken from "A Pocket Guide on Disability Equity" prepared by Disabled People South Africa (DPSA).

- What is Disability?
- Definitions
- Models
- Terminology

WHAT IS DISABILITY?

"Disability is the disadvantage or restriction of activity caused by a society that takes little or no account of people who have impairments and thus excludes them from mainstream activity" (*British Council of Organisations of Disabled People*).

DIFFERENT APPROACHES TO DEFINING DISABILITY

The definition and classification of disabled persons have gone through a number of changes over the centuries.

Biomedical Definition

Disability is identified with illness or impairment in the biomedical approach, with most emphasis falling on curing the disabled individual. If this fails, the person is removed from society.

Philanthropic Definition

Disability is regarded as a tragedy or object of sympathy and charity. People with disabilities are therefore pitied, given handouts and cared for in separate institutions.

Sociological Definition

This approach defines disability as a form of human difference or deviation from the social norms of the acceptable levels of activity performance.

Economic Definition

Disability is defined as a social cost caused both by extra resources that children and adults with disabilities require and by their limited productivity at work, relative to able-bodied people.

Socio-Political Definition as Adopted by the Integrated National Disability Framework

Disability needs to be defined within context, rather than focussing on the inability of people that inadvertently leads to stigmatisation and categorisation. The Integrated National Disability Framework has therefore adopted a socio-political approach to disability, whereby disability is located in the social environment. This takes cognisance of disabled people's viewpoint that disability is a social construct and most of its effects are inflicted upon people with disabilities by their social environment e.g. it is not the disability, nor the wheelchair that disables a person but it is the stairs leading to a building.

MEDICAL APPROACH VERSUS SOCIAL MODEL

Disabled people during the 1970s used their personal experience of disability and institutional life to show that it wasn't their impairments that caused the problem but the way in which society failed to make any allowances for their differences and instead locked them away.

Medical Model Approach

Traditional approaches say that the inability to carry out activities is caused by impairment or impairments; for example, you are not mobile because you have a spinal injury. This understanding of disability is said to be a medical model of disability because the causes of disability are attributed only to medical conditions.

Social Model Approach

This social model is not limited by such a narrow description of activities. It takes the wider view that the ability to take such activities is dependent upon social intervention. It can show that the limitation of activity is not caused by impairments but is a consequence of social organisation- hence the phrase 'social model'. In short, the social model says that a person is disabled if the world at large will not take into account their physical or mental differences.

PREFERRED TERMINOLOGY

Language reflects the social context in which it is developed and used. It therefore reflects the values and attitudes of that context, and plays an important role in reinforcing values and attitudes that lead to discrimination and segregation of particular groups in society. Language can therefore be used as a powerful tool to facilitate change and bring about new values, attitudes and social integration

Here are a few examples of the preferred terminology for English¹.

- Although some disabled people prefer the terms "physically challenged" or "differently abled", these should not generally be used. The disability rights movement of South Africa accepts both the terms "disabled person" and "people with disabilities".
- Avoid "suffers from," "afflicted with" or "victim of", all of which cast disabilities as a negative. "Suffers from" indicates ongoing pain and torment, which is no more the case for most people with disabilities as it is for most people without disabilities. "Afflicted with" denotes a disease, which most disabilities are not. "Victim of" implies that a crime is being committed on the person who has a disability.
- Do not use "wheelchair-bound" or "confined to a wheelchair". People see their wheelchairs as a convenient mode of transportation, not prisons, and the "bound/confined" phrase belies the fact that many people with motor disabilities engage in activities without their wheelchairs, including driving and sleeping. The proper phrase is "uses a wheelchair".
- Use "disability" not "handicap." The word "handicap" derives from the phrase "cap in hand", referring to a beggar, and is despised by most people with disabilities. Other terms to avoid are "physically/mentally challenged" (who isn't?) "cripple" or "crippled."
- Use "able-bodied" or "people without disabilities." The terms "normal" and "whole" are inappropriate and inaccurate.
- Most disabilities are not a disease. Do not call person with a disability a "patient" unless referring to a hospital setting. In an occupational and physical therapy context, "client" or "customer" is preferred.
- Some diseases by legal definition are considered disabilities. Victimization imagery ("AIDS victims") or defining the person by the disease ("she is a diabetic") is inappropriate. Use "person with diabetes" or "people living with AIDS".
- People who consider themselves as part of Deaf culture refer to themselves as "Deaf" with a capital "D". Because their culture derives from their language, they may be identified in the same way as other cultural groups, for example "Shangaan". Never use the terms "Deaf-mute" or "Deaf-and-Dumb".
- Avoid "deformed," "deformity" and "birth defect". A person may be "born without arms" or "has a congenital disability," but is probably not defective.
- Use "person with Down syndrome." Avoid "Mongol" or "mongoloid."
- Avoid "mentally retarded", "insane", "slow learner", "learning disabled" and "brain damaged". Use "person with an intellectual disability", or "person with a psychiatric disability".
- Avoid "cerebral palsied" and "spastic". Use "person with cerebral palsy".
- Use "person with epilepsy" or "child with a seizure disorder". Avoid "epileptic", either as noun or adjective.
- Avoid "dwarf" or "midget". Some groups prefer "little/short", but its best to use "person of short stature".
- Use "man with paraplegia" or "she has quadriplegia". Avoid "paraplegic" or "quadriplegic" as either a noun or adjective.

Negative and patronizing language produces negative and patronizing images. Words are important, so make sure your words do not offend or reinforce negative stereotypes.

¹ With acknowledgement to Patricia Digh of Real Work Group cited in the DPSA (Disabled People South Africa) - "A Pocket Guide on Disability Equity".



APPENDIX B: Principles of Universal Design (source: Center for Universal Design, 1997)

APPENDIX C: International Universal Design Declaration

"We, the participants of the UD2002 Conference, gathered here in Yokohama, and discussed focusing on the topic with the main theme "For All, For Everybody."

As our society has developed and become more complicated, we have tended to presume a uniform whole rather than recognizing the diversity of individuals. We propose to redefine the relationships between the designers/producers and users, thus giving priority to the diversity of individuals and cultures.

Design should create the social environments that respect and support the dignity of humans, which we propose to call Universal Design.

To realize this, it is urgent to establish user-centered system to be applicable to all aspects of society including infrastructures and legal systems.

We do not assume Universal Design to be a panacea. However, we think it possible going step by step toward the goal. In doing this, it is vital for the users to be part of the process, and the society to earnestly respond to them.

We would like to seek true globalization that respects difference in cultures.

The philosophy of Universal Design should be a foundation to respect limited natural resources and sustainable society.

We go forward.

We would like to continue our efforts toward building a society that will respect the natural variations among individuals, and the changes that we experience as we grow older, and give the highest priority to inclusion, participation, and independence for all.

We the participants from all over the world to the UD2002 Conference share this ideal as a result of this conference and declare that we devote ourselves to progress forward to this goal".

Yokohama, Japan December 4, 2002 APPENDIX D: Universal Design Audit Checklist (source: Levine, 2003:211-225)

UD Audit Checklist

How to Use this Checklist

To use the Checklist:

- Step 1 Check all the sections to be included in your evaluation.
- Step 2 Rate the level of usability for each element by checking the appropriate box in each section included. (either 1 point, 2 points, or 3 points.)
- Step 3
 Calculate the project score as described below.

 Remember, to rate 3 points, the element listed must have all the features listed.

How to Score this Checklist

Scores are computed as the percentage of the maximum score possible.

Calculation Steps:

Step 1	Count the number of sections you have rated, enter that number here:	
Step 2	Total the scores of all the sections rated.	
Step 3	The maximum possible score is the total in Step 1 multiplied by 3	
Step 4	UD Audit Score for the Project simply divide total in Step 2 by total for Step 3.	

Additional copies of this checklist are available online at www.ap.buffalo.edu/idea

APPENDIX E: Sample of Informed Consent Form (author's construct)

Study Title: Mainstreaming Universal Design in Cape Town: FIFA 2010 World Cup™related Activities as Catalysts for Social Change

Investigator: Mugendi K. M'Rithaa, Doctoral Candidate

Mr. M'Rithaa is a doctoral candidate studying the factors that influence the adoption and potential mainstreaming of Universal Design (UD) strategies in industrially developing contexts with special reference to FIFA 2010 World Cup[™]-related activities in Cape Town, South Africa. Although the study will not benefit you directly, it will provide information that might enable designers, educators, and policy planners to incorporate Universal Design features beneficial to the widest possible range of potential users of products, services, systems and built environments.

The study supervisors and other appropriate authorities at the Cape Peninsula University of Technology (CPUT), in the Western Cape Province, have approved the study and its procedures. The study procedures involve no foreseeable risk or harm to you. The procedures include:

[1] responding to a questionnaire about your response to design for all categories of people; and

[2] completing a supplementary demographic data sheet.

Participation in this study will take approximately 20 (twenty) minutes of your time.

Please feel free to ask any questions about the study or about being a participant/informant and you may call Mr. M'Rithaa at 021-460 3670 (work) or 072-655 8727 (mobile) during office hours if you have further questions. Additionally, you can contact him via facsimile: 021-460 3704, or email: <<u>MugendiM@cput.ac.za</u>>.

Your participation in this study is voluntary; you are under no obligation to participate and you have the right to withdraw at any time should you choose to do so.

The study data will be coded so they will not be linked to your name. Your identity will not be revealed while the study is being conducted or when the study is reported or published unless you give explicit consent for the same. To ensure anonymity and confidentiality, all study data will be collected by Mr. M'Rithaa, stored in a secure place, and not shared with any other person without your permission.

I have read this consent form and voluntarily consent to partici	pate in the study:
 Signature of Participant/Informant	Date
Signature of Witness	Date
I have explained this study to the above subject and have informed consent:	e sought his/her understanding for
Signature of Investigator	Date

DISCIPLINE	1 st Year	2 nd Year	3 rd Year	4 th Year	TOTAL
ARCHITECTURAL TECHNOLOGY	37	28	28	6	99
GRAPHIC DESIGN	77	70	70	22	239
INDUSTRIAL DESIGN	52	43	40	22	157
INTERIOR DESIGN	37	28	28	13	106
SURFACE DESIGN	33	17	22	8	80
TOWN & REGIONAL PLANNING	61	35	40	31	167
	297	221	228	102	848

APPENDIX F: Enrolments for Faculty of Informatics & Design (Cape Town Campus), 2007 (source: Institutional Planning Office, CPUT)

APPENDIX G: Judging Criteria and Weighting for the 2010 Green Goal Mouille Point Promenade & Beachfront Student Landscape Design Competition

UCT: focus on landscape design factors

	Weighting	Description	Marking			
1. Understanding the brief	15%	 Has the student followed the brief, and shown understanding 	 Each entry will be scored on a scale 			
2. Concept plan and contextualisation	15%	 Has the student provided and interesting, concept plan and put it into context of the bigger picture e.g. the "Green Corridor" Valuation of the concept plan 	 from 1 to 10 That rating will then be multiplied by the rating 			
3. Green Goal principles incorporated	15%	 Has the student considered "Green Goal" aspects; water, waste and energy minimisation/efficiency 	allocated to each category			
4. Visual quality of material	15%	 Valuation of the quality of the material and presentation of the poster/s 				
5. Design proposals	40%	 Valuation of the furnishing; feasibility, quality and creativity 				

CPUT: focus on furniture design factors

	Weighting	Description	Marking
1. Understanding the brief	5%	 Has the student followed the brief, and shown understanding 	 Each entry will be scored on a scale
2. Concept plan and contextualisation	5%	 Has the student provided and interesting, concept plan and put it into context of the bigger picture e.g. the "Green Corridor" Valuation of the concept plan 	from 1 to 10That rating will then be multiplied by the rating
3. Green Goal principles incorporated	20%	 Has the student considered "Green Goal" aspects; water, waste and energy minimisation/efficiency 	allocated to each category
4. Visual quality of material	10%	 Valuation of the quality of the material and presentation of the poster/s 	
5. Design proposals	60%	 Valuation of the furnishing; feasibility, quality and creativity 	

APPENDIX H: Questionnaire for Institutional Architect

<u>The Cape Peninsula University of Technology and Universal Design</u> (UD)

Name:							Эер	artment:						
Phone No.		-						Mobile		-				
Facsimile No.		-						Email						

Semi-Structured Interview Questions:

- 1. How does your Department/Section specifically cater for "extra-ordinary" individuals such as;
 - The Elderly; Women and Children; People with Disabilities; and Visitors (who do not speak any of the official languages)?
- 2. Do you have a specific policy that informs Design for Special Human Needs?
- 3. Kindly elaborate on Q.2.
- 4. In what ways can the Facility Management Department (of CPUT) incorporate UD for a more accessible and usable environment for all?
- 5. Who (individuals and/or departments/units) would best champion UD in the life of our esteemed institution (CPUT)?
- 6. With regards to Universal Design;
 - a. In your opinion, would your Department/Section be amenable in engaging UD?
 - b. Would you explicitly champion the concept of UD via the projects you are personally responsible for?
- 7. What obstacles do you anticipate in adopting UD as a strategy for enhancing accessibility at CPUT?
- 8. What is the best way to mainstream UD in CPUT?
- 9. What legacy will the FIFA 2010 World Cup[™] hopefully leave for accessibility in Cape Town once the mega-event is over?
- 10. Any other comments you might have...

Thank you for your time and valuable input. I would like to have your permission to cite you in my research.

Official use only	Interviewer's Na	me: Mugendi K. M'Rithaa	Signature	
Date	22 nd July 2008	Start Time	End Time	
Scrutinised	Yes I No 2	by (name)	Signature	
Questionnaire Serial	Number			

APPENDIX I: Questionnaire for Focus Group with Interior Design Students

Questions for Focus Group on 2010 Students' Design Competition

3rd Year Interior Design Students, CPUT:

I am carrying out some follow-up research on mainstreaming Universal Design (UD) in Cape Town and would greatly appreciate your honest responses to some questions relating to the 2010 Mouille Point Promenade & Beachfront Student Landscape Design Competition that was concluded on 8th May 2008:

A. Comments on 2010 Competition:

- 1. What worked best for you?
- 2. What did not work for you?
- 3. Would you have supported the involvement of students from other design disciplines in your team(s)?
- 4. Which disciplines would you have co-opted to work with?
- 5. Did you talk to people outside your own discipline during this project?
- 6. Do you consider competitions as effective platforms for promoting new design concepts/ideas?
- 7. How can future competitions be made more effective in promoting such concepts/ideas?

B. Comments on Universal Design (UD):

- 8. What worked best for you?
- 9. What did not work for you?
- 10. What aspects of UD would you wish to incorporate for future reference?
- 11. How can UD be mainstreamed in Interior Design?

C. Additional Comments:

- 12. Are you more confident about your professional competence after participating in the competition?
- 13. Do you intend to continue using UD Principles in future Interior Design projects?
- 14. Who would be the best champion for UD in the context of Cape Town?

Feel free to make any further comments...

Thanks for your valued time and input.

Mugendi K. M'Rithaa

22/05/2008 10.00-11.30 AM

APPENDIX J: Questionnaire for Indigenous Knowledge Expert

Indigenous Knowledge Systems: Disability, Ubuntu, and Universal Design

Semi-Structured Interview Questions: (20th March 2008; 09.00- 09.30; 2nd Floor, Admin Building, Bellville Campus, CPUT)

Q1: How did our people respond to the issue of Disability; for example, how was it defined, and what was the response to People with Disabilities?

Q2: How can the anthropocentric philosophy of *UBUNTU* inform current and future strategies to promote inclusion of disabled persons thereby countering exclusion (via stigma and discrimination)?

Q3: Who (individuals and/or institutions) would best champion UD in our different spheres of life?

Q4: In what ways can UD incorporate IKS for a more sustainable future for all?

Q5: Any other comments you might have...

Thank you for your valued time and input. I would like to cite you in my research.

APPENDIX K: Questionnaire for Politician

The DA, the City of Cape Town and Universal Design (UD)

Semi-Structured Interview Questions for DA Councillor:

- 1. How does the DA specifically engage with "extra-ordinary" individuals;
 - a. The Elderly;
 - b. Women and Children;
 - c. People with Disability; and
 - d. Foreigners (who do not speak any of the local languages)?
- 2. How can the anthropocentric philosophy of *UBUNTU* inform current and future strategies to promote inclusion of all residents and visitors and thereby countering exclusion (via stigma and discrimination)?
- 3. In what ways can the DA incorporate UD for a more sustainable future for all?
- 4. Who (individuals and/or institutions) would best champion UD in our different spheres of life?
- 5. With regards to Universal Design;
 - a. In your opinion, would the Ward you represent be amenable to engaging UD?
 - b. Would the Executive Mayor of Cape Town be interested in exploring the efficacy of UD?
 - c. Could you kindly facilitate a meeting whereby we can discuss/present the concept of UD to the Mayor?
- 6. What is the best way to mainstream UD in Cape Town?
- 7. What legacy will the FIFA 2010 World Cup[™] hopefully leave for residents (and visitors) in Cape Town once the mega-event is over?
- 8. Any other comments you might have...

Thank you for your time and valuable input. I would like to have your permission to cite you in my research.

Mugendi K. M'Rithaa <<u>MugendiM@cput.ac.za</u>> Dept of Industrial Design; Cape Peninsula University of Technology Tel: 021-460 3670; Fax: 021-460 3704; Cell: 072-655 8727

^{22&}lt;sup>nd</sup> April 2008: 10.00- 11.00 am; Goodwood, Cape Town

APPENDIX L: Questionnaire for Survey of Design Students



PROJECT UNIVERSAL DESIGN 2010+

Greetings. My name is Mugendi K. M'Rithaa. I am a doctoral student at the Department of Industrial Design, Cape Peninsula University of Technology in Cape Town. I am presently conducting a survey to gather opinions and ideas from designers on the subject of Universal Design (UD). Today I am conducting a survey on knowledge and usage of UD. I would be grateful if you could answer the following questions for me. Your responses will be kept confidential. If you would like me to contact you for further information, please indicate at the end of this questionnaire on Q.27.

	Questionnaire Serial Number													A								
Country	So	South Africa City/ Town									Cape Town											
Informant's Name											Ma	ale	I		Fem	Female 2						
Organisation's Name	с	PUT	Г								Specialisation*						Yea Stud	r of dy				
Phone No. (Landline)				-								Mobile				-						
Facsimile No.		- Email																				

SPECIALISATION* CODES

S pecialisation	Code		Code		Code
Industrial Design		Interior Design	2	Town & Regional Planning	3
Surface Design	4	Graphic Design	5	Architectural Technology	6

SECTION I: KNOWLEDGE

Q.1 Please indicate if you know anyone in any of the following groupings:									
"Extra-ordinary" population groups	Code	Go to							
Person with physical disability	I								
Person with visual disability	2]							
Person with hearing disability	3								
Person with learning disability	4								
Young person (14 years of age or younger)	5	Q.2							
Pregnant Woman/Mother with toddler (child aged 3 or younger)	6								
Elderly person (65 years of age or older)	7								
Foreign language speaker	8								
None of the above	9								

Q.2 Have you had any Universal Design-related training in the past six months?										
Previous UD training	Code	Go to								
Yes	I	Q.6								
No	2	Q.3								

Q.3 Which of the following concepts are you familiar with? You may select as many as you wish.									
Familiarity of concepts	Code	Go to							
Design for Disability	I								
Design-for-All	2								
Barrier-Free Design	3	Q.4							
Inclusive Design	4								
Universal Design	5								
None of the above	6	Q.6							

Q.4 Of the ones you have mentioned which one concept do you understand the best?				
Understanding of concepts	Code	Go to		
Design for Disability	I			
Design-for-All	2			
Barrier-Free Design	3	Q.5		
Inclusive Design	4			
Universal Design	5			
None of the above	6	Q.6		

Q.5 Please provide a brief definition (in your own words) of the concept you selected in Q.4 **Definition of concept**

Q.6 Please examine each pictogram carefully. Of the pictograms shown below, please indicate the number of the one that you feel is most suited to the principle described.

Pictograms of Universal Design principles	No.		Z
Simple and Intuitive Use			5
Perceptible Information		•C	
Size and Space for Approach and Use		2	6
Equitable Use		\mathbf{O}	İ,
Low Physical Effort		3	7
Toloranco for Error			Go to
			07
Flexibility in Use		4	~

SECTION 2: USAGE

Q.7 (Refer to Q.4). Is Universal Design used extensively in your design process?					
Extent of UD usage	Code	Go to			
Yes	I	Q.8			
No	2	Q.9			

Q.8 Which of the following principles of Universal Design do you use most often? You may select as				
many as you wish. Please refer to	the referer	nce poster on principles of Univer	rsal Design i	f you so wish.
Usage of Universal Design Code Code Go to				
principles				
Simple and Intuitive Use	I	Low Physical Effort	5	
Perceptible Information	2	Tolerance for Error	6	Q.9
Equitable Use	3	Size and Space for Approach	7	
Flexibility in Use	4	and Use		

Q.9 Which of the following statements best describes why you do not use any principles of Universal Design?

Non-usage of Universal Design principles	Code	Go to
I can use any principles I want to	I	
I have to choose from a different range of prescribed	2	
principles		Q.10
I have not heard of the principles of Universal Design before	3	
I am not interested in applying such principles in my work	4	
These principles do not apply to my work	5	

Q.10 If more information on Universal Design was available to you, how likely or unlikely would you be to use this information?

	normación.			
Very likely	Quite likely	Neither likely or unlikely	Quite unlikely	Very unlikely
5	4	3	2	—

Q.11 In what form do you feel should information on Universal Design be offered? You may select as many as you wish.

Preferred information	Code	Go to		Code	Go to
Exemplars/samples	1		Database of professionals/	٩	
Case studies	2		consultants	/	
Printed reference guidelines	3		Projects	10	
Useful internet links	4		Competitions	11	Q.13
Exchange programmes	5	Q.13	Academic courses	12	
Workshops/seminars	6		Training/short courses	13	
Industrial attachments	7		Discussion forums	14	
Posters	8		Other	15	Q.12

Q.12 Please provide other examples that you feel are not covered in Q.11 Other information

Q.13 Who do you feel would be most effective in championing implementation of Universal Design?					
			,		
Universal design champions	Code		Code		
Individual designers	1	Team/project leaders	3		
Political leaders	2	Design educators	4		

Please consider the following statements carefully. Based on your experience or perceptions of the following issues relating to Universal Design, please indicate how you rate your feeling for each statement.
SECTION 3: UNIVERSAL DESIGN IN CONTEXT

Q.14 How satisfied are you with the levels of accessibility within built environments/products/systems that you use most often?				
Very dissatisfied	Somewhat dissatisfied	Neither satisfied nor dissatisfied	Somewhat satisfied	Very satisfied
I	2	3	4	5

Q.15 If Universal Design considerations are used in the design process, how likely or unlikely would these considerations increase the total cost?

Very likely	Quite likely	Neither likely or unlikely	Quite unlikely	Very unlikely
5	4	3	2	

 Q.16 If you were given an opportunity to work on a multi-disciplinary team on a Universal Design project, how likely or unlikely would you be to do so?

 Very likely
 Quite likely
 Neither likely nor unlikely
 Quite unlikely
 Very unlikely

 5
 4
 3
 2
 I

Q.17 How important	Q.17 How important do you feel it is for designers working in South Africa to engage in socially responsible design					
to counter dis	scrimination in the form	of ageism, disablism and	sexism?			
Very important	Somewhat Neither Somewhat Not important at					
<i>,</i> .	important important nor unimportant all					
	unimportant					
5	4	3	2	I		

Q.18 How sure are you that you will be practicing design in South Africa beyond the 2010 FIFA World Cup™?					
Very confident/ sure	Very confident/ Somewhat Neither confident Somewhat unsure Very unsure sure confident/ sure nor sure				
5	4	3	2	I	

Q.19 To what extent do you agree that the philosophy of <i>ubuntu</i> is useful to design practice in South Africa?					
Strongly agree	Slightly agree	Neither agree nor disagree	Slightly disagree	Strongly disagree	Do not know
5	4	3	2	I	0

Q.20 If the 2010 FIFA World Cup [™] projects showcase the value of Universal Design, how likely or unlikely do you					
feel this would	feel this would have on greater adoption of Universal Design?				
Very likely	ry likely Quite likely Neither likely or Quite unlikely Very unlikely unlikely				
5	4	3	2	I	

Q.21 If legislation was used to compel designers to implement Universal Design in their projects, how likely or unlikely would you be to comply with the law?					
Very likely	likely Quite likely Neither likely or Quite unlikely Very unlikely				
5	4	3	2		

Q.22 To what degree do you agree that Universal Design courses be integrated into <u>all</u> design-related disciplines offered at university level?						
Strongly agreeSlightly agreeNeither agreeSlightlyStronglyDo not knownor disagreedisagreedisagreedisagreedisagree						
5	4	3	2	I	0	

Q.23 If incentives were available to reward designers who implement Universal Design in their projects, how likely					
or unlikely wo	ould you be to comply?				
Very likely	Quite likely Neither likely or Quite unlikely Very unlikely				
		unitery			
5	4	3	2	I	

Q.24 Please suggest some incentives that would appeal to you with reference to Q.23:
Possible incentives

SECTION 4: DEMOGRAPHIC DETAILS

Q.25 Please indicate the age group you fall into:				
Age	Code		Code	Go to
Up to 18 years	I	45 – 54 years	5	
19 – 24 years	2	55 – 64 years	6	0.26
25 – 34 years	3	65 years and above	7	Q.20
35 – 44 years	4		,	

Q.26 Please indicate the highest level of education you have attained:					
Education level attained	Code		Code		
Completed High School (Matric)	1	Completed Higher Diploma	6		
Part Certificate	2	Part university degree (Bachelor's or equivalent)	7		
Completed Certificate	3	Completed university degree (Bachelor's or equivalent)	8		
Part Ordinary Diploma	4	Post-graduate university degree (Master's	0		
Completed Ordinary Diploma	5	equivalent or higher)	7		

SECTION 5: FEEDBACK COMMENTS

Q.27 Kindly provide feedback on any additional questions or comments for consideration:									
Informant's feedback									

THANK YOU FOR YOUR TIME AND INPUT.

Official use only	Intervie	ewer'	s Name	e		Signature	
Date					Start Time	End Time	
Scrutinised	Yes I No 2				by (name)	Signature	

APPENDIX M: Questionnaire for Professional Designers

Questions for 'Mainstreaming UD in Cape Town' research

Greetings. My name is Mugendi K. M'Rithaa. I am a doctoral student at the Department of Industrial Design, CPUT, presently conducting a survey on the subject of Universal Design (UD). Subsequently, I wish to gain an insight into your views on UD in your personal and professional capacity. Kindly answer the following questions as candidly as possible. Your responses will be cited in my dissertation unless you explicitly object (refer to Section E).

Contact Details:

Name:												
Organisation & Position:												
Phone No.		-				Mobile		-				
Facsimile No.		-				Email						

A. FIFA 2010 World Cup[™]-related projects:

- 1. What is your specific role in the 2010-related projects?
- 2. Briefly describe the projects being carried out.
- 3. Would you enlist the participation of university students in your 2010 projects?
- 4. Would you support the involvement of independent professional designers in these projects?
- 5. Which design disciplines would you most likely co-opt to work with?
- 6. Kindly elaborate on your answer to Q. 5.
- 7. Do you consider competitions as effective platforms for promoting new design concepts/ideas?
- 8. What do you consider to be the key elements in a competition to make them more effective in promoting new concepts/ideas?
- 9. What legacy will the FIFA 2010 World Cup[™] hopefully leave for residents (and visitors) in Cape Town once the mega-event is over?

B. Familiarity with Universal Design (UD):

- 10. Are you familiar with the concept of Universal Design (UD)?
- 11. Kindly comment on how you first came to be involved with UD. Feel free to provide names, facts, figures and dates where appropriate.
- 12. Briefly comment on the reason(s) why UD has not found wider acceptance within your sphere of influence.
- 13. Is UD relevant to any of the projects/functions you are currently involved in?
- 14. With reference to the Principles of UD (refer to attached document) what principle(s) do you find most relevant to the projects you are involved in?
- 15. What aspects of UD do you consider to be most neglected presently?
- 16. How can UD be mainstreamed in your specific department?

C. Promotion of UD:

- 17. Briefly comment on your personal experiences with regards to UD promotion.
- 18. What media has been used to promote UD, and how would you rate the same in terms of effectiveness?
- 19. Who should be the main actors with regards to UD?
- 20. Which official/government policies would UD best align with?
- 21. Which local, regional, and national government departments would best accommodate UD?
- 22. Which individual/profession (in the private or public sectors) would be the ideal champion for UD? Kindly elaborate on your choice(s).
- 23. Who would be the best champion for UD in the context of the city of Cape Town?
- 24. What strategies would mainstream UD at local, regional, and national levels in an effective and sustainable manner?
- 25. Do you intend promoting the use of UD Principles in future?
- 26. Kindly elaborate on your answer to Q. 25.

D. Additional Comments:

27. Feel free to make any further comments...

E. Authorisation to publish responses:

I hereby request your authorisation to publish your responses in my dissertation, subject to the conditions mentioned above:

Signed:

Date:

Please send this authorisation to <u>MugendiM@cput.ac.za</u> by the 19th September 2008.

Thank you for your valued time and input.

Official use only	Intervi	ewe	er's N	ame	e: Mugendi K. M'Rith	naa	Signature		
Date					Start Time		End Time		
Scrutinised	Yes	Ι	No	2	by (name)		Signature		
Questionnaire Seria	l Numbe	er							

APPENDIX N: Questionnaire for Design Educators

Questions for 'Mainstreaming UD in Cape Town' project

Dear Colleague,

Greetings. My name is Mugendi K. M'Rithaa. I am a doctoral student at the Department of Industrial Design, CPUT, presently conducting a survey on the subject of Universal Design (UD). Subsequently, I wish to conduct a needs analysis on UD with respect to your Department. Kindly answer the following questions as candidly as possible. Your responses will be cited in my dissertation unless you explicitly object. Kindly confirm whether (or not) you wish to have your responses published and communicate the same as described in Section E.

Contact Details:

Name:												
Position:												
Phone No.		-				Mobile		-				
Facsimile No.		-				Email						

SPECIALISATION* CODES

Specialisation	Code		Code		Code
Industrial Design		Interior Design	2	Town & Regional Planning	3
Surface/Fashion Design	4	Graphic Design	5	Architecture	6

A. FIFA 2010 World Cup[™]-related projects:

- 1. Are there any 2010-related projects being done in your Department?
- 2. Who (students and/or staff) is carrying out such projects?
- 3. Briefly describe the projects being carried out.
- 4. Would you support the participation of your students in 2010 projects?
- 5. Would you support the involvement of students from other design disciplines in such projects?
- 6. Which disciplines/departments would you most likely co-opt to work with?
- 7. Kindly elaborate on your answer to Q. 6.
- 8. Do you consider competitions as effective platforms for promoting new design concepts/ideas?
- 9. How can competitions be made more effective in promoting new concepts/ideas?

B. Familiarity with Universal Design (UD):

- 10. Are you familiar with the concept of Universal Design (UD)?
- 11. Has UD been taught to any of your students to date?
- 12. Is UD relevant to any of the courses/modules you currently offer in your programme of study?
- 13. With reference to the Principles of UD (refer to attached document) what principle(s) do you find most relevant to your discipline?

- 14. What aspects of UD would you wish to incorporate in future?
- 15. How can UD be mainstreamed in your specific discipline?

C. Interest in Teaching of UD:

- 16. Would you be interested in having UD offered to your students in a formal module?
- 17. Which subject(s) would UD best align with?
- 18. In which year of study should UD first be introduced to your students?
- 19. Do you intend to continue using UD Principles in future discipline-specific projects?
- 20. Who would be the best champion for UD in the context of the city of Cape Town?

D. Additional Comments:

21. Feel free to make any further comments...

E. Authorisation to publish responses:

I hereby request your authorisation to publish your responses in my dissertation, subject to the conditions mentioned above.

Please send this authorisation to <u>MugendiM@cput.ac.za</u> by the 8th August 2008.

Thanks for your valued time and input.

Official use only	Intervi	ewer	's Nam	e: M u	ugendi K. M'Rithaa	ı	Signatur	e	
Date					Start Time		End Tim	e	
Scrutinised	Yes	Ι	No	2	by (name)		Signatur	e	
Questionnaire Seri	Questionnaire Serial Number								

APPENDIX O: Summary of Findings from UD Workshop at DEFSA 2007 Conference, CPUT, Cape Town (Boonzaier & M'Rithaa, 2007)

Workshop held from 11.30 -13.00 hrs, on Thursday, 4th October 2007 at CPUT Lecture Theatre Venue 2.

Question 1

Have you ever had a personal problem with any consumer Product(s)?

Yes: 19 No: 1 No Answer: 0

List of products with related problems:

- VEHICLE
 - Car one *headlight* not working
 - Cars too big, *seats* too high, uncomfortable

• **KITCHEN**

- Automated Coffee-machines poor *interaction*
- o Clover Milk Cartons New *lid* always breaks
- o Blender *ice-crushing* feature not effective
- Milk jug x2 *spout* spills milk
- Vegetable peeler *difficult to use*
- Kitchen appliances *difficult to keep clean*
- Can opener designed for the *right handed*
- FURNITURE
 - Chairs- not designed for short people, very **uncomfortable**

• ELECTRONICS

- o TV's Decoder *complex* programming, need manual even for simple functions
- Digital camera bad quality photos*
- Media equipment difficult to set up/ carry
- o Electronic Goods need *manual* to use
- Mac, Apple Computer crashed, *lost data*
- o Microsoft Windows and Excel Not enough functionality
- PERSONAL EFFECTS
 - o Reading glasses falls off, lens falls out
 - Pepper spray takes too long to use
- PACKAGING
 - Packaging x3 *difficult to open*
 - o Containers x2 difficult to open

Question 2

Have you ever had a personal problem with any built environments?

Yes: 19	No: 1	No Answer: 0
---------	-------	--------------

Built Environments and related problems:

- CPUT
 - CPUT Wheelchair friendly *toilets* available, but wheelchair can't enter *door* of restroom
 - Lifts available to solve *stairs*-problem, but person in wheelchair cannot operate *lift controls*.
 - o CPUT *stairs* are highly polished and *slippery*
 - o CPUT, Commerce Building *finding Exit*

• PUBLIC PLACES

- o Airports Long *passages* and problems with baggage handling
- Cape Town Airport No proper *signage*
- o Buildings, buses, subways etc. Difficulty manoeuvring child in stroller
- Public Buildings *doors*, "push" or "pull"?
- Shopping centre x2 steep *staircases*, very *loud/noisy*
- Entrances to buildings x2 glass *doors*, difficult to see. *Door handles* difficult to open
- Restaurants very *loud/noisy*

• **RESIDENTIAL**

- House of participant direction *doors* open does not work with space in rooms
- Home bad flow *design* though *rooms*
- Bathrooms x4 unmarked *faucets*, *wash basin* is badly designed, *bathroom* too small *cupboard doors* that don't fit.
- Kitchen c*upboard doors* that don't fit
- PUBLIC SPACES
 - Pavement lack of *pedestrian space*
 - o Unfamiliar places bad *signage*, difficult to navigate

Question 3

Have you ever had a personal problem with any design-related services?

Yes: 14 No: 5 No Answer: 1

Design-related services and related problems:

- Graphic Products, exhibitions lack of lateral thinking; creative search for solutions
- Business Cards *font* too small, *not readable*
- Multimedia software difficult to use, help instruction bad
- Design software difficult to use/ translate ideas to designs
- Websites, ATM, signage bad fonts used, contrasting colours making it difficult to read

- Architecture design bad *acoustics*/ sound proofing between rooms*
- ATM bad user interface design
- Taxi metering system, can be abused by operator

Question 4

Do you now have a greater insight into how consumer products can be improved?

Yes: 18 No: 1 No Answer: 1

Suggestions on what aspects of this improvement can be improved:

- Incorporating universal design principals, so that product can be used by a wider range of "disability" groups.
- Products designed for all people to use, across racial and cultural barriers.
- Simpler and more intuitive to use.
- Products must relate to all users
- Implementation and enforcement of design regulations, and a better attitude from designers
- Simplified design that can be used by the handicapped, young, old.
- Feasible products that take the user into consideration.
- Designing better ergonomic products.
- People willing to change the products they use, so that they can include more people.
- Talking more to users
- Having a greater awareness of possibilities and problems when a project is started.
- Consciousness of designers and users can be raised.
- More people need to be aware of the solutions
- Products to be tested before they are sold
- Know what issues need to be considered
- More inclusiveness of use on a universal scale

Question 5

Do you now have a greater insight into how built environments can be improved?

Yes: 15 No: 4 No Answer: 1

Suggestions on what aspects of this improvement can be improved:

- Improved communication design, thru signage and the environment
- Accessible buildings for all people.
- Incorporate cultural needs when designing
- User-centred design with pleasant easy to use spaces
- Research the public's needs before designing infrastructure
- Accessible design

Question 6

Do you now have a greater insight into how services can be improved?

Yes: 11 No: 7 No Answer: 2

Suggestions on what aspects of this improvement can be improved:

- Services offered by people, attitudes and perceptions*
- Innovative well researched products that benefit the service provider and the user
- Better systems design to make services easy to understand
- More research about the user
- Implementation of regulations in design

Question 7

Are we as South African designers currently taking Universal Design concepts seriously?

Yes: 3 No: 13 No Answer: 4

Elaborations on answers:

Yes:

- At an academic level, consciousness about Universal Design is increasing.
- Awareness is growing, but still needs to be applied practically
- Those who are aware do.

No:

- At some levels there is room for improvement
- Designers are not yet focusing on incorporating UD into design
- UD needs to be incorporated into the design process.
- Designers are closed minded and designing only for themselves
- Aesthetics/consumer-goods dominate consciousness/motivation
- Evidence on the ground lost opportunities
- No enough user involvement
- Client driven briefs
- Built environments not accessible by all

Number of participants interested in joining a user group: 10

Number of participants interested in joining a blog spot: 14

APPENDIX P: Sample SPSS Outputs

CROSSTABS /TABLES=Specialisation BY Q.6a Q.6b Q.6c Q.6d Q.6e Q.6f Q.6g BY YearofStudy /FORMAT=AVALUE TABLES /CELLS=COUNT /COUNT ROUND CELL /BARCHART.

Crosstabs

	Notes	
Output Created		2009-04-04T14:43:18.171
Comments		
Input	Data	C:\Documents and Settings\MugendiM\My Documents\Doctorate\Data Analysis\MKM SPSS\MKM SPSS1.sav
	Active Dataset	DataSet1
	Filter	<none></none>
	Weight	<none></none>
	Split File	<none></none>
	N of Rows in Working Data File	147
Missing Value Handling	Definition of Missing	User-defined missing values are treated as missing.
	Cases Used	Statistics for each table are based on all the cases with valid data in the specified range(s) for all variables in each table.
Syntax		CROSSTABS //TABLES=Specialization BY Q.6a Q.6b Q.6c Q.6d Q.6e Q.6f Q.6g BY YearofStudy /FORMAT=AVALUE TABLES /CELLS=COUNT /COUNT ROUND CELL /BARCHART.
Resources	Processor Time	0:00:10.531
	Elapsed Time	0:00:10.266
	Dimensions Requested	3
	Cells Available	142987

[DataSet1] C:\Documents and Settings\MugendiM\My Documents\Doctorate\Data Analysis\MKM SPSS\MKM SPSS1.sav

		ase i recessing	e annai y			
			С	ases		
	Va	lid	Mis	sing	Т	otal
	Ν	Percent	N	Percent	N	Percent
Specialization * Simple and Intuitive Use (2) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Perceptible Information (3) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Size and Space for Approach and Use (7) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Equitable Use (1) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Low Physical Effort (4) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Tolerance for Error (6) * Year of Study	147	100.0%	0	.0%	147	100.0%
Specialization * Flexibility in Use (5) * Year of Study	147	100.0%	0	.0%	147	100.0%

Case Processing Summary

Count										•
					Simp	le and Intuitive	Use (2)			
Year of	Study		1	2	3	4	5	6	7	Total
1	Specialization	1	2	4	0	1	2	1	0	10
		2	3	0	1	1	0	0	1	6
		3	2	0	1	0	1	1	0	5
		5	4	3	0	3	0	3	3	16
		6	3	2	0	1	0	0	0	6
		To tal	14	9	2	6	3	5	4	43
2	Specialization	1	1	3	2	1		3	0	10
		2	2	1	1	1		0	0	5
		5	3	7	1	1		1	1	14
		6	2	3	0	0		0	0	5
		To tal	8	14	4	3		4	1	34
3	Specialization	1	1	7	0	0	1	0	0	9
		2	2	1	0	0	0	1	1	5
		3	2	2	1	0	0	0	0	5
		5	3	3	0	1	0	0	0	7
		6	7	2	0	0	0	0	1	10
		To tal	15	15	1	1	1	1	2	36
4	Specialization	1	0	9	0	0	0	0	0	9
		2	3	0	0	1	0	1	0	5
		3	2	1	1	0	0	0	1	5
		5	1	4	1	0	0	1	0	7
		6	3	1	1	0	2	0	1	8
		To tal	9	15	3	1	2	2	2	34

Specialization * Simple and Intuitive Use (2) * Year of Study Crosstabulation









Year of Study=3





Specialization * Perceptible Information (3) * Year of Study Crosstabulation

Count										
Γ		-			-	Perceptibl	e Information	(3)	-	-
Year of	Study		1	2	3	4	5	6	7	Total
1	Specialization	1	0	1	6	0		2	1	10
		2	0	1	3	1		1	0	6
		3	0	2	1	1		1	0	5
		5	0	1	13	1		1	0	16
		6	2	0	3	0		1	0	6
		Total	2	5	26	3		6	1	43
2	Specialization	1	1	1	6			1	1	10
		2	0	1	3			0	1	5
		5	0	0	11			0	3	14
		6	0	0	5			0	0	5
		Total	1	2	25			1	5	34
3	Specialization	1	1	0	6	0		2	0	9
		2	0	1	3	0		0	1	5
		3	0	1	2	0		1	1	5
		5	0	1	6	0		0	0	7
		6	1	3	3	1		2	0	10
		Total	2	6	20	1		5	2	36
4	Specialization	1		0	9		0	0	0	9
		2		1	4		0	0	0	5
		3		0	4		0	1	0	5
		5		0	5		0	2	0	7
		6		2	3		1	1	1	8
		Total		3	25		1	4	1	34







Year of Study=3



Year of Study=4



Count										
				-	Size an	d Space for A	pproach and	Use (7)	-	-
Year of Study			1	2	3	4	5	6	7	Total
1	Specialization	1		3	0	1	0	0	6	10
		2		2	0	1	1	0	2	6
		3		1	1	1	0	1	1	5
		5		9	0	1	0	0	6	16
		6		0	0	0	1	0	5	6
		Total		15	1	4	2	1	20	43
2	Specialization	1	1	2		0	0	0	7	10
		2	0	1		1	0	2	1	5
		5	0	2		0	2	2	8	14
		6	0	1		2	0	0	2	5
		Total	1	6		3	2	4	18	34
3	Specialization	1	0	3	0	0	0	1	5	9
		2	0	2	0	1	0	0	2	5
		3	1	0	1	0	0	0	3	5
		5	0	0	0	0	1	1	5	7
		6	0	5	1	0	0	1	3	10
		Total	1	10	2	1	1	3	18	36
4	Specialization	1	0	0	0	0	0	0	9	9
		2	0	1	0	2	0	1	1	5
		3	2	0	0	0	1	1	1	5
		5	0	4	0	0	0	0	3	7
		6	0	1	2	0	1	2	2	8
		Total	2	6	2	2	2	4	16	34

Specialization * Size and Space for Approach and Use (7) * Year of Study Crosstabulation







Year of Study=3





Specialization * Equitable Use (1) * Year of Study Crosstabulation

Count		_	-					_	-			
			Equitable Use (1)									
Year of Study			1	2	3	4	5	6	7	Total		
1	Specialization	1	7	0	3		0	0	0	10		
		2	2	1	0		1	2	0	6		
		3	0	1	2		0	0	2	5		
		5	10	0	0		0	3	3	16		
		6	1	0	1		1	3	0	6		
		Total	20	2	6		2	8	5	43		
2	Specialization	1	4	3	1	0	0	2	0	10		
		2	3	0	0	1	0	0	1	5		
		5	4	1	0	0	4	4	1	14		
		6	3	0	0	0	0	1	1	5		
		Total	14	4	1	1	4	7	3	34		
3	Specialization	1	6	0	2		0	0	1	9		
		2	1	1	0		1	2	0	5		
		3	2	0	0		0	2	1	5		
		5	4	2	0		0	0	1	7		
		6	2	0	2		2	0	4	10		
		Total	15	3	4		3	4	7	36		
4	Specialization	1	9	0	0	0	0	0	0	9		
		2	0	1	1	0	1	1	1	5		
		3	1	1	0	1	0	1	1	5		
		5	6	0	0	1	0	0	0	7		
		6	3	2	0	0	2	1	0	8		
		Total	19	4	1	2	3	3	2	34		



Year of Study=2









Count										
-			-	_		Low Phy	sical Effort (4)			_
Year of Study			1	2	3	4	5	6	7	Total
1	Specialization	1	1	1	1	4	2	1	0	10
		2	0	0	2	2	1	1	0	6
		3	1	0	0	1	2	0	1	5
		5	0	3	2	8	1	1	1	16
		6	0	0	1	3	0	1	1	6
		Total	2	4	6	18	6	4	3	43
2	Specialization	1			1	7	1	0	1	10
		2			1	1	1	2	0	5
		5			1	10	0	3	0	14
		6			0	3	0	0	2	5
		Total			3	21	2	5	3	34
3	Specialization	1		0	1	7	0	0	1	9
		2		0	0	2	1	2	0	5
		3		1	0	1	2	1	0	5
		5		1	1	3	1	0	1	7
		6		1	2	5	0	1	1	10
		Total		3	4	18	4	4	3	36
4	Specialization	1	0	0		9	0	0	0	9
		2	2	1		1	0	0	1	5
		3	0	1		1	1	1	1	5
		5	0	0		4	1	1	1	7
		6	0	1		5	0	0	2	8
		Total	2	3		20	2	2	5	34

Specialization * Low Physical Effort (4) * Year of Study Crosstabulation







Year of Study=3





Specialization * Tolerance for Error (6) * Year of Study Crosstabulation Count

-	-	Tolerance for Error (6)									
Year of Study		1	2	3	4	5	6	7	Total		
1 Specialization	1	0	0	0	3	2	5	0	10		
	2	0	2	0	1	2	1	0	6		
	3	2	0	0	0	1	2	0	5		
	5	2	1	0	1	1	8	3	16		
	6	0	1	2	0	1	1	1	6		
	Total	4	4	2	5	7	17	4	43		
2 Specialization	1	2	1	0	1	3	2	1	10		
	2	1	1	0	0	2	0	1	5		
	5	4	2	1	0	3	2	2	14		
	6	0	1	0	0	1	3	0	5		
	Total	7	5	1	1	9	7	4	34		
3 Specialization	1	1		0	0	3	4	1	9		
	2	1		2	0	1	0	1	5		
	3	0		1	1	2	1	0	5		
	5	0		0	0	1	6	0	7		
	6	0		1	0	5	4	0	10		
	Total	2		4	1	12	15	2	36		
4 Specialization	1		0	0	0	0	9	0	9		
	2		2	0	0	3	0	0	5		
	3		1	1	0	2	0	1	5		
	5		1	1	1	0	1	3	7		
	6		0	1	1	1	4	1	8		
	Total		4	3	2	6	14	5	34		











Count										
_			-			Flexibili	ty in Use (5)			
Year of Study			1	2	3	4	5	6	7	Total
1	Specialization	1	1	0	0	2	4	1	2	10
		2	1	0	0	0	1	1	3	6
		3	0	0	0	3	1	0	1	5
		5	0	0	1	1	14	0	0	16
		6	0	2	0	1	3	0	0	6
		Total	2	2	1	7	23	2	6	43
2	Specialization	1	1	0		1	6	2		10
		2	0	1		0	4	0		5
		5	2	1		3	6	2		14
		6	0	0		0	4	1		5
		Total	3	2		4	20	5		34
3	Specialization	1		0		2	5	1	1	9
		2		0		2	2	0	1	5
		3		1		3	1	0	0	5
		5		0		3	4	0	0	7
		6		0		4	4	1	1	10
		Total		1		14	16	2	3	36
4	Specialization	1	0	0	0	0	9	0	0	9
		2	0	0	0	0	1	2	2	5
		3	0	0	0	3	1	1	0	5
		5	0	0	0	1	6	0	0	7
		6	2	1	1	1	1	1	1	8
		Total	2	1	1	5	18	4	3	34

Specialization * Flexibility in Use (5) * Year of Study Crosstabulation

Year of Study=1









